Mathematical Sciences Libitary

Ramsey Theory on the Integers

١

STUDENT MATHEMATICAL LIBRARY Volume 24

Ramsey Theory on the Integers

Bruce M. Landman Aaron Robertson



Editorial Board

David Bressoud, Chair Daniel L. Goroff Davide P. Cervone Carl Pomerance

2000 Mathematics Subject Classification. Primary 05D10.

For additional information and updates on this book, visit www.ams.org/bookpages/stml-24

Library of Congress Cataloging-in-Publication Data

Landman, Bruce M., 1951–
Ramsey theory on the integers / Bruce M. Landman, Aaron Robertson.
p. cm. — (Student mathematical library, ISSN 1520-9121; v. 24)
Includes bibliographical references and index.
ISBN 0-8218-3199-2 (softcover : alk. paper)
1. Ramsey theory. I. Robertson, Aaron, 1971– II. Title. III. Series.

QA166.L34 2003 511′.66—dc22

2003062937

Copying and reprinting. Individual readers of this publication, and nonprofit libraries acting for them, are permitted to make fair use of the material, such as to copy a chapter for use in teaching or research. Permission is granted to quote brief passages from this publication in reviews, provided the customary acknowledgment of the source is given.

Republication, systematic copying, or multiple reproduction of any material in this publication is permitted only under license from the American Mathematical Society. Requests for such permission should be addressed to the Acquisitions Department, American Mathematical Society, 201 Charles Street, Providence, Rhode Island 02904-2294, USA. Requests can also be made by e-mail to reprint-permission@ams.org.

> © 2004 by the American Mathematical Society. All rights reserved. The American Mathematical Society retains all rights except those granted to the United States Government. Printed in the United States of America.

∞ The paper used in this book is acid-free and falls within the guidelines established to ensure permanence and durability. Visit the AMS home page at http://www.ams.org/

10 9 8 7 6 5 4 3 2 1 09 08 07 06 05 04

To	To
E lean or	Elisa
Emma and Sarah	$Quinn \ and \ Ava$

-Bruce

-aaron

Contents

List of Tables	xi
Preface	xiii
Chapter 1. Preliminaries	1
§1.1. The Pigeonhole Principle	3
§1.2. Ramsey's Theorem	5
§1.3. Some Notation	9
§1.4. Three Classical Theorems	11
§1.5. A Little More Notation	14
§1.6. Exercises	16
§1.7. Research Problems	19
§1.8. References	20
Chapter 2. Van der Waerden's Theorem	21
§2.1. The Compactness Principle	25
$\S2.2.$ Alternate Forms of van der Waerden's Theorem	27
§2.3. Computing van der Waerden Numbers	29
	والمراجع المراجع الم

vii

viii		Contents
$\S{2.4.}$	Bounds on van der Waerden Numbers	35
$\S{2.5.}$	The Erdős and Turán Function	41
$\S{2.6.}$	Proof of van der Waerden's Theorem	43
$\S{2.7.}$	Exercises	49
$\S{2.8.}$	Research Problems	51
$\S{2.9.}$	References	52
Chapter	• 3. Supersets of AP	55
$\S{3.1.}$	Quasi-progressions	56
$\S{3.2.}$	Generalized Quasi-progressions	66
§3.3.	Descending Waves	70
$\S{3.4.}$	Semi-progressions	72
$\S{3.5.}$	Iterated Polynomials	81
$\S{3.6.}$	Arithmetic Progressions as Recurrence Solutions	91
$\S{3.7.}$	Exercises	93
$\S{3.8.}$	Research Problems	96
§ 3.9 .	References	100
Chapter	4. Subsets of AP	103
$\S4.1.$	Finite Gap Sets	105
$\S4.2.$	Infinite Gap Sets	111
$\S4.3.$	Exercises	129
$\S4.4.$	Research Problems	131
$\S4.5.$	References	133
Chapter	5. Other Generalizations of $w(k; r)$	135
$\S{5.1.}$	Sequences of Type $x, ax + d, bx + 2d$	135
$\S{5.2.}$	Homothetic Copies of Sequences	147

ents	Contents	ix
35	§5.3. Sequences of Type $x, x + d, x + 2d + b$	152
41	§5.4. Exercises	158
43	§5.5. Research Problems	159
49	5.6. References	160
51	Chapter 6. Arithmetic Progressions $(\mod m)$	163
52	§6.1. The Family of Arithmetic Progressions $(\mod m)$	164
55	§6.2. A Seemingly Smaller Family is More Regular	167
56	§6.3. The Degree of Regularity	173
66	§6.4. Exercises	176
70	§6.5. Research Problems	178
72	6.6. References	180
81	Chapter 7. Other Variations on van der Waerden's Theorem	181
91	§7.1. The Function $\Gamma_m(k)$	181
93	§7.2. Monochromatic Sets $a(S+b)$	185
96	§7.3. Having Most Elements Monochromatic	187
100	§7.4. Permutations Avoiding Arithmetic Progressions	191
103	§7.5. Exercises	195
105	§7.6. Research Problems	196
111	§7.7. References	197
129	Chapter 8. Schur's Theorem	199
131	§8.1. The Basic Theorem	200
133	§8.2. A Generalization of Schur's Theorem	211
135	§8.3. Refinements of Schur's Theorem	216
135	§8.4. Exercises	219
147	§8.5. Research Problems	221

x		Contents
§8.6. Reference	S	222
Chapter 9. Rado's	s Theorem	225
§9.1. Rado's Si	ngle Equation Theorem	225
§9.2. Some Rac	lo Numbers	239
§9.3. Generaliz	ations of the Single Equation Theorem	249
§9.4. Exercises		255
§9.5. Research	Problems	257
§9.6. Reference	98	258
Chapter 10. Othe	r Topics	261
§10.1. Folkman	i's Theorem	261
§10.2. Doublefr	ee Sets	265
§10.3. Diffseque	ences	266
§10.4. Brown's	Lemma	276
§10.5. Patterns	in Colorings	279
§10.6. Zero-sun	ns	281
10.7. Exercise	s	288
§10.8. Research	1 Problems	289
§10.9. Reference	es	291
Notation		293
Bibliography		297
Index		315

List of Tables

Table 2.1: Mixed van der Waerden values	34
Table 2.2: Lower bounds and values for $w(k; r)$	38
Table 3.1: Values and lower bounds for $Q_{k-i}(k)$	66
Table 3.2: Values and lower bounds for $GQ_{f(x)}(k)$	69
Table 3.3: Values and lower bounds for $SP_m(k)$	81
Table 3.4: Values of $R(S_n, k; r)$ and $R(AP \cup P_n, k)$	91
Table 4.1: Values of $w(f(x), 3; 2)$	121
Table 5.1: Values and lower bounds for $T(a, b)$	146
Table 5.2: Degree of regularity of (a, b) -triples	147
Table 5.3: Values of $R(c, AUG_b, 3)$	157
Table 6.1: Values and lower bounds for $R(AP_{(m)}, k; 2)$	166
Table 6.2: Degree of regularity of families of type $AP_{(m)}$	176
Table 10.1: Values of $\Delta(D, k)$	276
Table 10.2: Number of squarefree and cubefree colorings	281

xi

Preface

Ramsey Theory on the Integers covers a variety of topics from the field of Ramsey theory, limiting its focus to the set of integers – an area that has seen a remarkable burst of research activity during the past twenty years.

The book has two primary purposes: (1) to provide students with a gentle, but meaningful, introduction to mathematical research – to give them an appreciation for the essence of mathematical research and its inescapable allure and also to get them started on their own research work; (2) to be a resource for all mathematicians who are interested in combinatorial or number theoretical problems, particularly "Erdős-type" problems.

Many results in Ramsey theory sound rather complicated and can be hard to follow; they tend to have a lot of quantifiers and may well involve objects whose elements are sets whose elements are sets (that is not a misprint). However, when the objects under consideration are sets of integers, the situation is much simpler. The student need not be intimidated by the words "Ramsey theory," thinking that the subject matter is too deep or complex – it is not! The material in this book is, in fact, quite accessible. This accessibility, together with the fact that scores of questions in the subject are still to be answered, makes Ramsey theory on the integers an ideal subject for a student's first research experience. To help students find suitable Preface

Preface

projects for their own research, every chapter includes a section of "Research Problems," where we present a variety of unsolved problems, along with a list of suggested readings for each problem.

Ramsey Theory on the Integers has several unique features. No other book currently available on Ramsey theory offers a cohesive study of Ramsey theory on the integers. Among several excellent books on Ramsey theory, probably the most well-known, and what may be considered the Ramsey theory book, is by Graham, Rothschild, and Spencer (Ramsey Theory, 2^{nd} Edition [127]). Other important books are by Graham (Rudiments of Ramsey Theory [122]). McCutcheon (Elemental Methods in Ergodic Ramsey Theory [184]). Nešetřil and Rödl (Mathematics of Ramsey Theory [199]), Prömel and Voigt (Aspects of Ramsey Theory [207]), Furstenberg (Dynamical Methods in Ramsey Theory [111]), and Winn (Asymptotic Bounds for Classical Ramsey Numbers [274]). These books, however, generally cover a broad range of subject matter of which Ramsey theory on the integers is a relatively small part. Furthermore, the vast majority of the material in the present book is not found in any other book. In addition, to the best of our knowledge, ours is the only Ramsey theory book that is accessible to the typical undergraduate mathematics major. It is structured as a textbook, with numerous (over 150) exercises, and the background needed to read the book is rather minimal: a course in elementary linear algebra and a 1semester junior-level course in abstract algebra would be sufficient: an undergraduate course in elementary number theory or combinatorics would be helpful, but not necessary. Finally, Ramsey Theory on the Integers offers something new in terms of its potential appeal to the research community in general. Books offering a survey of solved and unsolved problems in combinatorics or number theory have been quite popular among researchers; they have also proven beneficial by serving as catalysts for new research in these fields. Examples include Old and New Problems and Results in Combinatorial Number Theory [92] by Erdős and Graham, Unsolved Problems in Number Theory [135] by Guy, and The New Book of Prime Number Records [220] by Ribenboim. With our text we hope to offer mathematicians an additional resource for intriguing unsolved problems. Although not

nearly exhaustive, the present book contains perhaps the most substantial account of solved and unsolved problems in Ramsey theory on the integers.

This text may be used in a variety of ways:

- as an undergraduate or graduate textbook for a second course in combinatorics or number theory;
- in an undergraduate or graduate seminar, a capstone course for undergraduates, or an independent study course;
- by students working under an REU program, or who are engaged in some other type of research experience;
- by graduate students looking for potential thesis topics;
- by the established researcher seeking a worthwhile resource in its material, its list of open research problems, and its somewhat enormous (often a fitting word when discussing Ramsey theory) bibliography.

Chapter 1 provides preliminary material (for example, the pigeonhole principle) and a brief introduction to the subject, including statements of three classical theorems of Ramsey theory: van der Waerden's theorem, Schur's theorem, and Rado's theorem. Chapter 2 covers van der Waerden's theorem; Chapters 3–7 deal with various topics related to van der Waerden's theorem; Chapter 8 is devoted to Schur's theorem and a generalization; Chapter 9 explores Rado's theorem; and Chapter 10 presents several other topics involving Ramsey theory on the integers.

The text provides significant latitude for those designing a syllabus for a course. The only material in the book on which other chapters depend is that through Section 2.2. Thus, other chapters or sections may be included or omitted as desired, since they are essentially independent of one another (except for an occasional reference to a previous definition or theorem). We do, however, recommend that all sections included in a course be studied in the same order in which they appear in the book.

Each chapter concludes with a section of exercises, a section of research problems, and a reference section. Since the questions contained in the Research Problem sections are still open, we cannot say with certainty how difficult a particular one will be to solve; some may actually be quite simple and inconsequential. The problems that we deem most difficult, however, are labeled with the symbol *. The reference section of each chapter is organized by section numbers (including the exercise section). The specifics of each reference are provided in the bibliography at the end of the book.

The material covered in this book represents only a portion of the subject area indicated by the book's title. Many additional topics have been investigated, and we have attempted to include at least references for these in the reference sections. Yet, for every problem that has been thought of in Ramsey theory, there are many more which that problem will generate and, given the great variety of combinatorial structures and patterns that lie in the set of integers, countless new problems wait to be explored.

We would like to thank Dr. Edward Dunne and the members of the AMS production staff for their assistance in producing this book. We also thank Tom Brown, Scott Gordon, Jane Hill, Dan Saracino, Dan Schaal, Ralph Sizer, and the AMS reviewers for their helpful comments and advice, which greatly improved the manuscript. We also express our gratitude to Ron Graham and Doron Zeilberger for their support of this project. We owe a big debt to the pioneers and masters of the field, especially Ron Graham, Jarik Nešetřil, Joel Spencer, Neil Hindman, Tom Brown, Timothy Gowers, Hillel Furstenberg, Vitaly Bergelson, Vojtěch Rödl, Endre Szemerédi, László Lovász (we had to stop somewhere), and of course Bartel van der Waerden, Issai Schur, Richard Rado, and Frank Ramsey. To all of the others who have contributed to the field of Ramsey theory on the integers, we extend our sincere appreciation. Finally, we want to acknowledge that this book would not exist without the essential contributions of the late Paul Erdős. But beyond the content of his achievements, he has personally inspired the authors as mathematicians. Our professional lives would have had far less meaning and fulfillment without his work and his presence in our field. For that pervasive, though perhaps indirect, contribution to this text, we are in his debt.

Chapter 1

Preliminaries

Unsolved problems abound, and additional interesting open questions arise faster than solutions to the existing problems. - F. Harary

The above quote, which appeared in the 1983 article "A Tribute to F. P. Ramsey," is at least as apropos today as it was then. In this book alone, which covers only a modest portion of Ramsey theory, you will find a great number of open research problems. The beauty of Ramsey theory, especially Ramsey theory dealing with the set of integers, is that, unlike many other mathematical fields, very little background is needed to understand the problems. In fact, with just a basic understanding of some of the topics in this text, and a desire to discover new results, the undergraduate mathematics student will be able to experience the excitement and challenge of doing mathematical research.

Ramsey theory is named after Frank Plumpton Ramsey and his eponymous theorem, which he proved in 1928 (it was published posthumously in 1930). So, what is Ramsey theory? Although there is no universally accepted definition of Ramsey theory, we offer the following informal description:

Ramsey theory is the study of the preservation of properties under set partitions.

In other words, given a particular set S that has a property P, is it true that whenever S is partitioned into finitely many subsets, one of the subsets must also have property P?

To illustrate further what sorts of problems Ramsey theory deals with, here are a few simple examples of Ramsey theory questions.

Example 1.1. Obviously, the equation x + y = z has a solution in the set of positive integers (there are an infinite number of solutions); for example, x = 1, y = 4, z = 5 is one solution. Here's the question: is it true that whenever the set of positive integers is partitioned into a finite number of sets S_1, S_2, \ldots, S_r , then at least one of these sets will contain a solution to x + y = z? The answer turns out to be yes, as we shall see later in this chapter.

Example 1.2. Is it true that whenever the set $\{1, 2, ..., 100\}$ is partitioned into two subsets A and B, then at least one of the two subsets contains a pair of integers which differ by exactly two? To answer this question, consider the partition consisting of

$$A = \{1, 5, 9, 13, \dots, 97\} \cup \{2, 6, 10, \dots, 98\}$$

and

 $B = \{3, 7, 11, \dots, 99\} \cup \{4, 8, 12, \dots, 100\}.$

We see that neither A nor B contains a pair of integers that differ by two, so that the answer to the given question is no.

Example 1.3. True or false: if there are 18 people in a group, then there must be either 4 people who are mutual acquaintances or 4 people who are mutual "strangers" (no two of whom have ever met)? (You will find the answer to this one later in the chapter.)

There is a wide range of structures and sets with which Ramsey theory questions may deal, including the real numbers, algebraic structures such as groups or vector spaces, graphs, points in the plane or in n dimensions, and others. This book limits its scope to Ramsey theory on the set of integers. (There is one exception – Ramsey's theorem itself – which is covered in this chapter.) In this chapter we introduce the reader to some of the most wellknown and fundamental theorems of Ramsey theory. We also present some of the basic terminology and notation that we will use.

1.1. The Pigeonhole Principle

Imagine yourself as a mailroom clerk in a mailroom with n slots in which to place the mail. If you have n + 1 pieces of mail to place into the n slots, what can we say about the amount of mail that will go into a slot? Well, we can't say much about the amount of mail a particular slot receives because, for example, one slot may get all of the mail. However, we can say that at least one slot must end up with at least two pieces of mail. To see this, imagine that you are trying to avoid having any slot with more than one piece of mail. By placing one piece of mail at a time into an unoccupied slot you can sort n pieces of mail. However, since there are more than n pieces of mail, you will run out of unoccupied slots before you are done. Hence, one slot must have at least two pieces of mail.

This simple idea is known as the pigeonhole principle, which can be stated this way:

If more than n pigeons are put into n pigeonholes, then some pigeonhole must contain at least two pigeons.

We now present the pigeonhole principle using somewhat more mathematical language.

Theorem 1.4 (Basic Pigeonhole Principle). If an *n*-element subset is partitioned into r disjoint subsets where n > r, then at least one of the subsets contains more than one element.

Example 1.5. In any class of 27 (or more) students, there must be two whose last name begins with the same letter.

Theorem 1.4 is a special case of the following more general principle.

Theorem 1.6 (Generalized Pigeonhole Principle). If more than mr elements are partitioned into r sets, then some set contains more than m elements.

1.2. Ramsey's Theorem

11.1

1. Preliminaries

Proof. Let S be a set with |S| > mr. Let $S = S_1 \cup S_2 \cup \cdots \cup S_r$ be any partition of S. Assume, for a contradiction, that $|S_i| \le m$ for all $i = 1, 2, \ldots, r$. Then

$$|S| = \sum_{i=1}^{r} |S_i| \le mr$$

a contradiction. Hence, for at least one *i*, the set S_i contains more than *m* elements, i.e., $|S_i| \ge m + 1$.

We see that Theorem 1.4 is a special case of Theorem 1.6 by taking m = 1. There are other common formulations of the pigeonhole principle; you will find some of these in the exercises.

Although the pigeonhole principle is such a simple concept, and seems rather obvious, it is a very powerful result, and it can be used to prove a wide array of not-so-obvious facts.

Here are some examples.

Example 1.7. For each integer n = 1, 2, ..., 200, let R(n) be the remainder when n is divided by 7. Then some value of R(n) must occur at least 29 times. To see this, we can think of the 200 integers as the pigeons, and the seven possible values of R(n) as the pigeonholes. Then, according to Theorem 1.6, since 200 > 28(7), one of the pigeonholes must contain more than 28 elements.

Example 1.8. We will show that within any sequence of $n^2 + 1$ integers there exists a monotonic subsequence of length n+1. (A sequence $\{x_i\}_{i=1}^m$ is called monotonic if it is either nondecreasing or nonincreasing). Let our sequence be $\{a_i\}_{i=1}^{n^2+1}$. For each $i \in \{1, 2, \ldots, n^2 + 1\}$, let ℓ_i be the length of the longest nondecreasing subsequence starting at a_i . If $\ell_i \geq n+1$ for some i we are done, so we may assume that $\ell_i \leq n$ for $1 \leq i \leq n^2 + 1$. Since each of the ℓ_i 's has a value between 1 and n, by the pigeonhole principle there exists $j \in \{1, 2, \ldots, n\}$ so that n+1 of the numbers ℓ_i equal j. Call these $\ell_{i_1}, \ell_{i_2}, \ldots, \ell_{i_{n+1}}$, where $i_1 < i_2 < \cdots < i_{n+1}$. Next, look at the subsequence. To see this, assume, for a contradiction, that it is not nonincreasing. Then $a_{i_k} < a_{i_{k+1}}$ for some k. Hence, the longest nondecreasing subsequence starting at a_{i_k} would be of length greater

than j, since there exists a subsequence of length j starting at $a_{i_{k+1}}$, a contradiction.

Next, we give another example for which the pigeonhole principle may not immediately appear to be applicable.

Example 1.9. Color each point in the xy-plane having integer coordinates either red or blue. We show that there must be a rectangle with all of its vertices the same color. Consider the lines y = 0, y = 1, and y = 2 and their intersections with the lines x = i, i = 1, 2, ..., 9. On each line x = i there are three intersection points colored either red or blue. Since there are only $2^3 = 8$ different ways to color three points either red or blue, by the pigeonhole principle two of the vertical lines, say x = j and $x = k \neq j$ must have the identical coloring (i.e. the color of (j, y) is the same as the color of (k, y) for y = 0, 1, 2). Using the pigeonhole principle again, we see that two of the points (j, 0), (j, 1), and (j, 2) must be the same color, say (j, y_1) and (j, y_2) . Then the rectangle with vertices $(j, y_1), (j, y_2), (k, y_1), \text{ and } (k, y_2)$ is the desired rectangle.

In the last example, we used colors as the "pigeonholes." Using colors to represent the subsets of a partition in this way is often convenient, and is quite typical in many areas of Ramsey theory.

1.2. Ramsey's Theorem

Ramsey's theorem can be considered a refinement of the pigeonhole principle, where we are not only guaranteed a certain number of elements in a pigeonhole, but we also have a guarantee of a certain relationship between these elements. It is a theorem that is normally stated in terms of the mathematical concept known as a graph. We will define what we mean by a graph very shortly, but before doing so, we consider the following example, known as the Party problem.

Example 1.10. We will prove the following: at a party of six people, there must exist either three people who have all met one another or three people who are mutual strangers (i.e., no two of whom have met). By the pigeonhole principle, we are guaranteed that for each person, there are three people that person has met or three people

4

that person has never met. We now want to show that there are three people with a certain relationship between them, namely, three people who all have met one another, or three people who are mutual strangers. First, assign to each pair of people one of the colors red or blue, with a red "line" connecting two people who have met, and a blue "line" connecting two people who are strangers. Hence, we want to show that for any coloring of the lines between people using the colors red and blue, there is either a red triangle or a blue triangle (with the people as vertices). Next, pick out one person at the party, say person X. Since there are five other people at the party, by the pigeonhole principle X either knows at least three people, or is a stranger to at least three people. We may assume, without loss of generality, that X knows at least 3 people at the party. Call these people A, B, and C. So far we know that the lines connecting X to each of A, B, and C are red. If there exists a red line between any of A, B, and C then we are done, since, for example, a red line between A and B would give the red triangle ABX. If the lines connecting A, B, and C are all blue, then ABC is a blue triangle.

Concerning the Party problem, another question we might ask is this: is 6 the lowest number of party members for the property we seek to hold? That is, does there exist a way to have five people at the party and not have either of the types of "triangles" discussed in the above example? To see that we cannot have only five people at the party and expect the same result, place five people in a circle and assume that each person knows the two people next to him/her, but no one else (draw a sketch to see that there is no red triangle and no blue triangle).

The fact that there is a solution to the Party problem is a special case of what is known as Ramsey's theorem. In order to state Ramsey's theorem we will use a few definitions from graph theory.

Definition 1.11. A graph G = (V, E) is a set V of points, called *vertices*, and a set E of pairs of vertices, called *edges*.

Definition 1.12. A subgraph G' = (V', E') of a graph G = (V, E) is a graph such that $V' \subseteq V$ and $E' \subseteq E$.

Definition 1.13. A complete graph on n vertices, denoted K_n , is a graph on n vertices, with the property that every pair of vertices is connected by an edge.

Definition 1.14. An *edge-coloring* of a graph is an assignment of a color to each edge of the graph. A graph which has been edge-colored is called a *monochromatic graph* if all of its edges are the same color.

We may now express the solution to the Party problem in graphtheoretical language. It says that for every 2-coloring, using the colors red and blue, of the edges of K_6 there must be either a red K_3 (a triangle) or a blue K_3 ; and furthermore, that there exists a 2-coloring of the edges of K_5 that fails to have this property.

We now state Ramsey's theorem for two colors.

Theorem 1.15 (Ramsey's Theorem for Two Colors). Let $k, \ell \geq 2$. There exists a least positive integer $R = R(k, \ell)$ such that every edgecoloring of K_R , with the colors red and blue, admits either a red K_k subgraph or a blue K_l subgraph.

Proof. First note that R(k, 2) = k for all $k \ge 2$, and $R(2, \ell) = \ell$ for all $\ell \ge 2$ (this is easy). We proceed via induction on the sum $k + \ell$, having taken care of the case when $k + \ell = 5$. Hence, let $k + \ell \ge 6$, with $k, \ell \ge 3$. We may assume that both $R(k, \ell - 1)$ and $R(k - 1, \ell)$ exist. We claim that $R(k, \ell) \le R(k - 1, \ell) + R(k, \ell - 1)$, which will prove the theorem.

Let $n = R(k - 1, \ell) + R(k, \ell - 1)$ and pick one vertex from K_n , call it v. Then there are n - 1 edges from v to the other vertices. Let A be the number of red edges and B be the number of blue edges coming out of v. Then, either $A \ge R(k - 1, \ell)$ or $B \ge R(k, \ell - 1)$, since if $A < R(k - 1, \ell)$ and $B < R(k, \ell - 1)$, then $A + B \le n - 2$, contradicting the fact that A + B = n - 1. We may assume, without loss of generality, that $A \ge R(k - 1, \ell)$. Let V be the set of vertices connected to v by a red edge, so that $|V| \ge R(k - 1, \ell)$. By the inductive hypothesis, K_V contains either a red K_{k-1} subgraph or a blue K_{ℓ} subgraph. If it contains a blue K_{ℓ} subgraph, we are done. If it contains a red K_{k-1} subgraph, then by connecting v to each vertex of this red subgraph we have a red K_k subgraph (since v is connected to V by only red edges), and the proof is complete.

The numbers $R(k, \ell)$ are known as the 2-color Ramsey numbers. The solution to the Party problem tells us that R(3,3) = 6. By Ramsey's theorem, we may extend the Party problem in various ways. For example, we know there exists a number n so that if there were n people at a party, then there would have to be either a group of four mutual acquaintances or a group of five mutual strangers. This number n is the Ramsey number R(4,5).

There are other ways to extend the Party problem. For example, in Exercise 1.11 we consider the case where people either love, hate, or are indifferent to, each other. In this situation we want to find three people who all love one another, three people who all hate one another, or three people who are all indifferent toward one another. Exercise 1.11 states that 17 people at the party will suffice (in fact 17 is the least such number with this property, but you cannot conclude this from Exercise 1.11). This is an example of a 3-color Ramsey number. More generally, Ramsey's theorem for two colors can easily be generalized to $r \geq 3$ colors (this is left as Exercise 1.17), in which case the Ramsey numbers are denoted by $R(k_1, k_2, \ldots, k_r)$. In case $k_i = k$ for $i = 1, \ldots, r$, we use the simpler notation $R_r(k)$. Thus, for example, in the "love-hate-indifferent" problem, we have R(3,3,3) = $R_3(3) = 17$.

The existence of the Ramsey numbers has been known since 1930. However, they are notoriously difficult to compute; the only known values are R(3,3) = 6, R(3,4) = 9, R(3,5) = 14, R(3,6) = 18, R(3,7) = 23, R(3,8) = 28, R(3,9) = 36, R(4,4) = 18, R(4,5) = 25, and R(3,3,3) = 17. (The fact that R(4,4) = 18 answers the question posed in Example 1.3.)

Obviously, Ramsey theory is named after Frank Ramsey. However, his famous theorem is the only result of Ramsey's in the field named after him. Unfortunately, Ramsey died of complications due to jaundice in 1930, a month before his 27^{th} birthday, but not before he left his mark.

1.3. Some Notation

In this section we go over some notation that we will frequently use.

We shall denote the set of integers by \mathbb{Z} , and the set of positive integers by \mathbb{Z}^+ . Most of our work will be confined to the set of integers. Hence, when speaking about an "interval" we will mean a set of the form $\{a, a + 1, \ldots, b\}$, where a < b are integers. Usually we will denote this interval more simply by [a, b].

When dealing with two sets X and Y, we will sometimes use the set S = X - Y, which we define to be the set of elements in X that are not in Y. Also, for S a set and a real number, a + S and aS will denote $\{a + s : s \in S\}$ and $\{as : s \in S\}$, respectively.

Sometimes we will find it convenient to use symbols such as 0,1, or 2 to stand for different "colors" rather than actual color names such as red or blue. We make this more formal in the following definition.

Definition 1.16. An *r*-coloring of a set S is a function $\chi : S \to C$, where |C| = r.

Typically, we will use $C = \{0, 1, ..., r-1\}$ or $C = \{1, 2, ..., r\}$. We can think of an *r*-coloring χ of a set *S* as a partition of *S* into *r* subsets $S_1, S_2, ..., S_r$, by associating the subset S_i with the set $\{x \in S : \chi(x) = i\}$.

The next definition will be used extensively.

Definition 1.17. A coloring χ is *monochromatic* on a set S if χ is constant on S.

Example 1.18. Let $\chi : [1,5] \rightarrow \{0,1\}$ be defined by $\chi(1) = \chi(2) = \chi(3) = 1$ and $\chi(4) = \chi(5) = 0$. Then χ is a 2-coloring of [1,5] that is monochromatic on $\{1,2,3\}$ and on $\{4,5\}$.

We will often find it convenient to represent a particular 2-coloring of an interval as a string of 0's and 1's. For example, the coloring in Example 1.18 could be represented by the string 11100. We may also abbreviate this coloring by writing $1^{3}0^{2}$. We may extend this notation to r-colorings for $r \geq 3$ by using strings with symbols belonging to the set $\{0, 1, 2, \ldots, r-1\}$. For example, define the 3-coloring χ on the interval [1,10] by $\chi(i) = 0$ for $1 \leq i \leq 5$, $\chi(i) = 1$ for $6 \leq i \leq 9$, and $\chi(10) = 2$. Then we may write $\chi = 0000011112$ or, equivalently, $\chi = 0^{5}1^{4}2$.

Sometimes we will want to describe the magnitude of functions asymptotically. For this purpose we mention two very commonly used symbols, called "Big-O" and "little-o."

Let f(n) and g(n) be functions that are nonzero for all n. We say that f(n) = O(g(n)) if there exist constants c, m > 0, independent of n, such that $0 < \left|\frac{f(n)}{g(n)}\right| \le c$ for all n > m. In other words, $\lim_{n\to\infty} \left|\frac{f(n)}{g(n)}\right| \le c$, if the limit exists. We say that f(n) = o(g(n)) if for all c > 0 there exists a constant m > 0, independent of n, such that $\left|\frac{f(n)}{g(n)}\right| < c$ for all n > m. In other words, $\lim_{n\to\infty} \left|\frac{f(n)}{g(n)}\right| = 0$.

If f and g are nonzero functions such that $\lim_{n\to\infty} \frac{f(n)}{g(n)}$ exists and is equal to ℓ , where $|\ell| \neq \infty$ and $\ell \neq 0$, then f(n) = O(g(n)). If $\ell = 0$ we have f(n) = o(g(n)). If $\ell = \infty$, then we have g(n) = o(f(n))by taking the reciprocal of the argument of the limit. Intuitively, if f(n) = O(g(n)) and g(n) = O(f(n)), then f(n) and g(n) have a similar growth rate; and if f(n) = o(g(n)), then f(n) is insignificant compared to g(n), for large n.

If f(n) and g(n) are functions with the same growth rate, i.e., if $\lim_{n\to\infty}\frac{f(n)}{g(n)}=1$, we may write $f(n)\sim g(n)$.

An example to explain these concepts is in order.

Example 1.19. Let $f(n) = \frac{n^2}{22} + 5n$ and $g(n) = n^2$. Then f(n) = O(g(n)), or, equivalently, $f(n) = O(n^2)$. We may also describe f(n)'s rate of growth by $f(n) = \frac{n^2}{22}(1 + o(1))$. To see this, we have

$$\frac{n^2}{22}(1+o(1)) = \frac{n^2}{22} + o(1)\frac{n^2}{22}.$$

Now, since $\frac{5n}{n^2/22} = \frac{110}{n}$ and $\lim_{n\to\infty} \frac{110/n}{1} = 0$, we have $5n = o(1)\frac{n^2}{22}$. We may also write $f(n) \sim \frac{n^2}{22}$ to describe the growth rate of f(n).

We will also use the following functions. For x a real number, we use $\lfloor x \rfloor$ to denote the greatest integer n such that $n \leq x$ (this is often called the "floor" function). The least integer function of a real number x, defined as the least integer n such that $n \geq x$, is denoted by $\lceil x \rceil$ (this is often referred to as the "ceiling function.")

1.4. Three Classical Theorems

Somewhat surprisingly, Ramsey's theorem was not the first, nor even the second, theorem in the area now known as Ramsey theory. The results that are generally accepted to be the earliest Ramsey-type theorems are due, in chronological order, to Hilbert, Schur, and van der Waerden. All of these results, which preceded Ramsey's theorem, deal with colorings of the integers, the theme of this book. Interestingly, even though Ramsey's theorem is a theorem about graphs, we will see later that it can be used to give some Ramsey-type results about the integers.

In this section we introduce three classical theorems concerning Ramsey theory on the integers. We will talk much more about each of these theorems in later chapters.

We start with a reminder of what an arithmetic progression is.

Definition 1.20. A k-term arithmetic progression is a sequence of the form a, a + d, a + 2d, ..., a + (k - 1)d, where $a \in \mathbb{Z}$ and $d \in \mathbb{Z}^+$.

We now state van der Waerden's theorem, which was proved in 1927.

Theorem 1.21 (Van der Waerden's Theorem). For all positive integers k and r, there exists a least positive integer w(k;r) such that for every r-coloring of [1, w(k;r)] there is a monochromatic arithmetic progression of length k.

The numbers w(k; r) are known as the van der Waerden numbers. Let's look at a simple case. Let k = r = 2. Hence, we want to find the minimum integer w = w(2; 2) so that no matter how we partition the interval $[1, w] = \{1, 2, \ldots, w\}$ into two subsets (i.e., 2-color [1, w]), we must end up with at least one of the two subsets containing a pair of elements a, a+d, where $d \ge 1$ (i.e., we must end up with a monochromatic 2-term arithmetic progression). Consider a 2-coloring of $\{1, 2\}$ where 1 and 2 are assigned different colors. Obviously, under a such a coloring, $\{1, 2\}$ does not contain a 2-term arithmetic progression that is monochromatic. Thus, w(2; 2) is not equal to 2 (not every 2-coloring of [1, 2] yields the desired monochromatic sequence). Does 3 work? That is, does every 2-coloring of [1, 3] yield a monochromatic

1. Preliminaries

2-term arithmetic progression. The answer is yes, by a simple application of the pigeonhole principle, since any 2-element set of positive integers is a 2-term arithmetic progression. Thus, we have shown that w(2;2) = 3.

Finding w(2; 2) was rather simple. All the van der Waerden numbers w(2; r) are just as easy to find (we leave this as an exercise in a later chapter). For $k \geq 3$, the evaluation of these numbers very quickly becomes much more difficult. In fact, the only known van der Waerden numbers are w(3; 2) = 9, w(3; 3) = 27, w(3; 4) = 76, w(4; 2) = 35 and w(5; 2) = 178. Besides trying to find exact values of the van der Waerden numbers, there is another open question that has been one of the most difficult, and most appealing, problems in Ramsey theory. Namely, finding a reasonably good estimate of w(k; r) in terms of k and r. We shall talk more about such questions, and the progress that has been made on them, in Chapter 2.

Van der Waerden's theorem has spawned many results in Ramsey theory. For this reason, and because the notion of an arithmetic progression is such a natural and simple concept, a large portion of this book is dedicated to various offshoots, refinements, extensions, and generalizations of van der Waerden's theorem.

The next two main results deal with solutions to equations and systems of equations. Let \mathcal{E} represent a given equation or system of equations. We call (x_1, x_2, \ldots, x_k) a monochromatic solution to \mathcal{E} if x_1, x_2, \ldots, x_k are all the same color and they satisfy \mathcal{E} .

We next theorem we present, proved by Issai Schur in 1916, is one of the earliest results in Ramsey theory.

Theorem 1.22 (Schur's Theorem). For any $r \ge 1$, there exists a least positive integer s = s(r) such that, for any r-coloring of [1, s], there exists a monochromatic solution to x + y = z.

The numbers s(r) are called the Schur numbers. As a simple example, we look at s(2). Here we want the least positive integer s so that whenever [1, s] is 2-colored, there will exist integers x, y, z (not necessarily distinct) satisfying x + y = z. Notice that s(2) must be greater than four, because if we take the 2-coloring χ of [1,4] defined by $\chi(1) = \chi(4) = 0$ and $\chi(2) = \chi(3) = 1$, then it is not possible to

1.4. Three Classical Theorems

find x, y, and z all of the same color satisfying x + y = z. Meanwhile, every 2-coloring of [1,5] (there are $2^5 = 32$ of them) does yield such a monochromatic triple (this is proved in Example 8.5). Thus, s(2) = 5.

As it turns out, the only Schur numbers that are currently known are s(1) = 2, s(2) = 5, s(3) = 14, and s(4) = 45. We will learn much more about Schur's theorem in Chapter 8.

The third classical theorem we mention is Rado's theorem, which is a generalization of Schur's theorem. In fact, Richard Rado was a student of Schur. The idea of Rado's theorem may be described as follows. Thinking of Schur's theorem as a theorem about the homogeneous linear equation x + y - z = 0, we ask the following more general question. Which systems, \mathcal{L} , of homogeneous linear equations with integer coefficients have the following property: for every $r \ge 1$, there exists a least positive integer $n = n(\mathcal{L}; r)$ such that every *r*-coloring of [1, n] yields a monochromatic solution to \mathcal{L} ?

In a series of articles published in the 1930's, Rado completely answered this question. Since Rado's theorem, in its most general form, is a bit complicated to describe, we will postpone stating the general theorem until Chapter 9, which is devoted to Rado's theorem. Instead, we mention here the special case of Rado's theorem in which the system consists of only a single equation.

We first need the following definition.

Definition 1.23. For $r \geq 1$, a linear equation \mathcal{E} is called *r*-regular if there exists $n = n(\mathcal{E}; r)$ such that for every *r*-coloring of [1, n] there is a monochromatic solution to \mathcal{E} . It is called *regular* if it is *r*-regular for all $r \geq 1$.

Example 1.24. Using Definition 1.23, Schur's theorem can be stated as "the equation x + y = z is regular."

We now state Rado's theorem for a single equation.

Theorem 1.25 (Rado's Single Equation Theorem). Let \mathcal{E} represent the linear equation $\sum_{i=1}^{n} c_i x_i = 0$, where $c_i \in \mathbb{Z} - \{0\}$ for $1 \le i \le n$. Then \mathcal{E} is regular if and only if some nonempty subset of the c_i 's sums to 0.

1. Preliminaries

Example 1.26. The equation x + y = z, i.e., x + y - z = 0, satisfies the requirements of Theorem 1.25. Hence, as noted before and proved by Schur, x + y = z is regular.

Example 1.27. It follows from Rado's theorem that the equation $3x_1 + 4x_2 + 5x_3 - 2x_4 - x_5 = 0$ is regular, since the sum of the first, fourth, and fifth coefficients is 0.

1.5. A Little More Notation

The three classical theorems mentioned above all have a somewhat similar flavor. That is, they have the following general form: there exists a positive integer n(r) such that for every r-coloring of [1, n(r)] there is a monochromatic set belonging to a particular family of sets. In one case, the family of sets was the k-term arithmetic progressions; in another case the family consisted of all solutions to a certain equation; and so on. Throughout this book we will be looking at this type of problem, and so it will be worthwhile to have a general notation that can be used for any such problem.

Let \mathcal{F} be a certain family of sets, and let k and r be positive integers. We denote by $R(\mathcal{F}, k; r)$ the least positive integer, if it exists, such that for any r-coloring of $[1, R(\mathcal{F}, k; r)]$, there is a monochromatic k-term member of \mathcal{F} . In the case where no such integer exists, we say $R(\mathcal{F}, k; r) = \infty$. Because our discussion will often be confined to the situation in which the number of colors is two, we often denote the function $R(\mathcal{F}, k; 2)$ more simply as $R(\mathcal{F}, k)$. If the length of the sequence is understood (as in Schur's theorem), we write $R(\mathcal{F}; r)$.

For certain Ramsey-type functions we deal with, it will be convenient to use a notation other than $R(\mathcal{F}, k; r)$. For example, later in the book we will encounter a type of sequence called a descending wave, for which we will use the notation DW(k; r) rather than something like R(DW, k; r) (where DW would represent the family of all descending waves). Similarly, since the notation w(k; r) is so standard, we will use w(k; r) instead of R(AP, k; r) and w(k) instead of R(AP, k), where AP is the family of all arithmetic progressions.

1.5. A Little More Notation

family \mathcal{F} here).

Finally, we remark that the notation $R(k, \ell)$ is reserved for the classical Ramsey numbers defined in Section 1.2 (note the absence of a

Throughout this book we will be considering various collections, \mathcal{F} , of sets of integers and, as with the three classical theorems of Section 1.4, wanting to know if, for a specified value of r and a particular set $M \subseteq \mathbb{Z}$, every r-coloring of M yields a monochromatic member of \mathcal{F} . For the case in which M is the set of positive integers, we have the following definition.

Definition 1.28. Let \mathcal{F} be a family of finite subsets of \mathbb{Z}^+ , and let $r \geq 1$. If for every *r*-coloring of \mathbb{Z}^+ and all $k \geq 1$, there is a monochromatic *k*-element member of \mathcal{F} , then we say that \mathcal{F} is *r*-regular. If \mathcal{F} is *r*-regular for all *r*, we say that \mathcal{F} is regular.

Sometimes we will replace the phrase "for all $k \geq 1$, there is a monochromatic k-element member of \mathcal{F} " by "there are arbitrarily large members of \mathcal{F} ."

Example 1.29. Let $\mathcal{F} = AP$, the collection of all arithmetic progressions. By van der Waerden's theorem, \mathcal{F} is regular since for every finite coloring of \mathbb{Z}^+ , there exists, for every $k \geq 1$, a monochromatic *k*-term arithmetic progression.

Whereas Definition 1.28 pertains to *all* colorings of a set, we will also want to consider whether or not a particular coloring of a set M yields a monochromatic member of the collection \mathcal{F} . For this we have the next definition.

Definition 1.30. Let \mathcal{F} be a family of subsets of \mathbb{Z} and let k be a positive integer. Let $r \geq 1$. An r-coloring of a set $M \subseteq \mathbb{Z}$ is called $(\mathcal{F}, k; r)$ -valid if there is no monochromatic k-element member of \mathcal{F} contained in M.

When the number of colors is understood, we may simply say that a coloring is (F, k)-valid. Also, when there is no possible confusion as to the meaning of \mathcal{F} or the value of k, we may simply say that a coloring is *valid*.

As an example, if \mathcal{F} is the family of sets of even numbers, then the 2-coloring of [1, 10] represented by the binary sequence 1110001110 is

1.6. Exercises

 $(\mathcal{F}, 4)$ -valid since there is no monochromatic 4-term sequence belonging to \mathcal{F} (i.e., there do not exists four even numbers that have the same color).

Let's consider another example.

Example 1.31. Let \mathcal{F} be the family of all subsets of \mathbb{Z}^+ . We will determine a precise formula for $R(\mathcal{F}, k; r)$. First, let χ be any r-coloring of [1, r(k-1)+1]. By the generalized pigeonhole principle, since we are partitioning a set of r(k-1)+1 elements into r sets, there must be, for some color, more than k-1 elements of that color. Thus, under χ , there is a monochromatic k-element member of \mathcal{F} . Since χ is an arbitrary r-coloring, we have that $R(\mathcal{F}, k; r) \leq r(k-1)+1$. On the other hand, there do exist r-colorings of the interval [1, r(k-1)] that are (\mathcal{F}, k) -valid. Namely, assign exactly k-1 members of the interval to each of the colors. Then no color will have a k-element member of \mathcal{F} . Thus, $R(\mathcal{F}, k; r) \geq r(k-1)+1$, and hence $R(\mathcal{F}, k; r) = r(k-1)+1$.

The fact that in Example 1.31 the numbers $R(\mathcal{F}, k; r)$ always exist is not very surprising. After all, \mathcal{F} is so plentiful that it is easy to find a monochromatic member. When the family of sets we are considering is not as "big," the behavior of the associated Ramsey function is much less predictable. For certain \mathcal{F} we will find that $R(\mathcal{F}, k; r) < \infty$ for all k and r, while for others this will only happen (for all k) provided r does not exceed a certain value. There will even be cases where $R(\mathcal{F}, k; r)$ never exists except for a few small values of k and r.

We will encounter many different results in this book, but the common thread will be an attempt to find answers, to whatever extent we can, to the following two questions.

1. For which \mathcal{F} , k, and r does the function $R(\mathcal{F}, k; r)$ exist?

2. If $R(\mathcal{F}, k; r)$ exists, what can we say about its magnitude?

1.6. Exercises

1.1 A bridge club has 10 members. Every day, four members of the club get together and play one game of bridge. Prove that

after two years, there is some particular set of four members that has played at least four games of bridge together.

- 1.2 Prove that if the numbers $1, 2, \ldots, 12$ are randomly positioned around a circle, then some set of three consecutively positioned numbers must have a sum of at least 19.
- 1.3 Prove the following versions of the pigeonhole principle.
 a) If a₁, a₂,..., a_n, c are real numbers such that ∑_{i=1}ⁿ a_i ≥ c, then there is at least one value of i such that a_i ≥ c/n.
 b) If a₁, a₂,..., a_n are integers, and c is a real number such that ∑_{i=1}ⁿ a_i ≥ c, then there is at least one value of i such that a_i ≥ [c/n].
- 1.4 With regard to Example 1.8, show that, given a sequence of only n^2 numbers, there need not be a monotonic subsequence of length n + 1.
- **1.5** Let $r \ge 2$. Show that there exists a least positive integer M = M(k;r) so that any *r*-coloring of M integers admits a monochromatic monotonic *k*-term subsequence. Determine M(k;r). (Note that from Example 1.8, $M(k;1) = k^2 + 1$.)
- **1.6** Let $r \geq 3$. Let χ be any r-coloring of the set

 $S = \{(x, y) \in \mathbb{R}^2 : x, y \in \mathbb{Z}\}$

(the members of S are sometimes called lattice points). Show that, under χ , there must exist a rectangle with all vertices the same color.

- **1.7** Explain how the Party problem fits the description of Ramsey theory offered on page 1.
- **1.8** Since R(3,3) = 6, we know that any 2-coloring of K_6 must admit at least one monochromatic triangle. In fact, any 2-coloring of K_6 must admit at least two monochromatic triangles. Prove this fact.
- **1.9** Show that any 2-coloring of K_7 must admit at least four monochromatic triangles.
- **1.10** Generalize Exercises 1.8 and 1.9 above to K_n . (Hint: Let $r_i, i = 1, 2, ..., n$, be the number of red edges connected to vertex *i*. Show that the number of monochromatic triangles

1. Preliminaries

is thus $\binom{n}{3} - \frac{1}{2} \sum_{i=1}^{n} r_i(n-1-r_i)$. Minimize this function to deduce the result.)

- 1.11 Show that any 3-coloring of K_{17} must admit at least one monochromatic triangle, via an argument similar to the one showing $R(3,3) \leq 6$, and using the fact that R(3,3) = 6.
- **1.12** Explain why $R(k, \ell) = R(\ell, k)$.
- **1.13** Prove that $R(k, \ell) < R(k-1, \ell) + R(k, \ell-1)$, if both $R(k-1, \ell)$ and $R(k, \ell 1)$ are even.
- **1.14** Show that $R(k, \ell) \leq {\binom{k+\ell-2}{k-1}}$ by showing that the recurrence $R(k, \ell) = R(k-1, \ell) + R(k, \ell-1)$ is satisfied by a certain binomial coefficient.
- 1.15 We can determine a lower bound for R(k, k) by using the probabilistic method (largely due to Erdős). Show that R(k, k) > k/e√2 2^{k/2} for large k via the following steps.
 a) Randomly color the edges of K_n either red or blue, i.e., each edge is colored red with probability 1/2. Show that for a given set of k vertices of K_n, the probability that the complete

graph on these k vertices is monochromatic equals $\frac{2}{\binom{k}{2}}$

b) Let p_k be the probability that a monochromatic K_k subgraph exists in our random coloring. Show that

$$p_k \le \sum_{i=1}^{\binom{n}{k}} 2^{1-\binom{k}{2}} = \binom{n}{k} 2^{1-\binom{k}{2}}$$

c) Use (b) to show that if $\binom{n}{k}2^{1-\binom{k}{2}} < 1$, then R(k,k) > n. d) Stirling's formula for the asymptotic behavior of n! says that $n! \sim \sqrt{2\pi n} \left(\frac{n}{e}\right)^n$. Use Stirling's formula to finish the problem.

1.16 Consider the following way to color the edges of K_n . Number the vertices of K_n in a counterclockwise fashion from 1 to n. Next, partition the numbers $\{1, 2, \ldots, n-1\}$ into two subsets. Call these sets R and B for red and blue. Now, each edge has two vertices, say i and j. Calculate |j - i| for that edge. If $|j - i| \in R$, then color the edge connecting i and j red. If $|j - i| \in B$, then color the edge connecting i and j blue. Such

a coloring is called a *difference coloring*. Since R(3, 4) = 9, we know that there is an edgewise 2-coloring of K_8 with no red K_3 and no blue K_4 . One such coloring is a difference coloring defined as follows. Color an edge red if $|j - i| \in \{1, 4, 7\}$, and blue if $|j - i| \in \{2, 3, 5, 6\}$. Show that this coloring does indeed prove that R(3, 4) > 8, i.e., that there is no red K_3 and no blue K_4 .

- **1.17** Prove Ramsey's theorem for r colors, where $r \geq 3$.
- **1.18** Show that $s(3) \ge 14$, where s(r) is the r-color Schur number.
- **1.19** Let $r \ge 1$. Show that for any integer a, there exists an integer M = M(a; r) such that, for any r-coloring of [1, M], there is a monochromatic solution to x + ay = z. Deduce Schur's theorem from this result.

1.7. Research Problems

Note: in this chapter and the next we present some problems which, although understandable and rather simple to state, are considered to be extremely difficult to solve. We include them primarily for illustrative purposes (and because they are intriguing problems). We suggest that the research problems from Chapters 3 through 10 are more suitable for beginning research in Ramsey theory.

*1.1 For $n \ge 3$, define g(n) to be the least positive integer with the following property. Whenever the set of lines through g(n) points satisfy both (a) no two lines are parallel, and (b) no three lines intersect in the same point, then the set of g(n) points contains the vertices of a convex *n*-gon. Prove or disprove: $g(n) = 2^{n-2} + 1$. It is known to hold for n = 3, 4, 5. It is also known that the existence of g(n) is equivalent to Ramsey's theorem.

References: [42], [90], [103]

- *1.2 Prove or disprove the following conjecture proposed by Paul Erdős and V. Sós: $R(3, n + 1) R(3, n) \to \infty$ as $n \to \infty$. References: [12], [89], [155]
- *1.3 Determine $\lim_{n\to\infty} R(n,n)^{1/n}$ if it exists. It is known that if this limit exists, then it is between $\sqrt{2}$ and 4. (The lower

18

bound comes from Exercise 1.15 and the upper bound is deduced from Exercise 1.14 using Stirling's formula, which is given in Exercise 1.15.) References: [99], [255], [256]

1.8. References

Harary's quote appears in [138], which also contains a biographical sketch of Ramsey.

§1.1. Example 1.8 is due to Erdős and Szekeres [101].

§1.2. Theorem 1.15 is proved in [218]. There is a much more general form of Ramsey's theorem (not confined to edge-colorings); for a proof, see, for example, [127]. Erdős and Szekeres rediscovered Ramsey's theorem, in an equivalent form, in [101] (see Research Problem 1.1). The Ramsey number R(4,5) = 25, discussed after Theorem 1.15, is the most recently discovered Ramsey number. See [187] for details. The Ramsey numbers R(3, 4), R(3, 5), R(4, 4), and R(3, 3, 3) were discovered by Greenwood and Gleason [132]. Kéry determined R(3, 6) in [154]. Graver and Yackel [130] determine R(3, 7) by matching the upper bound given by Kalbfleisch in [153]. McKay and Min [186] determined R(3, 8) by matching the lower bound given by Grinstead and Roberts [133]. Grinstead and Roberts [133] determined R(3, 9) by matching the lower bound given in [153]. For a survey of the best bounds to date on small Ramsey numbers see [216].

§1.4. Hilbert's result is in [142]. A proof using more modern language is in [124, p. 1368]. Recent work regarding Hilbert's theorem can be found in [57] and [134]. Van der Waerden's theorem and its original proof are in [270]. Schur's theorem and its original proof can be found in [247]. Rado's theorem was proved in the series of papers [215], [214], and [213]. For a summary of Rado's theorem see [211].

§1.7. The original work on g(n) from Research Problem 1.1 can be found in [90].

Additional References: Another brief account of the life and work of Ramsey can be found in [188]. The book [217] contain an account of Ramsey's work as it pertains to philosophy and logic. A brief history of Ramsey theory is given by Spencer in [258]. Chapter 2

Van der Waerden's Theorem

Perhaps the most fundamental Ramsey-type theorem on the integers is van der Waerden's theorem concerning arithmetic progressions. Loosely, it says that for any given coloring of \mathbb{Z}^+ , monochromatic arithmetic progressions cannot be avoided.

Let's consider arithmetic progressions of length three. We wish to find the least positive integer w such that regardless of how the integers $1, 2, \ldots, w$ are colored, using two colors, there will be a monochromatic 3-term arithmetic progression. This number w, denoted by w(3; 2) (or w(3)), is called a van der Waerden number. Using the notation introduced in Chapter 1, we may also denote it by R(AP, 3; 2), where AP is the family of all arithmetic progressions. Before finding w, we describe the standard methodology for finding the exact value of any particular Ramsey-type number $R(\mathcal{F}, k; r)$. The goal is to show that some number serves both as a lower bound and an upper bound for the minimum number we desire.

> (a) To establish that a certain value v is a lower bound for a specific Ramsey-type number $R(\mathcal{F}, k; r)$, it suffices to find *some* r-coloring of [1, v - 1] that yields no monochromatic k-element member of \mathcal{F} .

2. Van der Waerden's Theorem

2. Van der Waerden's Theorem

(b) To establish that v serves as an upper bound for $R(\mathcal{F}, k; r)$, it is necessary to show that every rcoloring of [1, v] yields a monochromatic k-element member of \mathcal{F} .

Back to the determination of w: we will establish that w = 9 by using the above method to prove that $w \ge 9$ and $w \le 9$.

According to (a), to show that $w \ge 9$ it suffices to exhibit a 2-coloring of [1,8] with no monochromatic 3-term arithmetic progression. One such coloring is the following: color 1, 4, 5, and 8 red, and color 2, 3, 6, and 7 blue. It is easy to check that this coloring avoids 3-term monochromatic arithmetic progressions.

To show that $w \leq 9$, we must show that every 2-coloring of [1,9] admits a monochromatic 3-term arithmetic progression. Assume, for a contradiction, that there exists a 2-coloring of [1,9] with no monochromatic 3-term arithmetic progression. Using red and blue as the colors, consider the possible ways in which the integers 3 and 5 may be colored. Can both 3 and 5 be red? If they were, then since (1,3,5) cannot be monochromatic, 1 must be blue. Likewise, since neither (3,4,5) nor (3,5,7) can be red, 4 and 7 must be blue. This situation is not possible because now (1,4,7) is blue. Thus we may conclude that 3 and 5 cannot both be red. The same argument, with the colors reversed, shows that 3 and 5 cannot both be blue. Hence, 3 and 5 are of different colors. Similarly, 5 and 7 cannot have the same color, and 4 and 6 cannot have the same color (explain why).

Without loss of generality, we assume the color of 3 is red. By the observations above, this leaves

$$\chi_1 = (\text{red, red, blue, blue, red})$$

and

 $\chi_2 = (red, blue, blue, red, red)$

as the only possible colorings of (3, 4, 5, 6, 7). If χ_1 is the coloring of (3, 4, 5, 6, 7) then, because of (2, 3, 4), the color of 2 must be blue. Then, because of (2, 5, 8), 8 must be red. Because of (1, 4, 7), 1 is blue. Finally, because of (1, 5, 9), 9 is red. From this we have that (7, 8, 9) is red, contradicting our assumption. Since χ_2 is the reverse of χ_1 , a symmetric argument will show that χ_2 also leads us to a contradiction. Thus, every 2-coloring of [1,9] yields a monochromatic arithmetic progression of length 3.

Now that we know that w exists (i.e., that w is finite), it is natural to ask whether the analogous least positive integer exists if we use more than two colors and/or require longer arithmetic progressions. The answer, as it turns out, is yes. This fact is known as van der Waerden's theorem. We start by presenting what is usually called the *finite* version of van der Waerden's theorem.

Theorem 2.1 (Van der Waerden's Theorem). Let $k, r \ge 2$ be integers. There exists a least positive integer w = w(k;r) such that for all $n \ge w$, for every r-coloring of [1,n] there is a monochromatic arithmetic progression of length k.

Van der Waerden's theorem is one of the most fundamental results in the area of Ramsey theory. However, because its proof can be somewhat difficult to follow, we postpone the proof until later in this chapter.

To help reinforce what van der Waerden's theorem says, we consider an application.

Example 2.2. Let a, b, k, and r be fixed positive integers. We use van der Waerden's theorem to show that every r-coloring of the set $\{a, a+b, a+2b, \ldots a+(w(k;r)-1)b\}$ admits a monochromatic k-term arithmetic progression. Let

$$\chi: \{a, a+b, \dots a+(w(k;r)-1)b\} \to \{0, 1, \dots, r-1\}$$

be any r-coloring, and define $\chi' : [1, w(k; r)] \to \{0, 1, \dots, r-1\}$ by $\chi'(j) = \chi(a + (j-1)b)$. By van der Waerden's theorem, χ' admits a monochromatic k-term arithmetic progression, say

$$\{c, c+d, \ldots, c+(k-1)d\}.$$

Hence $\{a + cb, a + (c + d)b, \ldots, a + (c + (k - 1)d)b\}$ is monochromatic under χ , by the definition of χ' . Rewriting this set, we have

$$\{(a+bc), (a+bc)+bd, (a+bc)+2(bd), \dots, (a+bc)+(k-1)(bd)\},\$$

the desired monochromatic k-term arithmetic progression.

2. Van der Waerden's Theorem

As we shall see, there are several other forms of van der Waerden's theorem that, although it is not at all obvious at first glance, are equivalent to Theorem 2.1 (the finite version). One alternate form is the statement that under any finite coloring of \mathbb{Z}^+ there exist arbitrarily long monochromatic arithmetic progressions. That this is equivalent to Theorem 2.1 requires some explanation.

Note that the existence of arbitrarily long monochromatic arithmetic progressions under a given coloring of \mathbb{Z}^+ does *not* imply that infinitely long monochromatic arithmetic progressions exist. What it does say is that for each *finite* number k we can find a monochromatic arithmetic progression of length k. We look at an example.

Example 2.3. Consider the following 2-coloring of \mathbb{Z}^+ , with the colors 0 and 1:

$$\underbrace{1}_{1} \underbrace{00}_{2} \underbrace{1111}_{4} \underbrace{00 \dots 0}_{8} \underbrace{11 \dots 1}_{16} 00 \dots$$

i.e., for $j \ge 0$, the interval $I_j = [2^j, 2^{j+1} - 1]$ is colored 1 if j is even, and colored 0 if j is odd. It is clear that for any k, there exists a monochromatic arithmetic progression of length k (take k consecutive integers in I_k , which has length 2^k). Thus, under this coloring there are arbitrarily long monochromatic arithmetic progressions.

We next show that there is no monochromatic arithmetic progression of infinite length. Assume that $A = \{a, a + d, a + 2d, ...\}$ is an infinitely long arithmetic progression. Then there is some n such that $2^n > d$ and $A \cap I_n$ is not empty. Since $d < 2^n$, we know that $A \cap I_{n+1}$ is also not empty. Since the color of I_n is different from the color of I_{n+1} , A is not monochromatic.

Now that we have clarified what we mean by arbitrarily long arithmetic progressions, we next show why the following two statements are equivalent: (a) every finite coloring of \mathbb{Z}^+ admits arbitrarily long monochromatic arithmetic progressions; and (b) w(k; r) exist for all k and r. This is the subject of the next section.

2.1. The Compactness Principle

2.1. The Compactness Principle

The compactness principle, also known as Rado's selection principle, in its full version, is beyond the scope of this book. We will use a simpler version of this principle and refer to it simply as the compactness principle.

The compactness principle, in very general terms, is a way of going from the infinite to the finite. It gives us a "finite" Ramseytype statement provided the corresponding "infinite" Ramsey-type statement is true. For example, we may conclude the "finite" version of van der Waerden's theorem for two colors:

> For all $k \geq 2$, there exists a least integer n = w(k)such that for every 2-coloring of [1, m], $m \geq n$, there is a monochromatic k-term arithmetic progression,

from the "infinite" version:

For every 2-coloring of \mathbb{Z}^+ , there are, for every $k \geq 2$, monochromatic k-term arithmetic progressions.

An alternative way to state the above "infinite" version of van der Waerden's theorem (using 2 colors) is to say that every 2-coloring of \mathbb{Z}^+ admits arbitrarily long monochromatic arithmetic progressions.

Before stating the compactness principle, we remark here that this principle does not give us any bound for the minimum number in the "finite" version; it only gives us its existence. The proof we give may seem somewhat familiar, as it is essentially what is known as Cantor's diagonal argument, the standard argument used to prove that the set of real numbers is uncountable.

Theorem 2.4 (The Compactness Principle). Let $r \ge 2$ and let \mathcal{F} be a family of finite subsets of \mathbb{Z}^+ . Assume that for every r-coloring of \mathbb{Z}^+ there is a monochromatic member of \mathcal{F} . Then there exists a least positive integer $n = n(\mathcal{F}; r)$ such that, for every r-coloring of [1, n], there is a monochromatic member of \mathcal{F} .

Proof. Let $r \ge 2$ be fixed and assume that every *r*-coloring of \mathbb{Z}^+ admits a monochromatic member of \mathcal{F} . Assume, for a contradiction,

that for each $n \ge 1$ there is an *r*-coloring

$$\chi_n: [1, n] \to \{0, 1, \dots, r-1\}$$

with no monochromatic member of \mathcal{F} . We proceed by constructing a specific *r*-coloring, χ , of \mathbb{Z}^+ .

Among $\chi_1(1), \chi_2(1), \ldots$ there must be some color that occurs an infinite number of times. Call this color c_1 and let $\chi(1) = c_1$. Now let \mathcal{T}_1 be the collection of all colorings χ_j with $\chi_j(1) = c_1$. Within the set of colors $\{\chi_j(2) : \chi_j(2) \in \mathcal{T}_1\}$, there must be some color c_2 that occurs an infinite number of times. Let $\chi(2) = c_2$ and let \mathcal{T}_2 be the collection of all colorings $\chi_j \in \mathcal{T}_1$ with $\chi_j(2) = c_2$. Continuing in this fashion, we can find, for each $i \geq 2$, some color c_i such that the family of colorings

$$\mathcal{T}_i = \{\chi_j \in \mathcal{T}_{i-1} : \chi_j(i) = c_i\}$$

is infinite. We define $\chi(i) = c_i$ for $i = 2, 3, \ldots$ The resulting coloring $\chi : \mathbb{Z}^+ \to \{0, 1, \ldots, r-1\}$ has the property that for every $k \ge 1, \mathcal{T}_k$ is the collection of colorings χ_i with $\chi(x) = \chi_i(x)$ for $x = 1, 2, \ldots, k$.

By assumption, χ admits a monochromatic member of \mathcal{F} , say S. Let $m = \max\{s : s \in S\}$. By construction, for every $\tau \in \mathcal{T}_m$ we have S monochromatic under τ . This contradicts our assumption that all of the χ_n 's avoid monochromatic members of \mathcal{F} .

It is clear that for m > n, if χ is an *r*-coloring of [1, n] that yields a monochromatic member of a specific family \mathcal{F} , and χ' is any extension of χ to [1, m], then χ' also yields a monochromatic member of \mathcal{F} . Thus, the conclusion of Theorem 2.4 could be replaced by the stronger-sounding (but equivalent) wording "there exists a least positive integer $n = n(\mathcal{F}; r)$ such that for every $m \ge n$ and every *r*-coloring of [1, m], there is a monochromatic member of \mathcal{F} ."

Note also that the converse of Theorem 2.4 is true. This is because if it is true that $n(\mathcal{F}; r)$ exists, then certainly it is the case that every *r*-coloring of \mathbb{Z}^+ yields a monochromatic member of \mathcal{F} .

By taking \mathcal{F} to be the family of all arithmetic progressions (of all finite lengths), we see, by the compactness principle, that if for each k, every r-coloring of \mathbb{Z}^+ admits a monochromatic arithmetic progression of length k, then w(k; r), as defined in Theorem 2.1, exists.

2.2. Alternate Forms of van der Waerden's Theorem 27

We remark here that we will often use the compactness principle or its converse without explicitly stating that we are doing so. Thus, we may use either the finite or infinite version of a particular Ramseytype theorem according to convenience.

2.2. Alternate Forms of van der Waerden's Theorem

In the last section we encountered two equivalent forms of van der Waerden's theorem (the "finite" and "infinite" versions). There are, in fact, several equivalent forms of van der Waerden's theorem. We state some of these in the following theorem.

Theorem 2.5. The following statements are equivalent.

(i) For $k \geq 2$, any 2-coloring of \mathbb{Z}^+ admits a monochromatic arithmetic progression of length k.

(ii) For $k \geq 2$, w(k; 2) exists.

(iii) For $k, r \geq 2$, w(k; r) exists.

(iv) Let $r \ge 2$. For any r-coloring of \mathbb{Z}^+ and any finite subset $S = \{s_1, s_2, \ldots, s_n\} \subseteq \mathbb{Z}^+$, there exist integers $a, d \ge 1$ such that $a + dS = \{a + s_1d, a + s_2d, \ldots, a + s_nd\}$ is monochromatic.

(v) For $k, r \geq 2$, any r-coloring of \mathbb{Z}^+ admits a monochromatic arithmetic progression of length k.

(vi) For $k \ge 2$, any infinite set of positive integers, $S = \{s_i\}_{i\ge 0}$, for which $c = \max\{|s_{i+1} - s_i| : i \ge 0\}$ exists, must contain an arithmetic progression of length k.

Proof. (i) \Rightarrow (ii) by the compactness principle.

Next, we prove (ii) \Rightarrow (iii) by induction on r. Statement (ii) is the initial case (r = 2) of (iii), so now let $r \ge 3$ and assume that w(k; r - 1) exists for all k. We will show that w(k; r) exists for all k. Let m = w(w(k; r - 1); 2) and let χ be an arbitrary r-coloring of [1, m], where the colors are red, $blue_1$, $blue_2$, $blue_3$, ..., $blue_{r-1}$. We prove $w(k; r) \le m$ by showing that χ admits a monochromatic k-term arithmetic progression. If we were to view the colors $blue_1, blue_2, \ldots, blue_{r-1}$ all as an identical shade of blue (so that we are only able to distinguish between the colors red and blue), then we would have a 2-coloring of [1, m] and hence, under this coloring, there is a "monochromatic" arithmetic progression of length $w(k; r-1) \geq k$. If this progression is red then we are done, so assume that it is "blue." However, our "blue" progression actually consists of the r-1 colors $\{blue_1, blue_2, \ldots, blue_{r-1}\}$. Hence, under χ , we have an (r-1)-coloring of some arithmetic progression $\{a, a + d, \ldots, a + w(k; r-1)d\}$. By Example 2.2, we are done.

To show that (iii) \Rightarrow (iv), let max $(S) = s_n$ and $w = w(s_n + 1; r)$. Under any *r*-coloring of [1, w] we have a monochromatic arithmetic progression $\{a, a + d, a + 2d, \ldots, a + s_n d\}$ for some $a, d \ge 1$. Since a + dS is a subset of this monochromatic arithmetic progression, we are done.

To prove that (iv) \Rightarrow (v), let k be fixed and take $S = \{1, 2, \dots, k\}$, so that $a + dS = \{a + d, a + 2d, \dots, a + kd\}$. Hence, by (iv), we are guaranteed that under any r-coloring of \mathbb{Z}^+ there exists a k-term monochromatic arithmetic progression.

Next, we show that $(v) \Rightarrow (vi)$. Let S and c be defined as in (vi). Let $T_0 = S$ and define the sets T_j , j = 1, 2, ..., c - 1, as

$$T_j = \{s+j : s \in S\} - \bigcup_{i=1}^{j-1} T_i,$$

i.e., just translate S by j, and make sure that T_j is disjoint from the previous T_i 's. Since $|s_{i+1} - s_i| \leq c$ for all $i \geq 0, T_1, T_2, \ldots, T_c$ is a partition of \mathbb{Z}^+ . Color each $z \in \mathbb{Z}^+$ with color j if $z \in T_j$. This defines a c-coloring of \mathbb{Z}^+ . By (v), there is an arithmetic progression $\{a + d, a + 2d, \ldots, a + kd\} \subseteq T_{j_0}$ for some j_0 . Hence,

 $\{(a-j_0)+d, (a-j_0)+2d, \dots, (a-j_0)+kd\} \subseteq T_0 = S,$

which proves (vi).

We complete the proof by showing that $(vi) \Rightarrow (i)$. Fix k. Consider any 2-coloring of \mathbb{Z}^+ using the colors red and blue. For one of the colors, say red, there must be an infinite number of positive integers $r_1 < r_2 < r_3 < \cdots$ with that color. If there exists an *i* such that $r_{i+1} - r_i > k$, then we have *k* consecutive blue integers, which gives us

a monochromatic arithmetic progression of length k. If no such i exists, then for all $i \ge 0$, $|r_{i+1} - r_i| \le k$ (so that $\max\{|r_{i+1} - r_i| : i \ge 0\}$ exists), and, by (vi), we have a red arithmetic progression of length k.

It is interesting to note that (ii) and (iii) are equivalent when, *a priori*, (ii) seems to be weaker than (iii). In fact, in many instances in this book, we will find statements which hold for 2 colors that don't hold for an arbitrary number of colors. As can be seen from the proof that (ii) implies (iii), the fact that we are dealing with arithmetic progressions is vital.

2.3. Computing van der Waerden Numbers

Thus far we have focused on the existence of w(k;r). The next natural step is to determine, if feasible, the values of w(k;r). Below we present all known (to date) nontrivial values of w(k;r). (We leave it as Exercise 2.2 to prove the trivial values w(2;r) = r + 1.)

$$w(3;2) = 9;$$
 $w(3;3) = 27;$ $w(4;2) = 35;$
 $w(3;4) = 76;$ $w(5;2) = 178.$

So why aren't more values known? Consider w(3; 5), which is obviously greater than w(3, 4) = 76. Say we want to check whether every 5-coloring of [1, 100] admits a monochromatic 3-term arithmetic progression. Using brute force, there are 5^{100} colorings to consider. Assuming (incorrectly) that it takes only one computer step to check a coloring for a monochromatic triple in arithmetic progression, we may need $5^{100} \approx 8 \times 10^{69}$ computer steps. At a trillion computer steps per second, if we had a trillion worlds, each with a trillion cities, each with a trillion computer labs, each with a trillion computers we could use, it would take 250,000 years to run this many steps. Then, if we should find success, we would need to check the 5-colorings of [1, 99], since so far we would know only that $w(3, 5) \leq 100$.

So, we can see that the brute force method is not at all feasible. However, there are algorithms for finding values of w(k;r) that are somewhat more efficient. We present three such algorithms below. We encourage the reader to try each of the algorithms, by hand,

29

2. Van der Waerden's Theorem

to calculate w(3; 2). In the algorithms, we use 1, 2, ..., r as the colors. Recall that we say a coloring is valid (in this circumstance) if it admits no monochromatic k-term arithmetic progression.

In Algorithm 1, g_j represents an *r*-coloring and $g_j(i)$ is the color of *i* under g_j . The final value of *n* equals w(k; r).

Algorithm 1.

STEP 1: Set n = 1, k = 1, and $g_1(1) = 1$ (at this point g_1 is defined only on $\{1\}$) STEP 2: Set $S = \{g_1, g_2, \dots, g_k\}$ and k = |S|STEP 3: Increment n by 1 STEP 4: Set $S = \emptyset$ and j = 0STEP 5: Set i = 0 and increment j by 1 STEP 6: Increment i by 1 STEP 7: Set $g_j(n) = i$ STEP 7: Set $g_j(n) = i$ STEP 8: If $g_j: [1, n] \rightarrow \{1, 2, \dots, r\}$ is valid then set $S = S \cup \{g_j\}$ STEP 9: If i < r, go to STEP 6 STEP 10: If j < k, go to STEP 5 STEP 11: If $S \neq \emptyset$, go to STEP 2 STEP 12: STOP and output n

This algorithm is more efficient than brute force because we know that all colorings in S avoid monochromatic arithmetic progressions of length k. Furthermore, the set S is built up along the way so as to weed out those colorings that do contain a k-term monochromatic arithmetic progression.

Even though Algorithm 1 is more efficient than the brute force method, its usefulness is limited to only two or three nontrivial (i.e., $k \geq 3$) values of w(k;r); this is due to memory restrictions, as the set S tends to have many members before the algorithm starts to efficiently weed out invalid colorings.

The next algorithm is also more efficient than the brute force method, without the memory restrictions of Algorithm 1. In Algorithm 2, g(i) represents the color of i. The final value of w equals w(k;r).

Algorithm 2.

```
STEP 1: Set i = 1, g(1) = 1, w = r + 1

STEP 2: Increment i by 1 and set g(i) = 1

STEP 3: If g: [1,i] \rightarrow \{1, 2, \dots r\} is valid, go to STEP 2

STEP 4: If g(i) = r, go to STEP 6

STEP 5: Increment g(i) by 1 and then go to STEP 3

STEP 6: If i > w, set w = i

STEP 7: Decrement i by 1

STEP 8: If i > 1, go to STEP 4

STEP 9: STOP and output w
```

Although Algorithm 2 does not have the memory problem of Algorithm 1, it still has the problem of being very time-intensive. It has been used successfully in finding all the known values of w(k; r). With the ever-increasing speed of new computers, one would wonder why we still do not know the exact values of, say, w(6;2) or w(4;3). Even with these somewhat more efficient algorithms, the number of steps required as a function of [1, w], the interval being colored, still grows at an exponential rate. For example, how might we make a rough comparison between the time it would take to calculate w(5; 2) = 178, and the time it would take to calculate w(6, 2)? It has been observed that the (admittedly few) known values of w(k; 2)approximate $\frac{3n!}{2}$, so perhaps a reasonable approximation for w(6,2)is 1080. Even if w(6; 2) = 720 (it is known that $w(6; 2) \ge 696$), the number of different 2-colorings of [1, 720] is 2^{720} , whereas the number of 2-colorings of [1, 178] is 2^{178} . To get some idea of how the number of computer steps required to find w(5; 2) would compare to the number required to find w(6,2), it would not be unreasonable to compare 2^{720} to 2^{178} ; the ratio of these two numbers is $2^{542}(!)$. Looking at how the computing times needed to calculate w(3; 2), w(4; 2), and w(5; 2)have compared may also help us understand why the calculation of w(6;2) has, thus far, been out of reach. For example, in one study, the computing times it took to calculate w(3; 2), w(4; 2), and w(5; 2)were instantaneous, 3 seconds, and more than a week, respectively.

The third algorithm we present is essentially a refinement of Algorithm 2, employing what we call the "culprit method." Given a

coloring χ of some interval [1, m] and an integer $k \geq 2$, we define a *culprit* of a positive integer n to be any integer a + (k-2)d with $a, d \geq 1$, such that $a, a + d, a + 2d, \ldots a + (k-2)d$ is monochromatic and a + (k-1)d = n. In other words, it is the $(k-1)^{\text{st}}$ term of an arithmetic progression whose k^{th} term is n, with the first k-1 terms all of the same color. To illustrate how the culprit can be used to cut down on the number of calculations, we consider an example.

Example 2.6. Assume that we are calculating w(4;2) (which equals 35) by means of Algorithm 2. Say that the program is running and we are at the following point in the computations: w = 30 (this means that so far we have discovered a valid coloring of [1,29] but not one of [1,30]; i = 26 and we are about to assign g(i) = 1; the integers 1, 2, 4, 6, 9, 10, 11, 13, 17, 18, 20, 22, 23 all have color 1; and the integers 3, 5, 7, 8, 12, 14, 15, 16, 19, 21, 24, 25 all have color 2. According to Algorithm 2, since 26 cannot be assigned either color to yield a valid coloring of [1, 26], we proceed to STEP 7, and then try various colors on 23, 24, and 25; should that fail, we will go back to 22, etc. However, the least culprit of 26 with color 1 is 18 (that is, as long as 2, 10, 18 have color 1, we will not be able to assign the color 1 to the integer 26); and the least culprit of 26 with color 2 is 19 (as long as 5, 12, 19 have color 2, we will not be able to assign the color 2 to the integer 26). Therefore, to try various color assignments on any of the integers in the set $\{20, 21, 22, 23, 24, 25\}$ at this point would be a waste of time, because we will not find any valid colorings of [1, 26] until we change the way [1, 19] is colored. Thus, we would save time by moving 19 into the set with color 2 at this point, rather than moving 23 into that set, which is the next move in Algorithm 2.

Algorithm 3 takes advantage of the culprit to give a more efficient algorithm. We use $cul_j(i)$ to represent the minimum of all culprits of i having color j.

2.3. Computing van der Waerden Numbers

Algorithm 3.

STEP 1: Set i = 1, g(1) = 1, w = r + 1STEP 2: Increment i by 1 and set g(i) = 1STEP 3: If i > w, set w = iSTEP 4: If $g: [1, i] \rightarrow \{1, 2, ..., r\}$ is valid, go to STEP 2 STEP 5: If g(i) = r, go to STEP 7 STEP 6: Increment g(i) by 1 and then go to STEP 4 STEP 7: Set $i = \max\{cul_j(i): 1 \le j \le r\}$ STEP 8: If g(i) < r, go to STEP 6 STEP 9: Decrement i by 1 STEP 10: If i = 1, STOP and output wSTEP 11: Go to STEP 8

Algorithm 3 is somewhat more efficient that Algorithm 2 but, it seems, not efficient enough to allow (at least thus far) the discovery of new van Waerden numbers.

We conclude this section by considering what are sometimes called "mixed" van der Waerden numbers and a table of all known (to date) computed values. Instead of requiring an arithmetic progression of length k to be of one of the colors (so that k is a constant that is independent of the color), the mixed van der Waerden numbers allow the required length to vary with the color. For example, if the colors are red and blue, then w(4) represents the least positive integer such that, for every 2-coloring of [1, w(4)], there is either a 4-term red arithmetic progression or a 4-term blue arithmetic progression. What if we "mix" the lengths so that we want the least positive integer nsuch that, for every 2-coloring of [1, n], there is either a 4-term red arithmetic progression or a 5-term blue arithmetic progression? We know that w(4) = 35 and w(5) = 178. So we must have $35 \le n \le 178$ (why?). More generally, we have the following as a corollary of van der Waerden's theorem. We leave the proof as Exercise 2.4.

Corollary 2.7. Let $r \ge 2$, and let $k_i \ge 1$ for $1 \le i \le r$. Then there exists a least positive integer $w = w(k_1, k_2, ..., k_r; r)$ such that, for every r-coloring $\chi : [1, w] \rightarrow [1, r]$, there exists, for some $i \in [1, r]$, a k_i -term arithmetic progression with color i.

Let's consider an example.

Example 2.8. According to Corollary 2.7, w(3, 4, 2; 3) is the least positive integer n such that for every 3-coloring (using the colors 1, 2, and 3) of [1, n], there is a 3-term arithmetic progression with color 1 or a 4-term arithmetic progression with color 2 or a 2-term arithmetic progression with color 3. It is easy to see that w(3, 4, 2; 3) = w(4, 3, 2; 2) and, in fact, that any re-ordering of the components of (k_1, k_2, \ldots, k_r) in Corollary 2.7 will have no effect on the value of n.

In Table 2.1 we present all known (to date) mixed van der Waerden numbers (including the classical van der Waerden numbers) $w(k_1, k_2, \ldots, k_r; r)$, where $k_i \ge 3$ for at least two values of *i*. The table gives, for r = 2, 3, 4, each *r*-tuple (k_1, k_2, \ldots, k_r) with the k_i 's ordered in nonincreasing order (as mentioned in Example 2.8, the order in which the k_i 's appear is irrelevant).

r	k_1	k_2	k_3	k_4	w	
2	3	3	-	-	9	
2	4	3	-	-	18	
2	4	4	-	-	35	
2	5	3	-	-	22	
2	5	4	-	-	55	
2	5	5	-	-	178	
2	6	3	-	-	32	
2	6	4	-	-	73	
2	7	3	-	-	46	
2	7	4	-	-	109	
2	8	3	-	-	58	
2	9	3	-	-	77	
2	10	3	-	-	97	
2	11	3	-	-	114	
2	12	3	-	-	135	
3	3	3	2	-	14	

1	κ_1	κ_2	<i>k</i> 3	ħ4	w
3	3	3	3	-	27
3	4	3	2	-	21
3	4	3	3	-	51
3	4	4	2	-	40
3	4	4	3	-	84
3	5	3	2	-	32
3	5	3	3	-	77
3	5	4	2	-	71
3	6	3	2	-	40
4	3	3	2	2	17
4	3	3	3	2	40
4	3	3	3	3	76
4	4	3	2	2	25
4	4	3	3	2	60
4	4	4	2	2	53

2 | 43

Table 2.1: Mixed van der Waerden numbers

5 | 3 | 2

4

2.4. Bounds on van der Waerden Numbers

2.4. Bounds on van der Waerden Numbers

Since we have seen that finding exact values for van der Waerden numbers is an extremely difficult problem, we turn to the problem of finding bounds on these numbers.

As we will see, there is a wide gap between the best known upper bounds on w(k) and the best known lower bounds. We begin with lower bounds.

The best known lower bound on w(k), due to Berlekamp, is presented in the following theorem. We do not include the proof, which requires a knowledge of field extensions.

Theorem 2.9. Let p be prime. Then $w(p+1) \ge p2^p$.

The person most responsible for the development of Ramsey theory as a branch of mathematics, in fact for the much broader area of mathematics known as combinatorial number theory, is Paul Erdős. The extremely insightful Erdős (also the most prolific mathematician of the twentieth century) conjectured that $\lim_{k\to\infty} \frac{w(k)}{2^k} = \infty$ (i.e., that w(k) grows significantly faster than 2^k). Whether this conjecture is true is uncertain, although Theorem 2.9 lends credence to it. Along these lines, we have the following theorem.

Theorem 2.10. Let $\epsilon > 0$. There exists $k_0 = k(\epsilon)$ such that for all $k \ge k_0$,

$$w(k) \ge rac{2^k}{k^\epsilon}.$$

We omit the proof of Theorem 2.10, but will prove Theorem 2.18, which gives an asymptotic lower bound for w(k). The proof makes use of a graph-theoretic result due to W.M. Schmidt (Lemma 2.17). The proof of Theorem 2.18 provides us with a method for obtaining lower bounds that may be applied to many of the other types of sequences mentioned in this book (these may be fruitful avenues of research to explore).

Before stating Schmidt's lemma, we need to give some definitions.

Definition 2.11. A hypergraph $\Gamma = (V, \mathcal{E})$ is a set of vertices V and a collection \mathcal{E} of subsets of V such that, for every $E \in \mathcal{E}$, $|E| \ge 2$. The members of \mathcal{E} are called hyperedges.

Note that if for every $E \in \mathcal{E}$ we have |E| = 2, then the hypergraph (V, \mathcal{E}) reduces to a graph as defined in Definition 1.11.

Example 2.12. Let $V = \{1, 2, 3, 4\}$ and consider the collection $\mathcal{E} = \{\{1, 2, 4\}, \{3, 4\}, \{2, 3, 4\}\}$. Then (V, \mathcal{E}) is a hypergraph with 3 hyperedges. If $\mathcal{G} = \{\{1, 2\}, \{1, 3\}, \{4\}\}$, then (V, \mathcal{G}) is not a hypergraph.

Definition 2.13. Let $\Gamma = (V, \mathcal{E})$ be a hypergraph. We say that Γ has *Property B* if there exists $V_1 \subseteq V$ such that for every $E \in \mathcal{E}$ we have $E \cap V_1 \neq \emptyset$ and $E \cap V_1 \neq E$.

We consider an example.

Example 2.14. Let $\Gamma_1 = (V, \mathcal{E}_1)$ with

$$V = \{1, 2, 3, 4, 5, 6, 7, 8, 9\}$$

and

 $\mathcal{E}_1 = \{\{1, 2, 3, 9\}, \{1, 4, 5, 6\}, \{3, 7, 8\}\}.$

Then, by letting $V_1 = \{1, 3\}$, as in Definition 2.13, we see that Γ_1 has Property B. Let $\Gamma_2 = (V, \mathcal{E}_2)$, where \mathcal{E}_2 is the collection of all 3-term arithmetic progressions contained in [1,9]. We show, by contradiction, that Γ_2 does not have Property B. Assuming Γ_2 does have Property B, there is a $V_2 \subseteq [1,9]$ such that $V_2 \cap E \neq \emptyset$ and $V_2 \cap E \neq E$ for every $E \in \mathcal{E}_2$. Consider the partition of [1,9] into the two sets V_2 and $V - V_2$. By the fact that w(3;2) = 9, we know there is some 3-term arithmetic progression E_0 such that either $E_0 \subseteq V_2$ or $E_0 \subseteq V - V_2$. If $E_0 \subseteq V_2$, then $E_0 \cap V_2 = E_0$, and if $E_0 \subseteq V - V_2$, then $E_0 \cap V_2 = \emptyset$. This contradicts our assumption about V_2 . Hence Γ_2 does not have Property B.

Definition 2.15. Let $\Gamma = (V, \mathcal{E})$ be a hypergraph. For $k \geq 2$, denote by $\mu(k)$ the minimal number of hyperedges $\mathcal{E} = \{E_1, E_2, \ldots, E_{\mu(k)}\}$ such that $|E_i| = k$ for $1 \leq i \leq \mu(k)$ and Γ does not have Property B.

Example 2.16. In Example 2.14 we saw that the set of all 3-term arithmetic progressions in [1,9] is a hypergraph on $\{1, 2, \ldots, 9\}$ not having Property B. The number of 3-term arithmetic progressions in [1,9] is

$$\left\lfloor \frac{9-1}{2} \right\rfloor + \left\lfloor \frac{9-2}{2} \right\rfloor + \left\lfloor \frac{9-3}{3} \right\rfloor + \dots + \left\lfloor \frac{9-7}{2} \right\rfloor = 16$$

Therefore $\mu(3) \leq 16$.

We now state Schmidt's lemma, the proof of which we omit.

Lemma 2.17 (Schmidt's Lemma). For k > 1,

$$\mu(k) \ge \frac{2^k}{1 + 2k^{-1}}$$

With the help of Lemma 2.17 we are able to obtain the following lower bound on w(k).

Theorem 2.18. $w(k;2) \ge \sqrt{k}2^{\frac{k+1}{2}}(1-o(1)).$

Proof. Let $m \ge w(k)$. Let \mathcal{E} be the collection of all k-term arithmetic progressions that are contained in [1, m]. Just as we saw in Example 2.14 that Γ_2 (for 3-term arithmetic progressions) does not have Property B, the hypergraph $\Gamma = (\{1, 2, \ldots, m\}, \mathcal{E})$ does not have Property B (we leave the details to the reader as Exercise 2.9).

To use Schmidt's lemma, we need to know something about the size of \mathcal{E} . For $E = \{a, a + d, a + 2d, ..., a + (k - 1)d\}$ to belong to \mathcal{E} , we must have $d \leq \frac{m-1}{k-1}$. We also must have $a \leq m - (k-1)d$. Since each k-term arithmetic progression is completely determined by the values of a and d, we have

$$\begin{aligned} |\mathcal{E}| &= \sum_{d=1}^{\left\lfloor \frac{m-1}{k-1} \right\rfloor} (m-(k-1)d) \\ &= m \left\lfloor \frac{m-1}{k-1} \right\rfloor - \frac{k-1}{2} \left\lfloor \frac{m-1}{k-1} \right\rfloor \left(\left\lfloor \frac{m-1}{k-1} \right\rfloor + 1 \right) \\ &\geq \frac{m^2}{2k}. \end{aligned}$$

Therefore, if m = w(k), then since Γ does not have Property B, by Schmidt's lemma we have

$$\frac{n^2}{2k} \ge \mu(k) \ge \frac{2^k}{1+2k^{-1}}$$

This implies that

$$m \ge \sqrt{rac{k2^{k+1}}{1+2k^{-1}}} \,,$$

2. Van der Waerden's Theorem

and therefore

$$w(k) \ge \sqrt{k}2^{\frac{k+1}{2}}(1-o(1)).$$

We next state two theorems on lower bounds for w(k;r), where r may be greater than two.

Theorem 2.19. Let $p \ge 5$ and q be primes. Then

 $w(p+1;q) \ge p(q^p - 1) + 1.$

Theorem 2.20. For all $r \ge 2$, $w(k;r) > \frac{r^k}{ekr}(1+o(1))$.

In the following table we present several of the best lower bounds (known to date), along with all of the known values (to date), for specific van der Waerden numbers.

$k \setminus r$	2	3	4	5
3	9	27	76	?
4	35	≥ 292	≥ 1048	≥ 2254
5	178	≥ 1210	≥ 10437	≥ 24045
6	≥ 696	≥ 8886	≥ 90306	≥ 93456
7	≥ 3703	≥ 43855	≥ 119839	?

Table 2.2: Lower bounds and values for w(k;r)

We now move on to upper bounds for the van der Waerden numbers. The main result in this direction is a remarkable theorem due to Timothy Gowers. In 1998, Gowers received a Fields Medal (the mathematical equivalent of the Nobel Prize). His work related to finding an upper bound on the van der Waerden numbers played a significant part in his winning of this prestigious award.

Before stating Gowers' result, we give some historical perspective on the search for an upper bound. First we define some very fastgrowing functions.

For a function f, and $n \in \mathbb{Z}^+$, denote by $f^{(n)}$ the composition of f with itself n times (i.e., $f^{(n)}(x) = \underbrace{f(f(\ldots(f(x))\ldots))}_{n \text{ times}}$). Let f_1 be the function $f_1(k) = 2k, k \ge 1$, and let f_2 be the function $f_2(k) = 2^k$,

2.4. Bounds on van der Waerden Numbers

 $k \ge 1$. Notice that $f_2(k) = f_1^{(k)}(1)$. Similarly, define $f_3(k) = f_2^{(k)}(1)$; and, more generally, for $i \in \mathbb{Z}^+$ define the functions

$$f_{i+1}(k) = f_i^{(k)}(1),$$

 $k \geq 1$. As the subscript increases, the growth of these functions accelerates at a phenomenal rate. For example,

$$f_3(k) = \underbrace{2^2}_{k \ 2's}^{2}.$$

In considering the magnitude of the functions $f_i(k)$, for reasons that are apparent, we will refer to the functions $f_2(k)$ and $f_3(k)$ as $exp_2(k)$ and tower(k), respectively.

Now consider $f_4(k)$. We call this function the *wow* function due to its incredible rate of growth, and we denote it by wow(k). To see how quickly wow(k) grows, consider wow(k) for k = 1, 2, 3, 4. We easily have wow(1) = 2, wow(2) = 4, and wow(3) = tower(4) = 65536. For k = 4, we have

 $wow(4) = f_4(4) = f_3(f_3(f_3(f_3(1)))) = f_3(65536) = tower(65536),$

A tower of 65536 2's.

Generalizing from $f_3(3)$ and $f_4(4)$, we now consider a function on \mathbb{Z}^+ known as the Ackermann function. The Ackermann function is defined as $ack(k) = f_k(k)$. This function is named after a similar function derived in 1928 by Wilhelm Ackermann, a high school math teacher, who received his Ph.D. under the direction of David Hilbert. This function is perhaps the fastest growing function you will ever encounter. To get an idea of how fast it grows, consider wow(5) and ack(5). We have

$$wow(5) = tower(tower(65536)),$$

for which calling it enormous would be an understatement. Meanwhile, we have

$$ack(5) = wow(wow(tower(65536)))$$

Given the incredibly immense size of wow(5), grasping the magniude of wow(tower(65536)), let alone ack(5), is no easy accomplishnent. Comparing $tower(5) = 2^{65536}$, a number with 19719 digits. 40

to $exp_2(5) = 2^5 = 32$, may give some insight into how wow(5) and ack(5) compare.

We introduced the above functions to show the enormity of the historical upper bounds on w(k). It turns out that the original proof of van der Waerden's theorem gives

$$w(k) \le ack(k),$$

which tells us, in particular, that $w(4) \leq tower(65536)$. Comparing this to the actual value w(4) = 35 certainly gives some reason to suspect that this upper bound might not be the best possible.

For many years, the much-revered mathematician Ronald Graham offered a \$1000 prize to anyone who could prove or disprove that $w(k) \leq tower(k)$. The idea of offering monetary prizes for what are considered difficult problems was championed by Paul Erdős. Graham and Erdős were good friends, and since Erdős' passing in 1996, Graham has kept this "prize money" tradition alive by honoring all of Erdős' prize problems and adding some of his own.

Since there is a very significant difference in magnitude between the functions tower(k) and ack(k), showing that $w(k) \leq tower(k)$ would seem to be quite a feat – some new method of proof would need to be used.

In 1987, the eminent logician Saharon Shelah, while not answering Graham's question, used an argument fundamentally different from earlier proofs of van der Waerden's theorem, to prove that

$w(k) \le wow(k),$

quite an improvement over the previous upper bound of ack(k).

Shelah's result is one of the most significant results in Ramsey theory. In fact, although Shelah did not answer Graham's "prize" question completely, Graham gave him half of the award money anyway.

When we consider the magnitude of wow(k), for example that wow(4) = tower(65536), and compare it to the best lower bounds known for w(k), it would be reasonable to think that this upper bound is not the best possible. In fact, Gowers' amazing result showed that the bound could be substantially improved. Here is Gowers' bound.

Theorem 2.21. For $k \geq 2$,

 $w(k;2) \le 2^{2^{2^{2^{k+9}}}}$

Gowers' bound is of a much smaller magnitude than tower(k), thereby settling Graham's question. As a consequence of Gowers' result, Graham asked in 1998 whether or not $w(k) < 2^{k^2}$ and currently offers \$1000 for an answer.

Remark 2.22. Gowers actually proved a more general result than what we state as Theorem 2.21. Namely, letting $f(k;r) = r^{2^{2^{k+9}}}$, he showed that

$$w(k;r) \le 2^{2^{j(k;r)}}$$

Recently, a new upper bound for w(3;r) for $r \ge 5$ has been obtained. We present this result without proof.

Theorem 2.23. Let $r \geq 5$. Then

 $w(3;r) < \left(\frac{r}{4}\right)^{3^r}.$

Even with Gowers' upper bound and the upper bound from Theorem 2.23, the best known upper and lower bounds on van der Waerden numbers are still nowhere near each other!

2.5. The Erdős and Turán Function

By van der Waerden's theorem we know that, for n large enough, whenever [1, n] is partitioned into a finite number of subsets, at least one of the subsets must contain a k-term arithmetic progression. In an effort to find out more about the van der Waerden numbers, Erdős and Turán defined a function that approaches the problem from what could be considered the opposite direction. That is, we may ask the following question: given a positive integer n, what is the maximum size of a subset of [1, n] that does not contain an arithmetic progression of length k? Erdős and Turán defined the following function.

Definition 2.24. For $k \ge 2$ and $n \ge 3$, let S be the collection of sets $S \subseteq [1, n]$ such that S does not contain an arithmetic progression of length k. Then

$$\nu_k(n) = \max\left\{|S| : S \in \mathcal{S}\right\}$$

2. Van der Waerden's Theorem

The function ν_2 is trivial, since $\nu_2(n) = 1$ for all n. In the following examples we consider $\nu_n(k)$ for some small values of k and n.

Example 2.25. Let k = 3 and n = 8. Since w(3) = 9, we know that there exists a 2-coloring of [1, 8] with no 3-term monochromatic arithmetic progression. One such coloring is 11001100. Hence, taking $S = \{1, 2, 5, 6\}$, we have $\nu_3(8) \ge 4$. Furthermore, there are only two other colorings, up to renaming the colors, of [1, 8] that avoid monochromatic 3-term arithmetic progressions, each of which has four integers of each color.

Hence, for any $T \subseteq [1,8]$ with $|T| \ge 5$, if we color the elements of T with one color and the elements of R = [1,8] - T with the other color, we are guaranteed to have a monochromatic 3-term arithmetic progression. If $|T| \ge 6$, then (since $|R| \le 2$) that arithmetic progression resides in T. This leaves us to consider |T| = 5 (so that |R| = 3). If R contains the monochromatic triple, then R is one of the following: $\{1,2,3\}, \{1,3,5\}, \{1,4,7\}, \{2,3,4\}, \{2,4,6\}, \{2,5,8\}, \{3,4,5\}, \{3,5,7\}, \{4,5,6\}, \{4,6,8\}, \{5,6,7\}, \{6,7,8\}$. It is easy to check that for each of these sets, T also contains a 3-term arithmetic progression. Hence, $\nu_3(8) \le 4$, and we have established that $\nu_3(8) = 4$.

Example 2.26. We show here that $\nu_3(26) \ge 9$. Since w(3;3) = 27, there exists a 3-coloring of [1,26] that avoids 3-term monochromatic arithmetic progressions. Clearly, there is some color c such that at least nine elements have color c. Since there is no 3-term arithmetic progression with color c, $\nu_3(26) \ge 9$.

The above examples illustrate the relationship between the Erdős and Turán function and the van der Waerden numbers. One important reason for studying $\nu_k(n)$ is that having an upper bound on these numbers would lead to an upper bound on w(k; r). To be more precise, we have the following theorem.

Theorem 2.27. Let $k \geq 3$, $r \geq 2$, and assume $\nu_k(m) \leq f(k,m)$, with $f(k,m) \leq \frac{m}{r}$. Then

$$w(k,r) \le rf(k,m) + 1$$

2.6. Proof of van der Waerden's Theorem

Proof. If $\nu_k(m) \leq f(k,m)$, then for any *r*-coloring of [1, rf(k,m)+1], the most used color, say *c*, would occur more than f(k,m) times. Since $f(k,m) \geq \nu_k(m)$, by the definition of $\nu_k(m)$ there must be a *k*-term arithmetic progression with color *c*.

To end this section, we state the following theorem; its proof is left to the reader as Exercise 2.13.

Theorem 2.28. Let $n, k \geq 3$. Let r(n) be the minimum number of colors required to color [1, n] so that no monochromatic k-term arithmetic progression exists. Then

$$\nu_k(n) \ge \left\lceil \frac{n}{r(n)} \right\rceil.$$

One of the most significant results in Ramsey theory, now known as Szemerédi's theorem, involves a conjecture made by Erdős and Turán in 1936. They conjectured that for every k,

$$\lim_{n \to \infty} \frac{\nu_k(n)}{n} = 0.$$

In 1952, Roth proved their conjecture for k = 3. In 1969, Szemerédi showed it holds for k = 4. Then, using an ingenious and very complex proof, Szemerédi fully settled the conjecture, establishing its truth, in 1975. Unfortunately his proof, which makes use of van der Waerden's theorem, does not yield any useful bounds for w(k).

2.6. Proof of van der Waerden's Theorem

For completeness, we restate the finite version of van der Waerden's theorem.

Van der Waerden's Theorem. Let $k, r \ge 2$ be integers. There exists a least positive integer w = w(k;r) such that for every r-coloring of [1, w] there is a monochromatic arithmetic progression of length k.

The proof we give of van der Waerden's theorem is constructed from two important lemmas. Before presenting these lemmas, we need some preliminaries.

The following proposition essentially tells us that the guarantee of a monochromatic arithmetic progression in an interval is unaffected

43

by translation (adding a constant integer) and/or dilation (multiplying by a positive integer) of the interval.

Proposition 2.29. Let k, r, m, a, and b be positive integers. Every r-coloring of [1,m] yields a monochromatic k-term arithmetic progression if and only if every r-coloring of

 $S = \{a, a + b, a + 2b, \dots, a + (m - 1)b\}$

yields a monochromatic arithmetic progression.

Remark. The "only if" portion of Proposition 2.29 immediately implies the result of Example 2.2.

We leave the proof of Proposition 2.29 to the reader as Exercise 2.14.

As an aside, it is worth mentioning that Proposition 2.29 may be easily extended to other sequences besides arithmetic progressions. Specifically, the following more general statement may be proved in essentially the same way. We leave the proof to the reader as Exercise 2.15.

Proposition 2.30. Let \mathcal{F} be a collection of sets and let $a, b \in \mathbb{Z}^+$. Assume the following: $S \in \mathcal{F}$ if and only if $a + bS \in \mathcal{F}$. Let $r \in \mathbb{Z}^+$. Then every r-coloring of [1, n] yields a monochromatic member of \mathcal{F} if and only if every r-coloring of

 $a + b[0, n - 1] = \{a, a + b, a + 2b, \dots, a + (n - 1)b\}$

yields a monochromatic member of \mathcal{F} .

We now give two definitions that are crucial to our proof of van der Waerden's theorem.

Definition 2.31. Let $r, m, n \ge 1$. Let γ be an r-coloring of [1, n+m]. Define $\chi_{\gamma,m}$ to be the r^m -coloring of [1, n] as follows: for $j \in [1, n]$, let $\chi_{\gamma,m}(j)$ be the m-tuple $(\gamma(j+1), \gamma(j+2), \ldots, \gamma(j+m))$. We call $\chi_{\gamma,m}$ a coloring derived from γ , or a derived coloring.

Note that $\chi_{\gamma,m}$ of Definition 2.31 is, in fact, an r^m -coloring since there are r^m possible *m*-tuples $(\gamma(j+1), \gamma(j+2), \ldots, \gamma(j+m))$, for $1 \leq j \leq n$. Also note that the above definition states that $i, j \in [1, n]$ have the same color under $\chi_{\gamma,m}$ precisely when [i + 1, i + m] and **Example 2.32.** Take r = 2, m = 3, and n = 18 in Definition 2.31. Define $\gamma : [1, 21] \rightarrow \{0, 1\}$ by the coloring

011000111100000111111.

To describe the 2³-coloring χ_3 of [1,18] derived from γ , we make, for convenience, the following correspondence between the set of all 3-tuples of two colors, $T = \{(i, j, k) : i, j, k \in \{0, 1\}\}$ (there are 8 of them), and the colors $0, 1, \ldots, 7$:

Note that any one-to-one correspondence between T and a set of eight "colors" would suffice.

Since $\chi_{\gamma,3}(1)$ corresponds to $(\gamma(2), \gamma(3), \gamma(4)) = (1, 1, 0)$, we see that $\chi_{\gamma,3}(1) = 4$. Similarly, $\chi_{\gamma,3}(2)$ corresponds to (1, 0, 0), so that $\chi_{\gamma,3}(2) = 1$. Evaluating the other 16 elements of [1, 18], we find that the derived coloring under $\chi_{\gamma,3}$ is

410367741000367777.

Definition 2.33. We say that a triple (k, t; r) is *refined* if there exists a positive integer m = m(k, t; r) such that for every *r*-coloring of [1, m], there exist positive integers z, x_0, x_1, \ldots, x_t such that each of the sets

$$T_{s} = \left\{ b_{s} + \sum_{i=0}^{s-1} c_{i} x_{i} : c_{i} \in [1, k] \right\},\$$

 $0 \leq s \leq t$, is monochromatic, where

$$b_s = z + (k+1) \sum_{i=s}^{t} x_i.$$

The above definition may seem rather cumbersome, but it will make the proof of van der Waerden's theorem easier to digest. So, before moving on, let's look at an example. Consider k = 2 and t = 2

in Definition 2.33. The sets are

 $T_{0} = \{b_{0}\},$ $T_{1} = \{b_{1} + x_{0}, b_{1} + 2x_{0}\},$ $T_{2} = \{b_{2} + x_{0} + x_{1}, b_{2} + 2x_{0} + x_{1}, b_{2} + x_{0} + 2x_{1}, b_{2} + 2x_{0} + 2x_{1}\},$

where $b_s = z + 3 \sum_{i=s}^{2} x_i$ for s = 0, 1, 2. Clearly, T_0 is monochromatic since it contains only a single point. Hence, in order for (2, 2; r) to be refined, there must exist m = m(2, 2; r) such that, under any *r*coloring χ of [1, m], we have

$$\chi(b_1 + x_0) = \chi(b_1 + 2x_0)$$

 and

 $\chi(b_2 + x_0 + x_1) = \chi(b_2 + 2x_0 + x_1) = \chi(b_2 + x_0 + 2x_1) = \chi(b_2 + 2x_0 + 2x_1)$

for some positive integers z, x_0, x_1, x_2 .

A subtle point of Definition 2.33, perhaps made clear by the above example, is that, even if (k, t; r) is a refined triple, we do not necessarily have that the T_s are all monochromatic of the same color, but rather that each T_s , individually, is monochromatic.

The following remark gives a glimpse as to how Definition 2.33 is used in the proof of van der Waerden's theorem.

Remark 2.34. By taking $c_0 = c_1 = \cdots = c_{s-1} = j$ for $j = 1, 2, \ldots, k$ in Definition 2.33, we have the arithmetic progression

 $\{a+jd: j=1,2,\ldots,k\}\subseteq T_s,$ where $a=b_s$ and $d=\sum_{i=0}^{s-1}x_i.$

We are now ready to present the lemmas that, when combined, will prove van der Waerden's theorem. Our approach will be proof by induction, showing that the existence of w(k; r) implies the existence of w(k + 1; r). In order to do this, we will be using refined triples.

Lemma 2.35. Let $k \ge 1$. If w(k;r) exists for all $r \ge 1$, then (k,t;r) is refined for all $r,t \ge 1$.

Proof. Let $r \ge 1$. The proof is by induction on t, starting with t = 1. To prove that (k, 1; r) is refined, we first show that we may take m = m(k, 1; r) (in Definition 2.33) to be 3w(k; r) + k + 1. Let

2.6. Proof of van der Waerden's Theorem

 χ be an arbitrary r-coloring of [1, 3w(k; r) + k + 1]. Since we are

assuming that w(k; r) exists, applying Proposition 2.29, the interval [w(k; r) + k + 2, 2w(k; r) + k + 1] must admit a monochromatic k-term arithmetic progression $S = \{a + d, a + 2d, \dots, a + kd\}$.

Using the notation of Definition 2.33, let z = a - (k+1), $x_0 = d$, and $x_1 = 1$. This gives $T_0 = \{a + (k+1)d\}$ and $T_1 = S$, which are both contained in [1, m] and are both (individually) monochromatic $(T_0 \text{ contains only one point})$, thereby proving that (k, 1; r) is a refined triple.

Now let $t \ge 1$ and assume that (k, t; r) is refined. We will show that (k, t+1; r) is refined. Let m = m(k, t; r) be as in Definition 2.33 and let $n = 2w(k; r^m)$.

We claim that we may take m(k, t+1; r) = n + m. Let γ be an *r*-coloring of [1, n+m]. Let $\chi = \chi_{\gamma,m}$ be the r^m -coloring of [1, n] derived from γ (see Definition 2.31). By the definition of n, and since $w(k; r^m)$ exists, there must be an arithmetic progression

$$\{a+d, a+2d, \dots, a+(k+1)d\} \subseteq [1, n]$$

with the first k terms monochromatic under χ_m . By the definition of χ , the k intervals $I_j = [a+jd+1, a+jd+m], 1 \leq j \leq k$, have identical colorings under γ . Since (k, t; r) is refined, there exist z, x_0, x_1, \ldots, x_t so that the T_i 's (as in Definition 2.33) are monochromatic under γ . Therefore, each I_j contains the monochromatic sets

$$\begin{aligned} S_s(j) &= T_s + (a + jd) \\ &= \{y + a + jd : y \in T_s\} \\ &= \left\{ (b_s + a + jd) + \sum_{i=0}^{s-1} c_i x_i : c_i \in [1, k] \right\} \end{aligned}$$

for $s = 0, 1, \ldots, t$ (where b_s is as in Definition 2.33). Furthermore, since the intervals have the same coloring under γ , $S_s(u)$ and $S_s(v)$ must have the same coloring under γ for $1 \leq u, v \leq k$. Hence, by construction, the set

$$Q_s = \left\{ (b_s + a) + \sum_{i=0}^{s-1} c_i x_i + jd : j, c_i \in [1, k] \right\}$$

is monochromatic under γ for each $s = 0, 1, \ldots, t$.

2. Van der Waerden's Theorem

We shall provide integers $z', x'_0, x'_1, \ldots, x'_{t+1}$, which produce monochromatic sets T'_s (analogous to the T_i 's in Definition 2.33), for $s = 0, 1, \ldots, t+1$, to show that (k, t+1; r) is refined. Let

$$z' = z + a,$$

 $x'_i = x_i \text{ for } 0 \le i \le s - 1,$
 $x'_s = d,$
 $x'_i = x_i \text{ for } s + 1 \le i \le t, \text{ and}$
 $z'_{t+1} = x_s.$

Since for each s = 0, 1, ..., t we have $T'_{s+1} = Q_s$, it follows that, under γ , T'_{s+1} is monochromatic for $0 \le s \le t$. Since $T'_0 = \{b_0 + a + (k+1)d\}$ consists of a single point (whose value is less than n+m), it is trivially monochromatic. Thus we have satisfied the conditions required to prove that (k, t+1; r) is refined, thereby proving the lemma. \Box

Lemma 2.36. If (k,t;r) is refined for all $r,t \ge 1$, then w(k+1;r) exists for all $r \ge 1$.

Proof. Let r be given and let χ be any r-coloring of \mathbb{Z}^+ . By assumption, (k, t; r) is refined for, in particular, t = r. Using the notation of Definition 2.33, there exist z, x_0, x_1, \ldots, x_r such that each of the sets T_0, T_1, \ldots, T_r is monochromatic under χ . By the pigeonhole principle, two of these sets must be the same color. Let T_v and $T_w, v < w$, be two such sets. We have

$$T_{v} = \left\{ z + (k+1) \sum_{i=v}^{r} x_{i} + \sum_{i=0}^{v-1} c_{i} x_{i} : c_{i} \in [1,k] \right\}$$

and

$$T_w = \left\{ z + (k+1) \sum_{i=w}^r x_i + \sum_{i=0}^{w-1} c_i x_i : c_i \in [1,k] \right\}.$$

Letting $a = z + \sum_{i=0}^{v-1} x_i + (k+1) \sum_{i=w}^{r} x_i$, we rewrite these as

$$T_{v} = \left\{ a + (k+1) \sum_{i=v}^{w-1} x_{i} + \sum_{i=0}^{v-1} (c_{i}-1) x_{i} : c_{i} \in [1,k] \right\}$$

and

$$T_w = \left\{ a - \sum_{i=0}^{\nu-1} x_i + \sum_{i=0}^{\omega-1} c_i x_i : c_i \in [1, k] \right\}.$$

Taking $c_0 = c_1 = \cdots = c_{v-1} = 1$ in T_w , we have

$$T'_w = \left\{ a + \sum_{i=v}^{w-1} c_i x_i : c_i \in [1,k] \right\} \subseteq T_w.$$

Letting $d = \sum_{i=v}^{w-1} x_i$, we have $a + (k+1)d \in T_v$ and, from Remark 2.34, $\{a + d, a + 2d, \ldots, a + kd\} \subseteq T'_w$. Hence, we have found a monochromatic arithmetic progression of length k+1, thereby proving the existence of w(k+1;r).

We now put the previous two lemmas together to prove van der Waerden's theorem.

Proof of van der Waerden's Theorem. Clearly w(1; r) exists for all $r \ge 1$, since we need only a single point. Lemma 2.35 shows that (1,t;r) is refined for all $r,t \ge 1$. This in turn, by Lemma 2.36, gives us the existence of w(2;r) for all $r \ge 1$. Hence, by repeated application of Lemmas 2.35 and 2.36 we have the existence of w(k;r) for any given k, for all $r \ge 1$.

2.7. Exercises

- **2.1** Show that within [1, n] there are $\frac{n^2}{4} + O(n)$ arithmetic progressions.
- **2.2** Prove that w(2; r) = r + 1 for all $r \ge 1$.
- **2.3** Show that the following 3-colorings yield no monochromatic 3-term arithmetic progression. Note that w(3;3) = 27, so that the following colorings are maximal.
 - a) 00110 01212 20200 10112 02212 1
 - b) 21001 01221 12210 10012 20020 2
- **2.4** Show that $w(k_1, k_2, \ldots, k_r; r)$, i.e., the mixed van der Waerden numbers, always exist.
- 2.5 For each of the following van der Waerden-like functions, explain why it is reasonable to say that "the function is not worth studying."

a) For positive integers k_1 and k_2 , define $f(k_1, k_2)$ to be the least positive integer n with the property that, for every 2-coloring of [1, n], there is a color i such that there is either

- 2.8. Research Problems
- 2. Van der Waerden's Theorem

a k_1 -term arithmetic progression with color i or a k_2 -term arithmetic progression with color i.

- b) For positive integers k_1 and k_2 , define $g(k_1, k_2)$ to be the least positive integer n such that for every 2-coloring (using the colors red and blue) of [1, n], there is a k_1 -term arithmetic progression with color red and a k_2 -term arithmetic progression with color blue.
- **2.6** Prove that, for all $k \ge 2$, w(k, 2; 2) = 2k 1 if k is even, and w(k, 2; r) = 2k if k is odd.
- **2.7** Prove that, for all $k \ge 2$, w(k, 2, 2; 3) = 3k if and only if $k \equiv \pm 1 \pmod{6}$.
- **2.8** We can easily obtain a lower bound for w(k; 2) by using the probabilistic method. Show that $w(k; 2) > 2^{k/2}$ for large k via the following steps.

a) Randomly color the integers in [1, n] either red or blue. Let A be an arbitrary arithmetic progression of length k within [1, n]. Show that the probability that A is monochromatic is $\frac{2}{2^k} = 2^{1-k}$.

b) Use Exercise 2.1 to deduce that the probability that a monochromatic arithmetic progression exists is at most

$$\sum_{i=1}^{\frac{n^2}{4}+O(n)} 2^{1-k} = \frac{n^2 + O(n)}{2^{k+1}}.$$

c) Argue that if $\frac{n^2 + O(n)}{2^{k+1}} < 1$ then w(k; 2) > n.

d) Conclude that $n = 2^{k/2}$ satisfies the first inequality in (c) for k large enough.

2.9 Fill in the details of the proof of Theorem 2.18.

- **2.10** Let $r = 2^n$. Show that for $3 \le n \le 12$, the bound given by Theorem 2.23 is better than Gowers' bound (Remark 2.22), but that for $n \ge 13$, Gowers' bound is better. (A computer will be helpful for this exercise.)
- **2.11** Find $\nu_3(7)$ and $\nu_3(9)$.
- **2.12** Exhibit a 2-coloring of [1, 14] which shows that $\nu_4(14) \ge 7$.

2.13 Prove Theorem 2.28.

- 2.14 Prove Proposition 2.29 (hint: see Example 2.2).
- 2.15 Prove Proposition 2.30.

2.8. Research Problems

- *2.1 Write a more efficient algorithm for determining w(k; r). References: [25], [261]
- *2.2 Find w(3;5), w(4;3), or w(6;2), or improve the known lower bounds.

References:[71], [210], [261]

2.3 Determine the value of a new "mixed" van der Waerden number $w(k_1, k_2, \ldots, k_r; r)$ (where at least two of the k_i 's are greater than 2), or obtain a lower bound, by using a computer program.

References: [25], [55], [71], [74]

2.4 Find the rate of growth, or improve the known bounds, on the mixed van der Waerden numbers $w(k, k_0)$, as a function of k, for any fixed $k_0 \geq 3$. Try similar problems using more than two colors, where all but one of the k_i 's $(1 \leq i \leq r)$ are fixed.

References: [25], [55], [71], [74]

- 2.5 Investigate the set of maximal length valid colorings for various van der Waerden (or mixed van der Waerden) numbers. References: [25], [55], [71], [74], [210], [261]
- **2.6** The first-named author has conjectured the following concerning mixed van der Waerden numbers:

$$egin{aligned} & w(k,k) & \geq w(k+1,k-1) \ & \geq w(k+2,k-2) \ & \geq \cdots \geq w(2k-2,2) \end{aligned}$$

for all $k \geq 3$. Prove or disprove this. Try extending this to mixed van der Waerden numbers where the number of colors is greater than two.

References: [25], [55], [71], [74]

1

***2.7** Let $A = \{a_1, a_2, ...\}$ be a sequence of positive integers such that $\sum_{i=1}^{\infty} \frac{1}{a_i} = \infty$. Erdős conjectured that A must contain

arbitrarily long arithmetic progressions. Prove or disprove this.

Reference: [275]

- *2.8 Erdős conjectured (in 1961) that $\lim_{k\to\infty} \frac{w(k;2)}{2^k} = \infty$. Prove or disprove this. References: [34], [263]
- *2.9 Improve on the known upper bound for w(k; 2). References: [121], [141]
- *2.10 Improve on the known lower bounds for w(k;r). Reference: [127]
- *2.11 Find the asymptotic value of w(3; r). Reference: [141]
- *2.12 Improve on the known upper bound for w(4; r). References: [121], [141]
- *2.13 Find an upper bound on $\nu_k(n)$ for general k and n. References: [229], [264], [265]

2.9. References

§2.1. The statement and proof of the full version of the compactness principle can be found in [127].

§2.2. Theorem 2.5 is taken primarily from [123, p. 1000], with statement (iv) from [72] and statement (vi) originally from [56]. More proofs of the equivalence of different forms of van der Waerden's theorem are given in [205] and [209], the latter of which also contains several corollaries of the theorem.

§2.3. The algorithms, and information about the computer time needed to find some exact values, may be found in [25] and [261]. The references for the values (except w(3,3;2) = 9) given in Table 2.1 are as follows:

w(4,3;2), w(4,4;2), w(5,3;2), w(5,4;2), w(6,3;2), w(7,3;2),and w(3,3,3;3) are from [71]; w(5;2) = 178 is from [261]; w(6,4;2), w(8,3;2), w(9,3;2), w(10,3;2), w(4,3,3;3), and w(3,3,3,3;4) are from [25];

w(11, 3; 2), w(12, 3; 2), w(4, 4, 3; 3), w(5, 3, 3; 3), and w(4, 3, 3, 2; 4) are from [74]; w(7, 4; 2) is from [24];

and the remaining values are from [55].

2.9. References

The mixed van der Waerden numbers such that all but one entry equals 2 are investigated in [74]. Bounds on numbers very similar to the mixed van der Waerden numbers can be found in [137].

§2.4. Theorem 2.9 is from [34]. Theorem 2.10 is from [263]. Erdős and Rado [95] showed, in 1952, that

$$w(k;r) > r^{k+1} - c\sqrt{(k+1)\ln(k+1)}$$

A series of improvements (but not as strong as Theorem 2.10) were subsequently made in [191], [246], and [5]. The bound

$$w(k;r) > \frac{kr^k}{e(k+1)^2}$$

which is sometimes better than that of Theorem 2.9, was proven by Everts [105]. Other results concerning Property B are found in [4], [6], [7], [8], [9], [20], [46], [85], [86], [87], [104], [107], and [150]. The function $\mu(k)$ (Definition 2.15) was defined by Erdős and Hajnal [93]. Schmidt's lemma is proven in [245], and later in [99], which also includes a simpler proof of a weaker bound. Erdős [86] gives an upper bound for $\mu(k)$. The proof of Theorem 2.18 is from [99]. The proof of Theorem 2.19 can be easily derived from a similar proof in [127, p. 96]. Theorem 2.20 is an application of the Lovász Local Lemma [94]. For a clear proof of the Lovász Local Lemma see [255]. For a proof of Theorem 2.20 see [127]. The lower bounds of Table 2.2 are from [210]. The given development of the *tower*, *wow*, and *ack* functions can be found in [127]. Gowers' results (Theorem 2.21 and Remark 2.22) can be found in [121]. Theorem 2.23 is from [141].

§2.5. The Erdős and Turán function is defined in [102]. A proof pf Roth's 1952 result can be found in [229]. Szemerédi's 1969 result and proof can be found in [264]. A proof of the conjecture of Erdős and Turán by Szemerédi is found in [265]. Furstenburg [112] gives a proof of this result using ergodic theory. A shorter proof is given in [267]. Bounds on $\nu_3(n)$ are found in [26], [192], and [227]. Rankin [219] obtains a lower bound on $\nu_k(n)$ for k > 3. Some relatively parly results on the Erdős and Turán function may be found in [230]

52

and [231]. Pomerance [204] and Riddell [221] consider analogues of Szemerédi's theorem in the set of lattice points in the plane (points having both coordinates in \mathbb{Z}).

§2.6. For a brief history of van der Waerden's theorem, see [68], which contains many references. The original proof of van der Waerden's theorem, which was referred to as Baudet's conjecture before it became a theorem, can be found in [270], with a recounting of its discovery in [271]. The proof presented here (namely, Lemmas 2.35 and 2.36) is derived from [17] and is similar to the one found in [127]. The shortest proof (to date) is found in [126]. A topological proof can be found in [116], while an algebraic proof can be found in [30]. Other proofs can be found in [78], [189], [206], [249], and [266].

 $\S2.7$. The colorings given in Exercise 2.3 can be found in [71]. Exercise 2.9 uses what is known as the probabilistic method, which is the subject of a book by Erdős and Spencer [99].

§2.8. Wróblewski [275] proved the following result: there exists a set of positive integers $\{x_1 < x_2 < \cdots\}$ with no 3-term arithmetic progression such that $\sum_{i=1}^{\infty} \frac{1}{x_i} > 3.00849$ (which may help to explain why Research Problem 2.7 is considered difficult).

Additional References: For biographies of Paul Erdős, see [148] and [243]. An algorithm that gives partitions of \mathbb{Z}^+ into an infinite number of subsets, none of which contain a 3-term arithmetic progression, is discussed in [118]. Earlier related work is covered in [119]. In [259], Spencer showed (among other things) that for each $k \geq 1$, there exists a set S of positive integers so that for every finite coloring of S, there is a monochromatic k-term arithmetic progression, but such that S is "sparse" enough that it itself does not contain any (k + 1)-term arithmetic progressions. An alternate proof, using a direct construction, is given in [198].

Chapter 3

Supersets of AP

As we discussed in Chapter 1, we can define a function analogous to the van der Waerden function w(k; r) by substituting for AP (the set of arithmetic progressions) some other set of sequences. That is, if \mathcal{F} is some specific collection of sequences, denote by $R(\mathcal{F}, k; r)$ the least positive integer m (if it exists) such that every r-coloring of [1, m]yields a monochromatic k-term member of \mathcal{F} .

Of course, there is no guarantee that $R(\mathcal{F}, k; r)$ exists, unless we choose the collection \mathcal{F} wisely. As a simple example, let r = 2 and let \mathcal{F} be the family of all sequences of positive integers that begin with the pattern i, i + 1 for some i. In order for $R(\mathcal{F}, 2; 2)$ to exist, there must exist a number $n = R(\mathcal{F}, 2; 2)$ such that for every 2-coloring of [1, n], there is a monochromatic pair i, i + 1. This is not true since, for example, the coloring of \mathbb{Z}^+ where all odd numbers are colored blue and all even numbers are colored red does not yield such a pair.

Let us assume that \mathcal{F} is some collection of sequences such that $\mathcal{R}(\mathcal{F}, k; 2)$ exists for all k. If \mathcal{G} is a family of sequences such that $\mathcal{F} \subseteq \mathcal{G}$, then since we are assured of finding k-term monochromatic members of \mathcal{F} in $[1, \mathcal{R}(\mathcal{F}, k; 2)]$, we are also assured of finding k-term monochromatic members of \mathcal{G} in $[1, \mathcal{R}(\mathcal{F}, k; 2)]$. In fact, if \mathcal{G} is larger than \mathcal{F} , then it may happen that we do not have to go out as far as the integer $\mathcal{R}(\mathcal{F}, k; 2)$ in order to guarantee a monochromatic k-term member of \mathcal{G} . Summarizing, we have the following fact.
3.1. Quasi-progressions

3. Supersets of AP

Theorem 3.1. Let \mathcal{F} be a collection of sequences and let k be a positive integer such that $R(\mathcal{F}, k; 2)$ exists. If $\mathcal{F} \subseteq \mathcal{G}$, then $R(\mathcal{G}, k; 2)$ exists and $R(\mathcal{G}, k; 2) \leq R(\mathcal{F}, k; 2)$.

Before proceeding, we remind the reader that we may denote R(F,k;2) more simply as R(F,k).

As we saw in Chapter 2, R(AP, k) exists for all k; these are just the van der Waerden numbers w(k). We noted in Chapter 2 that the determination of the rate of growth of w(k) is a very difficult problem. Can we alter the problem somewhat so as to make a solution easier to obtain? Theorem 3.1 suggests a method: replace AP with a different (appropriately chosen) collection \mathcal{G} such that $AP \subseteq \mathcal{G}$.

If $S \subseteq T$, T is called a *superset* of S.

In this chapter we will look at several different choices of \mathcal{G} such that \mathcal{G} is a superset of AP, and study the corresponding function $R(\mathcal{G}, k)$. In some cases we will see that there are interesting relationships between $R(\mathcal{G}, k)$ and w(k). The supersets of AP that we present here are by no means exhaustive, and we encourage the reader to consider other choices of \mathcal{G} .

3.1. Quasi-progressions

An arithmetic progression $\{a, a + d, a + 2d, ...\}$ can be thought of as an increasing sequence of positive integers such that the gap between each adjacent pair of integers is the constant d. One way of generalizing the notion of an arithmetic progression is to allow a little more variance in this gap. For example, we can require only that the gaps be either d or d + 1 for some positive integer d; or, that the gaps belong to the set $\{d, d + 1, d + 2\}$ for some $d \in \mathbb{Z}^+$. This prompts the next definition.

Definition 3.2. Let $k \ge 2$ and $n \ge 0$. A k-term quasi-progression of diameter n is a sequence of positive integers $\{x_1, x_2, \ldots, x_k\}$ such that there exists a positive integer d with the property that

$$d \le x_i - x_{i-1} \le d + n$$

for i = 2, 3, ..., k.

We call the integer d of Definition 3.2 a low-difference for the quasi-progression $X = \{x_1, x_2, \ldots, x_k\}$; and, when helpful to the discussion, we will say that X is a (k, n, d)-progression.

Example 3.3. Let A be any arithmetic progression with gap d. Then A is a quasi-progression of diameter 0. Note that A may also be considered a quasi-progression with diameter 1 (since all gaps belong to $\{d, d+1\}$ for some d). Any 1-element or 2-element sequence may be considered to have diameter 0.

Example 3.4. Let $B = \{1, 3, 5, 8\}$. We see that B is a 4-term quasi-progression with diameter 1 and low-difference 2; hence it is a (4, 1, 2)-progression. Note that B may also be considered a (4, 2, 2)-progression since all of the differences $x_i - x_{i-1}$ belong to the set $\{2, 3, 4\}$. Moreover, B is a (4, n, 2)-progression for all $n \ge 1$. Lastly, note that B is a (4, 2, 1)-progression since all of the gaps belong to $\{1, 2, 3\}$ – in fact, it is a (4, n, 1)-progression for each $n \ge 2$.

We see from Examples 3.3 and 3.4 that the diameter n and lowdifference d of a quasi-progression are not unique, but that the possible values for the low-difference depend on the choice of diameter. In particular, any quasi-progression of diameter $n \ge 0$ is also a quasiprogression of diameter m for all $m \ge n$. Obviously, for each $n \ge 0$, the set of quasi-progressions of diameter n is a superset of AP.

Although for a given sequence there are many choices for n and d, it will ordinarily be most advantageous for us to use the minimum value of n that works, and then to choose the minimum value for d corresponding to this n. Hence, we usually think of progression A of Example 3.3 as having low-difference d and diameter 0; and of progression B of Example 3.4 as having low-difference 2 and diameter 1.

We next mention a convenient notation for the function analogous to w(k), where we are concerned with quasi-progressions rather than arithmetic progressions.

Notation. For positive integers n and k, denote by $Q_n(k)$ the least positive integer m such that for every 2-coloring of [1, m] there is a monochromatic k-term quasi-progression of diameter n.

3. Supersets of AP

In the following theorem we list some rather obvious properties of $Q_n(k)$. We state these without proof and ask the reader to verify them.

Theorem 3.5. Let $n \ge 0$ and $k \ge 2$.

(i)
$$Q_n(k)$$
 exists.

- (*ii*) $Q_0(k) = w(k)$.
- (iii) $Q_0(k) \ge Q_1(k) \ge Q_2(k) \ge \cdots$.

Our goal is to find out as much as we can about the magnitude of the function $Q_n(k)$. We would hope that, for $n \ge 1$, a "reasonable" upper bound for $Q_n(k)$ is more easily obtainable than has been the case for the van der Waerden numbers. From Theorem 3.5(iii), we see that the larger the value of n, the more likely that we will be able to obtain a "not-so-big" upper bound on the function $Q_n(k)$.

We first look at the easiest case (where the diameter is at least k-1). In this case we are able to give the exact value of $Q_n(k)$.

We first give a lower bound for $Q_n(k)$ that holds for all $n \ge 1$ and all $k \ge 2$.

Theorem 3.6. Let $k \ge 2$ and $n \ge 1$. Then $Q_n(k) \ge 2k - 1$.

Proof. Consider the 2-coloring $\chi : [1, 2k - 2] \rightarrow \{0, 1\}$ defined by $\chi([1, k - 1]) = 1$ and $\chi([k, 2k - 2]) = 0$. This coloring admits no monochromatic k-element set. In particular, it yields no k-term monochromatic quasi-progression of diameter n. Therefore, we have $Q_n(k) \ge 2k - 1$.

We next show that the lower bound given in Theorem 3.6 also serves as an upper bound when n = k - 1.

Theorem 3.7. $Q_{k-1}(k) \leq 2k - 1$ for all $k \geq 2$.

Proof. Let χ be an arbitrary 2-coloring of [1, 2k-1]. Clearly there is some k-element set $X = \{x_1 < x_2 < \cdots < x_k\}$ that is monochromatic under χ . If for some $j, 2 \leq j \leq k$, we have $x_j - x_{j-1} > k$, then

$$x_k - x_1 = \sum_{i=2}^{\kappa} (x_i - x_{i-1}) > k + (k-2) = 2k - 2,$$

with is not possible. Thus X is a monochromatic

8.1. Quasi-progressions

which is not possible. Thus, X is a monochromatic (k, k - 1, 1)progression, since $1 \le x_i - x_{i-1} \le k$ for $2 \le i \le k$. This shows that $Q_{k-1}(k) \le 2k - 1$.

Note that by Theorems 3.5(iii), 3.6, and 3.7 we have the following corollary.

Corollary 3.8. For all $k \geq 2$ and all $n \geq k-1$, $Q_n(k) = 2k-1$.

As we alluded to earlier, it would be most desirable to obtain results in which the diameters are as small as possible. Although no "nice" upper bound is known for the function $Q_1(k)$, we look next at a lower bound for this function.

Theorem 3.9. $Q_1(k) \ge 2(k-1)^2 + 1$ for $k \ge 2$.

Proof. Define the 2-coloring χ of $[1, 2(k-1)^2]$ by the string

$$\underbrace{00\ldots 0}_{k-1}\underbrace{11\ldots 1}_{k-1}\underbrace{00\ldots 0}_{k-1}\underbrace{11\ldots 1}_{k-1}\ldots\underbrace{00\ldots 0}_{k-1}\underbrace{11\ldots 1}_{k-1},$$

where each of the (2k-2)-element blocks $\underbrace{00\ldots 0}_{k-1}\underbrace{11\ldots 1}_{k-1}$ appears k-1

times. To prove the theorem, it suffices to show that under this coloring there is no k-term monochromatic quasi-progression of diameter 1.

By way of contradiction, let $m = 2(k-1)^2$, and assume that $X = \{x_1, x_2, \ldots, x_k\} \subseteq [1, m]$ is a quasi-progression of diameter 1 that is monochromatic under χ . By the symmetry of χ , without any loss of generality, we may assume that $\chi(X) = 1$. Since each monochromatic block of color 1 has k-1 elements, there is some i, $2 \le i \le k$, where x_i and x_{i-1} belong to two different such blocks. For this i, we have $x_i - x_{i-1} \ge k$. Since X has diameter 1, this implies that X has a low-difference of at least k-1. Thus, each of the blocks of k-1 consecutive 1's contains no more than one member of X. Hence, X must have length at most k-1, a contradiction.

In the following theorem, we use a generalization of the coloring used in the proof of Theorem 3.9 to obtain a lower bound for $Q_{k-i}(k)$ in terms of i and k.

Theorem 3.10. Let $1 \leq i < k$ and let $m = 1 + \lfloor \frac{k-2}{i} \rfloor$. Then

(3.1)
$$Q_{k-i}(k) \ge 2\left(\left\lfloor \frac{k-1}{m} \right\rfloor (k-1-im) + i(k-1)\right) + 1$$

Before presenting the proof, first notice that this is indeed a generalization of Theorem 3.9, since if i = k - 1, we have m = 1, and thus Theorem 3.10 gives $Q_1(k) \ge 2((k-1)(0) + (k-1)^2) + 1$. At the other extreme, if we take i = 1, then m = k - 1, so that $Q_{k-1}(k) \ge 2((k-1-(k-1))+k-1)+1$, giving us Theorem 3.6.

Proof of Theorem 3.10. Let $s = 2\left(\lfloor \frac{k-1}{m} \rfloor (k-1-im) + i(k-1)\right)$. Define the 2-coloring χ of [1, s] by the string

$$1^{y}(0^{k-1}1^{k-1}0^{k-1}1^{k-1}\dots 0^{k-1}1^{k-1})0^{y},$$

where within the parentheses each of the blocks $0^{k-1}1^{k-1}$ occurs $\lfloor \frac{k-1}{m} \rfloor$ times, and where $y = i \left(k - 1 - m \lfloor \frac{k-1}{m} \rfloor\right)$. Note that this is, in fact, a string of length s. To establish (3.1) it is sufficient to show that, under χ , [1, s] contains no monochromatic k-term quasi-progression of diameter k - i. We proceed by contradiction.

Assume that $X = \{x_1, x_2, \ldots, x_k\} \subseteq [1, s]$ is a quasi-progression of diameter k-i that is monochromatic under χ . By the symmetry of χ , we may assume that $\chi(X) = 1$. Note that since $m \lfloor \frac{k-1}{m} \rfloor \geq k - m$ (why?),

$$y = i \left(k - 1 - m \left\lfloor \frac{k-1}{m} \right\rfloor\right)$$

$$\leq i(k - 1 - (k - m))$$

$$= i \left\lfloor \frac{k-2}{i} \right\rfloor$$

$$\leq k - 2.$$

Hence, there is no block of more than k-1 consecutive 1's. Thus, for some $j \in \{2, 3, \ldots, k\}$, we have $x_j - x_{j-1} \ge k$, which implies that X cannot have a low-difference that is less than i.

Since the low-difference of X is at least *i*, the first block of 1's (having length y), contains at most $\frac{y}{i} = k - 1 - m \lfloor \frac{k-1}{m} \rfloor$ members of X. Similarly, in any block of k - 1 consecutive 1's, there are at most $1 + \lfloor \frac{k-2}{i} \rfloor = m$ members of X. There are $\lfloor \frac{k-1}{m} \rfloor$ blocks of k - 1 consecutive 1's, so that, accounting for all blocks of 1's, we see that

3.1. Quasi-progressions

X has at most

$$k-1-m\left\lfloor\frac{k-1}{m}\right\rfloor+m\left\lfloor\frac{k-1}{m}\right\rfloor=k-1$$

elements, a contradiction.

Computing some actual values of Ramsey-type functions can be quite helpful in forming conjectures about the magnitude (or rate of growth) of the functions. As we saw in Chapter 2, for the classical van der Waerden numbers w(k; r), the computations can be quite prohibitive. In dealing with supersets of AP, however, we often find the computations much more reasonable. In fact, in some circumstances we have enough computed data to help us form "educated" conjectures. Such is the situation with certain special cases of Theorem 3.10. For these cases, computer-generated values of $Q_{k-i}(k)$ suggest that, whenever $i \leq \frac{k}{2}$, the right-hand side of (3.1) is the precise value of $Q_{k-i}(k)$ (see Table 3.1 at the end of this section). We single out these special cases in the following corollary.

Corollary 3.11. *Let* $1 \le i \le k - 1$ *.*

(i) If $k \equiv 0 \pmod{i}$, then $Q_{k-i}(k) \ge 2ik - 4i + 3$. (ii) If $k \equiv 1 \pmod{i}$, then $Q_{k-i}(k) \ge 2ik - 2i + 1$.

Proof. We will prove (ii), and leave the proof of (i) to the reader as Exercise 3.2. Letting k = ti + 1, then using the notation of Theorem 3.10, we have $m = 1 + \lfloor \frac{ti-1}{i} \rfloor = t$, and $\lfloor \frac{k-1}{m} \rfloor = i$. Therefore, (3.1) becomes

$$Q_{k-i}(k) \ge 2(t(k-1-it)+i(k-1))+1 = 2i(k-1)+1.$$

We see that Corollary 3.11(ii) implies Theorem 3.9, simply by letting i = k - 1. Also notice that by letting i = 1 in Corollary 3.11(i), we get that $Q_{k-1}(k) \ge 2k - 1$, as given by Theorem 3.6. So far, i = 1 is the only case we have presented for which we have a precise formula for $Q_{k-i}(k)$. In the next theorem, we give the formula for the only other nontrivial value of i for which an exact formula for $Q_{k-i}(k)$ is known.

60

3. Supersets of *AP*

Theorem 3.12. Let $k \ge 2$. Then

$$Q_{k-2}(k) = \begin{cases} 4k-5 & \text{if } k \text{ is even,} \\ 4k-3 & \text{if } k \text{ is odd.} \end{cases}$$

Proof. Letting i = 2 in Corollary 3.11, we obtain $Q_{k-2}(k) \ge 4k - 5$ when k is even, and $Q_{k-2}(k) \ge 4k - 3$ when k is odd. Hence we need only establish these as upper bounds for $Q_{k-2}(k)$.

To obtain the upper bounds, let $\chi : \mathbb{Z}^+ \to \{0, 1\}$ be any 2coloring. We will show that if k is even then there is a monochromatic k-term quasi-progression with diameter k - 2 in [1, 4k - 5], and that if k is odd then there exists such a progression in [1, 4k - 3].

First note that the upper bound obviously works for k = 2, since $Q_0(2) = w(2) = 3$. So we may assume that $k \ge 3$. Notice that from the proof of Theorem 3.7, [1, 2k - 3] contains a monochromatic (k - 1, k - 2, 1)-progression $\{x_1 < x_2 < \cdots < x_{k-1}\}$, say of color 1. Now if $x_{k-1} + j$ has color 1 for some $j \in \{1, 2, \ldots, k - 1\}$, then [1, 4k - 5] will contain the k-term monochromatic quasi-progression $\{x_1, x_2, \ldots, x_{k-1}\} \cup \{x_{k-1} + j\}$, having diameter k - 2, so in this case we are done. Thus, we may assume that $[x_{k-1} + 1, x_{k-1} + k - 1]$ has color 0.

If there is some integer b with color 0 that satisfies

$$x_{k-1} - (k-2) \le b < x_{k-1},$$

then $\{b\} \cup [x_{k-1}+1, x_{k-1}+k-1]$ forms a monochromatic (k, k-2, 1)progression, and we are done. Hence, we may further assume that $\chi([x_{k-1}-(k-2), x_{k-1}-1]) = 1$. For ease of notation, we let

$$a = x_{k-1} - (k-2),$$

so that $\chi([a, a + k - 2]) = 1$.

Note that $a \leq k-1$, since $x_{k-1} \leq 2k-3$. Hence we may assume that $\chi(x) = 0$ for all x < a, for otherwise for some $x_0 < a$ the set $\{x_0\} \cup [a, a+k-2]$ would be a monochromatic k-term quasi-progression with diameter k-2 (having color 1), and we would be done. Likewise, we assume that all of [a + k - 1, a + 2k - 3] has color 0 (why?), and hence that all of [a + 2k - 2, a + 3k - 4] has color 1. Finally, we may assume that [a + 3k - 3, a + 4k - 5] has color 0.

To complete the proof, we consider three cases. In each case we find a monochromatic quasi-progression with the desired properties. **Case 1.** k is even. In this case, the set

$$\left\{a+2i: 0 \le i \le \frac{k-2}{2}\right\} \bigcup \left\{a+2i: k-1 \le i \le \frac{3k-4}{2}\right\}$$

is monochromatic of color 1, has length k, and (since $a \le k - 1$) is contained in [1, 4k - 5]. It obviously is a quasi-progression having diameter k - 2 and low-difference 2.

Case 2. k is odd and a = 1. If $\chi(4k - 3) = 0$, then the interval [3k - 2, 4k - 3] is a monochromatic k-term quasi-progression with diameter k-2. If, on the other hand, $\chi(4k-3) = 1$, then the sequence $\{1, 4, 6, 8, \ldots, k-1, 2k-1, 2k+2, 2k+4, 2k+6, \ldots, 3k-3, 4k-3\}$ is a k-term quasi-progression with color 1, having diameter k-2.

Case 3. k is odd and $2 \le a \le k - 1$. Let

3.1. Quasi-progressions

$$\begin{vmatrix} A_1 = \{a - 1 - 2i : 0 \le i \le \lfloor \frac{a}{2} \rfloor - 1\}, \\ A_2 = \{a + k - 1\} \cup \{a + k + 2, a + k + 4, \dots, a + 2k - 5, a + 2k - 3\}, \\ A_3 = \{a + 3k - 3 + 2i : 0 \le i \le \lfloor \frac{k - a}{2} \rfloor\}. \end{vmatrix}$$

Then $A_1 \cup A_2 \cup A_3$ is contained in [1, 4k - 3] and forms a k-term quasi-progression of diameter k - 2. We leave the details as Exercise 3.3.

We next turn our attention to the problem of finding upper bounds for $Q_n(k)$.

The main result we have on upper bounds for $Q_n(k)$ relies on the following lemma. The proof of the lemma is fairly complicated, and we choose not to include it here.

Lemma 3.13. For $3 \le j \le \frac{n}{2}$, let E(n, j) be the least positive integer such that for every 2-coloring of [1, E(n, j)] there is a monochromatic $(n + j - 1, n, \ell)$ -progression for some $\ell \le j - 1$. Then

(3.2)
$$E(n, j+1) \le E(n, j) + 2\lambda(n+j-1),$$

where
$$\lambda = \left[\frac{n+j-2}{\lfloor \frac{n+j}{j} \rfloor}\right] + 1.$$

3.1. Quasi-progressions

With the help of Lemma 3.13, we are now able to obtain a bound on $Q_n(k)$, provided k does not exceed $\frac{3n}{2}$.

3. Supersets of AP

Theorem 3.14. Let $3 \le t \le \frac{n}{2}$. Then

(3.3)
$$Q_n(n+t) \le \frac{1}{2n}t^4 + \frac{4}{3}t^3 + (n+2)t^2 + 8nt$$

Proof. First, we note that by the definition of E(n, j) we have $Q_n(n+t) \leq E(n, t+1)$. To complete the theorem, we will show that E(n, t+1) is bounded above by the right-hand side of (3.3). Applying (3.2) t-2 times, we obtain

$$\begin{split} E(n,t+1) &\leq E(n,3) + 2\sum_{r=3}^{t} (n+r-1) \left(\left\lceil \frac{n+r-2}{\lfloor \frac{n+r}{r} \rfloor} \right\rceil + 1 \right) \\ &\leq E(n,3) + 2\sum_{r=3}^{t} (n+r-1) \left(\frac{n+r-3}{\lfloor \frac{n+r}{r} \rfloor} + 2 \right) \\ &\leq E(n,3) + 2\sum_{r=3}^{t} (n+r-1) \left(\frac{nr+r^2-3r}{n+1} + 2 \right). \end{split}$$

By the proof of Theorem 3.12, for every 2-coloring of [1, 4(n+2) - 3]there is a monochromatic (n + 2)-term quasi-progression of diameter n and low-difference at most 2. Thus, $E(n, 3) \le 4n + 5$. This gives

$$E(n,t+1) \le 4n+5+4\sum_{\substack{r=3\\r=1}}^{t}(n+r-1) + \frac{2}{n+1}\sum_{r=1}^{t-2}(n+r+1)(r^2+r(n+1)+2n-2).$$

Hence,

$$\begin{split} E(n,t+1) &\leq 4n+5+4nt+2t^2 \\ &+ \frac{2}{n} \left(\sum_{r=1}^{t-2} (r^3+r^2(2n+2)+r(n^2+4n-1)+2n^2-2) \right) \! . \end{split}$$

Simplifying, we get

$$E(n,t+1) \le 4n+5+\frac{2}{n}\left(\frac{t^4}{4}+\frac{2}{3}nt^3+\frac{(n^2+4n-1)(t-1)(t-2)}{2}\right)$$

(3.4) $+\frac{2}{n}\left(2(t-1)(n^2-1)+2n^2t+nt^2\right).$

To complete the proof, we note that since $3 \le t \le \frac{n}{2}$,

$$13 + 4t^2 + 2n + \frac{2}{n} \le \frac{t^2}{n} + 2tn + 8t + \frac{2t}{n}$$

(which may be shown in a straightforward manner by induction on n). Hence,

$$4n + 5 + (t - 1)(t - 2)(n + 4 - \frac{1}{n}) + 4(t - 1)(n - \frac{1}{n}) \le nt^2 + 4nt,$$

which implies that the right hand side of (3.4) is not greater than $\frac{t^4}{2n} + \frac{4}{3}t^3 + nt^2 + 8nt + 2t^2$. Since E(n, t+1) is bounded above by this expression, the proof is complete.

As mentioned before, we seek an upper bound on $Q_n(k)$ for the smallest possible value of n. The best result that Theorem 3.14 provides in this regard is for the case in which n is two-thirds of the value of k. We single out this bound in the next corollary, which is a direct consequence of Theorem 3.14; its proof is left as Exercise 3.4.

Corollary 3.15.

$$Q_{\lceil \frac{2k}{3} \rceil}(k) \le \frac{43}{324}k^3(1+o(1)).$$

We note that Theorem 3.10 gives a quadratic lower bound for $Q_{\lfloor \frac{2k}{3} \rfloor}(k)$, namely $\frac{2}{3}k^2(1+o(1))$, while the upper bound of Corollary 3.15 is a cubic. It would be interesting to know the actual rate of growth for this Ramsey-type function.

We end this section with a table. For a given pair (k, i) the corresponding entry in the table is the value (or the best known lower bound) for $Q_{k-i}(k)$. We only include those pairs where $i \ge 3$, since we have presented exact formulas for i = 1 and i = 2.

$k \setminus i$	3	4	5	6	7	8
4	19	-	-	-	-	-
5	29	33	-	-	-	-
6	27	49	67	-	-	-
7	37	65	73	≥ 124	-	-
8	39	51	93	?	≥ 190	-
9	45	65	≥ 115	?	?	≥ 287
10	55	67	83	?	?	?
11	57	75	101	?	?	?
12	63	83	≥ 103	?	?	?
13	73	97	≥ 108	?	?	?
14	75	99	≥ 119	?	?	?
15	81	?	≥ 133	?	?	?
16	91	?	?	?	?	?
17	93	?	?	?	?	?

Table 3.1: Values and lower bounds for $Q_{k-i}(k)$

3.2. Generalized Quasi-progressions

We now direct our attention to a type of sequence that generalizes the notion of a quasi-progression. The idea of this generalization is that we allow the "diameter" to vary with the terms of the sequence. Specifically, we give the following definition.

Definition 3.16. Let $\delta : \{2, 3, ...\} \rightarrow [0, \infty)$ be a function. A kterm generalized quasi-progression with diameter function δ (or a GQ_{δ} progression) is a sequence $\{x_1, x_2, ..., x_k\}$ such that, for some positive integer d, we have $d \leq x_i - x_{i-1} \leq d + \delta(i)$ for all i = 2, 3, ..., k.

We see that generalized quasi-progressions have "diameters" that may vary, and that a quasi-progression of diameter n may be considered a generalized quasi-progression, where δ is the constant function n. As we did with quasi-progressions, we will refer to d as the lowdifference of the progression.

Analogous to the notation $Q_n(k)$, we use the following notation for the Ramsey-type function associated with generalized quasiprogressions.

6.2. Generalized Quasi-progressions

Notation. Let $\delta : \{2, 3, ...\} \to [0, \infty)$ be a function and k a positive integer. Denote by $GQ_{\delta}(k)$ the least positive integer such that for every 2-coloring of $[1, GQ_{\delta}(k)]$ there is a monochromatic k-term GQ_{δ} -progression.

Example 3.17. Let δ be the function defined as the constant function $\delta(x) = 1$ for all x. Then in this case $GQ_{\delta}(k)$ has the same meaning as $Q_1(k)$. Likewise, if δ is the zero function, then $GQ_{\delta}(k)$ is simply the classical van der Waerden function w(k). Moreover, for any constant function $\delta = c$ with $c \ge 0$, $GQ_{\delta}(k)$ coincides with $Q_c(k)$.

When $\delta = \delta(x)$ is a specified function of x, it is often convenient to use the notation $GQ_{\delta(x)}$. For example if $\delta(x) = x^2$, then we may refer to a GQ_{δ} -progression as a GQ_{x^2} -progression and denote the associated Ramsey-type function by $GQ_{x^2}(k)$.

Example 3.18. Let δ be the function $\delta(x) = x-1$ for all $x \ge 2$. Then $\{x_1, x_2, \ldots, x_k\}$ is a GQ_{x-1} -progression provided there is a positive integer d such that $x_i - x_{i-1} \in [d, d+i-1]$ for $2 \le i \le k$. For example, $\{1, 3, 6, 7, 10, 16\}$ is a GQ_{x-1} -progression where d = 1, and $\{4, 6, 8, 10, 16\}$ is a GQ_{x-1} -progression where d = 2. Thus, unlike a quasi-progression, where the gaps between consecutive members of the progression (even an infinite progression) must belong to an interval [d, d+n] of fixed length (since n is fixed), the set of allowable gaps in a GQ_{x-1} -progression can grow as the terms of the sequence increase (so that if the progression is infinite, there is no upper bound on the entire set of possible gaps).

It is clear that the smaller the values of the function δ are, the closer to the classical van der Waerden numbers the function $GQ_{\delta}(k)$ **a**. Moreover, it is easy to see that $\delta_1(x) \leq \delta_2(x)$, for all x, implies $GQ_{\delta_1}(k) \geq GQ_{\delta_2}(k)$. For some functions δ , it is not difficult to obtain upper bounds for $GQ_{\delta}(k)$. Finding an upper bound for $GQ_{x-1}(k)$, the function discussed in Example 3.18, is left as Exercise 3.5. In the next theorem, we give an upper bound on $GQ_{x-2}(k)$. Our proof makes use of some known values of $GQ_{x-2}(k)$ (these and other values of $GQ_{q(x)}(k)$ are given in Table 3.2 at the end of this section). **Theorem 3.19.** Let $k \ge 2$ and $m_k = 1 + 2(k-1)^2 + 2\sum_{j=4}^{k-1} j(j-1)$. Then every 2-coloring of $[1, m_k]$ has a monochromatic k-term GQ_{x-2} -progression with low-difference $d \le 2(k-1)$. In particular, for $k \ge 5$, $GQ_{x-2}(k) \le \frac{2}{3}k^3 - \frac{8}{3}k - 13$.

Proof. The second claim of the theorem follows from the first via a straightforward computation, which we leave as Exercise 3.6.

To prove the first claim, we note the following values are known: $GQ_{x-2}(2) = 3$, $GQ_{x-2}(3) = 9$, and $GQ_{x-2}(4) = 19$. It is obvious from these values that for k = 2, 3, 4 it is also true that any k-term GQ_{x-2} -progression has a low-difference not exceeding 2(k-1). Hence the theorem is true for k = 2, 3, 4.

We shall complete the proof by induction on k. To this end, assume that $k \ge 4$ is an integer for which the statement is true, and let $\chi : [1, m_{k+1}] \rightarrow \{0, 1\}$ be any 2-coloring. Therefore, there exists a monochromatic GQ_{x-2} -progression $X = \{x_1 < x_2 < \cdots < x_k\}$ with low-difference $d \le 2(k-1)$ and $x_k \le m_k$. Without loss of generality, let $\chi(X) = 1$.

For $0 \le t \le 2k-1$, let $A_t = [x_k + d + kt, x_k + d + kt + k - 1]$, and let $A_{2k} = \{x_k + d + 2k^2\}$. Notice that

$$x_{k} + d + 2k^{2} \leq 1 + 2(k-1)^{2} + 2\sum_{j=4}^{k-1} j(j-1) + 2k^{2} + 2(k-1)$$
$$= 1 + 2k^{2} + 2\sum_{j=4}^{k} j(j-1)$$
$$= m_{k+1}.$$

Hence, $A_t \subseteq [1, m_{k+1}]$ for $0 \le t \le 2k$. Consider

$$A_0 = [x_k + d, x + d + k - 1].$$

If any $y \in A_0$ has color 1, then $X \cup \{y\}$ forms a monochromatic GQ_{x-2} -progression of low-difference d, and we are done (the difference between x_k and y does not exceed d + (k + 1) - 2). Hence, we may assume that all members of A_0 have color 0. From this we may assume that all members of $A_1 = [x_k + d + k, x_k + d + 2k - 1]$ have color 1 (or else, for some $z \in A_1$, the set $A_0 \cup \{z\}$ is a monochromatic (k + 1)-term sequence where the gap between the largest two terms lies in the

3.2. Generalized Quasi-progressions

interval [1, k], thus forming a GQ_{x-2} -progression with low-difference 1). Using this same argument, we may assume that all of A_2 has color 0. Continuing in this way, we are left with the situation in which all members of A_t have color 1 for t odd, and color 0 for t even. However, this gives us the monochromatic set $\{x_k + d + 2tk : 0 \le t \le k\}$ (it has color 0), which is a (k+1)-term GQ_{x-2} -progression (actually an arithmetic progression) with low-difference 2k, which completes the proof.

Theorem 3.19 gives a cubic polynomial in the variable k as an upper bound for $GQ_{x-2}(k)$. In Exercise 3.5, you are asked to establish a quadratic upper bound for $GQ_{x-1}(k)$. We might conjecture (although these two results offer quite limited evidence) that for large enough k, $GQ_{x-m}(k)$ is bounded above by a polynomial of degree m + 1. Noting that $GQ_{x-k}(k)$ has the same meaning as w(k), the truth of a conjecture of this type could have exciting ramifications for the van der Waerden numbers themselves.

There are other ways in which we could allow the "diameters" to grow. As one example, we could insist that the first few terms of the progression (say the first three terms) form an arithmetic progression, and then allow the diameters to increase as the terms of the progression increase. Such variations on topics covered in this book can be the seeds of interesting and meaningful research projects (and, we may hope, research projects that inspire further exploration by others).

We end this section with a table of known values of $GQ_{f(x)}(k)$ for some particular functions f(x). Notice that the classical 2-color van der Waerden numbers lie along a diagonal in the table.

$f(x)\setminus k$	3	4	5	6	7	8	9	10
$\begin{bmatrix} x \end{bmatrix}$	6	10	14	20	27	34	44	52
x-1	9	19	33	52	74	100	?	?
x-2	9	22	38	60	?	?	?	?
x-3	9	35	59	≥ 88	?	?	?	?
x-4	9	35	178	?	?	?	?	?

Table 3.2: Values and lower bounds for $GQ_{f(x)}(k)$

3. Supersets of AP

3.3. Descending Waves

In this section we consider another type of sequence that generalizes the notion of an arithmetic progression, called a *descending wave*. We begin with the definition.

Definition 3.20. For $k \ge 3$, an increasing sequence of positive integers $\{x_1, x_2, \ldots, x_k\}$ is a *descending wave* if $x_i - x_{i-1} \le x_{i-1} - x_{i-2}$ for $i = 3, 4, \ldots, k$.

As examples, the sequence $\{1, 4, 6, 8, 9\}$ is a 5-term descending wave, while $\{1, 5, 10\}$ fails to be a descending wave. We will denote by DW(k) the least positive integer such that for every 2-coloring of [1, DW(k)] there is a monochromatic k-term descending wave. It is obvious that every arithmetic progression is a descending wave, and therefore, by van der Waerden's theorem, DW(k) exists for all k. There does not appear to be any obvious implication between the property of being a descending wave and that of being a quasiprogression. Clearly, not every quasi-progression of diameter $n \ge 1$ is a descending wave; and conversely, for any given n > 1, it is easy to find a descending wave of length k > 3 that is not a quasi-progression of diameter n. For example, if n = 2, then $\{1, 8, 12\}$ is a descending wave, but is not a quasi-progression of diameter 2. Although here we will confine the discussion of descending waves to the existence and magnitude of the function DW(k), we mention that it is known that any infinite set of positive integers that contains arbitrarily long quasi-progressions of diameter d, for some fixed d, must also contain arbitrarily long descending waves.

We are able to give both upper and lower bounds for DW(k). We see from the bounds given by Theorems 3.21 and 3.24 (below) that DW(k) grows like a polynomial of degree three.

We start with a simple upper bound.

Theorem 3.21. For all $k \ge 3$, $DW(k) \le \frac{k^3}{2} - \frac{k^2}{2} + 1$.

Proof. Let $n = \frac{k^3}{2} - \frac{k^2}{2} + 1$. We will show that for any 2-coloring $\chi : [1, n] \to \{0, 1\}$ there is a monochromatic k-term descending wave. Assume, without loss of generality, that $\chi(1) = 0$. If there exist k

consecutive integers of color 1, we are done. So, assume that no k consecutive integers of color 1 exist under χ . Let $x_0 = 0$ and $x_1 = 1$ and define, for $i \ge 1$,

$$x_{i+1} = \min \{ y : y - x_i \ge x_i - x_{i-1}, \chi(y) = 0 \}$$

We leave it to the reader in Exercise 3.9 to show that x_1, x_2, \ldots, x_k is a monochromatic k-term descending wave and that $x_k \leq n$.

The best known upper bound is slightly better than that provided by Theorem 3.21. We state this result, without proof, as Theorem 3.22.

Theorem 3.22. For all $k \ge 3$, $DW(k) \le \frac{1}{2}k^3 - \frac{4}{2}k + 3$.

Turning to lower bounds, we have the following result.

Theorem 3.23. For all $k \ge 3$, $DW(k) \ge k^2 - k + 1$.

Proof. Let $k \ge 3$ and let χ be the 2-coloring of $[1, k^2 - k]$ defined by the string

$$\underbrace{\underbrace{00\ldots0}_{k-1}\underbrace{11\ldots1}_{k-1}\underbrace{00\ldots0}_{k-2}\underbrace{11\ldots1}_{k-2}\cdots\underbrace{00}_{2}\underbrace{11}_{2}\underbrace{0}_{1}\underbrace{1}_{1}}_{1}\cdot$$

We will show that there is no k-term descending wave with color 0. The proof that there is also none with color 1 is similar and is left as Exercise 3.10. Assume, for a contradiction, that

$$X = \{x_1 < x_2 < \dots < x_k\}$$

is a descending wave with color 0.

Notice that χ includes exactly k-1 different blocks of consecutive 0's – call these blocks $B_1, B_2, \ldots, B_{k-1}$, where $|B_i| = k - i$. By the pigeonhole principle, some block contains more than one member of X. Let t be the least integer such that $|B_t \cap X| \geq 2$. Let

$$X' = X \cap (B_1 \cup B_2 \cup \cdots \cup B_t).$$

Since for i < t, B_i contains at most one member of X, we have

 $|X'| \le (t-1) + k - t = k - 1.$

72

Hence, $X' = \{x_1 < x_2 < \dots < x_r\}$ with $x_{r-1}, x_r \in B_t$ and $r \le k - 1$. Therefore,

$$x_{r+1} - x_r \ge k - t + 1 > x_r - x_{r-1},$$

contradicting our assumption that X is a descending wave.

We now state, without proof, the best known lower bound, which shows, in part, that the asymptotic rate of growth of DW(k) is that of a cubic polynomial.

Theorem 3.24. For all $k \ge 3$, there exists a positive constant c such that $DW(k) \ge ck^3$.

3.4. Semi-progressions

Recall that a quasi-progression of diameter n is a sequence $\{x_i\}_{i=1}^k$ such that, for some $d \in \mathbb{Z}^+$, we have $x_i - x_{i-1} \in [d, d+n]$ for $2 \leq i \leq k$. We might think of this as a "loosening" of the property of being an arithmetic progression, by allowing the gaps between consecutive terms "some slack." Another way to generalize the notion of an arithmetic progression is to allow the gaps to vary by *multiples* of some d, rather than by *additions* to d. With this idea in mind, we give the following definition.

Definition 3.25. For $k, m \in \mathbb{Z}^+$, a k-term semi-progression of scope m is a sequence of positive integers $\{x_1, x_2, \ldots, x_k\}$ such that, for some $d \in \mathbb{Z}^+$, $x_i - x_{i-1} \in \{d, 2d, \ldots, md\}$ for all $i, 2 \leq i \leq k$.

We shall use the notation $SP_m(k)$ for the corresponding Ramseytype function. That is, we denote by $SP_m(k)$ the least positive integer such that for every 2-coloring of $[1, SP_m(k)]$ there is a monochromatic k-term semi-progression of scope m. Since for each positive integer m, any arithmetic progression is also a semi-progression of scope m, the collection of semi-progressions of scope m is a superset of AP. Hence, $SP_m(k)$ exists for all m and k. In fact, the following observation is an immediate consequence of the definition of a semi-progression.

Proposition 3.26. For all $k \ge 1$, $w(k) = SP_1(k) \ge SP_2(k) \ge \cdots$.

3.4. Semi-progressions

One of the motivations for studying the Ramsey properties of supersets of AP is the potential for gaining more information about w(k). In this regard, the functions $SP_m(k)$ may be a more relevant extension of the the van der Waerden numbers than are the functions $Q_m(k)$. Our reason for saying this has to do with the following definition.

Definition 3.27. For $m, k \in \mathbb{Z}^+$, define $\Gamma_m(k)$ to be the least positive integer s such that for every s-element set $S = \{x_1, x_2, \ldots, x_s\}$ with $x_i - x_{i-1} \in \{1, 2, \ldots, m\}$ for $2 \leq i \leq s$, there is a k-term arithmetic progression in S.

To illustrate this definition, we look at some examples.

Example 3.28. Let us find $\Gamma_2(3)$. We want the least s such that whenever $X = \{x_1, x_2, \ldots, x_s\}$ is a sequence with $x_i - x_{i-1} \in \{1, 2\}$ for each i, then X will contain a 3-term arithmetic progression. Does s = 3 work? No – the set $\{1, 2, 4\}$ is a 3-term sequence, with the property that the gaps between consecutive terms belong to $\{1, 2\}$, yet there is no 3-term arithmetic progression. Does s = 4 work? No – consider $\{1, 2, 4, 5\}$, whose gaps all lie in $\{1, 2\}$, but which also fails to have a 3-term arithmetic progression. As it turns out, $\Gamma_2(3) = 5$. To prove this, by a simple translation, it is sufficient to consider all 5-term increasing sequences whose first element is 1, and where each of the four gaps is either 1 or 2. There are then only $2^4 = 16$ possible sequences to check. We leave it to the reader to check that each of these sixteen sequences contains some 3-term arithmetic progression.

Example 3.29. Consider $\Gamma_1(k)$. Here we want to look at sets of consecutive integers, and ask for the size of such sets that would guarantee a k-term arithmetic progression. It is easy to see that we have $\Gamma_1(k) = k$, since any set of consecutive integers is itself an arithmetic progression.

The above examples prompt an obvious question: does $\Gamma_m(k)$ always exist? This question is answered, in the affirmative, by the following proposition.

Proposition 3.30. $\Gamma_r(k) \leq w(k;r)$ for all $k, r \in \mathbb{Z}^+$.

Proof. Let w = w(k; r) and let $X = \{x_i : 1 \le i \le w\}$ be a sequence of positive integers with $1 \le x_i - x_{i-1} \le r$ for i = 2, 3, ..., w. We wish to show that X contains a k-term arithmetic progression. As noted in Example 3.28, we may assume $x_1 = 1$.

Consider the coloring $\chi: [1, w] \to \{1, 2, \dots, r\}$ defined by

 $\chi(y) = j$ if and only if $j = \min\{x_i - y : x_i \ge y, x_i \in X\}.$

To see that χ is an *r*-coloring, note that no consecutive elements of X differ by more than *r*. Hence, since χ is an *r*-coloring of [1, w], by the definition of w(k; r) there is a monochromatic arithmetic progression $A = \{a + nd : 0 \le n \le k - 1\}.$

Say the color of A is j_0 . This tells us that for each $a + nd \in A$, $0 \le n \le k-1$, there is some $x_{i_n} \in X$ such that $x_{i_n} - (a + nd) = j_0$. That is, $\{x_{i_n} : 0 \le n \le k-1\} = \{(j_0 + a) + nd : 0 \le n \le k-1\}$, so that X contains a k-term arithmetic progression, as desired. \Box

Now that we have established the existence of $\Gamma_m(k)$, we are able to explain the significance of the function $SP_m(k)$ as it relates to the search for an upper bound on w(k). The explanation is simply this:

Proposition 3.31. For all $k, m \ge 0, w(k) \le SP_m(\Gamma_m(k))$.

This inequality holds because any 2-coloring of $[1, SP_m(\Gamma_m(k))]$ must contain a monochromatic semi-progression of scope *m* having $\Gamma_m(k)$ terms. By the definition of Γ_m , among these $\Gamma_m(k)$ terms there must be a *k*-term arithmetic progression.

We next present some results concerning the magnitude of the function $SP_m(k)$. We begin with a simple formula for $SP_m(k)$ when $k \leq m$ (we leave the proof as Exercise 3.11).

Theorem 3.32. If $k \le m$, then $SP_m(k) = 2k - 1$.

As made evident by Proposition 3.26, the lower the value of m, the more significant (and probably the more difficult to obtain) any upper bounds on $SP_m(k)$ will be. In the following theorem we give a nice upper bound for $SP_m(k)$, but only provided m is more than half the value of k.

Theorem 3.33. Let $m \ge 2$. Assume m < k < 2m. Let $c = \left\lceil \frac{m}{2m-k} \right\rceil$. Then $SP_m(k) \le 2c(k-1) + 1$.

3.4. Semi-progressions

Proof. Let $\ell = 2c(k-1) + 1$ and let $\chi : [1, \ell] \to \{0, 1\}$ be any 2coloring. We will show that under χ there is a monochromatic k-term semi-progression of scope m. It is clear that among the 2(k-1) + 1elements of $[1, \ell]$ that are congruent to 1 modulo c, there is a set Awith $|A| \ge k$ such that A is monochromatic. Assume $\chi(A) = 1$, and let $X = \{x_1 < x_2 < \cdots < x_k\}$ consist of the k smallest members of A.

Let $d_i = x_i - x_{i-1}$ for $2 \le i \le k$. If $d_i \le cm$ for $i \in [2, k]$, then since, by the definition of $A, d_i \in \{c, 2c, \ldots, mc\}$ for each i, X is the desired semi-progression of scope m. Thus, we may assume that there exists $j \in \{2, 3, \ldots, k\}$ such that $d_j = cs$ with $s \ge m + 1$.

Now let $S = \{x_{j-1} + ci : 1 \le i \le s-1\}$. Then $\chi(S) = 0$ and $|S| \ge m$. We consider two cases.

Case 1. $c \geq 3$. Notice that if $x_{j-1} \leq m-c$ and $x_j \geq \ell - (m-c) + 1$, then since $x_k - x_1 \leq \ell - 1$, this would imply that

$$\sum_{\substack{i=2\\i\neq j}}^{k} d_i \le (\ell-1) - (\ell-2m+2c+1) = 2m - 2c - 2.$$

which would contradict the fact that $\sum_{i \neq j} d_i \geq c(k-2)$ (since $c \geq 2$ and m < k). Hence, either $x_{j-1} > m-c$ or $x_j < \ell - (m-c) + 1$. We shall cover the case in which $x_{j-1} > m-c$; the case in which $x_j < \ell - (m-c) + 1$ may be done by a symmetric argument.

So, assume that $x_{j-1} > m - c$ and let

$$B = \{ b \not\equiv 1 \pmod{c} : x_{j-1} - m + c \le b < x_j \}.$$

For every $b \in B$, there is some $t \in S$ so that $|t-b| \leq \max\{c, m\} = m$. Hence, if there is a set $B_0 \subseteq B$ such that $|B_0| = k-m$ and $\chi(B_0) = 0$, then $S \cup B_0$ is a monochromatic semi-progression of scope m with length at least k, and we are done (it is a semi-progression of scope msince the gaps between adjacent members all belong to $\{1, 2, \ldots, m\}$). Thus, we may assume that at most k - m - 1 members of B have color 0. Let $Y = \{y_1 < y_2 < \cdots < y_n\}$ be those members of B having color 1. To finish the proof of Case 1, we will show that Y provides us with the monochromatic semi-progression we are seeking by showing that the following two statements are true:

3. Supersets of *AP*

 $(3.5) y_i - y_{i-1} \le m \text{ for } 2 \le i \le n$

and

(3.6)

For convenience, let r = k - m. To prove (3.5), first list the elements of B in increasing order: $b_1 < b_2 < \cdots$. Notice that for any two elements, b_h and b_{h+g} , of B,

n > k.

$$b_{h+g} - b_h \le g + \left\lceil \frac{g}{c-1} \right\rceil$$

Hence, because $|B - Y| \leq k - m - 1 = r - 1$, between any two consecutive y_i 's there are at most r - 1 elements from B - Y. Thus, for all *i* we have that for some h, $y_i - y_{i-1} \leq b_{h+r} - b_h$, and therefore

(3.7)
$$y_i - y_{i-1} \le r + \left\lceil \frac{r}{c-1} \right\rceil = \left\lceil \frac{rc}{c-1} \right\rceil$$

Also, we note that $c \geq \frac{m}{2m-k}$ implies that

(3.8) $\left\lceil \frac{rc}{c-1} \right\rceil \le m.$

By (3.7) and (3.8), we see that (3.5) is true.

To establish (3.6), we first observe that $|B| \ge (c-1)(m + \lceil \frac{m}{c} \rceil)$. Since $|B - Y| \le r - 1$, (3.6) will follow if we can prove that

(3.9)
$$(c-2)m + (c-1)\left\lceil \frac{m}{c} \right\rceil + 1 \ge 2r.$$

Since m > r, (3.9) is obviously true for $c \ge 4$. To finish Case 1, we will establish (3.9) for c = 3.

Using (3.8),

$$\begin{array}{rcl} m+2\left\lceil \frac{m}{3}\right\rceil +1 & \geq & \left\lceil \frac{3r}{2}\right\rceil +2\left\lceil \frac{\left\lceil \frac{3r}{2}\right\rceil}{3}\right\rceil +1 \\ & \geq & \frac{3r}{2}+2\left\lceil \frac{r}{2}\right\rceil +1 \\ & \geq & 2r, \end{array}$$

which completes the proof of Case 1.

Case 2. c = 2. We have $x_k - x_1 \le \ell - 1 = 4(k-1)$ and $3m \ge 2k$. Therefore, if $i \ne j$, then

$$(3.10) x_i - x_{i-1} \le m,$$

for otherwise we would have

$$\begin{aligned} x_k - x_1 &= x_j - x_{j-1} + \sum_{i \neq j} (x_i - x_{i-1}) \\ &\geq 2(m+1) + (m+1) + 2(k-3) \\ &= 3m + 2k - 3 \\ &\geq 4k - 3. \end{aligned}$$

Let $B' = \{x_{j-1} + 2i + 1 : 0 \le i \le s-1\}$. Clearly, for every $b \in B'$, there is a $t \in S$ such that $|t-b| \le 1$, so that, as in Case 1 for the set B, we may assume that at most k - m - 1 = r - 1 members of B' have color 0. Let $Y = \{y_1 < y_2 < \cdots < y_u\} = \{b \in B' : \chi(b) = 1\}$. So $u \ge s - (r-1)$. If for some $i \in \{2, 3, \ldots, u\}$ we have $y_i - y_{i-1} > 2r$, then

$$x_j - x_{j-1} > 2r + 2(s - r - 1) + 2 = 2s,$$

a contradiction. Thus, for each $i \in \{2, 3, \dots, u\}$,

 $(3.11) y_i - y_{i-1} \le 2r.$

By the same reasoning, the following hold:

(3.12) $x_j - y_u \le 2r,$ (3.13) $y_1 - x_{j-1} < 2r.$

By (3.10)–(3.13) and the fact that $m \ge 2r$ (since $3m \ge 2k$), $X \cup Y$ is a semi-progression of scope m. Since $X \cup Y$ is monochromatic and has at least k terms, the proof is complete.

It is interesting to note that the proof of Theorem 3.33 actually gives a stronger result. Namely, it shows that for every 2-coloring of [1, 2c(k-1)+1] (where c, k, and m are as in Theorem 3.33), there is a monochromatic sequence $\{x_1, x_2, \ldots, x_k\}$ such that either

$$x_i - x_{i-1} \in \{1, 2, \dots, m\}$$
 for $i = 2, 3, \dots, k;$

or

$$x_i - x_{i-1} \in \{c, 2c, \dots, mc\}$$
 for $i = 2, 3, \dots, k$.

3.4. Semi-progressions

That is, the value of d, from the definition of a semi-progression of scope m, belongs to the set $\{1, c\}$. It would be interesting to try to find similar results where $\{1, c\}$ is replaced by some other set.

Example 3.34. For a given m, the "best" (i.e., the largest) value of k for which Theorem 3.33 provides an upper bound is k = 2m - 1. For this case, the theorem gives

 $SP_m(2m-1) \le 2c(2m-2) + 1 = 4(m^2 - m) + 1.$

Theorem 3.32 gives the somewhat trivial result $SP_m(m) = 2m-1$. Precise formulae for Ramsey-type functions in less trivial cases are desirable, but typically somewhat difficult to come by. In the case of the function $SP_m(k)$, letting k = m + 1, an exact formula is known, which we now present.

Theorem 3.35. Let $m \ge 2$. Then

$$SP_m(m+1) = \begin{cases} 4m+1 & \text{if } m \text{ is even,} \\ 4m-1 & \text{if } m \text{ is odd.} \end{cases}$$

Proof. First note that setting k = m + 1 in Theorem 3.33 yields $SP_m(m+1) \leq 4m+1$. We next establish that $SP_m(m+1) \leq 4m-1$ if m is odd. To do this, let $m \geq 3$ be odd and let $\chi : [1, 4m-1] \rightarrow \{0, 1\}$ be any 2-coloring. From Proposition 2.30, it follows from Theorem 3.32 that in the interval [m+1, 3m-1] there is a monochromatic set $X = \{x_1, x_2, \ldots, x_m\}$ with $x_i - x_{i-1} \in \{1, 2, \ldots, m\}$ for $2 \leq i \leq m$. We may assume that $\chi(X) = 1$. We shall consider two cases.

Case 1. $x_i - x_{i-1} > 1$ for some $i \in \{2, 3, \ldots, m\}$. Let j be the least value of i such that $x_i - x_{i-1} > 1$. If $\chi(x_{j-1} + 1) = 1$, then $X \cup \{x_{j-1}+1\}$ forms an (m+1)-term monochromatic semi-progression of scope m, and we are done. So we shall assume that $\chi(x_{j-1}+1) = 0$. Similarly, we may assume that each of $x_1 - 1, x_1 - 2, \ldots, x_1 - m$ has color 0. Then the set

$$A = [x_1 - m, x_1 - 1] \cup \{x_{j-1} + 1\} \subseteq [1, 4m - 1]$$

is monochromatic. Also, since

$$(x_{j-1}+1) - (x_1-1) = j \le m,$$

A is a semi-progression of scope m.

79

Case 2. $x_i - x_{i-1} = 1$ for all $i \in \{2, 3, ..., m\}$. If any member of $Y = [x_1 - m, x_1 - 1] \cup [x_m + 1, x_m + m]$ has color 1, then clearly we have a monochromatic (m + 1)-term semi-progression of scope m in [1, 4m - 1]. So we may assume that $\chi(Y) = 0$. Now let

$$B = \left\{ x_1 - (2i-1) : 1 \le i \le \frac{m+1}{2} \right\} \bigcup \left\{ x_m + (2i-1) : 1 \le i \le \frac{m+1}{2} \right\}.$$

Then B consists of m + 1 elements, has color 0 (since it is a subset of Y), and each pair of consecutive elements of B has gap 2 or m + 1. Hence, since m + 1 is even and $m + 1 \leq 2m$, B is a monochromatic (m + 1)-term semi-progression of scope m.

We have thus far established that 4m - 1 and 4m + 1 serve as upper bounds for their respective cases. To complete the proof of the theorem, we need to show that they also serve as lower bounds. To do so we shall exhibit, for m odd, a 2-coloring of [1, 4m - 2] that avoids (m + 1)-term monochromatic semi-progressions of scope m; and, for m even, a 2-coloring of [1, 4m] that avoids such progressions.

Consider m odd. Color [1,4m-2] with the coloring α defined by the string

$$\underbrace{11\ldots 1}_{m-1}\underbrace{00\ldots 0}_{m}\underbrace{11\ldots 1}_{m}\underbrace{00\ldots 0}_{m-1}.$$

Let $C = \{x_1 < x_2 < \cdots < x_\ell\}$ be a maximum length monochromatic semi-progression of scope m. By the symmetry of α , we may assume $\alpha(C) = 1$. Let $d = \min\{x_i - x_{i-1} : 2 \le i \le \ell\}$. If d = 1, then $x_i - x_{i-1} \le m$ for each i. For this case, it is evident from the way α is defined that $\ell \le m$. If $d \ge 2$, then each of the two blocks of 1's in the representation of α contains at most $\frac{m}{2}$ members of C, so again $\ell \le m$. Hence, in all cases, α admits no monochromatic (m+1)-term semi-progressions of scope m.

Now assume m is even. Essentially the same reasoning as that used in the case of m odd shows that the coloring of [1, 4m] defined by the string

$$\underbrace{11\ldots 1}_{m}\underbrace{00\ldots 0}_{m}\underbrace{11\ldots 1}_{m}\underbrace{00\ldots 0}_{m}$$

has no monochromatic (m+1)-term semi-progression of scope m. We leave the details to the reader as Exercise 3.12.

 $m \setminus k$

1 2

3

8

We now derive a lower bound for $SP_m(k)$ that holds for all m and k.

Theorem 3.36. Let $k \geq 2$ and $m \geq 1$. Let $\lambda(k,m) = \left\lceil \frac{k-1}{\lceil k/m \rceil} \right\rceil$. Then

$$SP_m(k) \ge 2(k-1)\left(\left\lceil rac{k}{\lambda(k,m)}
ight
ceil - 1
ight) + 1.$$

Before giving the proof of Theorem 3.36, we note that for m fixed, the theorem gives us the asymptotic lower bound

$$SP_m(k) \ge \frac{2}{m}k^2(1+o(1)).$$

Proof of Theorem 3.36. Let k and m be given, let $\lambda = \lambda(k, m)$, and let $n = 2(k-1)(\lfloor \frac{k}{\lambda} \rfloor - 1)$. To prove the theorem we exhibit a specific coloring of [1, n] that avoids monochromatic k-term semiprogressions of scope m. We define this coloring by a string of 1's and 0's as follows. Let A represent a block of k-1 consecutive 1's and B a block of k-1 consecutive 0's. Color [1, n] with $ABAB \dots AB$, where A and B each occur $\begin{bmatrix} k \\ \lambda \end{bmatrix} - 1$ times (so there is a total of $2\left(\begin{bmatrix} k \\ \lambda \end{bmatrix} - 1 \right)$ blocks).

Assume, for a contradiction, that $X = \{x_1, x_2, \dots, x_k\}$ is a monochromatic semi-progression of scope m that is contained in [1, n]. Let $d \in \mathbb{Z}^+$ be such that $x_i - x_{i-1} \in \{d, 2d, ..., md\}$ for i = 2, 3, ..., k. Since each of the blocks A and B has length k-1, and since X is monochromatic, there is some i such that $x_i - x_{i-1} \ge k$. This implies that $d \geq \left\lceil \frac{k}{m} \right\rceil$, and hence each block contains at most $\left\lceil \frac{k-1}{d} \right\rceil \leq \lambda$ members of X. Thus, for each color, there are no more than $\lambda\left(\left\lceil \frac{k}{\lambda}\right\rceil-1\right)$ members of X having that color. Since $\lambda\left(\left\lceil \frac{k}{\lambda}\right\rceil - 1\right) \leq k - 1$ (this is true for any positive integers k, λ , we have a contradiction, and the proof is complete. \square

We conclude this section with a table of values and bounds of $SP_m(k)$ for small m and k, preceded by an exact formula for $SP_m(k)$ under the restriction that $k \leq \frac{3m}{2}$. A proof is outlined in Exercise 3.13.

Theorem 3.37. Let $m + 2 \le k \le \frac{3m}{2}$. Then $SP_m(k) = 4k - 3$.

2	3	4	5	6	7	8	9	10	11
3	9	35	178	?	?	?	?	?	?
3	9	17	33	55	≥ 87	≥ 125	≥ 177	?	?
3	5	11	19	31	71	97	≥ 117	?	?
3	5	7	17	21	35	44	65	≥ 75	≥ 84
3	5	7	9	19	25	33	49	56	≥ 69
3	5	7	9	11	25	29	33	55	61

27

15

33

33

37

37

47

41

13Table 3.3: Values and lower bounds for $SP_m(k)$

13

3.5. Iterated Polynomials

7

7

9

9

11

11

3 | 5 |

3 5

By the n^{th} iteration of the function f(x), we mean the composite function $f(f(\ldots(f(x))\ldots))$, often denoted $f^{(n)}(x)$. One way of thinking

of an arithmetic progression $S = \{a, a + d, a + 2d, \dots, a + (k-1)d\}$ is as a sequence that results from (k-1) iterations of the function f(x) = x + d; that is, $S = \{a, f(a), f^{(2)}(a), \dots, f^{(k-1)}(a)\}$. Looking at arithmetic progressions in this fashion leads us to some natural ways of forming supersets of AP.

We begin with a definition.

Definition 3.38. A p_n -sequence is an increasing sequence of positive integers $\{x_1, x_2, \ldots, x_k\}$ such that there exists a polynomial p(x) of degree n, with integer coefficients, so that $p(x_i) = x_{i+1}$ for $i = 1, 2, \ldots, k - 1.$

We shall call the polynomial p in Definition 3.38 a p_n -function and say that p(x) generates $\{x_1, x_2, \ldots, x_k\}$.

It will be convenient to adopt the following notation.

Notation. Denote by $P_{n,k}$ the family of all p_n -sequences of length k, and by P_n the family of all p_n -sequences (regardless of their length), i.e., $P_n = \bigcup_{k=1}^{\infty} P_{n,k}$.

Thus, for example, the sequence $\{1, 3, 7, 15\}$ is a member of $P_{1,4}$ because it is generated by p(x) = 2x + 1, a polynomial of degree one. Of course, it is also a member of P_1 . Similarly, each arithmetic progression belongs to P_1 , since for some d it is generated by x + d; that is, $AP \subseteq P_1$.

One easy way to produce p_n -sequences is to begin with any positive integer a, and any polynomial p with integer coefficients, and list the consecutive iterates of a by p; as long as this list is increasing, we have found a p_n -sequence. As one example, let $p(x) = x^2 - 2x + 3$ and a = 1. Since p is increasing on the interval $[1, \infty)$, then for each positive integer i, $\{1, 2, 3, 6, 27, \ldots, p^{(i)}(1)\}$ is a p_2 -sequence and is a member of $P_{2,i+1}$.

We shall give an explicit upper bound, in terms of k, for the Ramsey-type function $R(AP \cup P_{k-2}, k; 2)$ (i.e., we will want every 2-coloring to produce a monochromatic k-term sequence that is either a p_{k-2} -sequence or is an arithmetic progression). We will also show (in Theorem 3.51, below) that if this bound could be improved to a certain other (somewhat slower growing) function of k, then w(k) would be bounded above by a similar function.

We will sometimes want to consider the family of polynomials having degree not exceeding a specified number n. For this reason, we introduce the following notation:

$$S_n = \bigcup_{i=1}^n P_i$$
, and $S_{n,k} = \bigcup_{i=1}^n P_{i,k}$

Thus, for example, $S_{3,5}$ consists of all 5-term increasing sequences of positive integers that can be obtained by the iteration of some linear, quadratic, or cubic polynomial having integer coefficients.

Before getting to the main results on p_n -sequences, we need some lemmas. The first is a rather interesting fact in its own right: if Xis an increasing k-term sequence that is generated by a polynomial with integer coefficients, then there exists a polynomial with integer coefficients and degree not exceeding k-2 that also generates X; i.e., every k-term sequence that is a member of P_n for some n is also a member of S_{k-2} .

Lemma 3.39. Let $k \ge 3$. Then $\bigcup_{i=1}^{\infty} P_{i,k} = S_{k-2,k}$.

Proof. By the definition of $S_{k-2,k}$, we need only show that for all $k \geq 3$ and $n \geq 1$, every k-term p_n -sequence is a member of $S_{k-2,k}$.

8.5. Iterated Polynomials

Let $X = \{x_1, x_2, \ldots, x_k\}$ be generated by the polynomial $f(x) = \sum_{i=0}^{n} a_i x^i$, with $a_i \in \mathbb{Z}$ for each i and $a_n \neq 0$. To complete the proof we must show that there exist integers $b_0, b_1, \ldots, b_{k-2} \in \mathbb{Z}$ satisfying the system

3.14)
$$b_0 + b_1 x_1 + \dots + b_{k-2} x_1^{k-2} = x_2$$
$$b_0 + b_1 x_2 + \dots + b_{k-2} x_2^{k-2} = x_3$$
$$\vdots \qquad \vdots$$
$$b_0 + b_1 x_{k-1} + \dots + b_{k-2} x_{k-1}^{k-2} = x_k.$$

We see that (3.14) is a system of k-1 equations in the k-1 variables b_0, \ldots, b_{k-2} . The determinant of the coefficient matrix is the well-known Vandermonde determinant, known to equal

$$\prod_{1 \le \ell < m \le k-1} (x_m - x_\ell).$$

Since the x_i 's are distinct, this determinant is nonzero. By Cramer's rule, system (3.14) has the unique solution $(b'_0, b'_1, \ldots, b'_{k-2})$, where for each $j, 0 \leq j \leq k-2$,

$$b'_j = \frac{D_j}{\prod_{1 \le \ell \le m \le k-1} (x_m - x_\ell)}$$

where D_j is the determinant of the matrix

The proof is completed by showing that each b'_j is an integer. The proof of this fact requires a bit of abstract algebra, and is left to the exercises (see Exercise 3.16).

Remark 3.40. It is worth noting from the above proof that for every sequence of positive integers generated by a polynomial with integer coefficients, there exists a *unique* polynomial of degree at most k - 2 that generates the sequence.

84

3. Supersets of AP

The next lemma provides a useful characterization of the p_1 -sequences.

Lemma 3.41. A sequence is a p_1 -sequence of length $k \ge 3$ if and only if it has the form

$$\left\{ t, t+d, t+d+ad, t+d+ad+a^2d, \dots, t+d\sum_{i=0}^{k-2}a^i \right\}$$

for some positive integers t, d, a.

Proof. Let S be a set of the described form. It is easy to see that the function f(x) = ax + t(1-a)d generates S. Hence S is a p_1 -sequence.

Conversely, assume that $X = \{x_1, x_2, \ldots, x_k\}$ is an increasing sequence of positive integers for which there exist $a, b \in \mathbb{Z}$ such that the function f(x) = ax + b generates X. Let $t = x_1$ and $d = x_2 - x_1$. Then, since $ax_1 + b = x_2$ and $ax_2 + b = x_3$,

(3.15)
$$a = \frac{x_3 - x_2}{x_2 - x_1} = \frac{x_3 - t - d}{d}.$$

Therefore $x_2 = t + d$ and $x_3 = t + d(1 + a)$. Now assume that $n \ge 3$ and that for $j \in \{n - 1, n\}$, $x_j = t + d \sum_{i=0}^{j-2} a^i$. We shall complete the proof via induction on n by showing that $x_{n+1} = t + d \sum_{i=0}^{n-1} a^i$. Note that

$$a = \frac{x_{n+1} - x_n}{x_n - x_{n-1}}$$
$$= \frac{x_{n+1} - \left(t + d\sum_{i=0}^{n-2} a^i\right)}{\left(t + d\sum_{i=0}^{n-2} a^i\right) - \left(t + d\sum_{i=0}^{n-3} a^i\right)}$$

Therefore,

$$x_{n+1} = a \left(d \sum_{i=0}^{n-2} a^i - d \sum_{i=0}^{n-3} a^i \right) + t + d \sum_{i=0}^{n-2} a^i$$
$$= a da^{n-2} + t + d \sum_{i=0}^{n-2} a^i$$
$$= t + d \sum_{i=0}^{n-1} a^i,$$

and the proof is complete.

According to Lemma 3.41, every p_1 -sequence is completely determined by its first three terms. Extending this idea to an arbitrary 3-term sequence, we can pose the following "riddle": if x, y, and z are the first three numbers of a sequence, what is the the fourth number? Mathematically, the idea is very elementary. Using (3.15), letting $a = \frac{z-y}{y-x}$ and b = y - ax, we have a method for getting from one number in the sequence to the next: multiply by a and then add b. For example, given the sequence $-\frac{1}{9}, -\frac{1}{3}, \frac{5}{9}$, then by taking a = -4 and $b = -\frac{7}{9}$, we may answer the riddle by saying that the fourth number is $(-4)\frac{5}{9} - \frac{7}{9} = -3$. This solution to the riddle works for any three numbers x, y, z (even complex numbers) provided $x \neq y$.

The following two technical lemmas concerning the growth of iterated polynomials are stated without proof.

Lemma 3.42. Let $\{x_1 < x_2 < \cdots < x_k\}$ be a p_{k-2} -sequence, generated by the p_{k-2} -function g(x). Then

$$g(x_k) \le x_k + \prod_{i=1}^{k-1} (x_k - x_i).$$

Example 3.43. The sequence $\{1, 2, 3, 6\}$ is generated by the p_2 -function $f(x) = x^2 - 2x + 3$. Hence, $f(6) \le 6 + 3 \cdot 4 \cdot 5 = 126$ (in fact, f(6) = 27).

Lemma 3.44. Let $k \ge 3$ and let $\{x_1, x_2, \ldots, x_k\} \in S_{k-2}$ be generated by the p_n -function g(x) of degree $n \le k-2$. Then for each nonnegative integer j, the sequence

$$\{x_1, x_2, \ldots, x_k, g(x_k) + j \prod_{i=1}^{k-1} (x_k - x_i)\}$$

belongs to S_{k-1} and is generated by the function $h(x) = \sum_{i=1}^{k-1} b_i x^i$, where $b_{k-1} = j$. Furthermore, all members of S_{k-1} may be obtained from members of S_{k-2} in this way.

Remark 3.45. Note that, using the notation of Lemma 3.44, the lemma tells us that for $j \ge 1$, the sequence

$$\{x_1, x_2, \dots, x_k, g(x_k) + j \prod_{i=1}^{k-1} (x_k - x_{i-1})\}$$

belongs to P_{k-1} .

To help clarify what Lemma 3.44 is saying, we consider two examples.

Example 3.46. Using the notation of Lemma 3.44, if j = 0, then $\{x_1, x_2, \ldots, x_k, g(x_k)\}$ is generated by g(x), so it is, of course, a member of S_{k-2} (and hence of S_{k-1} , where we may take h(x) = g(x), with $b_{k-1} = 0$).

Example 3.47. Consider $\{1, 2, 3\}$, a p_1 -sequence generated by the polynomial g(x) = x + 1. Lemma 3.44 tells us that each sequence $\{1, 2, 3, 4 + 2j\}$, where $j \ge 0$, is a member of S_2 , and, for $j \ge 1$, is generated by a p_2 -function whose leading coefficient is j. For instance, according to Example 3.43, $\{1, 2, 3, 6\}$ is generated by the p_2 -function $x^2 - 2x + 3$. In turn, we can build 5-term p_3 -sequences from $\{1, 2, 3, 6\}$: each sequence $\{1, 2, 3, 6, 27 + j(5 \cdot 4 \cdot 3)\}, j \ge 1$, is a member of P_3 , generated by a cubic polynomial with leading coefficient j. Of more significance, perhaps, is what the last sentence of the lemma says: that every p_3 -sequence of length 5 may be found in this way – by starting with the 4-term members of S_2 .

Remark 3.48. Extending the previous example, notice that if $X = \{x_1, x_2, \ldots, x_k\}$ is any member of $S_{k-2} - AP$ whose first k-1 terms form an arithmetic progression $\{x_1 + id : 0 \le i \le k-2\}$, then X is generated by a member of P_{k-2} , i.e., $j \ne 0$. Furthermore, $x_k = x_1 + (k-1)d + j(d)(2d) \cdots ((k-2)d) = x_1 + (k-1)d + jd^{k-2}(k-2)!$.

3.5. Iterated Polynomials

We now are able to give an upper bound for $R(AP \cup P_{k-2})$.

Theorem 3.49. For all $k \ge 5$, $R(AP \cup P_{k-2}, k) \le k^{\frac{(k-2)!}{2}}$.

Proof. The inequality holds for k = 5, since it is known that $R(AP \cup P_3, 5) = 85$ (see Table 3.4 at the end of this section). We proceed by induction on k, letting $k \ge 6$ and assuming the theorem is true for k - 1. Let χ be any 2-coloring of $[1, k^{(k-2)!/2}]$, using the colors red and blue. By the induction hypothesis there exists a monochromatic (k - 1)-term sequence $X \in AP \cup P_{k-3}$ contained in $[1, (k-1)^{(k-3)!/2}]$. Say $X = \{x_1 < x_2 < \cdots < x_{k-1}\}$ has color red. By Remark 3.40, there is a unique polynomial $f \in S_{k-3}$ that generates X. Hence, by Remark 3.45, for each $j = 1, 2, \ldots, k$,

(3.16)
$$\left\{x_1, x_2, \dots, x_{k-1}, f(x_{k-1}) + j \prod_{i=1}^{k-2} (x_{k-1} - x_i)\right\} \in P_{k-2}.$$

For ease of notation, let us denote $\prod_{i=1}^{k-2} (x_{k-1} - x_i)$ by Π . Now let $A = \{f(x_{k-1}) + j\Pi : 1 \le j \le k\} = \{y_1 < y_2 < \cdots < y_k\}$. We wish to show that each member of A belongs to $[1, k^{(k-2)!/2}]$. To show this, note that by Lemma 3.42,

$$y_k \leq x_{k-1} + (k+1)\Pi \leq x_{k-1} + (k+1)x_{k-1}^{k-2}$$

$$(3.17) \leq (k-1)^{(k-3)!/2} + (k+1)(k-1)^{(k-2)!/2}.$$

That the expression (3.17) does not exceed $k^{(k-2)!/2}$ will follow if

$$\frac{(k-1)^{(k-3)!/2}}{(k-1)^{(k-2)!/2}} + (k+1) \le \left(\frac{k}{k-1}\right)^{\frac{(k-2)!}{2}} = \left(1 + \frac{1}{(k-1)}\right)^{\frac{(k-2)!}{2}}$$

This last inequality does indeed hold, since

(3.18)
$$k+2 \le \left(1+\frac{1}{k-1}\right)^{(k-2)!/2}$$

We leave the justification for (3.18) to the reader (hint: consider the binomial expansion of the right-hand side of the inequality). Thus, we have established that $A \subseteq [1, k^{(k-2)!/2}]$.

3. Supersets of *AP*

We consider two possibilities. If every member of A has color blue, then A is a monochromatic arithmetic progression. Otherwise, some member y_m of A has color red, in which case, by (3.16), $X \cup \{y_m\}$ is a red p_{k-2} -sequence. In either case we have a monochromatic k-term member of $AP \cup P_{k-2}$ that lies in $[1, k^{(k-2)!/2}]$. This completes the proof.

The upper bound given by Theorem 3.49, although it may be an interesting result, seems to involve a much larger collection of sequences than does AP; therefore, there is no obvious reason to think that having an upper bound on $AP \cup P_{k-2}$ could help us to find a reasonable upper bound on w(k) = R(AP, k). However, in the next theorem we show that if we were able to establish a certain improved upper bound on $R(AP \cup P_{k-2}, k)$, then we would also have a similar bound for w(k).

We first prove the following lemma.

Lemma 3.50. Let $k \ge 7$. If $\{x_1, x_2, \dots, x_k\} \in S_{k-2} - AP$, then $x_k \ge 2^{k-1}$.

Proof. Let $\{x_1, x_2, \ldots, x_k\} \in S_{k-2} - AP$, and let j be the largest integer such that $\{x_1, x_2, \ldots, x_j\}$ is an arithmetic progression (clearly, $2 \le j \le k-1$).

If j = k - 1, then by Remark 3.48, $x_k \ge k + (k - 2)! \ge 2^{k-1}$. If j < k-1, then by Lemma 3.44, $x_{j+1} \ge x_j + (x_j - x_{j-1}) + \prod_{i=1}^{j-1} (x_j - x_i)$. Now, x_{j-1}, x_j, x_{j+1} is a p_1 -sequence and hence is generated by the polynomial f(x) = ax + b, where

$$a = \frac{x_{j+1} - x_j}{x_j - x_{j-1}}$$

$$\geq \frac{x_j - x_{j-1} + \prod_{i=1}^{j-1} (x_j - x_i)}{x_j - x_{j-1}}$$

$$= 1 + \prod_{i=1}^{j-2} (x_j - x_i)$$

$$\geq 1 + (j-1)!.$$

3.5. Iterated Polynomials

Since $\{x_{j-1}, x_j, x_{j+1}, x_{j+2}\} \in S_2$, by Lemma 3.44 $x_{j+2} \ge f(x_{j+1})$ and hence, by Lemma 3.41,

$$x_{j+2} \ge x_j + 1 + (j-1)! + (1 + (j-1)!)^2.$$

Applying this same argument to each of the triples $\{x_{\ell}, x_{\ell+1}, x_{\ell+2}\}, j-1 \leq \ell \leq k-3$, we obtain

$$x_k \ge j + \sum_{i=1}^{k-j} [1 + (j-1)!]^i.$$

That this last expression is no less than 2^{k-1} is left as Exercise 3.18, which completes the proof.

Theorem 3.51. Let $\tau_k = 2^{k-2} + \prod_{i=1}^{k-3} (2^{k-3} - 2^{i-1})$. If $k \ge 7$ and $R(AP \cup P_{k-2}, k) \le \tau_k - 1$, then $w(k-1) \le \tau_k - 2 = 2^{k^2 - 4k} (1 + o(1))$.

Proof. To prove the result, we will show that, for all $k \geq 7$, if $X = \{x_1 < x_2 < \cdots < x_k\}$ is any k-term p_{k-2} -sequence such that $\{x_1, x_2, \ldots, x_{k-1}\}$ is not an arithmetic progression, then $x_k \geq \tau_k$. From this it follows by Lemma 3.44 that all k-term members of $AP \cup P_{k-2}$ that are contained in $[1, \tau_k - 1]$ are formed by adding $j \prod_{i=1}^{k-2} (x_{k-1} - x_i)$ to the k^{th} term of some arithmetic progression, for some nonnegative integer j. Hence, by the hypothesis, it follows that $w(k-1) \leq \tau_k - 2$, which is the desired conclusion.

Thus, we shall show that whenever X is a k-term p_{k-2} -sequence whose first k-1 terms do not form an arithmetic progression, then $x_k \ge \tau_k$. For the case of k = 7, we ask the reader to prove this by inspection of all 7-term members of $AP \cup P_5$, with the help of Lemma 3.44. Now let $k \ge 8$. If $\{x_1, x_2, \ldots, x_{k-1}\}$ is generated by the function g(x) = 2x, then $x_{k-1} \ge 2^{k-2}$, so that $x_k \ge \tau_k$ by Lemma 3.44.

If $\{x_1, x_2, \ldots, x_{k-1}\}$ is generated by a function other than 2x, then by Lemma 3.50, there is a least integer $j \leq k-1$ such that $x_j > 2^{j-1}$. Also, by Lemmas 3.44 and 3.50, if i < j, then

$$x_{k-1} - x_i \ge 2^{k-2} - 2^{i-1}.$$

If $j \leq i \leq k-2$, then since $x_j - x_{j-1} > 2^{j-1} - 2^{j-2} = 2^{j-2}$, and since $\{x_{i-1}, x_i, x_{i+1}\}$ is a p_1 -sequence, but not an arithmetic progression,

Π

$$x_{k-1} - x_i \ge \sum_{\ell=i-1}^{k-3} 2^{\ell} = 2^{k-2} - 2^{i-1}$$

Now let h be the p_{k-3} -function that generates $\{x_1, x_2, \ldots, x_{k-1}\}$. Then by Lemma 3.50, $h(x_{k-1}) \ge 2^{k-1}$. Since X is a p_{k-2} -sequence (but not a p_{k-3} -sequence), we know by Lemma 3.44 that

$$x_k \ge h(x_{k-1}) + \prod_{i=1}^{k-2} (x_{k-1} - x_i) > \tau_k,$$

completing the proof.

Other results that are similar to Theorem 3.51 have been found, where obtaining a seemingly mild improvement over the known bound on the Ramsey function corresponding to the superset of AP will lead to a like bound on w(k). References to such work are mentioned in Section 3.9.

We may also consider Ramsey functions involving iterated polynomials where the number of colors, r, is greater than two. As one example, the next theorem gives an upper bound for $R(P_1, 3; r)$ (hence an upper bound for $R(S_1, 3; r)$).

Theorem 3.52. $R(P_1, 3; r) \le r! r^2 + r$ for $r \ge 2$.

Proof. Direct calculation gives $R(P_1, 3; 2) = 7$, so that the theorem holds for r = 2. Now assume $r \ge 3$, and let χ be any *r*-coloring of $[1, r!r^2 + r + 1]$. For $i = 0, 1, \ldots, r^2$, define $M_i = [r!i + 1, r!i + r + 1]$. Clearly, each M_i contains a pair $a_i < b_i$, such that $\chi(a_i) = \chi(b_i)$. For each $i, 0 \le i \le r^2$, let h(i) be the ordered pair $(a_i - r!i, \chi(i))$. Note that there are a total of only r^2 possible ordered pairs (the set of possible ordered pairs is the same for all i). Thus, by the pigeonhole principle, there exist i < j such that h(i) = h(j). Since $a_j = a_i + r!(j - i)$ and $b_i - a_i \le r$,

 $a_j = a_i + \lambda (b_i - a_i),$

where $\lambda \geq 2$. Thus, by Lemma 3.41, $\{a_i, b_i, a_j\}$ is a p_1 -sequence. Since $\{a_i, b_i, a_j\}$ is monochromatic, and since $a_j < b_j \leq r!j + r + 1$, the theorem is proved. We end this section with a table. Note that by Lemma 3.39, there is no need to include values of $R(S_n, k; r)$ for which $n \ge k - 1$.

n	k	r	$R(S_n,k;r)$	$R(AP \cup P_n, k; r)$
1	3	2	7	7
1	4	2	23	?
2	4	2	20	21
1	5	2	76	76-177
2	5	2	72	?
3	5	2	67	85
4	6	2	$68 - 6^{12}$	$192 - 6^{12}$
1	3	3	14	14
1	4	3	≥ 71	?
2	4	3	≥ 71	≥ 71
1	3	4	24	24
1	3	5	38	38

Table 3.4: Values and bounds of $R(S_n, k; r)$ and $R(AP \cup P_n, k; r)$

3.6. Arithmetic Progressions as Recurrence Solutions

By a linear recurrence relation of order n, we mean an equation of the form

$$x_k = c_1 x_{k-1} + c_2 x_{k-2} + \dots + c_n x_{k-n},$$

where the c_i 's are given constants. Solving such a system means finding a closed formula for x_k that holds for all k. As a simple example, consider the following recurrence of order one: $x_k = 3x_{k-1}$, $k \ge 1$. A "general" solution is $x_k = 3^k x_0$. To get a "particular" solution, it suffices to know the initial value x_0 .

An arithmetic progression may be thought of as the solution to the recurrence

$$(3.19) x_k = 2x_{k-1} - x_{k-2}.$$

To get a particular solution, it is sufficient to be given two initial values. For example, if $x_1 = 1$ and $x_2 = 3$, then $x_3 = 2x_2 - x_1 = 5$,

 $x_4 = 7$, etc. This is simply another way of looking at the fact that arithmetic progressions $\{x_1, x_2, \ldots\}$ are completely determined by knowing x_1 and x_2 , since $x_k - x_{k-1} = x_{k-1} - x_{k-2}$ for all k.

Thus we see that the family AP may be considered a special subfamily of the family of those sets that occur as solutions to some linear recurrence. There are many ways that we can generalize (3.19). One way is the following. For $k \geq 3$, consider the family of those sets $\{x_1, x_2, \ldots, x_k\}$ with $x_1 < x_2$, having the property that there exists a set $\{a_i \geq 2 : 3 \leq i \leq k\}$ such that for $i = 3, 4, \ldots, k$, $x_i = a_i x_{i-1} + (1 - a_i) x_{i-2}$. Let us call this collection of sets R_1 . We see that every arithmetic progression belongs to R_1 (let $a_i = 2$ for every i). As another example, let us consider the 6-term sequence obtained by letting $a_3 = 2$, $a_4 = 3$, $a_5 = 2$, $a_6 = 7$, $x_1 = 1$, and $x_2 = 3$. Then we have $x_3 = 2x_2 - x_1$, $x_4 = 3x_3 - 2x_2$, $x_5 = 2x_4 - x_3$, and $x_6 = 7x_5 - 6x_4$, yielding the sequence $\{1, 3, 5, 9, 13, 37\}$, a 6-term member of R_1 .

It is not hard to obtain an upper bound on the Ramsey-type function $R(R_1, k)$. We do so in the following theorem.

Theorem 3.53. Let $k \ge 3$. Then $R(R_1, k) \le \frac{7}{24}(k+1)!$.

Proof. We may calculate directly that $R(R_1,3) = 7$, so that the result is true for k = 3. Proceeding by induction, assume that $k \ge 4$, and that $R(R_1, k - 1) \le \frac{7}{24}k!$. Let $m_k = \frac{7}{24}(k + 1)!$ and consider an arbitrary 2-coloring $\chi : [1, m_k] \to \{0, 1\}$. By the induction hypothesis, in $[1, m_{k-1}]$ there is a monochromatic (k-1)-term member of R_1 . Say $X = \{x_1, x_2, \ldots, x_{k-1}\}$ is such a sequence and that $\chi(X) = 0$. For $1 \le i \le k$, let $y_i = x_{k-1} + i(x_{k-1} - x_{k-2})$. Notice that, for each *i*, the set $\{x_1, x_2, \ldots, x_{k-1}, y_i\}$ is a *k*-term member of R_1 (where, according to the notation used in the definition of R_1 , $a_i = i + 1$). Also notice that every y_i is no greater than m_k , since

 $m_{k-1} + k(m_{k-1} - x_{k-2}) \le (k+1)m_{k-1} = m_k.$

To complete the proof, we consider two possibilities. If some $y_j, j \in \{1, 2, ..., k\}$, has color 0, then $X \cup \{y_j\}$ is a monochromatic member of R_1 that is contained in $[1, m_k]$. Otherwise $\{y_1, y_2, ..., y_k\}$

3.7. Exercises

is a monochromatic arithmetic progression (hence a member of R_1) contained in $[1, m_k]$, and the proof is complete.

There are many other generalizations of (3.19) that may be considered. For example, instead of having $b_i = 1 - a_i$, we can simply require that $b_i = -1$ for all *i*. Some examples are given in the exercises. We encourage the reader to experiment with other generalizations – it seems likely that there are some interesting ones that have yet to be explored.

3.7. Exercises

3.1 a) Find all 3-term quasi-progressions with diameter 1 that are contained in [1,6]. How many are contained in [1,10]? How many are contained in [1,m]?

b) How many 3-term quasi-progressions with diameter k are contained in [1, m]?

- **3.2** Prove Corollary 3.11(i).
- 3.3 Complete the details of Case 3 in the proof of Theorem 3.12.
- **3.4** Prove Corollary 3.15.
- **3.5** Show that $\frac{k(k+1)}{2}$ is an upper bound for $GQ_{x-1}(k)$ (see Example 3.18).
- **3.6** Show that the last sentence in the statement of Theorem 3.19 follows from the previous sentence.
- **3.7** Calculate DW(3) and DW(4).
- **3.8** How many descending waves of length three are contained in [1, 10]? in [1, m]?
- **3.9** Finish the proof of Theorem 3.21
- **3.10** Finish the proof of Theorem 3.23.
- **3.11** Prove Theorem 3.32.
- **3.12** Complete the proof of Theorem 3.35 by showing that for m even, the string $\underbrace{11\ldots 1}_{m} \underbrace{00\ldots 0}_{m} \underbrace{11\ldots 1}_{m} \underbrace{00\ldots 0}_{m}$ avoids mono-

chromatic (m + 1)-term semi-progressions of scope m.

3.13 This exercise provides the outline for a proof of Theorem 3.37.

94

a) Assume $m + 2 \le k \le \frac{3m}{2}$. Show that if k is even, then the 2-coloring $11 \underbrace{00 \dots 0}_{k-2} \underbrace{11 \dots 1}_{k-2} \underbrace{00 \dots 0}_{k-2} \underbrace{11 \dots 1}_{k-2}$ avoids mono-

chromatic k-term semi-progressions of scope m. Hence, for k even, $SP_m(k) \ge 4k-3$.

b) Use Theorem 3.36 to show that the lower bound of (a) also holds for k odd.

c) Use the results of (a) and (b), along with Theorem 3.33, to prove Theorem 3.37, i.e., that $SP_m(k) = 4k - 3$.

3.14 For *i* an integer, $0 \le i \le k-2$, let $\lambda(k, m, i) = \left\lceil \frac{k-i-1}{\lceil (k-i)/m \rceil} \right\rceil$. Prove that

$$SP_m(k) \ge 2(k-i-1)\left(\left\lceil \frac{k}{\lambda(k,m,i)} \right\rceil - 1\right) + 1.$$

3.15 a) Find specific values of m, k, and i for which the lower bound on $SP_m(k)$ given by Exercise 3.14 is better (greater) than that given by Theorem 3.36.

b) Find specific values of m and k for which the lower bound provided by Theorem 3.37 is better than any obtainable by Exercise 3.14.

- **3.16** Complete the proof of Lemma 3.39 by showing that each b'_j is an integer. (May require abstract algebra.)
- **3.17** Let $d \ge 1$. Use Lemma 3.41 to show that $X = \{x_1, x_2, \ldots, x_k\}$ is a member of P_1 if and only if $dX = \{dx_1, dx_2, \ldots, dx_k\}$ is a member of P_1 .
- **3.18** Complete the proof of Lemma 3.50 by proving that, for $2 \le j \le k-2$ and $k \ge 7$,

$$j + \sum_{i=1}^{k-j} [1 + (j-1)!]^i \ge 2^{k-1}.$$

(Hint: verify that the inequality holds whenever j = 2 and use $n! > (\frac{n}{e})^n$ to show that it is true for $j \ge 7$; then, show that it holds for each pair (j, k) in the set $\{(3, 7), (4, 7), (5, 7), (6, 8)\}$, and then use induction on k (where j is fixed).)

3.19 Define a C-sequence to be a sequence of distinct (but not necessarily increasing) positive integers $\{x_1, x_2, \ldots, x_k\}$ such

that there exists a function f(x) = ax+b, with $a \in \mathbb{Z}^+$, $b \in \mathbb{Z}$, where for each $i = 2, 3, \ldots, k$, $f(x_{i-1}) = x_i$. Obviously, every p_1 -sequence is a *C*-sequence. Prove that a sequence *S* is a *C*-sequence if and only if

$$S = \left\{ t + d, t + d + ad, t + d + ad + a^{2}d, \dots, t + d\sum_{i=0}^{k-2} a^{i} \right\}$$

for some $t, a, d \in \mathbb{Z}$, where $t, a \ge 1$ and $d \ne 0$.

- **3.20** Define an *E*-sequence of length *k* to be an increasing sequence of positive integers generated by a function p(x) = ax + b, where $a \ge 1$ and $b \ge 0$ are integers. It is clear that every arithmetic progression is an *E*-sequence, and that every *E*-sequence is a p_1 -sequence. Prove that (x_1, x_2, x_3) is an *E*-sequence if and only if it has the form $\{x, x + d, x + d + ad\}$, where $a \le \frac{d}{x} + 1$.
- **3.21** Denote by N(AP, k, m) the number of different k-term arithmetic progressions contained in [1, m]. Define $N(P_1, k, m)$ and N(C, k, m) analogously, replacing the family of arithmetic progressions with p_1 -sequences and C-sequences, respectively (see Exercise 3.19). Prove that

$$N(C,k,m) = 2N(P_1,k,m) - N(AP,k,m).$$

3.22 Define a *D*-sequence to be any sequence of distinct positive integers of the form $\{t, f(t), f^{(2)}(t), \ldots, f^{(k-1)}(t)\}$, where $f(x) = ax + b, a, b \in \mathbb{Z}$. It is clear that every *C*-sequence (see Exercise 3.19) is a *D*-sequence.

a) Prove that every D-sequence of length three is also a C-sequence.

b) Prove that a set of size $k \ge 4$ is a *D*-sequence but not a *C*-sequence if and only if it has the form

$$\left\{t, t+d, t+d-ad, t+d-ad+a^2d, \dots, t+d\sum_{i=0}^{k-2}(-1)^i a^i\right\},\$$

where $a, d \in \mathbb{Z}$, $a \ge 2$, and $d \ne 0$.

c) Using Exercise 3.21's notation, prove that for all $k \ge 4$,

$$N(D,k,m) = N(C,k,m) + 2\sum_{a=2}^{5}\sum_{d=1}^{5}(m-a^{k-2}d),$$

where j is the greatest integer such that $a^{k-2} \leq m-1$ and $s = \lfloor \frac{m-1}{a^{k-2}} \rfloor$.

- **3.23** Define R_2 to be the same as R_1 (as in Theorem 3.53), except instead of using the recurrence $x_i = a_i x_{i-1} b_i x_{i-2}$, where $b_i = 1 a_i$, we now use the recurrence $x_i = a_i x_{i-1} x_{i-2}$ for all *i*. Notice that R_2 is a superset of *AP*. Prove that $R(R_2, k) \leq \frac{1}{3}(k+1)!$ for $k \geq 3$.
- **3.24** As in Exercise 3.23, define R_3 to differ from R_1 only in that we require that $a_i = 2$ and $b_i \leq -1$ for all *i*. Find an upper bound on $R(R_3, k)$.

3.8. Research Problems

- **3.1** Let $B_2(k)$ be the least positive integer with the following property: for every 2-coloring of $[1, B_2(k)]$, there is a monochromatic sequence $\{x_1, x_2, \ldots, x_k\}$ such that $x_i - x_{i-1} \in \{d_1, d_2\}, 2 \leq i \leq k$, for some $d_1, d_2 \in \mathbb{Z}^+$. Write a computer program to calculate $B_2(k)$. Try to get bounds for $B_2(k)$. References: [58], [163], [167]
- **3.2** Repeat Research Problem 3.1 above, except consider $B_m(k)$, the Ramsey-type function for those sequences such that for some *m*-element set $D = \{d_1, d_2, \ldots, d_m\}$, the gaps must belong to *D*. Try to determine relationships that exist between $B_m(k)$ and the Ramsey-type functions discussed in this chapter. Write a computer program to calculate specific values of $B_m(k)$. Try to obtain bounds for $B_m(k)$. References: [58], [163], [167]
- **3.3** It has been conjectured that $Q_{k-i}(k) = 2ik 4i + 3$ if $k \not\equiv 1 \pmod{i}$ and $k \geq 2i \geq 2$. Prove or disprove this conjecture. As a special case, determine if $Q_{k/2}(k) = k^2 2k + 3$, for k even. Table 3.1 provides some known values of this function. One possible approach to solving this problem may come from the following conjecture (based on some known

computer output): if $k \equiv 0 \pmod{i}$ and k > i, then the only valid coloring of maximal length (i.e., of length 2ik-4k+2) is the coloring $A_1 \underbrace{B \dots B}_{i-1} A_0$, where A_1 is a string of 1's having length k - i, A_0 is a string of 0's having length k - i, and $B = \underbrace{00 \dots 011 \dots 1}_{i}$.

$$\overbrace{k=1}^{k-1} \overbrace{k=1}^{k-1}$$
References: [58], [167]

- **3.4** It has been conjectured that $Q_{k-i}(k) = 2ik 2i + 1$ if $k \equiv 1 \pmod{i}$ and $k \geq 2i \geq 1$. Prove or disprove this conjecture. Some specific values are given in Table 3.1. As in Research Problem 3.3, computer output suggests a rather regular pattern in the valid colorings of maximal length. References: [58], [167]
- **3.5** Denote by r^j a string of length rj consisting of j monochromatic blocks, each of length r, where the colors of the blocks alternate. For example, 3^4 represents the string 000111000111 or the string 111000111000, and 2^5 represents a string such as 1100110011. Prove or disprove the following conjecture: Whenever $k \geq 2i$ and $k \equiv 2 \pmod{i}$, the coloring represented by $(k-2)^{i-1}(k-1)^2(k-2)^{i-1}$ is a maximal valid coloring. A proof of this would support the conjecture of Research Problem 3.3.

References: [58], [167]

- * **3.6** Determine a relationship between w(k) and $Q_1(k)$ asymptotically (i.e., what happens to the ratio as $k \to \infty$?). References: [58], [167]
 - **3.7** It was mentioned after the proof of Theorem 3.33 that its proof establishes an upper bound for the Ramsey function associated with semi-progressions of scope m, where the value of d (as in the definition of semi-progression) is restricted to the set $D = \{1, c\}$. Try replacing the set D with various other sets, and attempt to find upper bounds. Perhaps a more general result can be obtained, where the case of $D = \{1, c\}$ would be one special case. Reference: [163]

- **3.8** Obtain an upper bound on $GQ_{x-3}(k)$. In particular, can we bound it above by a polynomial of degree four? Attempt to generalize this to $GQ_{x-t}(k)$. References: [82], [170]
- **3.9** Denote by $SP_m(k;r)$ the least positive integer M such that for every r-coloring of [1, M] there will be a monochromatic k-term semi-progression of scope m. As in the case of two colors, it is clear that $w(k;r) \leq SP_m(\Gamma_m(k);r)$. Study the function $SP_m(k;r)$ for values of r greater than two. Reference: **[163]**
- **3.10** In Table 3.3, we observe that for $m \in \{3, 4, 5, 6\}$,

$$SP_m(2m) = 6(2m-1) + \epsilon,$$

where $\epsilon = 1$ or 2. Determine if $SP_m(2m) = 12m(1 + o(1))$. Reference: [163]

- **3.11** We see from Table 3.3 that, for the pairs (m, k) = (2, 5) and (m, k) = (4, 9), $SP_m(k) = 8(k 1) + 1$. Determine if this is true for the pair (m, k) = (6, 13) or, more generally, for (m, k) = (2j, 4j + 1). Reference: [163]
- **3.12** Let us say a set of positive integers S has property QP if, for some fixed n, S contains a k-term quasi-progression of diameter n for every $k \ge 1$ (that, is S contains arbitrarily long quasi-progressions of diameter n).

a) Determine if the set of squares $\{i^2 : i = 1, 2, ...\}$ has property QP.

b) It is known that there are infinitely many 4-term quasiprogressions of diameter 1 among the set of squares. Determine if there exist any such progressions of length five. Is so, how many?

References: [58], [61], [167]

- **3.13** Improve on the lower bound given by Theorem 3.9. Reference: [163]
- **3.14** As in Theorem 3.12, obtain an exact formula for $Q_{k-3}(k)$. Reference: [163]

- **3.15** Improve on Theorem 3.14 by obtaining an upper bound on $Q_n(n+t)$ when t is a number that is greater than $\frac{n}{2}$. Reference: [163]
- * 3.16 Attempt to find the precise rate of growth of DW(k); i.e., find a such that $DW(k) = ak^3(1 + o(1))$. References: [15], [58], [179]
- **3.17** Find bounds on DW(k; r) for $r \ge 3$. References: [15], [58], [179]

3.8. Research Problems

- **3.18** Improve upon Table 3.4 (either by adding new entries, or by improving any of the bounds).References: [131], [171], [172]
- **3.19** Find an upper bound for R(C, k; r) (see Exercise 3.19). In particular, find an upper bound for R(C, 3; r) that is less than that of Theorem 3.52. Also, the following are known: R(C, 3; 2) = 5, R(C, 4; 2) = 20, R(C, 3; 3) = 13, R(C, 3; 4) = 17, $R(C, 4; 3) \ge 28$ and $R(C, 5; 2) \ge 53$. Find more of these values; in particular, find R(C, 5; 2). References: [131], [161], [162], [169], [171], [172], [177]
- **3.20** Referring to Exercise 3.20, find upper and/or lower bounds for the Ramsey-type function R(E,k;2) associated with kterm E-sequences. Find a lower bound for R(E,3;r) (it is obvious that $w(k;r) \ge R(E,k;r) \ge R(P_1,k;r)$). It is known that R(E,3;2) = 9, R(E,4;2) = 35, R(E,3;3) = 26, and $62 \le R(E,5;2) \le 177$. Find more exact values; in particular, find R(E,5;2). References: [131], [161], [162], [169], [171], [172], [177]
- **3.21** Repeat Research Problem 3.19, using *D* in place of *C* (see Exercise 3.22). Also, the following values and bounds are known: R(D,3;2) = 5, R(D,4;2) = 19, R(D,3;3) = 13, R(D,3;4) = 17, $R(D,4;3) \ge 28$, and $52 \le R(D,5;2) \le 177$. Find more exact values; in particular, find R(D,5;2). References: [131], [161], [162], [169], [171], [172], [177]
- **3.22** Look for other ways to define supersets of AP, and attempt to obtain non-trivial upper and/or lower bounds on the corresponding Ramsey-type function.

3.9. References

- 3. Supersets of AP
- **3.23** Calculate values and attempt to get bounds on any function analogous to the Erdős and Turán function, but where the family AP is replaced by one of the families discussed in this chapter. If any bound is obtained, use it to try to get a bound on the corresponding Ramsey-type function.

3.9. References

§3.1. The bounds on quasi-progressions are from [167], which also includes a proof of Lemma 3.13, as well as a table of computer-generated data. Brown, Erdős, and Freedman [58] study the question of which sets contain arbitrarily long quasi-progressions, and the relationship between this property and several other properties, including that of containing arbitrarily long descending waves. They also prove the equivalence of two properties, one a conjecture of Erdős' (a Ramsey-type statement involving infinite reciprocal sums; see Research Problem 2.3) and the other involving quasi-progressions. Other results on infinite reciprocal sums can be found in [59] and [73]. Brown, Freedman, and Shiue [61] consider the problem of finding quasi-progressions that are contained in the set of squares.

§3.2. The notion of generalized quasi-progressions is introduced in [170], which contains a proof of Theorem 3.19, some related results, and Table 3.2. Functions similar to $GQ_{\delta}(k)$ are considered in [82].

§3.3. Proofs of Theorems 3.21 and 3.24 may be found in [15]. A proof of Theorem 3.22 can be found in [58]. Some results concerning the notion of an ascending wave (the gaps increase rather than decrease) are also discussed in [15] and [58]. We can generalize descending waves to $DW(k, \ell)$, the least positive integer d such that [1, d] contains either an ascending wave of length k or a descending wave of length ℓ . See work by Erdős and Szekeres [101] and Lefmann [179] for more information. The fact that any infinite set of positive integers that contains arbitrary long quasi-progressions must also contain arbitrarily long descending waves is given in [58], where it is also shown that the reverse implication is false.

§3.4. The proof of Proposition 3.30 is from [195], which also has other related results. Rabung provides another proof of the existence

of $\Gamma_m(k)$ [209]. Further results on $\Gamma_m(k)$ are given in [63]. Theorems 3.32, 3.33, and 3.35–3.37 are from [163].

§3.5. Lemmas 3.42 and 3.44 are from [162]. A slightly weaker result than that of Theorem 3.49 appears in [131], which also contains proofs of Lemma 3.50 and Theorem 3.51. In [168], results like Theorem 3.51 are given for certain families of sequences that are properly contained in $AP \cup P_{k-2}$. Theorem 3.52 is from [161]. Other results involving Ramsey functions based on polynomial iteration appear in [172]. The values and bounds in Table 3.4 are from [131], [171], and [172]. Some properties on the growth of iterated polynomials, irrespective of Ramsey theory, are explored in [177]. Work on p_n -sequences involving more than two colors is considered in [169] and [171].

§3.6. Theorem 3.53 and related work may be found in [166].

§3.7. Exercise 3.5 is from [170]. Exercise 3.14 is proven in [163]. Exercises 3.19–3.22 are taken from [162]. Exercises 3.23 and 3.24 are from [166].

Chapter 4

Subsets of AP

In Chapter 3 we considered functions analogous to w(k;r) by replacing AP, the collection of arithmetic progressions, with certain collections \mathcal{F} such that $AP \subseteq \mathcal{F}$. The existence of the corresponding van der Waerden-type functions $R(\mathcal{F}, k; r)$ was guaranteed by van der Waerden's theorem, since each arithmetic progression is a member of \mathcal{F} . The purpose of this chapter is to consider the reverse situation, $\mathcal{F} \subseteq AP$. Thus, we wish to restrict, in some way, the allowable arithmetic progressions.

We note that if \mathcal{F} is a proper subset of AP, then van der Waerden's theorem does not guarantee the existence of $R(\mathcal{F}, k; r)$ for all kand r. In fact, any positive result we can uncover about such a function can be considered a strengthening of van der Waerden's theorem. For example, if $R(\mathcal{F}, k; r) < m$ for some $\mathcal{F} \subseteq AP$, then w(k; r) < m.

As a start, we show that for certain choices of \mathcal{F} , k, and r it is relatively easy to conclude that $R(\mathcal{F}, k; r)$ does not exist. We begin with some examples.

Example 4.1. Let \mathcal{F} be the collection of all arithmetic progressions $\{a, a + d, a + 2d, \ldots\}$ such that $a \geq 1$ and $d \in \{1, 2\}$. To show that $R(\mathcal{F}, k)$ does not exist, it is enough for us to find a 2-coloring of the positive integers that does not yield any k-term monochromatic members of \mathcal{F} . Consider the coloring $\chi : \mathbb{Z}^+ \to \{0, 1\}$ represented by the sequence 11001100.... It is easy to see that there do not exist

4.1. Finite Gap Sets

more than two consecutive numbers with the same color, nor do there exist x, x + 2 of the same color. Hence, $R(\mathcal{F}, 3)$ does not exist and therefore $R(\mathcal{F}, k)$ does not exist if $k \geq 3$.

Example 4.2. Let \mathcal{F} be the family of all arithmetic progressions $\{a, a + d, a + 2d, ...\}$ such that $a = 2^i$ for some $i \ge 0$. Consider the 2-coloring of \mathbb{Z}^+ defined by coloring all the powers of 2 red, and all other positive integers blue. Then obviously there is no blue member of \mathcal{F} . Also, if $X = \{x, y, z\}$ is red, where x < y < z, then X cannot be an arithmetic progression. This is due to the fact that for all $k \ge 0$, $\sum_{i=0}^{k} 2^i = 2^{k+1} - 1$, so that z - y > y - x. Hence, $R(\mathcal{F}, k)$ does not exist for $k \ge 3$.

Examples 4.1 and 4.2 may not be very surprising, since the collection \mathcal{F} consists of a rather small part of AP. The next example uses a much larger collection \mathcal{F} .

Example 4.3. Let \mathcal{F} be the collection of arithmetic progressions whose gaps are odd. Then the coloring of the positive integers defined by coloring the even numbers red and the odd numbers blue yields no monochromatic 2-term member of \mathcal{F} . Thus, $R(\mathcal{F}, 2)$ does not exist.

There are many ways that one may choose a subcollection of AP. Examples 4.1 and 4.3 illustrate one natural way, which is to require that the gap d of the arithmetic progressions belong to some prescribed set of positive integers. This type of subcollection of AP has been the subject of several recent research articles, and many interesting problems remain unanswered. For this reason, we devote most of this chapter to the idea of placing certain restrictions on the gaps that the desired arithmetic progressions may have.

Before proceeding, we introduce the following notation and language.

Notation. For D a set of positive integers, denote by A_D the collection of all arithmetic progressions whose gaps belong to D. We will refer to D as the gap set. We refer to an element of A_D as a D-a.p.. Also, if d is a positive integer, by a d-a.p. we shall mean an arithmetic progression whose gap is d.

We begin the study of the Ramsey properties of sets of type A_D by considering the case in which D is a finite set.

4.1. Finite Gap Sets

Example 4.1 hints that, for D finite, A_D may not be regular (see Definition 1.28). Indeed, the following theorem shows that this is true, even when only two colors are used.

Theorem 4.4. For D a finite set of positive integers, there is a k large enough so that $R(A_D, k; 2)$ does not exist.

Proof. Let $n = \max\{d : d \in D\}$. Define the 2-coloring of \mathbb{Z}^+ by the string

 $\underbrace{11\ldots 1}_n \underbrace{00\ldots 0}_n \underbrace{11\ldots 1}_n \underbrace{00\ldots 0}_n \cdots$

To prove the theorem, we will show that under this coloring there is no monochromatic (n + 1)-term arithmetic progression whose gap belongs to D. Obviously, any arithmetic progression with gap n alternates color, so there cannot exist even a 2-term monochromatic arithmetic progression with gap n.

Now assume X is an arithmetic progression with gap less than n. Then X cannot have more than n consecutive elements of the same color. Hence $R(A_D, n+1; 2)$ does not exist.

The proof of Theorem 4.4 shows that $R(A_D, k; 2)$ does not exist if $n = \max\{d : d \in D\}$. However, it does not say that n + 1 is the minimum value of k such that $R(A_D, k; 2)$ does not exist. So what is the minimum value? The answer seems to depend not only on the size of D, but also on the specific elements of D.

For example, the third van der Waerden number, w(3), is known to equal 9. Obviously, any 3-term arithmetic progression that is contained in [1,9] will have as its gap an element of $D = \{1,2,3,4\}$. Hence, for this choice of D, $R(A_D,3) = w(3) = 9$. On the other hand, if $E = \{1,3,5,7\}$, then (as discussed in Example 4.3) the coloring 101010... avoids 2-term arithmetic progressions with gaps in E; hence $R(A_E, 2) = \infty$.

4.1. Finite Gap Sets

4. Subsets of AP

For D such that $|D| \leq 3$, much is known about $R(A_D, k; 2)$. In particular, a complete answer is known if |D| = 1 or 2. We summarize these results in the next three theorems.

Theorem 4.5. If |D| = 1, then $R(A_D, 2; 2) = \infty$.

Proof. Let $D = \{d\}$. Define a 2-coloring χ of \mathbb{Z}^+ as follows. First, color [1, d] in any way. Then, for all x > d, define $\chi(x)$ such that $\chi(x) \neq \chi(x - d)$. Then χ is a 2-coloring of \mathbb{Z}^+ that does not have any 2-term monochromatic arithmetic progressions with gap d. This proves the theorem.

The next theorem makes use of Lemma 4.6, below, for which we remind the reader of the following notation.

Notation. For D a set of positive integers and t a real number, we write tD to denote $\{td : d \in D\}$.

Lemma 4.6. Let D be a set of positive integers, and let $k, t \ge 1$. Then $R(A_{tD}, k; r) = t[R(A_D, k; r) - 1] + 1$ (if $R(A_D, k; r) = \infty$, then $R(A_{tD}, k; r) = \infty$).

Proof. Let $m = R(A_D, k; r)$, and assume $m < \infty$. To show that $R(A_{tD}, k; r) \leq t(m-1)+1$, let χ be any *r*-coloring of [1, t(m-1)+1]. Define χ' on [1, m] as follows:

$$\chi'(x) = \chi(t(x-1) + 1).$$

By the definition of m, for some $d \in D$, within [1,m] there is a d-a.p., $\{x_i : 1 \leq i \leq k\}$, that is monochromatic under χ' . Then $\{t(x_i-1)+1: 1\leq i\leq k\}$ is monochromatic under χ and is a k-term td-a.p. Thus, $R(A_{tD}, k; r) \leq t(m-1)+1$.

To obtain the reverse inequality, note that by the definition of m, there exists an *r*-coloring ϕ of [1, m - 1] that avoids monochromatic *k*-term arithmetic progressions whose gaps belong to D. Define ϕ' on [1, t(m - 1)] as follows:

 $\phi'[t(j-1)+1, tj] = \phi(j)$ for each $j = 1, 2, \dots, m-1$.

Then ϕ' avoids monochromatic k-term arithmetic progressions whose gaps belong to tD. Hence, $R(A_{tD}, k; r) \ge t(m-1) + 1$.

It is easy to check that this same line of reasoning takes care of the cases when $R(A_D, k; r) = \infty$.

We now consider 2-element gap sets.

Theorem 4.7. Let $D = \{a, b\}$, let g = gcd(a, b), and let $k \ge 2$. If $\frac{a}{g}$ and $\frac{b}{g}$ are not both odd and if k = 2, then $R(A_D, k) = a + b - g + 1$. Otherwise, $R(A_D, k) = \infty$.

Proof. We first observe that it is sufficient to prove the theorem for the situation in which g = 1. To see this, assume that the statement of the theorem is true for all 2-element sets whose greatest common divisor equals one. Now let $D = \{a, b\}$ and let $a' = \frac{a}{g}$ and $b' = \frac{b}{g}$ not both be odd. Then gcd(a', b') = 1 and therefore by Lemma 4.6 we have

$$\begin{split} R(A_{\{a,b\}},2) &= g[R(A_{\{a',b'\}},2)-1] + 1 = g(a'+b'-1) + 1 = a+b-g+1.\\ \text{If both }a' \text{ and }b' \text{ are odd, then by our assumption, } R(A_{\{a',b'\}},2) = \infty,\\ \text{so that by Lemma 4.6, } R(A_{\{a,b\}},2) = \infty. \end{split}$$

We begin with the case in which $R(A_D, k)$ is finite. We prove $R(A_D, 2) \leq a + b$ by contradiction. Thus, we assume that a and b are not both odd, that k = 2, and that there exists a 2-coloring χ of [1, a+b] that yields no monochromatic 2-term arithmetic progression whose gap belongs to $\{a, b\}$.

Let \oplus represent addition modulo a + b. Then for all $i \in [1, a + b]$ we have $|i \oplus a - i| \in \{a, b\}$. Hence, by our assumption,

(4.1) $\chi(i) \neq \chi(i \oplus a) \text{ for all } i \in [1, a+b].$

Now,

$$1 \oplus (a+b-1)a \equiv (1-a) \pmod{(a+b)}$$

so that

(4.2) $1 \oplus (a+b-1)a \equiv (b+1) \pmod{(a+b)}$.

Also, since a + b - 1 is even, by repeated use of (4.1) we have

(4.3) $\chi(1 \oplus (a+b-1)a) = \chi(1).$

Combining (4.2) and (4.3), we have $\chi(b+1) = \chi(1)$, contradicting our assumption about χ .

To prove $R(A_D, 2) \ge a + b$, we give a 2-coloring of [1, a + b - 1]that yields no 2-term monochromatic *D*-a.p. Since gcd(a, a + b) = 1, for each $i \in [1, a + b - 1]$ there exists a unique $j \in [1, a + b - 1]$ such that $i \equiv ja \pmod{(a+b)}$. Hence the coloring χ of [1, a + b - 1] defined by

 $\chi(i) = \begin{cases} 1 & \text{if } i \equiv ea \pmod{(a+b)} \text{ with } e \text{ even,} \\ 0 & \text{if } i \equiv ua \pmod{(a+b)} \text{ with } u \text{ odd} \end{cases}$

is a well-defined 2-coloring of [1, a + b - 1].

Now, assume that y, z are both in [1, a + b - 1] with z = y + a. If $y \equiv ja \pmod{(a+b)}$, then $z \equiv (j+1)a \pmod{(a+b)}$, so that, by the way χ is defined, $\chi(y) \neq \chi(z)$. Also, if $y, z \in [1, a + b - 1]$ with z = y + b, then

(4.4) $y \equiv z \oplus a \,(\mathrm{mod}\,(a+b)),$

so that $y \equiv ta \pmod{(a+b)}$ where $t \in [2, a+b-1]$ (check that $t \neq 1$). So, from (4.4) and the definition of χ , we have $\chi(y) \neq \chi(z)$. Hence χ is $R(A_D, 2)$ -valid on [1, a+b-1], which completes the proof that $R(A_D, 2) = a + b$.

We now do the cases in which $R(A_D, k)$ is infinite. The case in which a and b are both odd is covered by Example 4.3. Now assume one of a and b is even (they cannot both be even). It remains only to show that $R(A_{\{a,b\}}, 3) = \infty$. We may assume that a is even.

To complete the proof, we shall exhibit a 2-coloring γ of \mathbb{Z}^+ that has period 2a and that has no monochromatic 3-term arithmetic progression with gap in D.

If $b \equiv i \pmod{2a}$ with $a \leq i < 2a$, then, by the periodicity of γ , if $X = \{x, x+b, x+2b\}$ were monochromatic under γ , then for any t > 0 with 2ta > b, the set $\{x+4ta, x+2ta+b, x+2b\}$ would be a monochromatic arithmetic progression with gap $2ta - b \equiv -i \pmod{2a}$. Hence, it is sufficient for us to consider only those b such that $b \equiv i \pmod{2a}$, where $1 \leq i \leq a$. Also, since γ has period 2a, and since b is odd (why?), we may assume $1 \leq b < a$.

We consider the following two cases, and define the coloring γ (with period 2a) according to the case. In each case we let B_j denote the interval [(j-1)a+1, ja] for $j \geq 1$.

Case 1. $1 \leq b < \frac{a}{2}$. For each $i \leq a$, let $\gamma(i) = 1$ if i is odd, and $\gamma(i) = 0$ if i is even; and for each i > a, let $\gamma(i) \neq \gamma(i-a)$. Then there is no 2-term monochromatic *a*-a.p., and γ has period 2*a*. If $\{x, x+b\}$ is monochromatic with $x \in B_j$, then by the way γ is defined, since b is odd, $x+b \in B_{j+1}$. Since $x+b \leq ja+\frac{a}{2}$, we have $x+2b \in B_{j+1}$, so that $\gamma(x+2b) \neq \gamma(x+b)$. Thus there is no 3-term monochromatic *b*-a.p. in \mathbb{Z}^+ , and γ has no monochromatic 3-term arithmetic progressions with gap in D.

4.1. Finite Gap Sets

Case 2. $\frac{a}{2} \leq b < a$. Define γ as follows. For every j, let $\gamma(B_{\ell}) = 1$ for ℓ odd, and $\gamma(B_{\ell}) = 0$ for ℓ even. Clearly, there is no monochromatic 2-term *a*-a.p. Also, if $\{x, x + b\}$ is monochromatic, with $x \in B_j$, then x + b must belong to B_j . Thus, $x + 2b \in B_{j+1}$ (or else 2b < a), so that $\gamma(x + 2b) \neq \gamma(x)$. So γ yields no monochromatic 3-term arithmetic progressions with gap in D.

If we look over the proof of Theorem 4.7, we see that we have proved something stronger than $R(A_{\{a,b\}},3) = \infty$: in certain cases in which $|D| = \{a,b\}$, it is possible to 2-color the positive integers so that there are neither monochromatic 2-term *a*-a.p.'s nor monochromatic 3-term *b*-a.p.'s. In fact, it is possible to prove several results about this type of "mixed" van der Waerden-type function. You will find some interesting results on this in the exercises of this chapter.

The last proof was rather long for a situation in which the structure seems quite uncomplicated and small. If we up the ante to |D| = 3, then, as you may have guessed, we do not have a theorem akin to Theorem 4.7 – but we can say something.

Theorem 4.8. Let |D| = 3. Then $R(A_D, 4) = \infty$.

Proof. Let $D = \{a, b, c\}$ and assume a < b < c. The proof splits naturally into two cases.

Case 1. $2b \leq c$. In this case let χ be the 2-coloring of \mathbb{Z}^+ defined recursively as follows:

(i) $\chi(i) = 1$ for $1 \le i \le a$, (ii) $\chi(i) \ne \chi(i-a)$ for $a < i \le b$, (iii) $\chi(i) \ne \chi(i-b)$ for $b < i \le c$, (iv) $\chi(i) \ne \chi(i-c)$ for i > c. Note that χ has period 2c. We will show that under χ there is no monochromatic 4-term arithmetic progression whose gap belongs to D.

Let $j = \lceil \frac{c}{b} \rceil$. For $1 \leq i < j$, let $B_i = \lfloor (i-1)b + 1, ib \rfloor$, and let $B_j = \lfloor (j-1)b + 1, c \rfloor$. Clearly, in each B_i , $1 \leq i \leq j$, there is no monochromatic 2-term *a*-a.p. Therefore there is no 3-term *a*-a.p. in $\lfloor 1, c \rfloor$. Notice that there are no monochromatic 2-term *a*-a.p.'s in $\lfloor 1, 2a \rfloor$, and therefore, for each nonnegative integer k, there also cannot be any in $\lfloor kc + 1, kc + 2a \rfloor$. Thus, in \mathbb{Z}^+ there is no monochromatic 4term *a*-a.p. By the same reasoning, there is no monochromatic 3term *b*-a.p. in \mathbb{Z}^+ (check this). Finally, it is clear from (iv) that there is no monochromatic 2-term *c*-a.p.

Case 2. 2b > c. Let χ' be the following 2-coloring of \mathbb{Z}^+ (note that χ' has period 4c):

- (i) $\chi'(i) = 1$ for $1 \le i \le a$,
- (ii) $\chi'(i) \neq \chi'(i-a)$ for $a < i \le b$,
- (iii) $\chi'(i) \neq \chi'(i-b)$ for $b < i \le 2c$,
- (iv) $\chi'(i) \neq \chi'(i-2c)$ for i > 2c.

Let d = 2c. Then in the same way that we used $2b \le c$ in Case 1, we may use $2b \le d$ to show that under χ' there is no monochromatic 4term *a*-a.p., no monochromatic 3-term *b*-a.p., and no monochromatic 2-term *d*-a.p. (we leave the details as Exercise 4.3). Since there is no monochromatic 2-term *d*-a.p., there cannot be a monochromatic 3-term *c*-a.p.

In each case, we have given a coloring of \mathbb{Z}^+ that yields no monochromatic 4-term arithmetic progression whose gap belongs to D, proving the theorem.

Based on the results obtained above, let us denote by m(n) the least positive integer m such that whenever D is a set of positive integers with |D| = n we have $R(A_D, m) = \infty$. The above theorems show that m(1) = 2, m(2) = 3, and $3 \le m(3) \le 4$. It would be nice to know more about the function m(k) (see Section 4.4 for some open problems).

4.2. Infinite Gap Sets

4.2. Infinite Gap Sets

Van der Waerden's theorem says that for all k and r, $R(A_{\mathbb{Z}^+}, k; r)$ exists (is finite). If we can find a proper subset D of \mathbb{Z}^+ such that $R(A_D, k; r)$ exists for all k and r, then we have strengthened van der Waerden's theorem. In the previous section we learned that no finite D will work. For infinite D, the answer is not so simple.

It is not hard to find certain sets D that strengthen van der Waerden's theorem. For example, by the next theorem we see that we may take D to be the set of all multiples of a fixed positive integer m. The theorem follows immediately from Lemma 4.6 by taking D(of Lemma 4.6) to be \mathbb{Z}^+ .

Theorem 4.9. Let m be a fixed integer with $m \ge 2$. Then for all positive integers k and r,

$$R(m\mathbb{Z}^+, k; r) = m(w(k; r) - 1) + 1$$

The following corollary is an easy consequence of Theorem 4.9. We leave its proof as Exercise 4.7.

Corollary 4.10. Let F be a finite set of positive integers. Then $R(\mathbb{Z}^+ - F, k; r) < \infty$.

We next consider a special case of Corollary 4.10 in which the results are quite interesting.

For c a positive integer, denote by w'(c, k; r) the least positive integer w' such that for every r-coloring of [1, w'] there is a monochromatic k-term arithmetic progression whose gap is at least c. Notice that if c = 1, then w'(c, k; r) coincides with the classical van der Waerden number w(k; r). We also see that $w'(c, k; r) = R(A_D, k; r)$, where $D = \{c, c + 1, c + 2, ...\}$, and by Corollary 4.10 we know that this number always exists. Let us take a closer look at the numbers w'(c, k; r).

For arithmetic progressions of length three, we have the following theorem, which is a generalization of the fact that w(3) = 9.

Theorem 4.11. Let c be a positive integer. Then w'(c,3) = 8c + 1.

4. Subsets of AP

Proof. Since w(3) = 9, by Proposition 2.29, any 2-coloring of

 $\{1, 1+c, 1+2c, \dots, 1+8c\}$

must have a monochromatic arithmetic progression of length three. Obviously, its gap is at least c. This shows that $w'(c,3) \leq 8c+1$.

To show that 8c + 1 is also a lower bound, consider the coloring of [1, 8c] represented by the string $A_1B_1A_2B_2$, where $A_i = 1^{2c}$ and $B_i = 0^{2c}$ for i = 1, 2.

We shall show, by contradiction, that under this coloring there is no monochromatic 3-term arithmetic progression whose gap is at least c. Assume there is such an arithmetic progression, $P = \{x, y, z\}$ with x < y < z. By the symmetry of the coloring, we may assume that P has color 1. Since $z - x \ge 2c$, one of the A_i 's must contain two members of P, and the other A_i must contain one member of P. Without any loss of generality, say $x, y \in A_1$ and $z \in A_2$. Then z - y > 2c, contradicting the fact that y - x < 2c. This shows that $w'(c, 3) \ge 8c + 1$, completing the proof.

We can use the same idea by which we obtained the lower bound in Theorem 4.11 to prove the following more general fact that has the lower bound of Theorem 4.11 as a special case.

Before stating the theorem, we mention some convenient notation.

Notation. For positive integers c, k, and r with $k, r \ge 2$, let $\lambda(c, k, r)$ denote the r-coloring $\lambda : [1, cr(k-1)^2] \rightarrow \{0, 1, \ldots, r-1\}$ defined by the string $(B_0B_1 \ldots B_{r-1})^{k-1}$, where for each $i = 0, 1, \ldots, r-1, B_i$ is a string of i's having length c(k-1), and where there are k-1 copies of the block $(B_0B_1 \ldots B_{r-1})$.

Example 4.12. The coloring $\lambda(3,5,2)$ is the coloring of [1,72] represented by the string $(0^{12}1^{12})^4 = 0^{12}1^{12}0^{12}1^{12}0^{12}1^{12}0^{12}1^{12}$.

Theorem 4.13. For all $c \ge 1$ and $k, r \ge 2$,

$$w'(c,k;r) \ge cr(k-1)^2 + 1.$$

Proof. To prove this, it suffices to show that, under $\lambda(c, k, r)$, there is no k-term monochromatic arithmetic progression whose gap is at least c. We leave the details as Exercise 4.8.

When dealing with the functions w'(c, k; r), the following modification of the terminology used for valid colorings will be useful.

Definition 4.14. An *r*-coloring that admits no monochromatic *k*-term arithmetic progressions with gap at least *c* is called a (c, k; r)-valid coloring. For r = 2, we call the coloring (c, k)-valid.

Theorem 4.13 gives a lower bound for w'(c, k; r). Of course, finding an upper bound would be much more significant, as it would provide an upper bound for the classical van der Waerden numbers w(k;r) (letting c = 1). This is one of the main reasons for investigating families \mathcal{F} that are subsets of AP. One possible approach to finding an upper bound on w'(c, k; r) would be to first find a fairly simple description of all maximal length (c, k)-valid colorings. Some conjectures along these lines have been formulated (see Sections 4.4 and 4.5). For the case of k = 3 and r = 2, it turns out that for each $c \geq 2$, the maximal length valid colorings are quite simple to describe – in fact, as we shall see in the next theorem, there is only one!

Notice that if χ is any valid *r*-coloring of the interval [a, b], then the *r*-coloring χ' obtained from χ by merely renaming the colors is still valid. Hence, when counting valid colorings we will not consider such pairs χ and χ' to be distinct colorings, unless otherwise stated.

It is well known that there are three distinct 2-colorings of [1, 8] for which there is no 3-term monochromatic arithmetic progression, i.e., which are (1, 3)-valid. These are represented by 11001100, 10011001, and 10100101 (the reader should verify this).

Note that the first of these colorings is the coloring $\lambda(1,3,2)$.

Theorem 4.17 below shows that for $c \ge 2$, $\lambda(c, 3, 2)$ is the only maximal length (c, 3)-valid 2-coloring of [1, 8c].

We shall need the following two lemmas.

Lemma 4.15. Let $c, k, m \in \mathbb{Z}^+$, and let χ be a (c, k)-valid 2-coloring of [1, mc]. Let $i \in \{1, 2, ..., c\}$ and let χ^* be the 2-coloring of [1, m] defined by $\chi^*(j) = \chi((j-1)c+i)$ for each j = 1, 2, ..., m. Then χ^* is (1, k)-valid on [1, m].

4. Subsets of AP

Proof. Assume χ^* is not (1, k)-valid on [1, m]. Then under χ^* there is a monochromatic arithmetic progression

 $\{x, x+d, \dots, x+(k-1)d\} \subseteq [1, m].$

Now consider the set $S = \{(x - 1 + jd)c + i : j = 0, 1, ..., k - 1\}$, an arithmetic progression contained in [1, mc]. It follows from the definition of χ^* that S is monochromatic under χ . Also, the gap of S is $cd \geq c$, contradicting the assumed validity of χ .

In the next two proofs we will be using the (AP, 3)-valid 2colorings (i.e., the (1, 3)-valid 2-colorings) of [1, 8], including ones that can be obtained from another coloring by interchanging the names of the two colors. Hence, there are six such colorings, which we denote as follows:

$\sigma = 11001100,$	$\tau = 10011001,$	$\mu = 10100101,$
$\sigma' = 00110011,$	$\tau' = 01100110,$	$\mu' = 01011010.$

Lemma 4.16. Let $c \geq 3$, and assume that $\chi : [1, 8c] \rightarrow \{0, 1\}$ is a (c, 3)-valid 2-coloring with $\chi(c) = 1$. Then $A = \{c, 2c, \ldots, 8c\}$ must have the color pattern $\sigma = 11001100$.

Proof. Define χ^* on [1,8] by $\chi^*(j) = \chi(jc)$. By Lemma 4.15 (taking i = c in its statement), χ^* is (1,3)-valid on [1,8]. Hence, because $\chi(c) = 1$, as noted before, χ^* has one of the color patterns σ , τ , or μ . Thus A has one of these three color patterns. To complete the proof, we shall show that it is impossible for A to have color pattern τ or μ .

We consider two cases.

Case 1. c is odd. Let $B = \{1, c + 1, 2c + 1, \dots, 7c + 1\}$. By Lemma 4.15, the function χ' defined on [1,8] by $\chi'(j) = \chi((j-1)c+1)$ has one of the six color patterns σ , σ' , τ , τ' , μ , μ' . Hence, under χ , B has one of these six color patterns.

Let us first assume that A has color pattern τ ; we will reach a contradiction.

If *B* has either coloring σ or μ' , then we have $\chi(c+1) = \chi(8c) = 1$. Hence, $\chi\left(\frac{9}{2}c + \frac{1}{2}\right) = 0$, for otherwise $\left\{c+1, \frac{9}{2}c + \frac{1}{2}, 8c\right\}$ would be a monochromatic arithmetic progression with gap at least *c*, which is not possible. This implies that $\{2c+1, \frac{9}{2}c+\frac{1}{2}, 7c\}$ is monochromatic under χ , a contradiction.

If B has one of the colorings σ' or τ , then $\chi(4c) = \chi(7c+1) = 1$. Hence, $\chi(c-1) = 0$. It follows that $\{c-1, 3c, 5c+1\}$ has color 0, a contradiction.

The remaining possibilities for coloring B are τ' and μ . For each of these cases, $\chi(2c+1) = \chi(5c) = 1$, so that $\chi(8c-1) = 0$. This implies that $\{4c+1, 6c, 8c-1\}$ is monochromatic under χ and its gap is at least c; again a contradiction.

To complete Case 1, now assume that A has color pattern μ .

If B has any of the color patterns σ , τ' , or μ , then $\chi(3c) = \chi(5c+1) = 1$. Therefore, we must have $\chi(c-1) = 0$ (by the validity of χ). This implies that $\{c-1, 2c, 3c+1\}$ is monochromatic, which is not possible. If B has either of the color patterns σ' or μ' , then $\chi(c) = \chi(6c+1) = 1$, which implies that $\chi(\frac{7}{2}c+\frac{1}{2}) = 0$; but then $\{2c, \frac{7}{2}c+\frac{1}{2}, 5c+1\}$ has color 0, a contradiction. Finally, if B has color pattern τ , then since $\chi(3c) = \chi(4c+1) = 1$, we must have $\chi(2c-1) = 0$; but then $\{2c-1, 4c, 6c+1\}$ is monochromatic, again a contradiction.

Case 2. *c* is even. This case has a very similar proof to that of Case 1, the main difference being that instead of using the set *B*, we use $C = \{2, c+2, 2c+2, \ldots, 7c+2\}$. We shall work out two of the subcases and leave the other subcases as Exercise 4.9.

Subcase i. A has color pattern τ and C has color pattern σ . In this case, $\chi(c+2) = \chi(8c) = 1$, so that $\chi(\frac{9}{2}c+1) = 0$. This then implies that $\{2c+2, \frac{9}{2}c+1, 7c\}$ must be monochromatic, which is not possible.

Subcase ii. A has color pattern τ and C has one of the color patterns σ' or τ . In this case, $\chi(4c) = \chi(7c+2) = 1$, and therefore $\chi(c-2) = 0$ (by assumption, $c-2 \ge 1$). Thus, all members of $\{c-2, 3c, 5c+2\}$ have color 0, a contradiction.

We are now ready to prove the theorem which shows that for all $c \ge 2$, $\lambda(c, 3, 2)$ is the only (c, 3)-valid coloring of [1, 8c].

4. Subsets of AP

Theorem 4.17. Let $c \geq 2$, and assume that χ is a (c,3)-valid 2-coloring of [1,8c] with $\chi(1) = 1$. Then $\chi = \lambda(c,3,2)$.

Proof. We can show that the theorem is true for c = 2 directly by checking that $\lambda(2,3,2) = 1111000011110000$ is the only (2,3)-valid 2-coloring of [1,16] (we leave this to the reader as Exercise 4.10).

Now let $c \geq 3$ and let χ be any (c, 3)-valid 2-coloring of [1, 8c] such that $\chi(c) = 1$. To complete the proof, it is sufficient to show that for each $i = 1, 2, \ldots, c$,

(4.5) $A_i = \{(j-1)c + i : 1 \le j \le 8\}$ has color scheme 11001100.

We know from Lemma 4.16 that (4.5) is true for i = c, so let us assume that $i \in \{1, 2, ..., c - 1\}$. Let χ_i be the coloring of [1,8] defined by $\chi_i(j) = \chi((j - 1)c + i)$. By Lemma 4.15, χ_i is (1,3)-valid on [1,8]. Therefore χ_i has one of the color patterns σ , σ' , τ , τ' , μ , μ' . Hence, we will complete the proof if we can show that A_i does not have any of the color patterns σ' , τ , τ' , μ , or μ' . We shall do this by contradiction.

First, assume A_i has one of the color patterns σ' , τ , or μ . Then $\chi(c+i) = \chi(3c) = 0$, so $\chi(5c-i) = 1$. Hence, $\{5c-i, 6c, 7c+i\}$ is monochromatic, contradicting the (c, 3)-validity of χ .

Next, assume A_i has the color pattern τ' . Then $\chi(2c + i) = \chi(5c) = 1$, implying that $\chi(8c-i) = 0$. This gives the monochromatic arithmetic progression $\{i, 4c, 8c - i\}$, a contradiction.

Finally, assume A_i has the pattern μ' . In this case, $\chi(i) = \chi(4c) = 0$, which implies that $\{3c + i, 5c, 7c - i\}$ is monochromatic, again impossible.

In the above discussion, for some constant c, the gaps in the arithmetic progressions are required to be no less than c. We can generalize this type of restriction if, instead of choosing c to be the same constant for all arithmetic progressions, the value of this minimum gap size is dependent on the particular arithmetic progression. Specifically, given a function $f : \mathbb{Z}^+ \to \mathbb{R}^+$, define w'(f(x), k; r) to be the least positive integer w', if it exists, such that for every r-coloring of [1, w'] there is a monochromatic k-term arithmetic progression $\{a, a+d, \ldots, a+(k-1)d\}$ with $d \ge f(a)$. Hence, the function

w'(c,k;r) represents the special case of w'(f(x),k;r) in which f is the constant function c.

Example 4.18. Let $f(x) = x^2$. Here we are interested in the collection of arithmetic progressions that consists of all those whose first term is 1, those whose first term is 2 with gap at least 4, those whose first term is 3 with gap at least 9, etc. Hence, for w'(f(x), 3; r) we would not consider progressions such as $\{2, 3, 4\}$, $\{2, 4, 6\}$, $\{2, 5, 8\}$, and $\{3, 11, 19\}$, but would consider those such as $\{2, 6, 10\}$, $\{2, 7, 12\}$, $\{3, 12, 21\}$, $\{3, 13, 23\}$, and $\{4, 20, 36\}$.

Definition 4.19. Let f be a function defined on the positive integers. If $X = \{a, a+d, \ldots, a+(k-1)d\}$ is an arithmetic progression such that $d \ge f(a)$, we shall call X an f-a.p. Further, if an r-coloring avoids monochromatic k-term f-a.p.'s, we will say that it is (f, k; r)-valid, or just (f, k)-valid if r = 2.

The existence of w'(c, k; r) is a direct consequence of van der Waerden's theorem. As we can see, for many choices of f(x), there is a much greater restriction placed on the arithmetic progressions than there is if we are only requiring that the gap be at least c. Hence, as you might guess, the existence of w'(f(x), k; r) is not as easy to show. In fact, as we shall see, for many functions f(x) it does not exist.

For 2-term arithmetic progressions, the question of the existence of w'(f(x), k; r) is relatively easy to answer. According to the next theorem, not only does it always exist, but we are able to give a precise formula for its value. For the purpose of keeping the statement of the theorem as simple as possible, we will assume that f is integer-valued; however, it is easy to see that w'(f(x), 2; r) will still exist in the more general case.

Before stating the theorem we remind the reader of the following notation.

Notation. If g is a function, we denote by $g^{(n)}(x)$ the n^{th} iterate of g (or the composition of g with itself n times). That is,

$$g^{(2)}(x) = g(g(x)), \ g^{(3)}(x) = g(g(g(x))), \ \text{etc.}$$

Also, we define $g^{(0)}(a)$ to equal a (that is, $g^{(0)}$ is the identity function).

Theorem 4.20. Let $f : \mathbb{Z}^+ \to \mathbb{Z}^+$ be a nondecreasing function. Let g(x) = f(x) + x. Then $w'(f(x), 2; r) = g^{(r)}(1)$.

Before delving into the proof, we look at why the existence of w'(f(x), 2; r) is not surprising. Obviously, for any *r*-coloring of \mathbb{Z}^+ , some color must occur an infinite number of times. Hence, if integer x is colored with this color, then clearly there is another integer y of the same color such that $y - x \ge f(x)$ for any function f.

Proof of Theorem 4.20. To show that $g^{(r)}(1)$ is an upper bound, let χ be any *r*-coloring of $[1, g^{(r)}(1)]$. Applying the pigeonhole principle, we see that there exist two members of the (r + 1)-element set $\{1, g(1), g^{(2)}(1), \ldots, g^{(r)}(1)\}$ that have the same color. Say

$$\chi(g^{(i)}(1)) = \chi(g^{(j)}(1)),$$

where $0 \le i < j \le r$. Since

$$(4.6) g^{(j)}(1) - g^{(i)}(1) \ge g^{(i+1)}(1) - g^{(i)}(1) = f(g^{(i)}(1)),$$

 $\{g^{(i)}(1),g^{(j)}(1)\}$ is a 2-term monochromatic f-a.p. This proves the upper bound.

To show that $g^{(r)}(1)$ is also a lower bound for w'(f(x), 2; r), we give a 2-coloring of $[1, g^{(r)}(1) - 1]$ that has no monochromatic pair $\{a, a + d\}$ with $d \ge f(a)$. Note that by (4.6), for each $i = 1, 2, \ldots, r$, the interval $A_i = [g^{(i-1)}(1), g^{(i)}(1) - 1]$ is not empty. Now define $\chi(A_i) = i$ for each $i = 1, 2, \ldots, r$. By (4.6), no two members of A_i differ by more than $f(g^{(i-1)}(1)) - 1$. Since f is nondecreasing, for each i there does not exist $\{a, a + d\} \in A_i$ with $d \ge f(a)$. This completes the proof.

We now consider the existence of w'(f(x), 3; 2) when f is a function from \mathbb{Z}^+ to \mathbb{R}^+ . The next theorem shows that w'(f(x), 3; 2)always exists.

Theorem 4.21. For any function $f : \mathbb{Z}^+ \to \mathbb{R}^+$, $w'(f(x), 3; 2) < \infty$.

Proof. Obviously, if the theorem is true for a function f_1 and if $f_1(x) \ge f_2(x)$ for all $x \in \mathbb{Z}^+$, then the theorem is also true for f_2 . Hence, there is no loss of generality if we assume that f is a nondecreasing function.

We will show that for every 2-coloring of \mathbb{Z}^+ there is a monochromatic 3-term arithmetic progression of the desired type. The theorem then follows by the Compactness Principle (see Section 2.1).

Let χ be a 2-coloring of \mathbb{Z}^+ . We can think of χ as a sequence of **Q**'s and 1's: $\chi = \chi(1)\chi(2)\chi(3)\dots$ We consider two cases.

Case 1. χ does not include infinitely many copies of the pattern 001 or the pattern 110 (i.e., there do not exist infinitely many t such that $\chi(t) = \chi(t+1)$ and $\chi(t) \neq \chi(t+2)$). In this case, there is some n so that $\chi(n)\chi(n+1)\chi(n+2)\ldots$ is one of the infinite binary sequences 000..., 111..., or 101010.... For each of these three possibilities, there is obviously a 3-term arithmetic progression whose first term is n and whose gap is at least f(n).

Case 2. One of the patterns 001 or 110 occurs infinitely often. Without loss of generality, say there are infinitely many occurrences of 001. Let $\chi(a) = 0$, $\chi(a + 1) = 0$, and $\chi(a + 2) = 1$ be one of these occurrences. Then there exists another occurrence, $\chi(b) = 0$, $\chi(b+1) = 0$, and $\chi(b+2) = 1$, where $b - a \ge f(a+2)$ (otherwise there would not be infinitely many occurrences of 001). Let d = b - a. So $\{a, a + 1, a + d, a + d + 1\}$ has color 0, and $\{a + 2, a + d + 2\}$ has color 1.

If $\chi(a + 2d + 2) = 1$, then $\{a + 2, a + d + 2, a + 2d + 2\}$ is a monochromatic arithmetic progression with gap $d \ge f(a + 2)$. If $\chi(a + 2d + 2) = 0$, then $\{a, a + d + 1, a + 2d + 2\}$ is a monochromatic arithmetic progression with gap $d + 1 \ge f(a + 2) \ge f(a)$.

In both cases we have produced the desired type of arithmetic progression. $\hfill \Box$

The following theorem, which also tells us that w'(f(x), 3; 2) exists, is more useful than Theorem 4.21 because it provides more information about the magnitude of w'(f(x), 3; 2). We do not include the proof.

Theorem 4.22. Let $f : \mathbb{Z}^+ \to \mathbb{R}^+$ be a nondecreasing function. Let $\beta = 1 + 4 \left\lceil \frac{f(1)}{2} \right\rceil$. Then

$$w'(f(x),3;2) \leq \left\lceil 4f\left(\beta + 4\left\lceil \frac{f(\beta)}{2} \right\rceil\right) + 14\left\lceil \frac{f(\beta)}{2} \right\rceil + \frac{7\beta - 13}{2} \right\rceil$$

120

4. Subsets of AP

Theorem 4.22 gives an upper bound for w'(f(x), 3; 2). The next theorem provides a lower bound. To keep the notation simpler, we assume $f : \mathbb{Z}^+ \to \mathbb{Z}^+$, although a more general result can be obtained by the same method of proof.

Theorem 4.23. Let $f : \mathbb{Z}^+ \to \mathbb{Z}^+$ be a nondecreasing function with $f(n) \ge n$ for all $n \in \mathbb{Z}^+$. Let h = 2f(1) + 1. Then

 $w'(f(x), 3; 2) \ge 8f(h) + 2h + 2 - t,$

where t is the largest integer such that $f(t) + t \leq 4f(h) + h + 1$.

Proof. Let m = 8f(h) + 2h + 1 - t. To prove the theorem we shall give a 2-coloring of [1, m] under which there is no monochromatic 3-term f-a.p.

We begin by partitioning [1, m] into the following four intervals:

$$\begin{split} I_1 &= [1, h-1], \\ I_2 &= [h, 2f(h) + h - 1], \\ I_3 &= [2f(h) + h, 4f(h) + h] \\ I_4 &= [4f(h) + h + 1, m]. \end{split}$$

Now define the 2-coloring χ on [1, m] as follows: let $\chi(I_1 \cup I_3) = 1$ and $\chi(I_2 \cup I_4) = 0$.

Assume that there is a monochromatic 3-term arithmetic progression $P = \{a, b, c\}$ with $d = b - a = c - b \ge f(a)$. Since $\chi(a) = \chi(b)$, we have the following six cases, each of which leads to a contradiction.

Case 1. $a, b \in I_1$. Then $d \leq h-2$, and therefore $c = b + d \leq 2h - 3$. Also, since $d \geq f(1)$, we have $c \geq 1 + 2f(1) = h$. Thus, $c \in I_2$, which contradicts the fact that P is monochromatic.

Case 2. $a \in I_1$ and $b \in I_3$. In this case $d \ge 2f(h) + 1$, and therefore $c \ge 4f(h) + h + 1$. This implies $\chi(c) = 0$, a contradiction.

Case 3. $a, b \in I_3$. Then $c \ge a + 2f(a) \ge 3a \ge 6f(h) + 3h$, and therefore $\chi(c) = 0$, so P is not monochromatic.

Case 4. $a, b \in I_2$. In this case we have $d \leq 2f(h) - 1$, and hence $c \leq b + 2f(h) - 1 \leq 4f(h) + h - 2$. Also, $c \geq a + 2f(a) \geq h + 2f(h)$. So $c \in I_3$, a contradiction.

$$\begin{aligned} c &\geq a+2f(a) \\ &\geq t+1+2f(t+1) \\ &> 4f(h)+h+1+f(t+1) \\ &> 8f(h)+2h+2-(t+1)=m, \end{aligned}$$

which is not possible.

Case 6. $a, b \in I_4$. Then $c \ge a + 2f(a) \ge 3a > m$, which is not possible.

Example 4.24. It is not hard to compute bounds based on Theorems 4.22 and 4.23 for the function f(x) = x + c, where c is a nonnegative integer. If c is odd, we obtain

$$\frac{1}{2}(43c+49) \le w'(x+c,3;2) \le 64c+61$$

Similar bounds may be obtained if c is even; we leave this as Exercise 4.11.

By the next theorem, we can improve upon the upper bound that is supplied by Theorem 4.22 for w'(x+c,3;2). The proof makes use of the fact that w'(x,3;2) = 24; this and some other computer-generated values are presented in the following table.

f(x)	w'(f(x), 3; 2))			
x	24			
x + 1	46			
x+2	67			
x + 3	89			
x + 4	110			
x + 5	132			
2x	77			
2x + 1	114			

Table 4.1: Values of w'(f(x), 3; 2)

Interestingly, every one of these values agrees exactly with the lower bound given by Theorem 4.23.

Theorem 4.25. Let c be a nonnegative integer. Then $\frac{43}{2}c + 24 + \delta \le w'(x + c, 3; 2) \le 23c + 24,$

where $\delta = 0$ if c is even and $\delta = \frac{1}{2}$ if c is odd.

Proof. The lower bound follows immediately from Example 4.24. For the upper bound, let χ be any 2-coloring of [1, 23c+24]. Let χ' be the 2-coloring of [1,24] defined by $\chi'(i) = \chi((c+1)(i-1)+1)$. Since, as noted above, w'(x,3;2) = 24, under χ' there exists a monochromatic arithmetic progression $\{a, a + d, a + 2d\}$ with $d \ge a$. Therefore

$$\{(c+1)(a-1)+1, (c+1)(a+d-1)+1, (c+1)(a+2d-1)+1\}$$

is an arithmetic progression that is monochromatic under χ with a gap that is no less than (c+1)a = (c+1)(a-1) + 1 + c. Hence, it is a monochromatic *f*-a.p. where f(x) = x + c, which establishes the upper bound.

We know from Theorem 4.21 that w'(f(x), 3; 2) always exists. In contrast, w'(f(x), 4; 2) does not always exist; in fact, it only exists if f is a rather slowly growing function. The situation is similar when more than two colors are used, even for 3-term arithmetic progressions. The details are found in the following theorem.

Theorem 4.26. For $k \geq 3$ and $r \geq 2$, let

$$c = \frac{\sqrt[r-1]{2} - 1}{k - 1}.$$

If $k \ge 4$ or $r \ge 3$, then w'(cx, k; r) does not exist.

Proof. Assume $k \ge 4$ or $r \ge 3$ and let $b = \sqrt[r-1]{2}$. To prove the theorem, we give an *r*-coloring of \mathbb{Z}^+ for which there does not exist a monochromatic arithmetic progression $\{a + jd : 0 \le j \le k - 1\}$ with $d \ge ca$.

Let $\chi : \mathbb{Z}^+ \to \{0, 1, \dots, r-1\}$ be defined as follows. Whenever $x \in B_i = [b^i, b^{i+1} - 1]$, where *i* is a nonnegative integer, let $\chi(x) = \overline{i}$ where $i \equiv \overline{i} \pmod{r}$ and $0 \leq \overline{i} \leq r-1$.

Let us assume that $\{a, a + d, ..., a + (k-1)d\}$ is monochromatic under χ . We will complete the proof by showing that d must be less than ca. Let n be such that $b^n \leq a + d < b^{n+1}$. Then since $d < b^{n+1}$, we know that $a + 2d < 2b^{n+1} = b^{n+r}$. Since $\chi(a + 2d) = \chi(a + d)$, the only B_i that a + 2d can belong to is B_n .

Now let us look at a + 3d. Since $a + 2d \in B_n$, we have

$$a + 3d = (a + 2d) + d < 2b^{n+1} = b^{n+r}$$

Since $\chi(a+3d) = \chi(a+d)$, this implies $b^n \le a+d < a+3d < b^{n+1}$, so that $a+3d \in B_n$.

We see that by applying this same line of reasoning, we will obtain $a + 4d, a + 5d, \ldots, a + (k-1)d \in B_n$. Since

$$b^n \le a + d < a + (k - 1)d < b^{n+1},$$

 $d < \frac{b^{n+1} - b^n}{k - 2}.$

we have

4.2. Infinite Gap Sets

Thus,

$$a \ge b^n - d > b^n \left(1 - \frac{b-1}{k-2}\right) \ge \frac{b^n}{2} = b^{n-(r-1)}$$

Since $\chi(a) = \chi(a+d)$, it follows that $a \in B_n$.

We have shown that $b^n \leq a < a + (k-1)d < b^{n+1}$, so that

$$d < \frac{b^{n+1} - b^n}{k-1} = cb^n \le ca.$$

Thus, we have shown that there is no monochromatic k-term arithmetic progression $\{a + jd : 0 \le j \le k - 1\}$ whose gap is at least ca, thereby proving the theorem.

For the case in which r = 2, the following theorem, which we offer without proof, improves upon Theorem 4.26.

Theorem 4.27. Let $k \geq 4$. Then

$$w'\left(\frac{x}{k^2-4k+3},k;2\right)$$

does not exist.

Theorems 4.11, 4.13, 4.17, 4.20–4.23, 4.26 and 4.27 deal with van der Waerden-type problems in which the set of allowable gaps is restricted by insisting that the gaps exceed a certain value. In particular, in Theorems 4.11, 4.13, and 4.17 we restricted the allowable

gaps to belong to sets of the form $D = \{c, c+1, c+2, ...\}$ for a fixed positive integer c, and found, among other things, that in this case the associated van der Waerden-type numbers always exist. There are, of course, other ways to restrict the set of allowable gaps and ask if the corresponding van der Waerden-type function exists. It would be very nice if we could characterize which sets D have the property that $R(A_D, k; r)$ exists for all k and r. Such a characterization is not known. However, we will give a partial answer by addressing the following questions:

What general properties must a set D have in order for R(A_D, k; r) to exist for all k and r?;
 Are there any properties that preclude A_D from having this property?

It will be convenient to introduce some terminology.

Definition 4.28. Let $D \subseteq \mathbb{Z}^+$. For a fixed positive integer r, we say that D is r-large if $R(A_D, k; r)$ exists for all k. We call D large if it is r-large for all $r \geq 1$.

Note that if $s \ge r$, then any set that is s-large must also be r-large.

Let us consider Question 1 above. That is, what properties are necessary for D to be large? One such property is fairly easy to prove. We state it in the following theorem. Note that the theorem gives a condition that is necessary in order for a set to be 2-large. By the previous paragraph, it is also a necessary condition in order for the set to be large.

Theorem 4.29. If D is 2-large, then for each positive integer m, D contains a multiple of m.

Proof. Assume that D is a set not containing a multiple of every positive integer. Let $n \in \mathbb{Z}^+$ be such that D contains no multiple of n. The proof is completed by showing that D is not 2-large, i.e., that there is a 2-coloring of \mathbb{Z}^+ under which there do not exist arbitrarily long monochromatic arithmetic progressions with gap belonging to D. This may be done by an argument that is essentially the same

as the proof of Theorem 4.4; we leave the details to the reader as Exercise 4.16. $\hfill \Box$

Note that the condition of Theorem 4.29 – that D must contain a multiple of every positive integer – is equivalent to the strongersounding condition that D must contain an infinite number of multiples of every positive integer.

As it turns out, a sequence of positive integers that grows too fast cannot be large. Here is one result along these lines. For other interesting results of this type, we refer the reader to the references given in Section 4.5.

Theorem 4.30. Let $D = \{d_i\}_{i=1}^{\infty}$ be an increasing sequence of positive integers such that d_i divides d_{i+1} for all *i*. Then D is not 2-large.

Proof. Define a 2-coloring $\chi : \mathbb{Z}^+ \to \{0,1\}$ recursively as follows. First, for all $x \in [1, d_1]$ let $\chi(x) = 1$. Once χ has been defined on $[1, d_i]$, we extend χ to $[1, d_{i+1}]$ by having $\chi(x) \neq \chi(x - d_i)$ for each $x \in [d_i + 1, d_{i+1}]$.

To complete the proof we will show that for each $i \ge 1$ and each $j \ge 1$, there is no 3-term monochromatic d_i -a.p. contained in $[1, d_j]$.

First note that, by the way χ is defined on each of the intervals $[d_i + 1, d_{i+1}]$, for every $i \ge 1$ there can be no 2-term monochromatic d_i -a.p. contained in $[1, d_{i+1}]$.

Now assume that $j \ge i+2$ and that $x_1 < x_2 < x_3$ is a monochromatic d_i -a.p. that is contained in $[1, d_j]$. Since d_{i+1} divides d_j , by the way χ is defined we see that every subinterval of $[1, d_j]$ of the form $[kd_{i+1}+1, (k+1)d_{i+1}], k \ge 1$, either has the same color pattern as $[1, d_{i+1}]$, or has the pattern obtainable from that of $[1, d_{i+1}]$ by replacing all 1's by 0's and vice versa. Hence, since there is no 2-term monochromatic d_i -a.p. in $[1, d_{i+1}]$, neither of the pairs $\{x_1, x_2\}$ or $\{x_2, x_3\}$ can be contained in any one interval $[kd_{i+1} + 1, (k+1)d_{i+1}]$. This implies that $x_3 - x_1 > d_{i+1}$, and hence $x_3 - x_1 \ge 2d_i$, which contradicts the fact that $\{x_1, x_2, x_3\}$ is a d_i -a.p.

Example 4.31. Let $G = \{a, ar, ar^2, ar^3, ...\}$ be an infinite geometric sequence. Then $R(A_G, k; 2)$ does not exist for large enough k. In other words, there exists a 2-coloring of \mathbb{Z}^+ that does not have, for

4. Subsets of AP

some k, a monochromatic k-term arithmetic progression with difference from G.

We now shift our attention to finding sufficient conditions for a set to be large. We begin with a very useful theorem that tells us that if a union of a finite number of sets is large, then at least one of the sets itself must be large. This theorem is itself a Ramsey theorem in the sense in which we described Ramsey theory in Chapter 1: Ramsey theory is the study of the preservation of properties under set partition. That is, this theorem states that if we have a set that is large, and if we partition this set into n subsets, then one of the subsets must be a large set, i.e., if we n-color a large set, then there must exist a monochromatic large subset.

Theorem 4.32. Let D be a large set, and let $n \ge 1$. If

$$D = D_1 \cup D_2 \cup \cdots \cup D_n,$$

then some D_i is large.

Proof. We will prove the theorem for n = 2. It is then a simple induction argument to extend it to general n. Let $D = D_1 \cup D_2$, and assume that neither D_1 nor D_2 is large. We will prove the theorem by showing that D is not large.

Since D_1 is not large, there exist positive integers k_1 and r, and some *r*-coloring χ of \mathbb{Z}^+ , under which there is no monochromatic k_1 term D_1 -a.p. Similarly, there exist positive integers k_2 and s, and an *s*-coloring ϕ of \mathbb{Z}^+ , under which there is no monochromatic k_2 -term D_2 -a.p.

Now, let σ to be the *rs*-coloring of \mathbb{Z}^+ given by $\sigma(i) = (\chi(i), \phi(i))$. Thus, the "colors" of σ consist of all ordered pairs whose first coordinate is one of the *r* colors of χ and whose second coordinate is one of the *s* colors of ϕ . Let $k = \max\{k_1, k_2\}$. To show that *D* is not large it will suffice to show that if *P* is any *k*-term arithmetic progression that is monochromatic under σ , then the gap of *P* does not belong to *D*.

Let $P = \{x_1, x_2, \ldots, x_k\}$ be an arithmetic progression that is monochromatic under σ . Then $\chi(x_1) = \chi(x_2) = \cdots = \chi(x_k)$. Hence, by our assumptions about χ and the fact that $k \ge k_1$, the gap of P cannot belong to D_1 . Similarly, $\phi(x_1) = \phi(x_2) = \cdots = \phi(x_k)$, which implies that the gap of P cannot belong to D_2 . Hence the gap of P does not belong to $D_1 \cup D_2$. Since $D = D_1 \cup D_2$, the proof is complete.

The following fact is an immediate consequence of Theorems 4.4 and 4.32.

Corollary 4.33. Let D be large, and F be finite. Then D - F is large.

Note that the proof of Theorem 4.32 actually tells us more than the statement of the theorem. It says that if D_1 is not *r*-large, and if D_2 is not *s*-large, then $D_1 \cup D_2$ is not *rs*-large. Extending this statement to *n* sets, it tells us that if, for each i = 1, 2, ..., n, D_i is not r_i -large, then $D_1 \cup D_2 \cup \cdots \cup D_n$ is not $r_1r_2 \cdots r_n$ -large. Looking at the contrapositive of this, and speaking in somewhat more general terms, it describes a way of going from a set that has a certain degree of largeness, to a subset of the set that has another (smaller) degree of largeness (that is, going from $r_1r_2 \cdots r_n$ -largeness to the r_i -largeness of D_i for some i). On the other hand, it is generally more difficult to prove results that involve showing that the *r*-largeness of a certain set implies the *r*-largeness of a proper subset of that set. One such result that is relatively easy to prove, however, is the following refinement of the above corollary.

Theorem 4.34. Let r be a positive integer, and assume D is r-large. If F is finite, then D - F is r-large.

Proof. It is sufficient to show that for every $d \in D$, the set $D - \{d\}$ is *r*-large (why?). We shall do this by contradiction. So assume that $d_0 \in D$, and that $D - \{d_0\}$ is not *r*-large. Thus, there exist an *r*-coloring χ of \mathbb{Z}^+ , and a positive integer *k*, such that under χ there is no monochromatic *k*-term arithmetic progression whose gap belongs to $D - \{d_0\}$. Since *D* is *r*-large, this implies that under χ there are arbitrarily long monochromatic d_0 -a.p.'s. By Theorem 4.29, $md_0 \in D$ for some $m \geq 2$. It is clear that under χ there are arbitrarily long monochromatic md_0 -a.p.'s. Since $md_0 \in D - \{d_0\}$, we have a contradiction.
4.3. Exercises

128

In this text we are limiting our methods to what are considered elementary in the sense that we are relying on the basic techniques of combinatorics and number theory on the set of integers. Some very useful and powerful theorems in Ramsey theory have also been proven by entirely different techniques. For example, several important improvements over van der Waerden's original theorem have been found by the use of ergodic theory and the methods of measure-preserving systems. These are very deep results, and we will not be including any of their proofs here, but we would like to mention at least one important special case of one of these results, which we shall phrase in the language of large sets.

4. Subsets of AP

Theorem 4.35. Let p(x) be a polynomial with integer coefficients, with positive leading coefficient, and whose constant term is 0. Then $\{p(i) > 0 : i \in \mathbb{Z}^+\}$ is large.

Theorem 4.35 tells us that the range of any polynomial with the stated conditions gives us a large set. Denoting, for a polynomial f, the set $\{f(x) : x \in \mathbb{Z}^+\}$ by range(f), we see, for example, that range(p) is large when $p(x) = x^2$. That is, $\{1^2, 2^2, 3^2, \ldots\}$ is a large set. Even a set as sparse as $\{x^{1000} : x \in \mathbb{Z}^+\}$ is large.

Theorem 4.35 is a generalization of van der Waerden's theorem, because van der Waerden's theorem tells us that \mathbb{Z}^+ itself is large, which may be considered as the range of the polynomial f(x) = x, $x \in \mathbb{Z}^+$.

One hypothesis of Theorem 4.35 is that the constant term of p(x) equals 0, i.e., that x|p(x). We may extend Theorem 4.35 to polynomials with any linear factor x + a, where a is an integer, as given by the following corollary.

Corollary 4.36. Let p(x) be a polynomial with integer coefficients, with leading coefficient positive, and such that x + a divides p(x) for some integer a. Then $\{p(i) : i \in \mathbb{Z}^+\}$ is large.

Proof. Let p(x) = (x + a)q(x), and let $\hat{p}(x) = p(x - a)$. Then $\hat{p}(x) = xq(x - a)$.

By Theorem 4.35, range $(\hat{p}) \cap \mathbb{Z}^+$ is large. If $a \leq 0$, then we have range $(\hat{p}) \subseteq \operatorname{range}(p)$, and therefore range $(p) \cap \mathbb{Z}^+$ is large. If a > 0,

then range $(p) = \text{range}(\hat{p}) - F$, where F is a finite set, and hence by Corollary 4.33, range $(p) \cap \mathbb{Z}^+$ is large.

There are some other interesting results, both positive and negative, concerning monochromatic arithmetic progressions that satisfy certain properties. We mention these in Section 4.5.

4.3. Exercises

- **4.1** Show that if |D| = 1, then $R(A_D, 2; 2) = \infty$, by exhibiting a specific 2-coloring of the positive integers.
- **4.2** Let s be a positive integer. Prove that if $R(A_D, k; 2)$ is finite, then

 $R(A_{sD}, k; 2) = s[R(A_D, k; 2) - 1] + 1;$

and that if $R(A_D, k; 2) = \infty$, then $R(A_{sD}, k; 2) = \infty$.

- **4.3** Fill in the details of the proof of Case 2 of Theorem 4.8 (i.e., explain why there is no 4-term monochromatic *a*-a.p., no 3-term monochromatic *b*-a.p., and no 3-term monochromatic *c*-a.p.).
- **4.4** Define $w((d_1, d_2), k_1, k_2)$ to be the least positive integer n such that for every 2-coloring of [1, n] there is either a monochromatic k_1 -term arithmetic progression whose gap is d_1 , or a monochromatic k_2 -term arithmetic progression whose gap is d_2 . Prove that for all $d_2, k_2 \geq 2$,

$$w((1, d_2), 2, k_2) = d_2(k_2 - 1) + 1.$$

4.5 In this exercise, we extend the notation of Exercise 4.4 above by denoting by w((d₁, d₂, d₃), k₁, k₂, k₃) the least positive integer n such that for every 2-coloring of [1, n] there is some i ∈ {1, 2, 3} such that there is a k_i-term monochromatic arithmetic progression with gap d_i. Let g = gcd(d₁, d₂), and assume that exactly one of the numbers d_{1/g} and d_{2/g} is even.
a) Use Theorem 4.7 to prove that

 $w((d_1, d_2, d_3), 2, 2, 3) \le d_1 + d_2 - g + 1.$

b) Assuming further that $d_3 = \max\{d_1, d_2, d_3\}$, prove that $w((d_1, d_2, d_3), 2, 2, 3) = d_1 + d_2 - g + 1$.

4.6 Using the notation of Exercise 4.5, prove that

$$w((1, d_2, d_3), 2, k_2, k_3) = (k_2 - 1)d_2 + 1$$

if exactly one of d_2 and d_3 is odd.

- 4.7 Prove Corollary 4.10.
- 4.8 Complete the proof of Theorem 4.13.
- 4.9 Complete the proof of Case 2 of Lemma 4.16.
- **4.10** Let χ be a (2, 3)-valid 2-coloring of [1,16] with $\chi(1) = 1$. Show that $\chi = 1111000011110000$. Do this by direct computation; do not use Theorem 4.17.
- 4.11 Finish Example 4.24 for the case when c is even.
- **4.12** Use Theorems 4.22 and 4.23 to find upper and lower bounds for w'(mx, 3; 2) if m is an odd integer and $m \ge 3$.
- **4.13** Use the method employed in the proof of Theorem 4.25 to prove that for any positive integer b,

 $w'(bx + bc, 3; 2) \le (w'(bx, 3; 2) - 1)c + w'(bx, 3; 2).$

- **4.14** It is known that w'(2x, 3; 2) = 77. Use this to find an upper bound for w'(2(x + c), 3; 2) in terms of c.
- **4.15** Prove that for all k and r,

$$w'\left(rac{x}{w(k;r)-k+1},k;r
ight)=w(k;r),$$

- so that there is at least some constant c = c(k; r) such that w'(cx, k; r) exists (w(k; r) here represents the ordinary van der Waerden function).
- 4.16 Complete the proof of Theorem 4.29.
- **4.17** Prove that if D is large, and m is a positive integer, then mD is large.
- **4.18** Prove that if D is large and m is a positive integer, then $D \{x : m \nmid x\}$ is large.
- **4.19** Let us call a set of positive integers "small" if it is not large. Must the complement (with respect to \mathbb{Z}^+) of a small set be large? Must the complement of a large set be small?

- 4.4. Research Problems
- **4.20** Prove: if D is r-large, and if all elements of D are multiples of the positive integer m, then $\frac{1}{m}D$ is r-large. Hence if D is large, then $\frac{1}{m}D$ is large.

4.4. Research Problems

- 4.1 Do a more complete study of $R(A_D, k; 2)$. In particular, define m = m(n) to be the least positive integer such that for all D with |D| = n, $R(A_D, m; 2) = \infty$. By the work discussed in this chapter, we know that m(1) = 2, m(2) = 3, and m(3) = 3 or 4. Determine the exact value of m(3) (it has been conjectured that m(3) = 3). Reference: [164]
- **4.2** Using the notation of Research Problem 4.1, find an upper bound on m(n) as a function of n. In particular, is it true that $m(n) \leq n$ for all n? Reference: [164]
- **4.3** Using the notation of Exercises 4.4–4.6, characterize those triples (d_1, d_2, d_3) for which $w((d_1, d_2, d_3), 2, 3, 3) < \infty$. Reference: [164]
- **4.4** Find a formula for $w((d_1, d_2, d_3), 2, 2, 3)$ (see Exercise 4.5). Reference: [164]
- **4.5** Prove or disprove: the lower bound of Theorem 4.25 is the precise value of w'(x+c, k; 2) (computer output suggests that this may well be the precise value; see Table 4.1). Reference: [64]
- **4.6** It is known that there is a function f(x) that tends to infinity as x goes to infinity, such that for each r, w'(f(x), k; r) exists for all k. This function grows very slowly (something like an inverse of the classical van der Waerden function w(k; r)). Try to find a faster growing function that still has this property.

References: [53], [64]

4.7 Find results analogous to Theorem 4.11 or 4.17 for w'(c, 4; 2) or w'(c, 3; 3). Reference: **[64**]

- 4. Subsets of AP
- **4.8** Let A(k;r) be the set of all positive real numbers a such that $w'(ax, k; r) < \infty$. It has been discovered that $w'(\frac{x}{4}, 4; 2)$ exists (it happens to equal 134), so that $\frac{1}{4} \in A(4; 2)$. As noted in this chapter, $w'(\frac{x}{3}, 4; 2)$ does not exist. Thus, letting $\beta(k;r)$ denote $\sup\{a: a \in A(k;r)\}$ we have $\frac{1}{4} \leq \beta(4; 2) \leq \frac{1}{3}$. Find the exact value of $\beta(4; 2)$, or improve its bounds. Do the same for $\beta(3; 3)$ (it is known that $\frac{1}{25} \leq \beta(3; 3) \leq \frac{\sqrt{2}-1}{2}$) and for $\beta(3; 4)$ (here it is known that $\frac{1}{74} \leq \beta(3; 4) \leq \frac{\sqrt{2}-1}{2}$). Reference: [64]
- **4.9** The following is known: if $w'(c, k; 2) = 2c(k-1)^2 + 1$ and j is a positive integer, then $w'(jc, k; 2) = 2jc(k-1)^2$. It would be desirable to know if the following stronger statement holds: if $w'(c, k; 2) = 2c(k-1)^2 + 1$ and $m \in \mathbb{Z}^+$, then

$$w'(c+m,k;2) = 2(c+m)(k-1)^2 + 1.$$

References: [64], [165]

- **4.10** Theorem 4.27 prompts us to ask whether the fastest growing function f, such that w'(f(x), k; 2) exists for large enough k, must grow like the function $\frac{x}{k^2}$. If we cannot answer this, or if the answer is no, then one might try to answer the following question: does $w'(\frac{x}{2^k}, k; 2)$ exist for k large enough? Writing a computer program to calculate various values of w'(f(x), k; 2) is likely to be a big help with these types of questions. Reference: [64]
- **4.11** Theorem 4.32 says that if a finite union of sets is large, then at least one of the sets must be large. Is the same true if we replace the word "large" with "*r*-large?" In particular, is it true that if $D \cup E$ is 2-large, then either D or E must be 2-large? The proof of Theorem 4.32 shows that if D is not 2-large and E is not 2-large, then $D \cup E$ is not 4-large, but it does not tell us about the 2-largeness or 3-largeness of $D \cup E$. It follows from Theorems 4.29 and 4.30 that neither $\{2n-1: n \in \mathbb{Z}^+\}$ nor $\{n!: n \in \mathbb{Z}^+\}$ is 2-large. Hence, the union of these two sets is not 4-large. Is the union 2-large?

Reference: [62]

- **4.12** Which sets *B* have the property that some translation of *B* is large? In other words, for which sets *B* does there exist an integer *t* such that $D = B + t = \{b + t : b \in B\}$ is large? By Theorem 4.35, the range of a polynomial with integer coefficients and positive leading coefficient has this property, because if f(x) is the polynomial, then f(x) f(0) is a polynomial with a zero constant term. References: [62], [65]
- **4.13** Let p(x) be any polynomial with integer coefficients, positive leading coefficient, and p(0) = 0, and let D be a large set. Determine if $\{p(d) : d \in D\}$ must be large? In particular, must $\{d^2 : d \in D\}$ be large? References: [62], [65]
- **4.14** For m a positive integer and $0 \le a < m$, denote by $S_{a(m)}$ the set of all arithmetic progressions whose gaps are not congruent to $a \pmod{m}$. Do a study of $R(S_{a(m)}, k; r)$. References: [64], [65], [160]

4.5. References

§4.1. Theorems 4.7 and 4.8 are found in [164]. More generally, [164] considers the function $w(\vec{d}, \vec{k})$, where $\vec{d} = (d_1, d_2, \ldots, d_n)$ and $\vec{k} = (k_1, k_2, \ldots, k_n)$, defined as the least positive integer m such that for every 2-coloring of [1, m] there will be, for some i, a monochromatic k_i -term d_i -a.p.

§4.2. The proof of Theorem 4.11 is from [165]. Theorems 4.13, 4.17, 4.20–4.23, and 4.25–4.27 come from [64], which also gives a result that is slightly more general than Theorem 4.9. Several conjectures, and some evidence for them, are also found in [64]. The proofs of Theorems 4.29, 4.30, 4.32, 4.34, and Corollary 4.36 are found in [62], along with some other conditions that are necessary, or are sufficient, for a set to be large. For instance, if there exists $\epsilon > 0$ such that the ratio $\frac{d_{i+1}}{d_i}$ exceeds $1 + \epsilon$ (asymptotically), then $\{d_i : i \ge 1\}$ is not large. Some examples and questions are also presented in [62]. Some additional results and examples on large sets are given in [151]. Theorem 4.35 is a special case of a result due to Bergelson and Leibman

134

[33]. They employ methods of ergodic theory and measure theory. Extensions of this, and related results, may be found in [112], [113], [114], [115], [185], and [233]. See [28] and [184] for excellent extensive surveys of work in ergodic Ramsey theory. Walters [272] provides a combinatorial proof of the polynomial extensions of van der Waerden's theorem due to Bergelson and Leibman.

§4.3. Exercises 4.4–4.6 come from [164]. Exercise 4.14 is from [64]. Exercises 4.17–4.20 are from [62].

§4.4. Research Problems 4.1–4.4 come from [164]. Research Problems 4.5–4.10 are discussed in [64], and 4.11–4.13 are from [62].

Additional References: Brown [53] finds a 2-coloring of \mathbb{Z}^+ and a function h such that if $P = \{a, a+d, \ldots a+(k-1)d\}$ is monochromatic then $k \leq \min\{h(a), h(d)\}$. That is, in order to have the van der Waerden property hold, we cannot require d or a to be too small as a function of k (one corollary of this is Theorem 4.4). Other work along these lines is done in [21] and [152]. This general problem is mentioned in the excellent monograph of Erdős and Graham [92, p. 17]; several other intriguing problems concerning monochromatic arithmetic progressions are discussed in Chapter 2 of [92].

Chapter 5

Other Generalizations of w(k;r)

In Chapter 3 we looked at ways of generalizing the van der Waerden function w(k;r) by using supersets of the family of arithmetic progressions. Thus, we ended up with functions, such as $Q_n(k)$, which, under special circumstances (in this case when n = 0), coincide with the classical van der Waerden function itself. In this chapter we will consider the Ramsey-type functions for some other generalizations of arithmetic progressions, constructed by introducing other parameters, so that the number w(k;r) is a special case of a more general function.

5.1. Sequences of Type x, ax + d, bx + 2d

In this section we consider the following generalization of a 3-term arithmetic progression.

Definition 5.1. Let $a \le b$ be fixed positive integers. An (a,b)-triple is any set $\{x, ax + d, bx + 2d\}$, where x and d are positive integers.

It is clear from the definition that we have a generalization of 3-term arithmetic progressions: (x, y, z) is a 3-term arithmetic progression if and only if it is a (1,1)-triple. As further examples, $\{1,3,8\}$

5.1. Sequences of Type x, ax + d, bx + 2d

is a (1,4)-triple (with x = 1 and d = 2), {2, 7, 10} is a (3,4)-triple (with d = 1), and {1, 3, 4} is a (2,2)-triple (with d = 1).

Notation. Denote by $T_{a,b}$ the set of all (a, b)-triples. Since in this discussion we shall be dealing only with sets of size three, for ease of notation, we shall denote the Ramsey-type function $R(T_{a,b}, k; r)$ more simply as T(a, b; r).

We see that T(1,1;r) has the same meaning as w(3;r). Thus, from the known exact values of w(k;r), we have T(1,1;2) = 9, T(1,1;3) = 27, and T(1,1;4) = 76. However, the existence of T(a,b;r) for the general pair (a,b) would not seem to follow from van der Waerden's theorem. In fact, as we shall see, T(a,b;r) does not always exist. We seek to answer two natural questions:

1. For which values of a, b, and r does T(a, b; r) exist?

2. When it does exist, what can we say about the value of T(a, b; r)?

One case for which we are able to give a rather complete answer to these questions is that in which b = 2a - 1 and r = 2. The result is given by Proposition 5.2 below. We first investigate the relationship between the problem at hand and some families of arithmetic progressions that were discussed in Chapter 4.

Note that $\{1,3,5\}$ is not only a (1,1)-triple, but also a (2,3)-triple (taking d = 1). Thus, $(a_1, b_1) \neq (a_2, b_2)$ does not imply that T_{a_1,b_1} and T_{a_2,b_2} are disjoint. Is there more we can say about the relationship between such a pair of families?

To help answer this question, let's look more closely at the particular case of (1,1)-triples versus (2,3)-triples. In the example of $\{1,3,5\}$ mentioned above, it is not a coincidence that this triple belongs to both $T_{2,3}$ and AP. Actually, if (x, y, z) is any (2,3)-triple, then

(5.1) z - y = (3x + 2d) - (2x + d) = x + d = y - x,

so that (x, y, z) is an arithmetic progression. Hence $T_{2,3} \subseteq T_{1,1}$. On the other hand, the reverse inclusion does not hold since, by (5.1), for any (2,3)-triple (x, y, z), we must have $z - y = x + d \ge 2$.

More generally, we have the following.

Proposition 5.2. Let $a \ge 1$. Then $(x, y, z) \in T_{a,2a-1}$ if and only if (x, y, z) is an arithmetic progression with gap $y - x \ge (a - 1)x + 1$.

Proof. $X = \{x, y, z\} \in T_{a, 2a-1}$ if and only if

z - y = (2a - 1)x + 2d - (ax + d) = (a - 1)x + d = y - x

for some positive integer d, i.e., if and only if X is an arithmetic progression with gap at least (a-1)x + 1.

As a result of Proposition 5.2, we see that T(a, 2a - 1; 2) is a special case of the Ramsey-type function w(g(x), 3; 2) we studied in Chapter 4, which, you may recall, corresponds to arithmetic progressions $\{y, y+d, y+2d\}$ such that $d \ge g(y)$. Making use of bounds given for this function in Chapter 4, we are able to give upper and lower bounds for T(a, 2a - 1; 2) (for the particular case in which a = 1, we know that T(1, 1; 2) = w(3; 2) = 9; see Table 5.1 at the end of this section for other values of T(a, b; 2)).

Theorem 5.3. For all $a \geq 2$,

$$16a^2 - 12a + 6 \le T(a, 2a - 1; 2) \le \begin{cases} 16a^3 - 2a^2 + 4a - 3 & \text{for a even,} \\ 16a^3 + 14a^2 + 2a - 3 & \text{for a odd.} \end{cases}$$

Proof. By Proposition 5.2, we have T(a, 2a - 1; 2) = w'(g(x), 3; 2) where g(x) = (a - 1)x + 1. Applying Theorem 4.23, we obtain

$$T(a, 2a - 1; 2) \ge 8g(2a + 1) + 2(2a + 1) + 2 - c$$

where c is the greatest integer such that

$$(a-1)c + 1 + c \le 4g(2a+1) + 2a + 2$$
,

i.e., such that $ac + 1 \le 4[(a-1)(2a+1) + 1] + 2a + 2$. This implies $c = \lfloor \frac{8a^2 - 2a + 1}{a} \rfloor = 8a - 2$. Therefore,

 $T(a, 2a - 1; 2) \ge 8((a - 1)(2a + 1) + 1) + 4a + 4 - 8a + 2,$

which gives us the desired lower bound.

For the upper bound, we give the proof for the case of a even, and leave the case in which a is odd as Exercise 5.2. Assuming a is even and q(x) = (a - 1)x + 1, then, using the notation of Theorem

4.22, we have $\beta = 1 + 2a$ and $g(\beta) = 2a^2 - a$. Hence, by Proposition 5.2, Theorem 4.22 yields

$$T(a, 2a - 1; 2) \leq 4g \left(1 + 2a + 4\frac{2a^2 - a}{2} \right) + 7(2a^2 - a) + \frac{7}{2}(1 + 2a) - \frac{13}{2}$$

= 4g(4a^2 + 1) + 14a^2 - 3
= 4[(a - 1)(4a^2 + 1) + 1] + 14a^2 - 3,

which equals the desired upper bound.

By virtue of Proposition 5.2, we may also use the results of Chapter 4 to tell us about the function T(a, 2a - 1; r) when $r \ge 3$; the following theorem gives us a simple answer.

Theorem 5.4. Let $a \ge 2$ and $r \ge 3$. Then T(a, 2a - 1; r) does not exist.

Proof. By Proposition 5.2, we have T(a, 2a - 1; r) = w(g(x), 3; r), where g(x) = (a - 1)x + 1. From Theorem 4.26, we know that $w(\frac{r-\sqrt{2}}{2}x, 3; r)$ does not exist. Since $g(x) \ge \frac{r-\sqrt{2}}{2}x$ for all x, we see that w(g(x), 3; r) does not exist. \Box

As we see from the last two theorems, the existence of T(a, b; r) is dependent on the value of r. This prompts the next definition. For convenience we will stray a bit from the way that "regular" was presented in Chapter 1 (where it is associated with a collection of sets).

Definition 5.5. Let a and b be positive integers with $a \leq b$. If T(a,b;r) does not exist for some positive integer r, the degree of regularity of (a,b), denoted dor(a,b), is the largest value of r such that T(a,b;r) exists. If T(a,b;r) exists for all positive integers r, we say that $dor(a,b) = \infty$, and also say that (a,b) is regular.

Obviously, for every pair (a, b), $dor(a, b) \ge 1$. By van der Waerden's theorem (1,1) is regular, while Theorems 5.3 and 5.4 tell us that dor(a, 2a - 1) = 2 for each $a \ge 2$.

As a brief aside, it is worth noting that the notion of regularity may be applied to any type of sequence. Thus, for example, any of the types of supersets of AP discussed in Chapter 3 may be considered

5.1. Sequences of Type x, ax + d, bx + 2d

to be regular. Meanwhile, as we saw in Chapter 4, certain types of sets may yield a finite Ramsey-type value only when the number of colors is not too great (see, for example, Theorem 4.26), so that they would have a finite degree of regularity. The general question of which sets are regular and, if not, *how* regular, is rather intriguing. In later chapters we shall see some other interesting cases of families of sequences that are r-regular up to some specific finite value of r, but which fail to be regular.

Getting back to the degree of regularity for (a, b)-triples, we would like to know whether there are pairs besides those of the form (a, 2a - 1) whose degree of regularity is greater than one. The following theorem provides a complete answer to this question. It also gives an upper bound for T(a, b; 2) whenever dor(a, b) > 2.

Theorem 5.6. Let $a, b \in \mathbb{Z}^+$ with $a \leq b$. Then dor(a, b) = 1 if and only if b = 2a. Furthermore,

$$T(a,b;2) \leq \begin{cases} 4a(b^3 - 3b - 3) + 2b^3 + 4b^2 + 6b & \text{if } b > 2a, \\ 4a(b^3 + 2b^2 + 2b) - 4b^2 & \text{if } b < 2a. \end{cases}$$

Proof. First, assume b = 2a. To show dor(a, b) = 1, we need only exhibit a 2-coloring of \mathbb{Z}^+ that admits no monochromatic (a, 2a)-triple. Note that for any (a, 2a)-triple (x, y, z), we have z = 2y. Thus any 2-coloring γ such that for each even number 2n, $\gamma(2n) \neq \gamma(n)$, avoids monochromatic (a, 2a) triples (color the odd numbers first, arbitrarily, and then color the even numbers appropriately).

Now assume b > 2a. Let

$$m = 4a(b^3 + b^2 - 3b - 3) + 2b^3 + 4b^2 + 6b,$$

and let $\chi : [1, m] \to \{0, 1\}$ be an arbitrary 2-coloring. We will show that under χ there is a monochromatic (a, b)-triple. Assume, for a contradiction, that there is no such (a, b)-triple. In particular, $\{2, 2a + 2, 2b + 4\}$ is not monochromatic. Thus, there exist $x, x + 2 \in \{2, 4, 6, \ldots, 2b + 4\}$ such that $\chi(x) \neq \chi(x + 2)$. Without loss of generality, let $\chi(x) = 0$ and $\chi(x + 2) = 1$.

Let z be the least integer greater than a(x+2) such that z is a multiple of b-2a. Hence

 $z \le a(x+2) + b - 2a - 1 \le a(2b+4) + b - 2a - 1 = 2a(b+1) + b - 1.$ Thus

(5.2) $bz + 2(b - 2a) \le 2a(b^2 + b - 2) + b^2 + b \le m.$

Let S be the (a, b)-triple (z, az + (b - 2a), bz + 2(b - 2a)). By (5.2), $S \subseteq [1, m]$, so by our assumption some member of S must have color 1. Let $s \in S$ with $\chi(s) = 1$, and let

$$T = \left\{ s + i(b - 2a) : 0 \le i \le \frac{s(b - 1)}{b - 2a} + 2 \right\}$$

Note that since (b-2a) divides s and since $a \leq b$, we have that

$$s + \left[\frac{s(a-1)}{(b-2a)} + 1\right](b-2a) = as + (b-2a) \in T.$$

Note also that the largest member of T is bs + 2(b - 2a), and that by (5.2) we have

$$bs + 2(b - 2a) \leq b(bz + 2(b - 2a)) + 2(b - 2a)$$

$$\leq 2a(b^3 + b^2 - 2b - 2) + b^3 + b^2 + 2b$$

$$\leq m.$$

Thus, $T \subseteq [1, m]$, and since $\{s, as + (b - 2a), bs + 2(b - 2a)\}$ is an (a, b)-triple contained in T (hence cannot be monochromatic), some member of T must have color 0.

Let t be the least member of T with color 0. Thus, since t > sand $\chi(s) = 1$, we have $\chi(t - (b - 2a)) = 1$. Since $x \le 2b + 2$ and $t \le bs + 2(b - 2a)$, from (5.2) we have

$$b(x+2) + 2(t - ax - b) = 2t + x(b - 2a)$$

$$\leq 4a(b^3 + b^2 - 2b - 2)$$

$$+2(b^3 + b^2 + 2b) + (2b + 2)(b - 2a)$$

$$= m.$$

Let $(\alpha, \beta, \gamma) = (x+2, t-(b-2a), b(x+2)+2(t-ax-b))$. We claim that (α, β, γ) is an (a, b)-triple. To see this, first note that by the definition of $t, \beta = t - (b-2a) > a(x+2)$. Now let $d = \beta - a(x+2)$. Then

5.1. Sequences of Type x, ax + d, bx + 2d

 $\beta = a\alpha + d$ and $\gamma = b\alpha + 2d$, establishing the claim. Since we assumed that $\chi(\alpha) = 1$, and since $\chi(\beta) = 1$, we know that $\chi(\gamma) = 0$. However, this gives the monochromatic (a, b)-triple (x, t, γ) , a contradiction (we leave it to the reader to show that (x, t, γ) is, in fact, an (a, b)-triple). This finishes the case in which b > 2a.

Now assume b < 2a, and let $m = 4a(b^3 + 2b^2 + 2b) - 4b^2$. Assume, by way of contradiction, that there is a 2-coloring χ of [1, m] under which there is no monochromatic (a, b)-triple. As in the previous case, there exist $y - 2, y \in \{2, 4, \dots, 2b + 4\}$ that are not of the same color; say $\chi(y-2) = 1$ and $\chi(y) = 0$. Let z be the least integer greater than ay - (2a - b) that is a multiple of 2a - b, and let S be the (a, b)-triple $\{z, az + (2a - b), bz + 2(2a - b)\}$. As in the previous case, there is an $s \in S$ having color 1, and if

$$T = \{s, s + (2a - b), s + 2(2a - b), \dots, bs + 2(2a - b)\},\$$

then $as+(2a-b) \in T$ and $T \subseteq [1, m]$. It follows that T contains a least member, t, with color 0 (why?). Then $\chi(t - (2a - b)) = 1$ and, since $(\alpha, \beta, \gamma) = (y-2, t - (a-2b), b(y-2) + 2(t - ay + b))$ is an (a, b)-triple that is contained in [1, m] (the verification of this is left to the reader as Exercise 5.3), we must have $\chi(\gamma) = \chi(by + 2(t - ay)) = 0$. This gives the monochromatic (a, b)-triple (y, t, γ) , a contradiction. \Box

Theorem 5.6 provides an upper bound for T(a, b; 2). The next theorem gives a lower bound.

Theorem 5.7. Let $a, b \in \mathbb{Z}^+$ with $a \leq b$. Then

$$T(a,b;2) \ge \begin{cases} 2b^2 + 5b - 2a + 4 & \text{if } b > 2a, \\ 3b^2 + 5b - 4a + 4 & \text{if } b < 2a. \end{cases}$$

Proof. For each of the two cases we shall exhibit a 2-coloring (of the appropriate interval) that avoids monochromatic (a, b)-triples.

For the case of b > 2a, let $\chi : [1, 2b^2 + 5b - 2a + 3] \rightarrow \{0, 1\}$ be the coloring defined by $\chi([1, b + 1]) = 0$, $\chi([b + 2, b^2 + 2b + 1]) = 1$, and $\chi([b^2 + 2b + 2, 2b^2 + 5b - 2a + 3]) = 0$. There cannot be a monochromatic (a, b)-triple of color 1, because the largest term of any (a, b)-triple whose least term lies in $[b + 2, b^2 + 2b + 1]$ would lie outside of this interval.

140

Now assume that (x, y, z) is an (a, b)-triple of color 0. Obviously, $z \notin [1, b+1]$. Also, if $x, y \in [1, b+1]$, then $d \leq b+1-ax$ (where ax + d = y), so that $z \leq bx + 2(b+1-ax) \leq b^2 + 2b$. Hence, we must have $y \geq b^2 + 2b + 2$. It follows that if $x \leq b+1$, then $d \geq b^2 + 2b + 2 - ax$, so that

$$bx + 2d \ge 2b^2 + 4b + 4 - 2ax + bx$$

= $2b^2 + 5b + 4 - 2a + (x - 1)(b - 2a)$
> $2b^2 + 5b - 2a + 3$,

which is not possible. Thus, we assume that $x \ge b^2 + 2b + 2$. Then, since $d \ge 1$ and $b \ge 3$,

$$z \ge b^3 + 2b^2 + 2b + 2 \ge 2b^2 + 5b - 2a + 4,$$

which is also impossible. This completes the case in which b > 2a.

To establish the lower bound for b < 2a, consider the 2-coloring of $[1, 3b^2 + 5b - 4a + 3] = [1, m]$ defined by coloring $[b + 2, b^2 + 2b + 1]$ with color 1, and its complement in [1, m] with color 0. We leave the proof that this coloring avoids monochromatic (a, b)-triples as Exercise 5.4.

Now that we have established upper and lower bounds for the function T(a, b; 2), we look at a few examples.

Example 5.8. Consider T(1, b; 2) where b is some fixed positive integer. Thus we are concerned with the family of triples that have the form (x, x + d, bx + 2d) for some $d \in \mathbb{Z}^+$, a natural generalization of AP. We know that T(1, 1; 2) = w(3; 2) = 9, and (by Theorem 5.6) T(1, 2; 2) does not exist. For $b \ge 3$, Theorem 5.6 gives $4b^3(1 + o(1))$ as an asymptotic upper bound for T(1, b; 2), while Theorem 5.7 gives $2b^2(1+o(1))$ as an asymptotic lower bound (Exercise 5.5 gives a lower bound for T(1, b; 2) that is slightly better than that of Theorem 5.7, but which is asymptotically the same).

Example 5.9. Let b = a. Theorems 5.6 and 5.7 give the asymptotic bounds $3a^2(1 + o(1)) \leq T(a, a; 2) \leq 4a^4(1 + o(1))$. It is known, in fact, that $T(a, a; 2) = O(a^2)$, so that in this case the actual value of T(a, a; 2) is very close to the bound of Theorem 5.7. Moreover, it has

5.1. Sequences of Type x, ax + d, bx + 2d

been shown that $T(a, a; 2) \leq 3a^2 + a$ for all even $a \geq 4$, and that $T(a, a; 2) \leq 8a^2 + a$ for all odd a.

Example 5.10. Theorems 5.6 and 5.7 yield the asymptotic bounds $12a^2(1 + o(1)) \leq T(a, 2a - 1; 2) \leq 32a^4(1 + o(1))$. However, these bounds are not as good as the bounds provided by Theorem 5.3: $16a^2(1 + o(1)) \leq T(a, 2a - 1; 2) \leq 16a^3(1 + o(1))$.

We now return to the problem of determining the degree of regularity of (a, b)-triples. So far we know, by Theorem 5.6 that

$$dor(a,b) \ge 2$$
 unless $b = 2a$;

by Theorems 5.3 and 5.4 that

$$dor(a, 2a - 1) = 2$$
 for $a > 1$;

and by van der Waerden's theorem that $dor(1,1) = \infty$.

It has been conjectured that (1,1)-triples (the arithmetic progressions) are the only (a, b)-triples that do not have a finite degree of regularity. In the next theorem, we show that, whenever b is large enough in comparison to a, the degree of regularity is indeed finite.

Theorem 5.11. Let $1 \le a < b$, with $b \ge (2^{3/2} - 1)a - (2^{3/2} - 2)$. Then $dor(a, b) \le \lceil 2 \log_2 c \rceil$, where $c = \lceil \frac{b}{a} \rceil$.

Proof. Let a and b be as stated and let $r = \lceil 2 \log_2 c \rceil + 1$. To prove the theorem, we will provide an r-coloring of \mathbb{Z}^+ that yields no monochromatic (a, b)-triple. For convenience of notation, let $\beta = \sqrt{2}$.

We define $\chi : \mathbb{Z}^+ \to \{0, 1, \dots, r-1\}$ as follows. For each $i \ge 1$, whenever $\beta^i \le x < \beta^{i+1}$, let $\chi(x) = \overline{i}$ where $i \equiv \overline{i} \pmod{r}$ and $0 \le \overline{i} \le r-1$. For example, (assuming $r \ge 5$), $\chi(1) = 1$, $\chi(2) = 2$, $\chi(3) = 3$, $\chi([4,5]) = 4$, and $\chi([\lceil \beta^r \rceil, \lfloor \beta^{r+1} \rfloor]) = 0$.

Assume, for a contradiction, that there exists an (a, b)-triple (x, y, z) = (x, ax + d, bx + 2d) that is monochromatic under χ . Let $j \geq 1$ be the integer such that $\beta^j \leq y < \beta^{j+1}$. Since $c \geq 2$, we have $z \leq cy$, so that from the meaning of r it follows that $z < \beta^{r-1}\beta^{j+1} = \beta^{j+r}$. Hence, by the way χ is defined and the fact that $\chi(y) = \chi(z)$, we must have

$$(5.3) \qquad \qquad \beta^j \le y < z < \beta^{j+1}.$$

We consider two cases.

Case 1. $b \leq 2a$. By the hypothesis of the theorem,

$$b-a \ge 2a(\beta-1) - 2(\beta-1)$$

Therefore,

(5.4)

$$\frac{(a-1)x}{(b-a)x} \leq \frac{1}{2(\beta-1)}$$

Since $b \leq 2a - 1$, we have that $(b - a)x \leq (a - 1)x$. Hence, by (5.4),

$$\frac{(a-1)x+d}{(b-a)x+d} \leq \frac{1}{2(\beta-1)}.$$

Thus, using (5.3),

$$y - x \le \frac{z - y}{2(\beta - 1)} < \frac{\beta^{j+1} - \beta^j}{2(\beta - 1)} = \frac{\beta^j}{2} = \beta^{j-2}.$$

Because $y \geq \beta^j$, this implies $x \geq \beta^j - \beta^{j-2} = \beta^{j-2}$. Since, in this case, r = 3, and since $\chi(x) = \chi(y)$, by the way χ is defined we must have $\beta^j \leq x < \beta^{j+1}$. Therefore, $x, y, z \in [\beta^j, \beta^{j+1})$. Hence,

$$z - x = (b - 1)x + 2d < \beta^{j}(\beta - 1) \le x(\beta - 1),$$

contradicting the fact that $b-1 > \beta - 1$.

Case 2. b > 2a. In this case,

y - x = (a - 1)x + d < (b - a)x + d = z - y.

Hence, $\beta^{j}(\beta-1) \leq \beta^{j}\left(1-\frac{1}{\beta^{r-1}}\right)$. Since $y > \beta^{j}$, this implies

$$x \ge \beta^j - \beta^j \left(1 - \frac{1}{\beta^{r-1}}\right) = \beta^{j-r+1}.$$

Since $\chi(x) = \chi(y)$, by the definition of χ we must have

$$\beta^j \le x < y < \beta^{j+1}.$$

Thus, all three numbers x, y, z belong to the interval $[\beta^j, \beta^{j+1})$. As in Case 1, this yields a contradiction.

We may extend the idea of (a, b)-triples to k-tuples. That is, for fixed positive integers $a_1 \leq a_2 \leq \cdots \leq a_{k-1}$, we may ask about the Ramsey-type functions corresponding to k-tuples of the form $\{x, a_1x + d, a_2x + 2d, \ldots, a_{k-1}x + (k-1)d\}$. (Analogously to the notation T(a, b), these more general Ramsey-type functions are denoted by $T(a_1, a_2, \ldots, a_{k-1})$; see, for example, Research Problem 5.6.)

For this discussion, we shall limit ourselves to the special case in which $a_1 = a_2 = \cdots = a_{k-1}$. Therefore, for convenience, we use the notation $dor_k(a)$ to denote the largest integer r (or possibly ∞) such that for every r-coloring of the positive integers there is a monochromatic k-term progression of the form

(5.5)
$$\{x, ax + d, ax + 2d, \dots, ax + (k-1)d\}.$$

Noting that for a = 1 we are simply dealing with arithmetic progressions, by van der Waerden's theorem $dor_k(1) = \infty$ for each k. It is also known that $2 \leq dor(2,2) \leq 5$ and $2 \leq dor(3,3) \leq 5$ (see Table 5.2 at the end of this section), i.e, that $2 \leq dor_3(a) \leq 5$ if a = 2 or a = 3. In contrast to van der Waerden's result, the following theorem shows that for all $a \neq 1$, and large enough k, $dor_k(a) < \infty$ (in fact, the degree of regularity is not greater than three).

Theorem 5.12. For all $a \ge 2$ and all $k \ge \frac{a^2}{a+1} + 2$, $dor_k(a) \le 3$

Proof. To prove the theorem, it suffices to show that if a and k satisfy the given hypotheses, then there exists a 4-coloring of \mathbb{Z}^+ that avoids monochromatic k-term sequences having the form of (5.5). Clearly, we may assume $k = \lfloor \frac{a^2}{a+1} \rfloor + 2$.

Define χ to be the 4-coloring of the positive integers defined by coloring each of the intervals [1, a - 1], $[a, a^2 - 1]$, $[a^2, a^3 - 1]$, ... as follows: $\chi([a^j, a^{j+1} - 1]) = \overline{j}$, where $j \equiv \overline{j} \pmod{4}$ and $0 \leq \overline{j} \leq 3$. We will complete the proof by showing that if $\chi(x) = \chi(ax + d)$, then $\chi(ax + (k - 1)d) \neq \chi(x)$.

Assume that x and ax + d have the same color under χ , and let i be the integer such that $x \in [a^i, a^{i+1})$. Obviously, $ax + d \notin [a^i, a^{i+1})$. Hence, by the way χ is defined, there is some $m \in \mathbb{Z}^+$ such that $ax + d \in [a^{i+4m}, a^{i+4m+1})$. From this it follows that

(5.6)
$$a^i(a^{4m}-a^2) \le d \le a^{i+1}(a^{4m}-1).$$

Note that, by the way m and χ are defined, if we can show that (5.7) $a^{i+4m+1} \leq ax + (k-1)d < a^{i+4(m+1)}$,

then ax + (k - 1)d must be colored differently from x and ax + d, thereby completing the proof. Hence, we proceed to prove (5.7).

To prove $ax + (k-1)d < a^{i+4(m+1)}$, first note that $k < a^3 + 1$ for all $a \ge 2$. Hence, $1 + (k-2)(1-a^{-4m}) < a^3$, and therefore

(5.8)
$$a^{i+4m+1} + (k-2)a^{i+1}(a^{4m}-1) < a^{i+4(m+1)}$$

By (5.6), we have

$$ax + (k-1)d = ax + d + (k-2)d$$

 $\leq a^{i+4m+1} + (k-2)a^{i+1}(a^{4m}-1).$

This, together with (5.8), implies $ax + (k-1)d < a^{i+4(m+1)}$, as desired.

To complete the proof, we show that $a^{i+4m+1} \leq ax + (k-1)d$. Since $k \geq \frac{a^2}{a+1} + 2$, we have $(k-2)(a^2-1) \geq a^3 - a^2$, and therefore $(k-2)(a^2 - a^{-4(m-1)}) \geq a^3 - a^2$. From this we know that

(5.9)
$$a^{i+4m} + (k-2)a^i(a^{4m} - a^2) \ge a^{i+4m+1}.$$

Also, from (5.6), we have

(5.10)
$$ax + (k-1)d \ge a^{i+4m} + (k-2)a^i(a^{4m} - a^2)$$

By (5.9) and (5.10) we have $a^{i+4m+1} \leq ax + (k-1)d$, and the proof is complete.

The following two tables give the known values and lower bounds for T(a, b; 2), and the known degrees of regularity for some small a and b, respectively.

$a \setminus b$	1	2	3	4	5	6	7
1	9	∞	39	58	81	≥ 108	≥ 139
2		16	46	∞	139	≥ 106	≥ 133
3			39	60	114	∞	≥ 135
4				40	87	≥ 124	≥ 214
5					70	100	≥ 150
6						78	≥ 105
7							95

Table 5.1: Values and lower bounds for T(a, b; 2)

ĺ	(a,b)	dor(a, b)	(a,b)	dor(a, b)
	(1,1)	∞	(2,5)	2 - 3
	(1,2)	1	(2,6)	2 - 3
	(1,3)	2 - 3	(2,7)	2 - 4
	$(1,\!4)$	2 - 4	(2,8)	2 - 4
	$(1,\!5)$	2 - 5	(2,9)	2 - 5
	$(1,\!6)$	2 - 6	(3,3)	2 - 5
	(1,7)	2 - 6	(3,4)	2 - 5
	(1,8)	2 - 6	(3,5)	2
	(1, 9)	2 - 7	(3,6)	1
	(2,2)	2 - 5	(3,7)	2-4
	(2,3)	2	(3,8)	2-3
	$(2,\!4)$	1	(3, 9)	2-3

Table 5.2: Degree of regularity of (a, b)-triples

5.2. Homothetic Copies of Sequences

Definition 5.13. Let $s_1, s_2, \ldots, s_{k-1}$ be positive integers. A homothetic copy of the k-tuple $(1, 1 + s_1, 1 + s_1 + s_2, \ldots, 1 + \sum_{i=1}^{k-1} s_i)$ is a k-tuple $(a, a + bs_1, a + b(s_1 + s_2), \ldots, a + b \sum_{i=1}^{k-1} s_i)$, where a and b are any positive integers.

How are homothetic copies related to arithmetic progressions? Well, consider the collection of all homothetic copies of (1, 2, ..., k), i.e., where $s_i = 1$ for all *i* in Definition 5.13. Then this is the collection of all sequences of the form $\{a, a+b, a+2b, ..., a+(k-1)b\}$; in other words, the family of all *k*-term arithmetic progressions.

For fixed $s_1, s_2, \ldots, s_{k-1}$, we shall denote by $H(s_1, s_2, \ldots, s_{k-1})$ the 2-color Ramsey-type function associated with the family of all homothetic copies of $(1, 1+s_1, \ldots, 1+\sum_{i=1}^{k-1} s_i)$. Hence, the classical van der Waerden function w(k) has the same meaning as $H(1, 1, \ldots, 1)$. It follows easily from van der Waerden's theorem that for all k and all (k-1)-tuples $(s_1, s_2, \ldots, s_{k-1})$, the number $H(s_1, s_2, \ldots, s_{k-1})$ exists (this is left to the reader as Exercise 5.6).

We know that H(1,1) = w(3) = 9. We now examine the function H(s,t) for general pairs (s,t). That is, for a fixed pair (s,t), we want the least positive integer h = H(s,t) such that for every 2-coloring of

[1, h] there is a monochromatic triple of the form (a, a+bs, a+bs+bt) for some positive integers a and b. As we will see, this is a Ramsey-type function for which we are able to come rather close to providing a precise formula.

We begin with an upper bound for H(s,t). We shall make use of the following lemmas. We leave the proof of Lemma 5.14 as Exercise 5.7.

Lemma 5.14. For all $s, t \ge 1$, H(s, t) = H(t, s).

For convenience, from now on we will refer to a homothetic copy of $\{1, 1 + s, 1 + s + t\}$ as an (s, t)-progression.

Lemma 5.15. Let s, t, c be positive integers. Then

$$H(cs, ct) = c(H(s, t) - 1) + 1$$

Proof. Let m = H(s, t). By the definition of H(s, t), every 2-coloring of [0, m - 1] yields a monochromatic (s, t)-progression. Hence, every 2-coloring of $\{0, c, 2c, \ldots, (m - 1)c\}$ yields a monochromatic (cs, ct)-progression, giving c(m - 1) + 1 as an upper bound for H(cs, ct).

To prove the reverse inequality, first note that there exists a 2coloring χ of [1, m-1] that yields no monochromatic (s, t)-progression. Now define χ' on [1, c(m-1)] by

$$\chi'([c(i-1)+1,ci]) = \chi(i),$$

for $1 \le i \le m-1$. To complete the proof, we will show that χ' avoids monochromatic (cs, ct)-progressions.

Assume, for a contradiction, that $\{x_1 < x_2 < x_3\} \subseteq [1, c(m-1)]$ is a (cs, ct)-progression that is monochromatic under χ' . Then there exists u > 0 such that $x_3 - x_2 = uct$ and $x_2 - x_1 = ucs$. For j = 1, 2, 3, let $y_j = \lceil \frac{x_j}{c} \rceil$. Then

$$y_3 - y_2 = \left\lceil \frac{x_3}{c} \right\rceil - \left\lceil \frac{x_2}{c} \right\rceil = ut.$$

Similarly, $y_2 - y_1 = us$.

Hence $\{y_1 < y_2 < y_3\}$ is an (s,t)-progression. Furthermore, $\chi(y_j) = \chi\left(\left\lceil \frac{x_j}{c} \right\rceil\right) = \chi'(x_j)$ for each j. This contradicts our assumption that there is no monochromatic (s,t)-progression under χ . \Box

5.2. Homothetic Copies of Sequences

Theorem 5.16. For all $s, t \ge 1$, $H(s, t) \le 4(s + t) + 1$.

Proof. By Lemma 5.14, we may assume that $s \leq t$. We may also assume that gcd(s,t) = 1. To justify this last statement, assume that the theorem is true for all pairs (s_1, t_1) such that $gcd(s_1, t_1) = 1$. Let d = gcd(s, t). Then by Lemma 5.15, we have

$$H(s,t) = d\left(H\left(\frac{s}{d}, \frac{t}{d}\right) - 1\right) + 1$$

$$\leq d\left(4\left(\frac{s}{d} + \frac{t}{d}\right) + 1 - 1\right) + 1$$

$$= 4(s+t) + 1.$$

We noted before that H(1, 1) = 9. We may also calculate directly that H(1, 2) = 13 and H(1, 3) = 17, so that the theorem holds for the pairs (1,1), (1,2), and (1,3). Now assume gcd(s,t) = 1, $s \leq t$, and $(s,t) \notin \{(1,1), (1,2), (1,3)\}$. Consider the following collection of subsets of [1,15]:

$$\begin{split} \mathcal{C} &= & \left\{ \{1,2,6\},\{2,3,7\},\{3,4,8\},\{4,5,9\},\{1,3,10\},\{2,4,11\}, \\ & \{3,5,12\},\{1,4,13\},\{2,5,14\},\{1,5,15\},\{6,7,10\},\{7,8,11\}, \\ & \{8,9,12\},\{6,8,13\},\{7,9,14\},\{6,9,15\},\{10,11,13\}, \\ & \{11,12,14\},\{10,12,15\},\{13,14,15\} \right\}. \end{split}$$

We leave it to the reader (Exercise 5.10a) to check that every 2coloring of [1,15] yields a monochromatic triple from this list of twenty triples in C.

We now make a one-to-one correspondence between the members of C and twenty different (s, t)-triples that are in [1, 4(s + t) + 1] by means of the following associations (that the twenty resulting triples of elements of [1, 4(s + t) + 1] are, in fact, distinct (s, t)-triples is left to the reader as Exercise 5.10b):

$1 \leftrightarrow 1$	$2 \leftrightarrow s+1$	$3\leftrightarrow 2s+1$
$4 \leftrightarrow 3s + 1$	$5 \leftrightarrow 4s + 1$	$6 \leftrightarrow s+t+1$
$7 \leftrightarrow 2s + t + 1$	$8 \leftrightarrow 3s + t + 1$	$9 \leftrightarrow 4s + t + 1$
$10 \leftrightarrow 2s + 2t + 1$	$11 \leftrightarrow 3s + 2t + 1$	$12 \leftrightarrow 4s + 4t + 1$
$13 \stackrel{.}{\leftrightarrow} 3s + 3t + 1$	$14 \leftrightarrow 4s + 3t + 1$	$15 \leftrightarrow 4s + 4t + 1.$

Since every 2-coloring of [1,15] admits a monochromatic member of C, it is clear that every 2-coloring of [1,4(s+t)+1] must admit a corresponding monochromatic triple from among the twenty (s,t)triples that result from the above association scheme. This yields the desired upper bound on H(s,t).

As it turns out, the upper bound given in Theorem 5.16 is known to be the actual value of H(s,t) for a rather broad class of pairs (s,t), although an exact formula that holds for all cases is still unknown. The next theorem provides part of the answer.

Theorem 5.17. Let $s \le t$ be positive integers. If neither $\frac{s}{\gcd(s,t)}$ nor $\frac{t}{\gcd(s,t)}$ is divisible by 4, then H(s,t) = 4(s+t) + 1.

Proof. By Theorem 5.16, we see that it is sufficient to show that $H(s,t) \ge 4(s+t) + 1$. We shall split the problem into two cases, in each one providing a specific 2-coloring of [1, 4(s+t)] that avoids monochromatic (s, t)-progressions.

Let $d = \gcd(s, t)$. Of course, not both $\frac{s}{d}$ and $\frac{t}{d}$ are even.

Case 1. $\frac{s}{d}$ and $\frac{t}{d}$ are both odd. As demonstrated in the proof of Theorem 5.16, we may assume d = 1. Color [1, 4(s+t)] according to the string

$$\underbrace{1010\ldots10}_{2(s+t)}\underbrace{0101\ldots01}_{2(s+t)}.$$

Assume, for a contradiction, that $\{x < y < z\}$ is a monochromatic (s,t)-progression. Then there exists a positive integer b such that y = x + bs and z = y + bt. Let B_1 and B_2 represent the intervals [1, 2(s + t)] and [2(s + t) + 1, 4(s + t)], respectively.

In case b is odd, then (since we are under the assumption that s is odd), x and y have different parities (one is even and the other odd). Similarly, y and z have different parities. Since x and y have the same color, yet opposite parity, it is evident from the way the coloring is defined that $x \in B_1$ and $y \in B_2$. Hence $z \in B_2$, from which it follows that y and z cannot have the same color, a contradiction.

5.2. Homothetic Copies of Sequences

If b is even, then x, y, and z are all of the same parity. Hence, either they all belong to B_1 or they all belong to B_2 . Thus,

$$b(s+t) = z - x \le 2(s+t),$$

which means that b = 1, a contradiction.

Case 2. One of $\frac{s}{d}$ and $\frac{t}{d}$ is even. By the hypotheses, and without loss of generality, we assume that $\frac{s}{d} \equiv 2 \pmod{4}$. As in Case 1, we shall assume that d = 1.

Let λ be the 2-coloring of [1, 4(s + t)] represented by the string 1100 1100...1100; that is, it consists of s + t consecutive occurrences of the string 1100.

We shall show, by contradiction, that λ avoids monochromatic (s,t)-progressions. Thus, assume that $\{x < y < z\}$ is an (s,t)-progression that is monochromatic under λ . Then there exists $b \ge 1$ such that y - x = bs and z - y = bt. Now, since

$$z - x = b(s + t) \le 4(s + t) - 1,$$

we have that $b \leq 3$.

First assume b = 2. Then z - x = b(s + t) is even. Note that by the definition of λ , we know that the only way that two integers i and j can have an even difference and be of the same color is for j - i to be divisible by 4. Hence 4 divides z - x. However, by our assumptions of Case 2, this is impossible since exactly one of s and t is even. Now assume that b = 1 or b = 3. Because $s \equiv 2 \pmod{4}$, y - x is even but is not divisible by 4; this is also impossible, since bsis not a multiple of 4.

By Lemmas 5.14 and 5.15, and Theorem 5.17, we see that we would know all values of H(s,t) if we knew the value of H(4j,t) when t is odd and gcd(j,t) = 1. For many such pairs it is known that H(s,t) = 4(s+t) + 1. Meanwhile, it has been conjectured that H(4j,1) = 4(4j+1) for all $m \ge 1$, and it has been proven that $4(4j+1) \le H(4j,1) \le 4(4j+1) + 1$ for all positive integers j.

5.3. Sequences of Type x, x + d, x + 2d + b

One simple way to form a generalization of an arithmetic progression is expressed in the following definition.

Definition 5.18. Let $b \ge 0$. A k-term augmented progression with tail b is a sequence of the form

$$\{x, x+d, x+2d, \dots, x+(k-2)d, x+(k-1)d+b\}$$

for any $x, d \in \mathbb{Z}^+$.

Denote the family of all augmented progressions with tail b by AUG_b . We see that this provides us with a generalization of w(k), since $AUG_0 = AP$ and therefore $w(k) = R(AUG_0, k)$. We shall limit the discussion to the case of k = 3, so that we are interested in sequences of the form $\{x, x+d, x+2d+b\}$ and the function $R(AUG_b, 3)$. We are unaware of any significant work that has been done for $k \ge 4$ (this sounds like a research problem that is wide-open for exploration).

An obvious question we should ask is: does $R(AUG_b, 3)$ exist (in other words, is it finite?) for every b? We answer this, and more, in the following theorem.

Theorem 5.19. For b odd, $R(AUG_b, 3)$ does not exist. For b even, $R(AUG_b, 3) \leq \lfloor \frac{9}{4}b \rfloor + 9$.

Proof. First assume b is odd. Consider the coloring of the positive integers represented by the string 101010.... Then no triple of the form $\{x, x + d, x + 2d + b\}$ can be monochromatic because the first and third elements of this triple differ by an odd number.

Now let b be even. We consider two cases.

Case 1. $b \equiv 0 \pmod{4}$. Since $R(AUG_0, 3) = w(3) = 9$, the theorem is true when b = 0. Hence, it is sufficient to show that for $b \ge 4$, and $m = \left\lceil \frac{9}{4}b \right\rceil + 9$, every 2-coloring of [1, m] admits a monochromatic 3-term augmented progression with tail b. Let $\chi : [1, m] \rightarrow \{0, 1\}$ be any 2-coloring and assume, by way of contradiction, that χ admits no such monochromatic set.

Let $A = \{i \in [1, m] : \chi(i) = 1\}$ and $B = \{i \in [1, m] : \chi(i) = 0\}$. By the pigeonhole principle, some 3-element subset, S, of [1, 5] is

5.3. Sequences of Type x, x + d, x + 2d + b

monochromatic. Since there are $\binom{5}{3} = 10$ possibilities for S, we shall consider ten subcases. We will present four of the subcases here, and leave the rest to the reader as Exercise 5.12. Without loss of generality, we shall assume $S \subseteq A$.

Subcase i. $S = \{1, 2, 3\}$. Since $1, 2 \in A$, and since we are assuming that there is no 3-term monochromatic augmented progression with tail b, we must have $3+b \in B$. Likewise, $2, 3 \in A$ implies $4+b \in B$; and $1, 3 \in A$ implies $5+b \in B$. Using the same line of reasoning, we have the following sequence of implications (note that each of the integers occurring in the implications below belongs to [1, m]; note also that since $b \geq 4$, in each implication the 3-term augmented progression alluded to is, in fact, a set of 3 *distinct* integers):

$$\begin{array}{l} 3+b,4+b\in B \text{ implies } 5+2b\in A,\\ 3+b,5+b\in B \text{ implies } 7+2b\in A,\\ 1,5+2b\in A \text{ implies } 3+\frac{b}{2}\in B,\\ 3,5+2b\in A \text{ implies } 4+\frac{b}{2}\in B,\\ 3,7+2b\in A \text{ implies } 5+\frac{b}{2}\in B,\\ 3+\frac{b}{2},4+\frac{b}{2},5+\frac{b}{2}\in B \text{ implies } 5+\frac{3b}{2},7+\frac{3b}{2}\in A,\\ 1,5+\frac{3b}{2}\in A \text{ implies } 3+\frac{b}{4}\in B,\\ 3,5+\frac{3b}{2}\in A \text{ implies } 5+\frac{b}{4}\in B,\\ 3,5+\frac{3b}{2}\in A \text{ implies } 5+\frac{b}{4}\in B,\\ 3+\frac{b}{4},4+\frac{b}{4},5+\frac{b}{4}\in B \text{ implies } 5+\frac{5b}{4},6+\frac{5b}{4}\in A,\\ 5+\frac{5b}{4},6+\frac{5b}{4}\in A \text{ implies } 7+\frac{9b}{4}\in B,\\ 3+\frac{b}{4},7+\frac{9b}{4}\in B \text{ implies } 5+\frac{3b}{4}\in A,\\ 5+\frac{b}{4},7+\frac{9b}{4}\in B \text{ implies } 5+\frac{3b}{4}\in A,\\ 5+\frac{b}{4},6+\frac{3b}{4}\in B \text{ implies } 6+\frac{3b}{4}\in A,\\ 5+\frac{b}{4},6+\frac{3b}{4}\in A \text{ implies } 7+\frac{7b}{4}\in B.\\ \end{array}$$

Thus $\{3 + \frac{b}{4}, 5 + \frac{b}{2}, 7 + \frac{7b}{4}\}$, an augmented progression with tail b, is monochromatic, a contradiction.

Subcase ii. $S = \{1, 2, 4\}$. By Subcase i, we may assume $3 \in B$. Using the same idea as in the proof of Subcase i, we have the following sequence of implications:

 $\begin{array}{l} 1,2,4\in A \text{ implies } 6+b,7+b\in B,\\ 6+b,7+b\in B \text{ implies } 8+2b\in A,\\ . 2,4,8+2b\in A \text{ implies } 5+\frac{b}{2},6+\frac{b}{2}\in B,\\ 3,5+\frac{b}{2}\in B \text{ implies } 7+2b\in A, \end{array}$

5.3. Sequences of Type x, x + d, x + 2d + b

5. Other Generalizations of w(k;r)

 $\begin{array}{l} 1,7+2b\in A \text{ implies } 4+\frac{b}{2}\in B,\\ 4+\frac{b}{2},5+\frac{b}{2},6+\frac{b}{2}\in B \text{ implies } 6+\frac{3b}{2},8+\frac{3b}{2}\in A,\\ 2,4,6+\frac{3b}{2},8+\frac{3b}{2}\in A \text{ implies } 4+\frac{b}{4},5+\frac{b}{4},6+\frac{b}{4}\in B,\\ 4+\frac{b}{4},5+\frac{b}{4},6+\frac{b}{4}\in B \text{ implies } 6+\frac{5b}{4},7+\frac{5b}{4}\in A,\\ 6+\frac{5b}{4},7+\frac{5b}{4}\in A \text{ implies } 8+\frac{9b}{4}\in B,\\ 4+\frac{b}{4},6+\frac{b}{4},8+\frac{9b}{4}\in B \text{ implies } 6+\frac{3b}{4},7+\frac{3b}{4}\in A,\\ 6+\frac{3b}{4},7+\frac{3b}{4}\in A \text{ implies } 8+\frac{7b}{4}\in B.\\ \end{array}$

Thus $\left\{4 + \frac{b}{4}, 6 + \frac{b}{2}, 8 + \frac{7b}{4}\right\}$, an augmented progression with tail *b*, is monochromatic, a contradiction.

Subcase iiii. $S = \{1, 2, 5\}$. By Subcases i and ii, we may assume that $3, 4 \in B$. By the method used in those subcases, it can be shown that there exists a monochromatic augmented progression with tail b whose largest element does not exceed $\max\{8 + \frac{9b}{4}, 10 + 2b\} \leq m$. We leave the details to the reader as Exercise 5.11.

Subcase iv. $S = \{2, 3, 4\}$. That the result holds for this case is a consequence of Subcase i, by a simple translation of length 1 (see Proposition 2.30), because the proof of Subcase i yields a monochromatic augmented progression with tail *b* that is contained in [1, m-1].

Case 2. $b \equiv 2 \pmod{4}$. We consider the same ten subcases as for Case 1. As the proofs are quite similar to those of Case 1, we present here only one subcase, and leave the the proofs of the other subcases to the reader.

Subcase i. $S = \{1, 3, 5\}$. Then $5 + b, 7 + b \in B$. Thus, $9 + 2b \in A$. We then have the following sequence of implications:

$$\begin{array}{l} 3,5,9+2b\in A \text{ implies } 6+\frac{b}{2},7+\frac{b}{2}\in B,\\ 6+\frac{b}{2},7+\frac{b}{2}\in B \text{ implies } 8+\frac{3b}{2}\in A,\\ 1,3,5,8+\frac{3b}{2}\in A \text{ implies } \frac{9}{2}+\frac{b}{4},\frac{11}{2}+\frac{b}{4},\frac{13}{2}+\frac{b}{4}\in B,\\ \frac{9}{2}+\frac{b}{4},\frac{11}{2}+\frac{b}{4},\frac{13}{2}+\frac{b}{4}\in B \text{ implies } \frac{13}{2}+\frac{5b}{4},\frac{15}{2}+\frac{5b}{4}\in A,\\ \frac{13}{2}+\frac{5b}{4},\frac{15}{2}+\frac{5b}{4}\in A \text{ implies } \frac{17}{2}+\frac{9b}{4}\in B,\\ \frac{9}{2}+\frac{b}{4},\frac{13}{2}+\frac{b}{4},\frac{17}{2}+\frac{9b}{4}\in B \text{ implies } \frac{13}{2}+\frac{3b}{4},\frac{15}{2}+\frac{3b}{4}\in A,\\ \frac{13}{2}+\frac{3b}{4},\frac{15}{2}+\frac{3b}{4}\in A \text{ implies } \frac{17}{2}+\frac{7b}{4}\in B.\\ \end{array}$$

Then the augmented progression $\left\{\frac{11}{2} + \frac{b}{4}, 7 + \frac{b}{2}, \frac{17}{2} + \frac{7b}{4}\right\}$ is contained in *B*; this is a contradiction since all of the 3-term augmented progressions occurring in the argument are contained in [1, m]. We next provide a lower bound for $R(AUG_b, 3)$ (for b even). Note that this lower bound agrees precisely with all known values of $R(AUG_b, 3)$ (see Table 5.3 at the end of this section).

Theorem 5.20. For b = 2 and for all even $b \ge 10$,

$$R(AUG_b, 3) \ge 2b + 10.$$

For $b \in \{0, 4, 6, 8\}$, $R(AUG_b, 3) \ge 2b + 9$.

Proof. The coloring 1100011110000 avoids 3-term mono-chromatic augmented progressions with tail 2, so that $R(AUG_2,3) \ge 14$. For $b \in \{0,4,6,8\}$, let s_b denote the alternating string 1010...101 of length b + 3. It is easy to check that the coloring defined by the string s_b00s_b , which has length 2b + 8, avoids monochromatic 3-term augmented progressions with tail b.

Now let $b \ge 10$ be even. Define the coloring χ of [1, 2b + 9] as follows. Let

$$B_{1} = \{1, 2\},$$

$$B_{2} = \{3, 4, 5\},$$

$$B_{3} = [6, b + 2],$$

$$B_{4} = \{b + 3\},$$

$$B_{5} = [b + 4, b + 7],$$

$$B_{6} = [b + 8, 2b + 9]$$

Let $\chi(B_1) = \chi(B_3) = \chi(B_5) = 1$ and $\chi(B_2) = \chi(B_4) = \chi(B_6) = 0$. We assume that $P = \{x < y < z\}$ is a monochromatic augmented progression with tail b, and seek a contradiction. Note that no single B_i can contain all elements of P, since $z - x \ge b + 2$.

First assume $\chi(P) = 1$. It is clear that $y \neq 2$, since otherwise z = b + 3, which is of a different color. Thus, if $x \in B_1$, then $y \ge 6$; but then $z \in B_6$, which is not possible. If $x \in B_3$, then again we have $z \in B_6$.

Now assume $\chi(P) = 0$. If $x, y \in B_2$, it then follows that $z \in B_5$, a contradiction. If $x \in B_2$ and $y \notin B_2$, then $z \ge 3b + 1 > 2b + 9$, which is impossible. Finally, if x = b + 3, then z is again outside of [1, 2b + 9].

From the discussion above we know that $R(AUG_b, 3; 2)$ exists for every even b (that is, when two colors are used). What happens when we increase the number of colors? As the next theorem shows, the situation is different when r = 3, and quite different when $r \ge 4$.

Theorem 5.21. If $b \not\equiv 0 \pmod{6}$, then $R(AUG_b, 3; 3)$ does not exist. Furthermore, for all b, $R(AUG_b, 3; 4)$ does not exist.

Proof. If b is odd, then this is covered by Theorem 5.19. Hence, we let b be even but not divisible by 6. Assume, for a contradiction, that $R(AUG_b, 3; 3)$ exists. Color the positive integers according to the string 012012012.... Then any monochromatic augmented progression $\{x, x + d, x + 2d + b\}$ must have d a multiple of 3; but then b must also be a multiple of 3. Since b is even, it must be a multiple of 6, a contradiction. This proves the first statement.

To see that the second statement holds, color the positive integers with the coloring $\chi = 1^{k}2^{k}3^{k}4^{k}1^{k}2^{k}3^{k}4^{k}\dots$ By the case for three colors, we assume that 6 divides b. Let (x, y, z) = (x, x+d, x+2d+b); we will show that (x, y, z) is not monochromatic. Note that if s < t, then $\chi(s) = \chi(t)$ if and only if, for some positive integer i, we have $4ki - (k-1) \le t - s \le 4ki + (k-1)$. However, z - y = d + b is in [4kj - (k-1) + 6k, 4kj + (k-1) + 6k] = [4k(j+1) + k + 1, 4k(j+1) + 3k - 1]for some positive integer j. Hence $\chi(y) \ne \chi(z)$.

We mention, without proof, that it is known that when b is a multiple of 6, $R(AUG_b, 3; 3) \leq \frac{55}{6}b + 1$, so that $R(AUG_b, 3; 3)$ exists if and only if $b \equiv 0 \pmod{6}$.

Recall that in Chapter 4 we discussed a generalization of w(k), denoted w'(c, k), which pertained to arithmetic progressions whose gaps were no less than c. In the same way, we may generalize the function $R(AUG_b, k)$ by defining, for each positive integer c, $R(c, AUG_b, k)$ to be the least positive integer m such that for every 2-coloring of [1, m] there is a monochromatic augmented triple $\{x, x+d, x+2d\}$ with the added restriction that $d \geq c$. Since $R(c, AUG_0, k)$ has the same meaning as w'(c, k), we know by Theorem 4.11 that $R(c, AUG_0, 3) = 8c+1$. By modifying the proofs of Theorems 5.19 and 5.20, it is not all that difficult (although it is a bit tedious) to obtain generalizations

5.3. Sequences of Type x, x + d, x + 2d + b

of those two theorems, where the parameter c is included. We do not present the proofs of these generalizations here, but do state the results in the next two theorems. The proof of Theorem 5.22 imitates the proof of Theorem 5.19, except instead of considering the possible colorings of [1, 5], one considers the possible colorings of $\{1, 1 + c, 1 + 2c, 1 + 3c, 1 + 4c\}$. Notice that Theorems 5.19 and 5.20

Theorem 5.22. Let $b \ge 0$ be even, $c \ge 1$, and a = 2b + 9c. Then

$$R(c, AUG_b, 3) \leq \begin{cases} \max\left\{\frac{9b}{4} + 8c + 1, a + 1\right\} & \text{if } 4|b, \\ \max\left\{\frac{9b}{4} + \frac{17c}{2} + 1, a + 1\right\} & \text{if } 4 \nmid b \text{ and } c \text{ is } odd, \\ \max\left\{\frac{9b}{4} + \frac{17c}{2} + \frac{19}{2}, a + 10\right\} & \text{if } 4 \nmid b \text{ and } c \text{ is } even. \end{cases}$$

Theorem 5.23. Let $b \ge 2$ be even, and let $c \ge 1$. Then

are, indeed, special cases of Theorems 5.22 and 5.23.

$$R(c, AUG_b, 3) \ge \begin{cases} 2b + 7c + 3 & \text{if } b \ge 6c + 3, \\ 2b + 7c + 2 & \text{if } c + 2 \le b \le 6c + 2, \\ b + 8c + 4 & \text{if } 2 \le b \le c + 1. \end{cases}$$

The following table gives the known values of $R(c, AUG_b, 3)$. Note that $R(AUG_b, 3) = R(1, AUG_b, 3)$.

$b \setminus c$	1	2	3	4	5	6	7	8	9
0	9	17	25	33	41	49	57	65	73
2	14	22	30	38	46	54	62	70	78
4	17	24	32	40	48	56	64	72	80
6	21	28	35	42	50	58	66	74	?
8	25	32	39	46	53	60	68	?	?
10	30	36	43	50	57	64	71	?	?
12	34	40	47	54	61	68	?	?	?
14	38	44	51	58	65	?	?	?	?
16	42	49	55	62	69	?	?	?	?
18	46	53	59	66	?	?	?	?	?
20	50	57	63	?	?	?	?	?	?
22	54	61	?	?	?	?	?	?	?
24	58	?	?	?	?	?	?	?	?
26	62	?	?	?	?	?	?	?	?

Table 5.3: Values of $R(c, AUG_b, 3)$

5.5. Research Problems

5. Other Generalizations of w(k;r)

5.4. Exercises

- **5.1** How many (1,3)-triples are contained in [1,100]? How many (a,b)-triples are contained in [1,100]? How many are contained in [1,n]?
- **5.2** Prove Theorem 5.3 for the case in which a is odd.
- **5.3** At the end of the proof of Theorem 5.6 it is claimed that (α, β, γ) is an (a, b)-triple that resides in [1, m]. Verify this.
- **5.4** Complete the proof of Theorem 5.7 by showing that the coloring of $[1, 3b^2 + 5b 4a + 3] = [1, m]$ defined by coloring $[b+2, b^2+2b+1]$ with color 1, and its complement in [1, m] with color 0, avoids monochromatic (a, b)-triples.
- **5.5** By Theorem 5.7, we know that $T(1, b; 2) \ge 2b^2 + 5b + 2$ for $b \ge 3$. Show that this lower bound can be tightened to $T(1, b; 2) \ge 2b^2 + 5b + 6$ (hint: color

 $S = [1, b+1] \cup \{b+3\} \cup [b^2 + 2b + 4, 2b^2 + 5b + 5]$

with the color 1, and its complement in $[1, 2b^2 + 5b + 5]$ with the color 0).

- **5.6** Use van der Waerden's theorem to prove that for all k and all (k-1)-tuples $(s_1, s_2, \ldots, s_{k-1})$ of positive integers, the number $H(s_1, s_2, \ldots, s_{k-1})$ exists (is finite).
- 5.7 Prove Lemma 5.14.
- **5.8** Assume that s_1, s_2, \ldots, s_k , and c are positive integers and that $H(s_1, s_2, \ldots, s_k) = m$. Prove that

 $H(cs_1, cs_2, \dots, cs_k) = c(m-1) + 1.$

- **5.9** Verify that H(1,2) = 13 and H(1,3) = 17.
- **5.10** a) Show that every 2-coloring of [1,15] yields a monochromatic member of the family C as defined in the proof of Theorem 5.16.

b) Show that the 20 triples of elements of [1, 4(s+t)+1] that correspond to the 20 members of C (via the correspondence between elements of [1,15] and elements of [1,4(s+t)+1] given in the proof of Theorem 5.16) are distinct (s,t)-triples.

5.11 Complete the proof of Subcase iii of Case 1 of Theorem 5.19.

- 5.12 Prove the subcases of the proof of Theorem 5.19 that were not done in the text.
- **5.13** For given nonnegative integers a and b, define $\hat{f}(a, b)$ to be the least positive integer such that for every 2-coloring of $[1, \hat{f}(a, b)]$ there is a monochromatic set $\{x, x+d+a, x+2d+b\}$ for some positive integers x and d. Prove that if $b \ge a$, then $\hat{f}(a, b) = \hat{f}(b a, b)$.
- **5.14** Using the notation of Exercise 5.13, prove that if $b \ge 2a$, then $\hat{f}(a, b) = R(a + 1, AUG_{b-2a}, 3).$
- **5.15** Use Exercises 5.13 and 5.14 to show that if $\frac{b}{a} < a \le b$, then $\hat{f}(a,b) = R(b-a+1, AUG_{2a-b}, 3).$

5.5. Research Problems

- **5.1** Improve the known bounds for T(a, a; 2). Reference: [175]
- **5.2** Improve the known bounds for T(a, 2a 1; 2). Reference: [175]
- **5.3** Improve the known bounds for T(1, b; 2). Reference: [175]
- **5.4** Prove or disprove: (1,1) is the only pair (a,b) such that $dor(a,b) = \infty$. Reference: [175]
- **5.5** Find the degree of regularity of some pair (a, b) for which the degree of regularity is still unknown (see Table 5.2). Reference: [175]
- **5.6** Extend the study of T(a, b; 2) to T(a, b, c; 2) by considering 4-tuples of the form (x, ax + d, bx + 2d, cx + 3d). Extend it to $T(a_1, a_2, \ldots, a_{k-1}; 2)$. Reference: [175]
- **5.7** Let a > 1 and r > 3. Define $\tau = \tau(a; r)$ to be the least positive integer such that $dor_{\tau}(a) \leq r$. Prove or disprove: there exists s > r such that $\tau(a; s) < \tau(a; r)$. Reference: [175]

- 5.6. References
- 5. Other Generalizations of w(k;r)
- **5.8** Investigate the function H(s, t, u), i.e., for homothetic copies of 4-term sequences. Reference: [66]
- **5.9** Investigate the function H(s, t; 3); that is, where three colors are used instead of two. Reference: [66]
- **5.10** Prove or disprove the conjecture that H(4j, 1) = 4(4j+1) for all $j \ge 1$. Reference: [**66**]
- **5.11** The only cases for which the exact value of H(s,t) have not been determined are those pairs (s,t) = (t+e,t) such that 0 < e < t < 2e and either t + e or t is a multiple of 4. Determine the exact value of H(s,t) for these cases. Reference: [66]
- **5.12** Investigate the function $R(AUG_b, k)$ for k > 3; in particular, consider sequences of the form $\{x, x + d, x + 2d, x + 3d + b\}$. References: [39], [165]
- **5.13** Let $\hat{f}(a, b), a \leq b$, be defined as in Exercise 5.13. Prove or disprove the conjecture that for any fixed even value of b, the maximum of $\hat{f}(a, b)$ occurs when $a = \frac{b}{2}$. Reference: [165]
- 5.14 Study the Ramsey properties for sequences of the form

$$\{x, x+d, x+2d+b_1, x+3d+b_2\}.$$

References: [39], [165]

5.15 Improve the upper bound for $R(AUG_b, 3; 3)$. Find a lower bound for this function. References: [**39**], [**165**]

5.6. References

§5.1. The results of this section may be found in [175]. That article also contains other results on T(a, b; 2), including a quadratic upper bound for T(a, a; 2). It also mentions a result on an extension to k-tuples that is analogous to a result on arithmetic progressions from [127], as well as a conjecture on $dor_k(a)$.

§5.2. The work on homothetic copies is from [66], which contains a more thorough discussion. In [170] and [172], Ramsey-type functions are considered for the collection of sequences that, for some $t \ge 1$, are homothetic copies of the k-tuple

 $\{1, 2, 2+t, 2+t+t^2, \dots, 2+t+t^2+\dots+t^{k-2}\}.$

§5.3. The function $R(AUG_b, 3; r)$ was first considered in [39], where it is shown that $R(AUG_b, 3)$ does not exist for b odd, and that, for beven, $R(AUG_b, 3) \leq \frac{13}{2}b + 1$. The authors of [39] also provide proofs of Theorems 5.20 and 5.21, and show that $R(AUG_b, 3; 3) \leq \frac{55}{6}b + 1$ for $b \equiv 0 \pmod{6}$. Proofs of Theorems 5.22 and 5.23 are found in [165], for which Theorems 5.19 and 5.20 are special cases. That paper also investigates the function \hat{f} (see Exercises 5.13–5.15), and its relationship to the function $R(AUG_b, k)$.

Additional References: An important result, known as the Hales-Jewett theorem [136], is a generalization of van der Waerden's theorem.

Chapter 6

Arithmetic Progressions $(\mod m)$

An arithmetic progression is a sequence in which the gaps between successive terms are all equal to some positive integer d. In this chapter we shall consider sequences that are analogous to arithmetic progressions but where, instead of all of the gaps being equal integers, they are all congruent modulo m, where m is some prescribed integer. Another way to think of this analogy is that the gaps of an arithmetic progression are equal elements of \mathbb{Z} , whereas the gaps of one of the integer sequences discussed here, although possibly non-identical integers, are equal when considered as elements of the additive group \mathbb{Z}_m . We will discover some rather interesting Ramsey properties in this setting.

We begin with some basic definitions and notation.

Definition 6.1. Let $m \ge 2$ and $0 \le a < m$. A k-term $a \pmod{m}$ -progression is a sequence of positive integers $\{x_1 < x_2 < \cdots < x_k\}$ such that $x_i - x_{i-1} \equiv a \pmod{m}$ for $2 \le i \le k$.

For fixed m and a, denote the family of all $a \pmod{m}$ -progressions by $AP_{a(m)}$.

Definition 6.2. Let $m \ge 2$. An arithmetic progression (mod m) is a sequence that is an $a \pmod{m}$ -progression for some $a \in \{1, 2, ..., m-1\}$.

Note that in Definition 6.2 we do not allow a = 0 (in the context of the congruence classes modulo m, having a = 0 is somewhat like having a gap of 0 in an arithmetic progression).

6. Arithmetic Progressions (mod m)

For a given $m \ge 2$, we denote the family of all arithmetic progressions (mod m) by $AP_{(m)}$. Obviously, for any $m \ge 2$,

$$AP_{(m)} = \bigcup_{a=1}^{m-1} AP_{a(m)}$$

Example 6.3. The sequence $\{1, 7, 33, 44, 70\}$ is a 5-term 1 (mod 5)progression. The sequence $\{3, 41, 49, 67\}$ is a member of $AP_{8(10)}$ as well as a member of $AP_{3(5)}$ and $AP_{0(2)}$. Both sequences are members of $AP_{(5)}$.

6.1. The Family of Arithmetic Progressions $(\mod m)$

Van der Waerden's theorem tells us that $R(AP, k; r) < \infty$ for all k and r; in other words, that the family of arithmetic progressions, AP, is regular. In contrast, the next theorem tells us that for every $m \ge 2$, the family of arithmetic progressions (mod m) is not regular (it is not even 2-regular). The proof makes use of elementary group theory.

Theorem 6.4. Let $m \ge 2$ and $k > \lfloor \frac{m}{2} \rfloor$. Then $R(AP_{(m)}, k; 2) = \infty$.

Proof. Let m and k be as in the statement of the theorem. It is sufficient to find a 2-coloring of \mathbb{Z}^+ that admits no monochromatic k-term arithmetic progression (mod m).

Let the following string represent $\chi : \mathbb{Z}^+ \to \{0, 1\}$:

$$\underbrace{11\dots 1}_{\lceil \frac{m}{2}\rceil}\underbrace{00\dots 0}_{m-\lceil \frac{m}{2}\rceil}\underbrace{11\dots 1}_{m-\lceil \frac{m}{2}\rceil}\underbrace{00\dots 0}_{m-\lceil \frac{m}{2}\rceil}$$

We will prove that for each $a, 1 \le a \le m-1$, the maximum size of a monochromatic $a \pmod{m}$ -progression does not exceed $\left\lceil \frac{m}{2 \operatorname{gcd}(a,m)} \right\rceil$, which is bounded above by $\left\lceil \frac{m}{2} \right\rceil$. It is clear that the theorem follows from this.

Let $a \in \{1, 2, ..., m - 1\}$ be fixed. Let $d = \operatorname{gcd}(a, m)$ and let $q = \frac{m}{d}$. We will consider the integers 1, 2, ..., m as the elements of \mathbb{Z}_m , where \mathbb{Z}_m is the *m*-element cyclic group of order *m*, under the operation of addition modulo *m* (note: it is more typical to call the identity element 0, but under addition modulo *m*, using the element *m* as the identity works just as well; recall that addition in $\mathbb{Z}_m = \{0, 1, ..., m - 1\}$ is defined by $i \oplus j = \overline{i + j}$, where, for any $x \in \mathbb{Z}$, \overline{x} denotes the (unique) integer such that $x \equiv \overline{x} \pmod{m}$ and $0 \leq \overline{x} \leq m - 1$).

From elementary group theory we know that, since q is a divisor of m, there is a unique q-element cyclic subgroup H of \mathbb{Z}_m , with $H = \{d, 2d, \ldots, qd = m\}$. Also, since gcd(a, m) = d, H is generated by a, so that

(6.1) $H = \{d, 2d, \dots, qd\} = \left\{\overline{a}, \overline{2a}, \dots, \overline{qa}\right\}.$

Now assume that $X = \{x_1 < x_2 < \cdots < x_q\}$ is an arbitrary *q*-term *a* (mod *m*)-progression (in \mathbb{Z}^+). Therefore, we have, for each $i, 1 \leq i \leq q-1$,

$$x_{i+1} = x_i + d_i,$$

where $d_i \equiv a \pmod{m}$. It follows that

$$\{\overline{x_1}, \overline{x_2}, \dots, \overline{x_q}\} = \left\{\overline{x_1}, \overline{x_1 + a}, \dots, \overline{x_1 + (q - 1)a}\right\}$$
$$= \overline{x_1} + H.$$

Furthermore, from (6.1) we have that $Y = \overline{x_1} + H$, a subset of [1, m], is an arithmetic progression with gap d and length q. Among the q members of this arithmetic progression, at most $\left\lceil \frac{q}{2} \right\rceil$ of them can belong to the interval $\left[1, \left\lceil \frac{m}{2} \right\rceil\right]$, and the same holds for the interval $\left[\left\lceil \frac{m}{2} \right\rceil + 1, m\right]$. Hence, by the way χ is defined, no more than $\left\lceil \frac{q}{2} \right\rceil$ members of Y can be monochromatic. Further, since $\chi(x_i) = \chi(\overline{x_i})$ for each $i, 1 \leq i \leq q$, we see that at most $\left\lceil \frac{q}{2} \right\rceil$ members of X can form an arithmetic $a \pmod{m}$ -progression, which gives the desired result.

We end this section with a table.

$m\setminus k$	2	3	4	5	6	7
2	∞	∞	∞	∞	∞	∞
3	3	∞	∞	∞	∞	∞
4	3	∞	∞	∞	∞	∞
5	3	∞	∞	∞	∞	∞
6	3	9	∞	∞	∞	∞
7	3	9	∞	∞	∞	∞
8	3	9	∞	∞	∞	∞
9	3	9	22	∞	∞	∞
10	3	9	27	∞	∞	∞
11	3	9	22	∞	∞	∞
12	3	9	∞	∞	∞	∞
13	3	9	27	∞	∞	∞
14	3	9	27	∞	∞	∞
15	3	9	27	∞	∞	∞
16	3	9	27	53	∞	∞
17	3	9	27	58	∞	∞
18	3	9	28	54	≥ 97	≥ 101
19	3	9	28	∞	∞	∞
20	3	9	30	∞	∞ .	∞
21	3	9	32	66	∞	∞
22	3	9	35	∞	∞	∞
23	3	9	32	64	∞	∞
24	3	9	34	67	≥ 100	?
$\overline{25}$	3	9	34	79	∞	∞
26	3	9	35	65	≥ 102	?
27	3	9	35	63	≥ 110	?
28	3	9	35	65	?	?
29	3	9	35	65	∞	∞
30	3	9	35	75	∞	∞

Table 6.1: Values and lower bounds of $R(AP_{(m)}, k; 2)$

6.2. A Seemingly Smaller Family is More Regular

167

6.2. A Seemingly Smaller Family is More Regular

It is obvious from Theorem 6.4 that any subset of $AP_{(m)}$ also fails to be 2-regular. In particular, for any fixed $a \in \{1, 2, \ldots, m-1\}$, the collection $AP_{a(m)}$ of all $a \pmod{m}$ -progressions is not 2-regular. This is not unlike the fact that the family of all arithmetic progressions whose gaps must be the fixed integer d (in Chapter 4, we denoted this family by A_D , where $D = \{d\}$) is not 2-regular. Loosely speaking, we might say that families such as $AP_{a(m)}$ ($a \neq 0$) or $A_{\{d\}}$ are too "small" to be 2-regular. This is not surprising, especially for $A_{\{d\}}$, since such families restrict the allowable gap (or the gap $(\mod m)$) to just one number. Yet, it turns out that every collection of the form

is, in fact, 2-regular. This statement seems to run counter to our intuition, since it surely seems that the family $AP_{(m)}$ is, in some sense, much "larger" than the family of (6.2). We proceed to establish the 2-regularity of these "smaller" families.

For families of the form (6.2) in which a = 0, it is fairly easy to show regularity. We begin with this case. It is clear from the definitions that any member of $A_{\{m\}}$ is also a member of $AP_{0(m)}$. Hence $AP_{0(m)} \cup A_{\{m\}} = AP_{0(m)}$.

Theorem 6.5. Let $k, r \ge 1$ and $m \ge 2$. Then

$$R(AP_{0(m)}, k; r) = rm(k-1) + 1.$$

Proof. Consider any r-coloring of I = [1, rm(k-1) + 1]. Exactly r(k-1) + 1 elements of I are congruent to $1 \pmod{m}$. By the pigeonhole principle, at least k of these r(k-1) + 1 integers have the same color. Since these k elements are mutually congruent modulo m, they form a $0 \pmod{m}$ -progression. This proves that rm(k-1) + 1 is an upper bound for $R(AP_{0(m)}, k; r)$.

To show the reverse inequality, let χ be the coloring represented by the string

$$(1^m 2^m \dots r^m) (1^m 2^m \dots r^m) \dots (1^m 2^m \dots r^m)$$

6. Arithmetic Progressions (mod m)

where the string $(1^m 2^m \dots r^m)$ appears k-1 consecutive times. Since χ is an *r*-coloring of [1, rm(k-1)] that avoids *k*-term monochromatic 0 (mod *m*)-progressions, the proof is complete.

We now turn our attention to families of the form (6.2) where $a \neq 0$. For brevity, from now on, we will use the symbol $AP_{a(m)}^*$ to denote $AP_{a(m)} \cup A_{\{m\}}$. We shall obtain upper and lower bounds for $R(AP_{a(m)}^*, k)$. As it turns out, the behavior of $R(AP_{a(m)}^*, k)$ when $\frac{m}{\gcd(a,m)}$ is even is somewhat different from the behavior when $\frac{m}{\gcd(a,m)}$ is odd. Therefore, we shall handle these two cases separately.

We will establish bounds on $R(AP_{a(m)}^*, k)$ by considering a certain generalization of this Ramsey-type function. The generalization involves adding one more parameter. While it is often the case that proving a generalization of a theorem is more difficult than proving the theorem, an extra parameter can also give us some "leeway" in the proof. In this particular instance, proving the generalization appears better suited to the mode of proof than proving the less complex theorem ($k = \ell$ in Theorem 6.7 below). Also, if we can get a more sweeping result without more work, all the better. Here is the generalization of the function.

Definition 6.6. Let $k, \ell, r \geq 2$. For $1 \leq a < m$, let $R(AP^*_{a(m)}, k, \ell; r)$ denote the least positive integer n (if it exists) such that for every r-coloring of [1, n] there is either a k-term monochromatic $a \pmod{m}$ -progression or an ℓ -term monochromatic arithmetic progression with gap m.

As usual, $R(AP^*_{a(m)}, k, \ell; 2)$ is abbreviated by $R(AP^*_{a(m)}, k, \ell)$. We begin with the case in which $\frac{m}{\gcd(a,m)}$ is even.

Theorem 6.7. Let $1 \le a < m$ and let $k \ge 2$ and $\ell \ge 3$. Assume $\frac{m}{\gcd(a,m)}$ is even. Let $c = m(k-1)(\ell-1)$ Then

(6.3) $c+1 \le R(AP^*_{a(m)}, k, \ell) \le c+a(k-2)+1.$

Proof. To establish the lower bound, we will present a 2-coloring of $[1, n] = [1, m(k - 1)(\ell - 1)]$ that avoids both monochromatic k-term $a \pmod{m}$ -progressions and monochromatic ℓ -term arithmetic

progressions with gap m. To describe this coloring, we first partition $[1, m(k-1)(\ell-1)]$ into the k-1 blocks

$$B_i = [m(\ell - 1)(i - 1) + 1, m(\ell - 1)i],$$

 $1 \leq i \leq k-1$. Letting $d = \gcd(a, m)$, we further partition each B_i into $\frac{m(\ell-1)}{d}$ smaller blocks:

$$B_i = \bigcup_{j=1}^{\frac{m(\ell-1)}{d}} C_{i,j},$$

where

$$C_{i,j} = [m(\ell-1)(i-1) + d(j-1) + 1, m(\ell-1)(i-1) + dj]$$

Note that each $C_{i,j}$ contains d integers and that

 $[1,n] = \bigcup_{j=1}^{\frac{m(\ell-1)}{d}} \bigcup_{i=1}^{k-1} C_{i,j}.$

Now define the 2-coloring λ of [1, n] as follows. Let $\lambda : C_{1,1} \to \{0, 1\}$ be defined arbitrarily. For $(i, j) \neq (1, 1)$, color $C_{i,j}$ by the following rule: for each $t \in [1, d]$, $\lambda (m(\ell - 1)(i - 1) + d(j - 1) + t) = \lambda(t)$ if and only if *i* and *j* have the same parity.

Since $|B_i| = m(\ell - 1)$ for each *i*, no single B_i can contain an ℓ -term arithmetic progression with gap *m*. So if there exists an ℓ -term monochromatic arithmetic progression with gap *m*, there must exist x, y, i_1, j_1, j_2 with y - x = m, $\lambda(x) = \lambda(y), x \in C_{i_1,j_1}$, and $y \in C_{i_1+1,j_2}$. Since 2*d* divides *m* and $|C_{i,j}| = d$ for all *i* and *j*, $j_2 - j_1$ must be even. Therefore, by the definition of $\lambda, \lambda(x) \neq \lambda(y)$, a contradiction.

To complete the proof of the lower bound, we need to show that, under λ , there is no monochromatic $a \pmod{m}$ -progression of length k. Assume, for a contradiction, that such a progression exists. Then for some i, B_i contains at least two terms of this progression. So assume $x, y \in B_i$, with x < y, $\lambda(x) = \lambda(y)$, and $y - x \equiv a \pmod{m}$. It follows by the definition of λ that if $y \in C_{i,j}$ and $x \in C_{i,j'}$, then j - j' is even. However, since $\frac{m}{d}$ is even, $\frac{a}{d}$ is odd; this implies that y - x = ud where u is odd, contradicting the fact that j - j' is even. This establishes the lower bound. For the upper bound, we use induction on k. First assume k = 2and let χ be any 2-coloring of $[1, m(\ell - 1) + 1]$. Let

 $S = \{ im + 1 : 1 \le i \le \ell - 1 \}.$

If S is not monochromatic, then some member of S has the same color as m + 1 - a, and we have a 2-term monochromatic $a \pmod{m}$ -progression. If S is monochromatic, and if $\chi(1) = \chi(S)$, then we have an ℓ -term monochromatic arithmetic progression with gap m. Finally, if S is monochromatic, but $\chi(1) \neq \chi(S)$, then we must have either $\chi(m+a+1) = \chi(1)$ or $\chi(m+a+1) = \chi(m+1)$, each of which gives us a monochromatic 2-term $a \pmod{m}$ -progression.

Now assume that $k \ge 2$ and that the upper bound of (6.3) holds for k and all $\ell \ge 3$. Let χ be any 2-coloring of

$$I = [1, mk(\ell - 1) + a(k - 1) + 1].$$

We wish to show that, under χ , there is either a (k + 1)-term monochromatic $a \pmod{m}$ -progression or an ℓ -term monochromatic arithmetic progression with gap m. By the inductive hypothesis, we may assume that there is some k-term monochromatic $a \pmod{m}$ progression X contained in $[1, m(k-1)(\ell-1) + a(k-2) + 1]$ (or else we would have the desired monochromatic ℓ -term arithmetic progression). Let x_k denote the largest member of X. If any member z of $Y = \{x_k + a + im : 0 \le i \le \ell - 1\} \subseteq I$ has the same color as X, then $X \cup \{z\}$ is a monochromatic (k+1)-term $a \pmod{m}$ -progression, as desired. If no member of Y has the same color as X, then Y is a monochromatic ℓ -term arithmetic progression with gap m, which completes the proof.

For k = 2, Theorem 6.7 gives a precise formula for the associated Ramsey function, which we state as the following corollary.

Corollary 6.8. Let $1 \le a < m$, and assume $\frac{m}{\gcd(a,m)}$ is even. Then $R(AP_{a(m)}^*, 2, \ell) = m(\ell - 1) + 1.$

Under a certain strengthening of the hypotheses of Theorem 6.7, it can be shown that the lower bound of (6.3) is the exact value of $R(AP_{a(m)}^*, k, \ell)$. The proof being rather complex, we state this result without proof.

Theorem 6.9. Let $1 \le a < m$, where a divides m and $\frac{m}{a}$ is even. Let $k, \ell \ge 3$ with $\frac{k-2}{\ell-2} \le \frac{m}{a}$. Then

$$R(AP_{a(m)}^{*}, k, \ell) = m(k-1)(\ell-1) + 1$$

For $k = \ell$, we have the following result as an immediate corollary of Theorem 6.9.

Corollary 6.10. Let $1 \le a < m$ where a divides m and $\frac{m}{a}$ is even. Then for all $k \ge 3$, $R(AP^*_{a(m)}, k) = m(k-1)^2 + 1$.

We now examine the situation in which $\frac{m}{\gcd(a,m)}$ is odd. In the next theorem we give an upper bound for the associated Ramsey-type function. We first mention a lemma that will be useful.

Lemma 6.11. Let $1 \leq a < m$. Let $k, \ell \geq 2$ and $c \geq 1$. Let $n = R(AP^*_{a(m)}, k, \ell; r)$. Then

$$R(AP_{ca(cm)}^{*}, k, \ell; r) = c(n-1) + 1.$$

Proof. Notice that for any $c \ge 1$, $X = \{x_1 < x_2 < \cdots < x_k\}$ is an *m*-a.p. if an only if $1 + cX = \{1 + cx_i : 1 \le i \le k\}$ is an arithmetic progression with gap *cm*. Also, X is an $a \pmod{m}$ -progression if and only if 1 + cX is a $ca \pmod{cm}$ -progression. Therefore, by the meaning of *n*, any *r*-coloring of the set $\{1, c+1, 2c+1, \ldots, (n-1)c+1\}$ must contain a *k*-term monochromatic $ca \pmod{cm}$ -progression or an ℓ -term monochromatic arithmetic progression with gap *ca*. Hence, $R(AP_{ca(cm)}^*, k, \ell; r) \le c(n-1) + 1$.

For the reverse inequality, we know there exists an r-coloring χ of [1, n - 1] that avoids both monochromatic k-term $a \pmod{m}$ -progressions and monochromatic ℓ -term arithmetic progressions with gap m. Now define the r-coloring χ' of [1, c(n - 1)] as follows: for each $j, 1 \leq j \leq n - 1$,

$$\chi'([c(j-1)+1, cj]) = \chi(j)$$

We complete the proof by showing that χ' avoids monochromatic k-term $ca \pmod{cm}$ -progressions, and also avoids monochromatic ℓ -term arithmetic progressions with gap cm. Assume, for a contradiction, that $\{s_i : i = 1, 2, ...\}$ is a sequence of one of these types that is monochromatic with respect to χ' . Let $t_i = \begin{bmatrix} s_i \\ c \end{bmatrix}$ for each i. Then, by

the definition of χ' , using the reasoning of the previous paragraph, the sequence $\{t_i : i = 1, 2, ...\} \subseteq [1, n - 1]$ is either a *k*-term $a \pmod{m}$ -progression or an ℓ -term arithmetic progression with gap *m* that is monochromatic with respect to χ , a contradiction.

We remark that the above proof also shows that if, for a particular triple $(k, \ell; r)$, one of the two Ramsey-type functions mentioned in the statement of Lemma 6.11 is infinite, then so is the other.

We now give an upper bound for $R(AP^*_{a(m)},k,\ell)$ when $\frac{m}{\gcd(a,m)}$ is odd.

Theorem 6.12. Let $1 \le a < m$ and let $k, \ell \ge 2$. Let $d = \gcd(a, m)$ and assume $\frac{m}{d}$ is odd. Then

$$R(AP^*_{a(m)}, k, \ell) \le m\left((k-2)(\ell-1) + 1\right) + ka - d + 1$$

Proof. Note that it is sufficient to prove the result for d = 1. To see this, assume the theorem holds whenever d = 1, and let $a' = \frac{a}{d}$ and $m' = \frac{m}{d}$. Note that gcd(a', m') = 1. Hence, using Lemma 6.11, we have

$$\begin{aligned} R(AP_{a(m)}^*, k, \ell) &= d\left(R(AP_{a'(m')}^*, k, \ell) - 1\right) + 1 \\ &\leq d\left(m'((k-2)(\ell-1) + 1) + ka' - 1\right) + 1 \\ &= m\left((k-2)(\ell-1) + 1\right) + ka - d + 1. \end{aligned}$$

We complete the proof by using induction on k. For k = 2 we must show that for any $\ell \geq 2$, $R(AP^*_{a(m)}, 2, \ell) \leq m + 2a$. Apparently, for k = 2 the value of ℓ is irrelevant; this is so, as we will show that $R(AP_{a(m)}, 2) \leq m + 2a$. By way of contradiction, assume there exists a 2-coloring χ of [1, m + 2a] that avoids 2-term monochromatic $a \pmod{m}$ -progressions. Let \oplus represent the operation of addition modulo m+2a in the group \mathbb{Z}_{m+2a} , where we take the identity element to be m+2a. Notice that if $m+a < i \leq m+2a$, then $i \oplus a = i-m-a$, so that $\{i \oplus a, i\}$ is a 2-term $a \pmod{m}$ -progression. Thus, by our assumption, for every $i \in [1, m + 2a]$,

(6.4)
$$\chi(i \oplus a) \neq \chi(i).$$

6.3. The Degree of Regularity

Since d = 1, m is odd and therefore m + 2a is odd. Hence, by (6.4) we must have $\chi(1 \oplus (m+2a)a) \neq \chi(1)$. However, $1 \oplus (m+2a)a = 1$, a contradiction.

Now assume that $k \ge 2$ is fixed and that the upper bound holds for k and every $\ell \ge 2$. We will show that it holds for k + 1. Let $\ell \ge 2$ and let χ be any 2-coloring of $[1, m((k-1)(\ell-1)+1)+(k+1)a]$. By the inductive hypothesis we may assume there is a monochromatic kterm $a \pmod{m}$ -progression X within $[1, m((k-2)(\ell-1)+1)+ka]$ (or else we have the desired ℓ -term progression). The proof is completed in the very same manner as the last part of the proof of Theorem 6.7; we leave the details to the reader as Exercise 6.8.

We next state, without proof, the best known lower bound for $R(AP_{a(m)}^*, k, \ell)$ when $\frac{m}{\gcd(a,m)}$ is odd.

Theorem 6.13. Let $1 \leq a < m$ and assume that $\frac{m}{\gcd(a,m)}$ is odd. Then for all $k, \ell \geq 3$,

$$R(AP_{a(m)}^*, k, \ell) \ge (k-2)(m(\ell-2)+a)+1.$$

From Theorems 6.7, 6.12, and 6.13, the following result is immediate. \cdot

Corollary 6.14. If $1 \le a < m$, then

$$R(AP_{a(m)}^{*}, k; 2) = mk^{2}(1 + o(1)).$$

Note that the magnitude of the Ramsey-type function given by Corollary 6.14 is significantly smaller than that of w(k; 2) (which grows at least exponentially – see Theorem 2.18).

6.3. The Degree of Regularity

Van der Waerden's theorem tells us that the family of arithmetic progressions is regular, i.e., the associated Ramsey-type function is finite for all k regardless of the number of colors being used. According to Corollary 6.14, for $1 \leq a < m$ the family $AP_{a(m)}^*$ is 2-regular. However, unlike the family of arithmetic progressions, in most cases $AP_{a(m)}^*$ has degree of regularity only two, and never more than three. What makes this especially interesting is that the associated Ramsey

6. Arithmetic Progressions (mod m)

function of this family, using two colors, is a much slower growing function than that of the family of arithmetic progressions. The next theorem shows that when $a \neq \frac{m}{2}$, this family is not 3-regular.

Theorem 6.15. Let $m \ge 3$ and $1 \le a < m$. Assume $a \ne \frac{m}{2}$. Then $R(AP_{a(m)}^*, k; 3) = \infty$ whenever $k > \lfloor \frac{2m}{3} \rfloor$.

Proof. We actually prove a stronger result: for m, a, and k as in the statement of the theorem, $R(AP^*_{a(m)}, k, 2; 3) = \infty$. That is, we show that there exists a 3-coloring of \mathbb{Z}^+ that avoids monochromatic k-term $a \pmod{m}$ -progressions and monochromatic 2-term arithmetic progressions with gap m.

As was explained in the proof of Theorem 6.12, by Lemma 6.11 we may assume that gcd(a,m) = 1. Let $s = \lceil \frac{2m}{3} \rceil$ and $t = \lceil \frac{4m}{3} \rceil$. Define $\chi : \mathbb{Z}^+ \to \{1,2,3\}$ to be the 3-coloring represented by the string

$$1^{2}2^{t-2}3^{2m-t}1^{2}2^{t-2}3^{2m-t}1^{2}2^{t-2}3^{2m-t}$$

Note that χ is periodic with period 2m.

We next show that χ yields no 2-term monochromatic arithmetic progression with gap m. Since $m \geq 3$, we have s < m < t. Thus if $\overline{j} \in [1, s]$, then $s+1 \leq \overline{j+m} \leq 2m$, so that $\chi(j+m) \neq \chi(j)$. Likewise if $\overline{j} \in [s+1,t]$, then $\overline{j+m} \notin [s+1,t]$, and if $\overline{j} \in [t+1,2m]$, then $\overline{j+m} \notin [t+1,2m]$. So for every positive integer $j, \chi(j) \neq \chi(j+m)$, i.e., there is no 2-term monochromatic member of $A_{\{m\}}$.

To complete the proof, let S be any (s + 1)-term $a \pmod{m}$ -progression; we show that S is not monochromatic under χ . We see that S has the form $\{x, x + a + c_1m, x + 2a + c_2m, \ldots, x + sa + c_sm\}$ with $0 \le c_1 \le c_2 \le \cdots \le c_s$. Let

 $\overline{S} = \left\{ \overline{x}, \overline{x+a+c_1m}, \overline{x+2a+c_2m}, \dots, \overline{x+sa+c_sm} \right\}.$

Note that \overline{S} consists of s+1 distinct elements modulo m, for otherwise there exist i < j with $x+ia+c_im \equiv x+ja+c_jm \pmod{m}$; then, since $gcd(a,m)=1, i \equiv j \pmod{m}$, contradicting the fact that $0 \le i < j < m$. Clearly, these s+1 elements are also distinct modulo 2m. By the way χ is defined, and since each of [1, s], [s+1, t],and [t+1, 2m] contains no more than s elements, S cannot be monochromatic under χ . \Box

6.3. The Degree of Regularity

In the above proof we needed the fact that $\frac{m}{\gcd(a,m)} \geq 3$. The proof does not work if $\frac{m}{\gcd(a,m)} = 2$, i.e., if $a = \frac{m}{2}$. In this one case, $AP_{a(m)}^*$ is 3-regular. In fact, an exact formula for the associated Ramsey-type function using three colors is known. We state it, without proof, as the next theorem.

Theorem 6.16. For all $m, k \geq 2$, if $\frac{m}{2} \in \mathbb{Z}^+$ then

$$R(AP^*_{\frac{m}{2}(m)}, k; 3) = \frac{m}{2}(6k^2 - 13k + 5) + 1.$$

So far we know that the family of sequences $AP_{a(m)}^*$ is regular if a = 0, has degree of regularity at least three if $a = \frac{m}{2}$, and has degree of regularity two otherwise. We will know the degree of regularity for this family in all cases if we can determine the degree of regularity when $a = \frac{m}{2}$. The next theorem provides us with the solution: the degree of regularity is exactly three if $a = \frac{m}{2}$. What is even more striking is that although the associated Ramsey function using three colors exists (is finite) for all values of k (the length of the sequence), when four colors are used it does not exist even for k = 2.

Theorem 6.17. Let $1 \leq a < m$ and assume that $\frac{m}{\gcd(a,m)}$ is even. Then $R(AP^*_{a(m)}, 2; 4) = \infty$.

Proof. By Lemma 6.11 it suffices to prove this when gcd(a, m) = 1. Hence we may assume that m is even and a is odd. To prove the theorem we provide a 4-coloring of \mathbb{Z}^+ that admits no monochromatic 2-term $a \pmod{m}$ -progressions and no monochromatic 2-term arithmetic progressions with gap m.

Let χ be the 4-coloring of \mathbb{Z}^+ represented by

$$\underbrace{1212\ldots 12}_{m}\underbrace{3434\ldots 34}_{m}\underbrace{1212\ldots 12}_{m}\underbrace{3434\ldots 34}_{m}\ldots$$

It is easy to see that χ admits no monochromatic 2-term arithmetic progression with gap m. Also, since m is even, by the way χ is defined, whenever |y - x| is odd, $\chi(x) \neq \chi(y)$. Thus, there is no monochromatic 2-term $a \pmod{m}$ -progression.

We conclude this chapter with a table that summarizes what is known about the degree of regularity of the families in this chapter.

6.4. Exercises

6. Arithmetic Progressions (mod m)

Throughout the table, we assume that 1 < a < m is fixed. Whenever a family is *r*-regular, we also give the order of magnitude (as a function of the length *k* of the sequence, with *r* fixed) of the best known upper bound for the associated function $R(\mathcal{F}, k; r)$. For comparison reasons, we also include the family AP, but do not include Gowers' bounds for R(AP, k; r) (see Section 2.4).

\mathcal{F}	Restriction	2-regular	3-regular	<i>r</i> -regular $(r \ge 4)$
AP		yes	yes	yes
$AP_{(m)}$		no	no	no
$AP_{a(m)}$		no	no	no
$AP_{0(m)}$		yes; $2mk$	yes; $3mk$	yes; rmk
$AP_a^*(m)$	$\frac{m}{a} \neq 2$	yes; mk^2	no	no
$AP_a^*(m)$	$\frac{m}{a} = 2$	yes; mk^2	yes; $3mk^2$	no

Table 6.2: Degree of regularity of families of type $AP_{(m)}$

6.4. Exercises

6.1 Let $1 \le a < m$.

a) Show that there exists a set of positive integers that contains arbitrarily long members of $AP_{a(m)}^*$ but that fails to contain arbitrarily long members of $AP_{a(m)}$.

b) Show that there exists a set of positive integers that contains arbitrarily long members of $AP_{a(m)}$ but does not contain arbitrarily long descending waves (see Definition 3.20).

c) Show that there exists a set of positive integers that contains arbitrarily long arithmetic progressions but does not contain arbitrarily long members of $AP_{a(m)}^*$.

- **6.2** Can you find a relationship among a, b, and m, so that any set containing arbitrarily long members of $AP^*_{a(m)}$ must also contain arbitrarily long members of $AP^*_{b(m)}$?
- **6.3** Define g(n) to be the largest value of k such that for every 2-coloring of the group $\mathbb{Z}_n = \{0, 1, 2, \ldots, n-1\}$ there is a monochromatic k-term arithmetic progression (with distinct members). Note: we are not insisting that the members of the arithmetic progression be increasing, just distinct. Find

g(6). Also, If it is known that $g(n) \leq k$, is there anything we can say about w(k+1)?

- **6.4** Define h(n) to be the same as g(n) in Exercise 6.3, except the members of the arithmetic progression are not required to be distinct. Find h(6). Also, find an infinite set S such that if $s \in S$, then g(s) = h(s).
- **6.5** Prove that if $R(AP_{(m)}, k) > (m-1)(k-1) + 1$, then

$$w(k) > (m-1)(k-1) + 1.$$

- **6.6** Prove that if $m \ge \left\lceil \frac{w(k)-1}{k-1} \right\rceil$, then $R(AP_{(m)}, k; 2) \le w(k)$.
- **6.7** Prove the following implication: if $R(AP_{(m)}, k; 2) = \infty$ for $k \ge \log m$, then $w(\log m) \ge m \log m$ and hence $w(k) \ge ke^k$.
- 6.8 Complete the last part of the proof of Theorem 6.12.
- **6.9** Let $1 \le a < m$. Prove that $R(AP^*_{a(m)}, 2; 3) \le 3m$. (Hint: assume there exists a 3-coloring $\chi : [1, 3m] \to \{1, 2, 3\}$ yielding no appropriate monochromatic 2-element set; assume that $\chi(m+1) = 1$ and $\chi(2m+1) = 2$, and arrive at a contradictric., tion.)

6.10 Let 1 ≤ a < m, let m/gcd(a,m) be odd, and let k ≥ 2.
a) Prove the following, which is a slight improvement over Theorem 6.12 in certain cases when l = 3: R(AP^{*}_{a(m)}, k, 3; 2) ≤ 2m(k - 1) + a(k - 2) + 1.
b) Prove the following, which gives a slight improvement over Theorem 6.12 when l = 2: R(AP^{*}_{a(m)}, k, 2; 2) ≤ m(k - 1) + a(k - 1) + 1 - ε, where ε = 0 if a is odd, and ε = gcd(a, m) is a is even.

6.11 Consider the following generalization of $AP_{a(m)}^*$. For T a set of positive integers, let $AP_{a(m),T}^* = AP_{a(m)} \cup A_T$; that is, it consists of all $a \pmod{m}$ -progressions and all arithmetic progressions whose gaps are in T. Now define $R(AP_{a(m),T}^*, k, \ell)$ to be as in Definition 6.6, except we replace " ℓ -term member of $A_{\{m\}}$ " with " ℓ -term member of A_T ". Prove the following. a) $R(AP_{a(m),T}^*, k) < \infty$ if and only if $cm \in T$ for some $c \in \mathbb{Z}^+$.

6. Arithmetic Progressions (mod m)

b) Let $d = \gcd(a, m)$, and let $c \ge 1$, $k \ge 2$ and $\ell \ge 3$. Assume that $cm \in T$. If $\frac{m}{d}$ is even, then $R(AP_{a(m),T}^*, k, \ell) \le cm(k-1)(\ell-1) + (k-2)a + 1$. c) Under the hypothesis of (b), except assuming $\frac{m}{d}$ is odd, $R(AP_{a(m),T}^*, k, \ell) \le cm(k-2)(\ell-1) + m + ak - d + 1$.

6.5. Research Problems

- 6.1 Investigate which pairs of integers m and k have the property that $R(AP_{(m)}, k) < \infty$. Exercise 6.6 handles the situation in which $m \ge \left\lceil \frac{w(k)-1}{k-1} \right\rceil$. For those values of m and k that do not satisfy this inequality, the situation is unclear. For example, it is known that $R(AP_{(22)}, 5) = \infty$, while $R(AP_{(21)}, 5)$ and $R(AP_{(23)}, 5)$ are both finite. In particular, find a function h(m) (as small as possible) such that $R(AP_{(m)}, h(m)) = \infty$ for all $k \ge h(m)$. (See Table 6.1.) References: [65], [173]
- **6.2** Prove or disprove: if m and k are such that $R(AP_{(m)}, k) < \infty$, then $R(AP_{(m)}, k) \le w(k)$. (See Table 6.1.) References: [65], [173]
- **6.3** Find bounds for $R(AP_{(m)}, k)$ when it exists. (See Table 6.1.) References: [65], [173]
- **6.4** Computer calculations suggest that the upper bound of 3m from Exercise 6.9 is not the best possible. Improve on this bound (or show it *is* the best possible). References: [173], [176]
- **6.5** It is known that if $\frac{m}{\gcd(a,m)}$ is odd, then $R(AP^*_{a(m)}, 2; 6) = \infty$. For certain such pairs a and m the least value of r such that $R(AP^*_{a(m)}, 2; r) = \infty$ is 6; for some it is 5. Determine if the least r is ever 4 (it is never 3 by Exercise 6.9). Reference: [176]
- **6.6** It has been conjectured that Theorem 6.9 is true when the restriction $\frac{k-2}{\ell-2} \leq \frac{m}{a}$ is removed. Prove or disprove this conjecture. (Note: it is known that this formula does not work if we loosen the requirement to $\frac{m}{\gcd(a,m)}$ being even. For example, computer calculations have shown that $R(AP_{9(10)}^*, 3, 3) = 49$,

which is much closer to the upper bound of 50 given by Theorem 6.7 than to the lower bound of 41. The conjecture has been proven when $\frac{m}{a} = 2$, but not for other cases.) Reference: [173]

6.7 Improve the lower bound given by Theorem 6.13. Perhaps it can be tightened to $m(k-2)(\ell-1) + 1$. Reference: [173]

6.5. Research Problems

- **6.8** Corollary 6.8 gives a precise formula for $R(AP^*_{a(m)}, k, \ell)$ when k = 2 and $\frac{a}{\gcd(a,m)}$ is even. Formulae have also been found for this function when k = 2 and $\frac{a}{\gcd(a,m)}$ is odd. No formulae are known for $\ell = 2$ and general k. Find such formulae. Reference: [173]
- **6.9** Consider the variation of $R(AP_{a(m)}^*, k, \ell)$ that follows. Let $R'(AP_{a(m)}^*, k, \ell)$ denote the least positive integer n so that every 2-coloring of [1, n] either admits an ℓ -term monochromatic arithmetic progression with gap m, or else in *each* color there is a k-term monochromatic $a \pmod{m}$ -progression. It has been shown that $R'(AP_{1(2)}^*, k, \ell) \leq 2(k-1)(\ell-1)+k-2$. Does this generalize to arbitrary a and m?
- Reference: [173] 6.10 Families analogous to type $AP_{a(m)}$ have been considered, where more than one congruence class is allowed. For example, the families $AP_{a(m),b(n)} = AP_{a(m)} \cup AP_{b(n)}$ were studied, and a characterization was given for those pairs of congruence classes $a \pmod{m}$ and $b \pmod{n}$ for which such a family is 2-regular. An upper bound was also given when 2-regularity occurs. As one special case, if m is even and b = (c-1)m for some $c \ge 2$, then $R(AP_{a(m),b(cm)}, 3) \le 4cm$.

b = (c-1)m for some $c \ge 2$, then $R(AP_{a(m),b(cm)},3) \le 4cm$. Prove or disprove the conjecture that under the same hypotheses, $R(AP_{a(m),b(cm)},3) = 4(c-1)m+1$. Reference: [65]

6.11 Given $1 \le a < m$ fixed, determine the Ramsey properties of sequences of the form $\{x_1 < x_2 < \cdots < x_k\}$ such that $x_i - x_1 \equiv a \pmod{m}$ for all $i = 2, \ldots, k$. References: [65], [160], [173]

6.12 Let $1 \leq a < m$. Consider the family of all sequences of positive integers of the form $\{x_1 < x_2 < \cdots < x_k\}$ satisfying $x_i - x_{i-1} = t_i d$, with $t_i \equiv a \pmod{m}$ for each *i*, and where *d* is some fixed positive integer. For example, if $t_i = 1$ for all *i*, then these are the arithmetic progressions. For m = 2 and a = 1, it has been calculated that for this family (call it \mathcal{C}), $R(\mathcal{C}, 3) = 9$, $R(\mathcal{C}, 4) = 15$, and $R(\mathcal{C}, 5) = 21$. Study \mathcal{C} . One specific question is this: what happens for a = 1 and arbitrary m? Reference: [65], [160], [173]

6.6. References

§6.1. The proof of Theorem 6.4 is from [65], which also discusses the Ramsey-type functions for $AP_{a(m)} \cup AP_{b(n)}$, as well as unions of more than two such families. It also contains work related to Exercise 6.11. §6.2. The family $AP_{a(m)}^*$ is introduced by Landman and Long [173], and its 2-color Ramsey-type functions are studied. This paper contains proofs of Theorems 6.5, 6.7, 6.9, 6.12, and 6.13, and of Lemma 6.11. It also gives an exact formula for $R(AP_{a(m)}^*, 2, \ell)$. It discusses several computer-generated values and patterns which leave us with several unanswered, but intriguing questions.

§6.3. Proofs of Theorems 6.15–6.17 are in [176]; that paper also includes more work on $R(AP_{a(m)}^*, 2; r)$. The generalization of $AP_{a(m)}^*$ to families of the type $AP_{a(m)} \cup A_T$ is examined in [160], which also includes a table summarizing what is known about the regularity of these and related families, as well as the asymptotic values of their associated Ramsey-type functions. For an extended version of Table 6.2, see [176].

Additional References: Arithmetic progressions contained in the group \mathbb{Z}_n (where the elements of the arithmetic progression are distinct in \mathbb{Z}_n) are studied in [254] (we believe that the very last inequality in the article is incorrect) with upper and lower bounds given for the associated Ramsey-type functions. A relationship between such Ramsey functions and the classical van der Waerden numbers yields a lower bound for w(6).

Chapter 7

Other Variations on van der Waerden's Theorem

The notion of an arithmetic progression, being so basic, has naturally led to many intriguing mathematical questions. Here we mention a few selected topics dealing with arithmetic progressions, not covered in the previous chapters, that fall under the general heading of Ramsey theory on the integers.

7.1. The Function $\Gamma_m(k)$

In Chapter 3 the function $\Gamma_m(k)$ was introduced. Here is a reminder of its definition.

Definition 7.1. For $m, k \geq 2$, $\Gamma_m(k)$ denotes the least positive integer s such that for every set $S \subseteq \mathbb{Z}^+$ with $S = \{x_1, x_2, \ldots, x_s\}$ and $x_i - x_{i-1} \in \{1, 2, \ldots, m\}, 2 \leq i \leq s, S$ contains a k-term arithmetic progression.

We showed in Chapter 3 that $\Gamma_m(k)$ exists for all m and k. We also discussed the relevance of $\Gamma_m(k)$ to the goal of finding an upper bound on the van der Waerden numbers: $w(k) \leq SP_m(\Gamma_m(k))$, where SP_m is the Ramsey-type function associated with semi-progressions of scope m. In this section we look more closely at the Γ_m function.

We first define a related function.

Definition 7.2. For $m, k \ge 2$, $\Omega_m(k)$ is the least positive integer n such that whenever $X = \{x_1, x_2, \ldots, x_n\}$ and $x_i \in [(i-1)m, im-1]$, there is a k-term arithmetic progression in X.

We consider an example.

Example 7.3. Let m = 2 and k = 3. We wish to find the least integer n so that every sequence x_1, x_2, \ldots, x_n satisfying the conditions $x_1 \in \{0, 1\}, x_2 \in \{2, 3\}, \ldots, x_n \in \{2n - 2, 2n - 1\}$, will contain a 3-term arithmetic progression. The sequence $\{0, 2, 5, 6, 9, 11\}$, since it does not contain a 3-term arithmetic progression, shows that $\Omega_2(3) > 6$. If we check directly, we find that every 7-term sequence x_1, x_2, \ldots, x_7 with each $x_i \in \{2i - 2, 2i - 1\}$ does contain a 3-term arithmetic progression. Therefore $\Omega_2(3) = 7$.

One obvious question we want to ask is whether $\Omega_m(k)$ always exists. The next theorem, which describes a fundamental relationship between the values of the Ω functions and w(k; r), answers this question in the affirmative.

Theorem 7.4. For all $k, r \geq 1$,

$$\Omega_r(k) \le w(k;r) \le \Omega_r(r(k-1)+1).$$

Proof. Let w = w(k; r). We first show that $\Omega_r(k) \leq w$. To this end, let $X = \{x_1, x_2, \ldots, x_w\}$ with $x_n \in [(n-1)r, nr-1]$ for $n = 1, 2, \ldots, w$. Thus, for each $n \in [1, w]$, $x_n = (n-1)r + c_n$, where $c_n \in [0, r-1]$. Now *r*-color [1, w] as follows: $\chi(n) = j$ if and only if $c_n = j$. From the definition of w, there is some monochromatic *k*-term arithmetic progression under χ . So assume $\{a + id : 0 \leq i \leq k - 1\} \subseteq [1, w]$ has color j_0 for some $a, d \geq 1$. Hence, $c_{a+id} = j_0$ for each $i \in [0, k - 1]$, so that

$$x_{a+id} = (a+id-1)r + c_{a+id}$$

= $(a+id-1)r + j_0$
= $(a-1)r + j_0 + idr$,

for $0 \le i \le k-1$. Therefore, $\{x_{a+id} : 0 \le i \le k-1\}$ is a k-term arithmetic progression contained in X, and hence, $\Omega_r(k) \le w(k;r)$.

Now let $m = \Omega_r(r(k-1)+1)$. To complete the proof, we show that every *r*-coloring of [0, m-1] yields a monochromatic *k*-term 7.1. The Function $\Gamma_m(k)$

arithmetic progression (this is equivalent to having $w(k;r) \leq m$). Let $\lambda : [0, m-1] \rightarrow \{0, 1, \ldots, r-1\}$ be an arbitrary *r*-coloring of [0, m-1]. Define the sequence $A = \{a_i : 0 \leq i \leq m-1\}$ by letting $a_i = ir + \lambda(i)$. Notice that for each $i, a_i \in [ir, (i+1)r-1]$. By the definition of m, A must contain an arithmetic progression of length r(k-1) + 1, say

$$B = \{a_{i_j} : 0 \le j \le r(k-1)\} = \{a_{i_0} + jd : 0 \le j \le r(k-1)\}$$

for some $d \ge 1$. Then the set $\{a_{ijr} : 0 \le j \le k-1\}$ is a k-term arithmetic progression with gap rd. Hence, for each $j, 1 \le j \le k-1$, we have

(7.1)
$$rd = a_{i_{jr}} - a_{i_{(j-1)r}} = r(i_{jr} - i_{(j-1)r}) + \lambda(i_{jr}) - \lambda(i_{(j-1)r}).$$

Since the range of λ is $\{0, 1, \ldots, r-1\}$, and since the right-hand side of (7.1) must be a multiple of r, we have that $\lambda(i_{jr}) = \lambda(i_{(j-1)r})$ for each $j, 1 \leq j \leq k-1$. Hence, by (7.1), $i_{jr} - i_{(j-1)r} = d$ for each $j \in \{0, 1, \ldots, k-1\}$. Thus, $\{i_{jr} : 0 \leq j \leq k-1\}$ is a k-term arithmetic progression that is monochromatic, and is contained in $[1, a_{r(k-1)}] \subseteq [1, m]$.

In the next theorem we show that $\Gamma_r(k)$ is bounded above by a simple function involving $\Omega_r(k)$.

Theorem 7.5. $\Gamma_r(k) \leq r\Omega_r(k)$ for all $k, r \geq 2$.

Proof. Let $m = \Omega_r(k)$. It is necessary to show that every sequence $\{x_1 < x_2 < \cdots < x_{rm}\}$ satisfying $x_i - x_{i-1} \leq r$ for each $i, 2 \leq i \leq rm$, contains a k-term arithmetic progression. Let X be such a sequence. We begin with the case in which $x_1 = 1$, so that for each $j, 1 \leq j \leq m$, the interval [(j-1)r, jr-1] must contain some element y_j of X. By the definition of m, the sequence $Y = \{y_j : 1 \leq j \leq m\}$ contains a k-term arithmetic progression. Since $Y \subseteq X$, we are done.

Now assume $x_1 > 1$. Let $x'_i = x_i - x_1 + 1$ for each $i, 1 \le i \le rm$. Since $x'_1 = 1$, we know that $\{x'_i : 1 \le i \le rm\}$ contains a k-term arithmetic progression $\{y_j : 1 \le j \le k\}$. Hence,

 $\{y_j + x_1 - 1 : 1 \le j \le k\}$

is a k-term arithmetic progression contained in X, which completes the proof. $\hfill \Box$

184 7. Other Variations on van der Waerden's Theorem

Theorem 7.5 bounds Γ_r from above by an expression involving the Ω function. The Ω function may also be used to bound the Γ function from below. The proof being rather straightforward, we leave it as Exercise 7.2.

Theorem 7.6. $\Omega_r(k) \leq \Gamma_{2r-1}(k)$ for all $k, r \geq 2$.

We are now able to tie the van der Waerden numbers more closely to the Γ function.

Theorem 7.7. Let $k, r \geq 2$. Then

$$\Gamma_r(k) \le w(k;r) \le \Gamma_{2r-1}(r(k-1)+1).$$

Proof. We first show that $\Gamma_r(k) \leq w(k;r)$. Let w = w(k;r) and let $X = \{x_i : 1 \leq i \leq w\}$ with $x_i - x_{i-1} \in \{1, 2, \ldots, r\}$. We wish to show that X contains a k-term arithmetic progression. As in the proof of Theorem 7.5, it suffices to assume that $x_1 = 1$.

Define the coloring χ on [1, w] as follows: $\chi(n) = j$ if and only if $j = \min\{x_i - n : x_i \ge n\}$. Obviously, $w \le x_w$. Therefore, χ is an *r*-coloring of [1, w] using the colors $\{0, 1, \ldots, r-1\}$. By the definition of w, there is a monochromatic k-term arithmetic progression under χ ; say $\{a+jd: 0 \le j \le k-1\}$ has color t. Note that for any $n \in [1, w]$ such that $\chi(n) = t$, we have $t + n = x_i$ for some $i, 1 \le i \le w$. So for each $j, 0 \le j \le k-1$, we have $t + a + jd = x_{ij} \in X$. Thus X contains a k-term arithmetic progression, as desired.

The second inequality follows immediately from Theorems 7.4 and 7.6, since

$$w(k;r) \le \Omega_r(r(k-1)+1) \le \Gamma_{2r-1}(r(k-1)+1).$$

Example 7.8. Taking r = 2 and using Theorems 7.4–7.6, we obtain

(7.2)
$$\frac{1}{2}\Gamma_2(k) \le \Omega_2(k) \le w(k;2) \le \Omega_2(2k-1) \le \Gamma_3(2k-1).$$

In particular, for k = 3, the leftmost two inequalities agree with our previous calculations of w(3;2) = 9, $\Omega_2(3) = 7$ (Example 7.3), and $\Gamma_2(3) = 5$ (Example 3.28), while the other two inequalities tell us that $9 \leq \Omega_2(5) \leq \Gamma_3(5)$.

7.2. Monochromatic Sets a(S+b)

185

The above series of theorems, although providing a good picture of how the different functions are interrelated, does not actually provide any specific upper or lower bound for any of these functions. We next mention, without proof, a lower bound for $\Gamma_r(k)$ due to Alon and Zaks. Note that by (7.2) this also gives a lower bound for $\Omega_2(k)$

Theorem 7.9. For every $r \ge 2$, there exists a constant c > 0 (dependent upon r) such that

$$\Gamma_r(k) > r^{k-c\sqrt{k}}$$

for all $k \geq 3$.

7.2. Monochromatic Sets a(S+b)

In Chapter 2 we gave several equivalent forms of van der Waerden's theorem. One of these is the following: for every *r*-coloring of \mathbb{Z}^+ and every finite $S \subseteq \mathbb{Z}^+$, there exist integers $a, b \ge 1$ such that aS + b is monochromatic. We can think of aS + b as being derived from *S* via the operations, in order, of multiplication and addition. What happens if we reverse the order of these two operations? More explicitly, is it true that for every finite *S* and every *r*-coloring of \mathbb{Z}^+ , there must be a monochromatic set of the form a(S+b)? As we shall see, the answer is no. This suggests the question: for which *S* and *r* does this Ramsey property hold?

We adopt the following terminology and notation.

Definition 7.10. For S a finite set of positive integers and $r \ge 2$, we say that S is *reverse* r-regular if for every r-coloring of \mathbb{Z}^+ there exist $a \ge 1$ and $b \ge 0$ such that a(S + b) is monochromatic. We say S is reverse regular is S is reverse r-regular for all $r \ge 2$.

Notation. If S is reverse r-regular, denote by RR(S;r) the least positive integer m such that for every r-coloring of [1, m] there is a monochromatic set a(S + b) for some $a \ge 1$ and $b \ge 0$.

Interestingly enough, the regularity properties of sets of the form a(S+b) are far different from those of the form aS+b. As mentioned above, by van der Waerden's theorem, the latter type are always regular. The following theorem describes, quite plainly, which sets

are reverse regular. We omit the proof, but we do prove Theorem 7.12, below, which has the "if" part of Theorem 7.11 as an immediate corollary.

Theorem 7.11. A set S of positive integers is reverse regular if and only if $|S| \leq 2$.

It is obvious that every 1-element set is reverse regular. For sets S such that |S| = 2, it is not difficult to give a formula for RR(S) = RR(S; 2).

Theorem 7.12. Let $s, t \ge 1$ with s < t. Then $RR(\{s, t\}; 2) = 2t$.

Proof. For convenience we will denote $RR(\{s,t\};2)$ more simply by RR(s,t). To show that RR(s,t) > 2t-1, let d = t-s and consider the 2-coloring χ of [1, 2t-1] defined by $\chi(i) = 1$ if $\lfloor \frac{i}{d} \rfloor$ is odd and $\chi(i) = 0$ if $\lfloor \frac{i}{d} \rfloor$ is even. Clearly, if $a(S+b) \subseteq [1, 2t-1]$ with $b \ge 0$, then a = 1 and $b \le t-1$. So it suffices to show that there is no $0 \le b \le t-1$ such that S+b is monochromatic. Assume, by way of contradiction, that $\{s+b,t+b\}$ is monochromatic. Then $\lfloor \frac{s+b}{t-s} \rfloor$ and $\lfloor \frac{t+b}{t-s} \rfloor$ have the same parity. That is, $\lfloor \frac{s+b}{d} \rfloor$ and $\lfloor \frac{s+d+b}{d} \rfloor$ have the same parity. That is, $\lfloor \frac{s+b}{d} \rfloor + 1$, we have arrived at a contradiction.

To show that $RR(s,t) \leq 2t$, consider any 2-coloring of [1, 2t]. At least two elements of $\{2s, s+t, 2t\}$ must have the same color. We consider three cases, depending on which two of these elements are monochromatic. If 2s and s+t have the same color, then taking b = s and a = 1, we have $a(\{s, t\} + b)$ monochromatic. We leave the other two cases as Exercise 7.4.

Although the only reverse regular sets (those that are reverse r-regular for every r) are those with two or less elements, the next theorem tells us that for any finite set S and any positive integer r, there exists some positive integer c such that cS is reverse r-regular. That is to say, for every finite S, for arbitrarily large r there is a multiple of S that is reverse r-regular (which multiple of S this is, however, depends on r).

7.3. Having Most Elements Monochromatic

Theorem 7.13. Let S be a finite set of positive integers, and let $r \ge 1$. Then there exists a positive integer c (depending on S and r) such that cS is reverse r-regular.

Proof. Let S and r be fixed. From the equivalent form of van der Waerden's theorem mentioned in the first paragraph of this section, by using the compactness principle, we know there is a positive integer m such that for every r-coloring of [1, m] there is a monochromatic set of the form aS + b, $a \ge 1$, $b \ge 0$. Let $c = \operatorname{lcm}\{1, 2, \ldots, m\}$.

Let χ be any *r*-coloring of \mathbb{Z}^+ . We wish to show that there exist $a' \geq 1$ and $b' \geq 0$ such that a'(cS + b') is monochromatic under χ . Consider the *r*-coloring α of [1, m] defined by $\alpha(i) = \chi(ci)$. We know that there exist $a > 0, b \geq 0$ such that aS + b is monochromatic under α . Therefore, caS + cb is monochromatic under χ . Clearly, $a \leq m$, and hence *a* divides *c*. Let c = da. Then caS + dab = a(cS + db) is monochromatic under χ . Hence, letting a' = a and b' = db, the proof is complete.

7.3. Having Most Elements Monochromatic

In this section we look at a modification of van der Waerden's theorem by loosening the requirement that there is a k-term arithmetic progression having *all* of its terms be of the same color. What if we instead require only that, in any 2-coloring, there be an arithmetic progression that is predominantly of one color? We formalize this idea in the following definition.

Definition 7.14. Let $k \ge 1$ and let $0 \le j < k$. Let $w^*(k, j)$ be the least positive integer w such that for every $\chi : [1, w] \to \{0, 1\}$ there is a *k*-term arithmetic progression with the property that the number of elements of color 0 and the number of elements of color 1 differ by more than j.

It is clear from van der Waerden's theorem that $w^*(k,j)$ always exists.

Example 7.15. If j = k - 1, then $w^*(k, j)$ has the same meaning as w(k), since there must be k elements of one color and none of the other color. At the opposite extreme, if k is any odd positive integer,

188 7. Other Variations on van der Waerden's Theorem

then $w^*(k,0) = k$, because in any 2-coloring of [1,k] (which itself is a k-term arithmetic progression), the number of integers of one color must exceed the number of the other color.

We saw in Example 7.15 that the case in which k is odd and j = 0 is rather trivial. This is not so when k is even and j = 0.

Theorem 7.16. If $k \ge 2$ is even, then $w^*(k,0) = 2^j(k-1)+1$, where j is the largest positive integer such that 2^j divides k.

Proof. Let $m = 2^{j}(k-1) + 1$, and let $\chi : [1,m] \to \{1,-1\}$ be an arbitrary 2-coloring of [1,m]. To show that $w^{*}(k,0) \leq m$, assume, for a contradiction, that, under χ , every k-term arithmetic progression has exactly $\frac{k}{2}$ of its elements in each color. In particular, for each $a \in [1, m-k]$,

(7.3)
$$\sum_{i=0}^{k-1} \chi(a+i) = 0$$

 and

(7.4)

 $\sum_{i=1}^{n} \chi(a+i) = 0.$

Likewise,

(7.5)
$$\sum_{i=0}^{k-1} \chi \left(1 + i 2^{j} \right) = 0.$$

Note that from (7.3) and (7.4) we have

(7.6)
$$\chi(a) = \chi(a+k) \text{ for all } a \in [1, m-k]$$

We next show that (7.5) and (7.6) lead to a contradiction.

7.3. Having Most Elements Monochromatic

Let $q = \frac{k}{2^j}$. Note that from the meaning of j, q must be odd. Since $k - 1 = q2^j - 1$, we have

$$\sum_{i=0}^{j-1} \chi \left(1 + i2^{j} \right) = \sum_{s=0}^{2^{j}-1} \left(\sum_{i=sq}^{(s+1)q-1} \chi \left(1 + i2^{j} \right) \right)$$
$$= \sum_{s=0}^{2^{j}-1} \left(\sum_{i=0}^{q-1} \chi (1 + (i+sq)2^{j}) \right)$$
$$= \sum_{s=0}^{2^{j}-1} \left(\sum_{i=0}^{q-1} \chi \left(1 + i2^{j} + sk \right) \right).$$

By (7.6), for all s and i such that $0 \le i \le q-1$ and $0 \le s \le 2^j-2$, we have

$$\chi\left(1+i2^{j}+sk\right)=\chi\left(1+i2^{j}+(s+1)k\right).$$

Therefore

(7.7)
$$\sum_{i=0}^{k-1} \chi \left(1 + i2^{j} \right) = 2^{j} \sum_{i=0}^{q-1} \chi \left(1 + i2^{j} \right).$$

From (7.5) and (7.7), we obtain

$$\sum_{i=0}^{q-1} \chi \left(1 + i 2^j \right) = 0,$$

which gives us a contradiction because q is odd.

We still must show that $w^*(k,0) \ge 2^j(k-1)+1$. It suffices to find one 2-coloring $\lambda : [1, m-1] \to \{-1, 1\}$ such that for every arithmetic progression $\{a, a + d, \ldots, a + (k-1)d\} \subseteq [1, m-1]$, we have

$$\sum_{i=1}^{k} \lambda(a + (i-1)d) = 0$$

Define λ as follows. Let $\lambda(x) = 1$ for $1 \le x \le \frac{k}{2}$, and let $\lambda(x) = -1$ for $\frac{k}{2} < x \le k$. Finally, for x > k, let $\lambda(x) = \lambda(\overline{x})$, where $\overline{x} \in [1, k]$ and $x \equiv \overline{x} \pmod{k}$.

Let $\{a + (i-1)d : 1 \leq i \leq k\}$ be any arithmetic progression contained in [1, m-1]. Let $e = \gcd(d, k)$ and $k' = \frac{k}{e}$. Since $d < 2^{j}$,

190 7. Other Variations on van der Waerden's Theorem

k' is even. Therefore, by the way λ is defined and the fact that k'd is a multiple of k,

$$\begin{split} \sum_{i=1}^{k} \lambda(a+(i-1)d) &= \sum_{s=0}^{e-1} \left(\sum_{i=sk'+1}^{(s+1)k'} \lambda(a+(i-1)d) \right) \\ &= e \sum_{i=1}^{k'} \lambda(a+(i-1)d) \\ &= e \sum_{i=1}^{k'_2} \left(\lambda(a+(i-1)d) + \lambda \left(a+(i-1)d + \frac{k'd}{2}\right) \right) \\ &= 0, \end{split}$$

because $\frac{k'd}{2} \equiv \frac{k}{2} \pmod{k}$. This completes the proof.

Although very little is known about $w^*(k, j)$ for $j \ge 1$, progress has been made on a related question. Before stating the question, we adopt the following notation. For a 2-coloring $\chi : [1, n] \to \{1, -1\}$, denote by $[1, n]_1$ and $[1, n]_{-1}$ the sets of elements of [1, n] having the colors 1 and -1, respectively. The question we wish to ask may now be stated this way: for n a positive integer, can we determine some minimum number j, depending on n, such that for every 2-coloring $\chi : [1, n] \to \{1, -1\}$, there must be some arithmetic progression Aso that $A \cap [1, n]_1$ and $A \cap [1, n]_{-1}$ differ in size by at least j? The following theorem gives an asymptotic lower bound for such a j as a function of n. We omit the proof, which is beyond the scope of this book.

Theorem 7.17. Let

$$j(n) = \min_{\chi} \max_{a,d,k} \left\{ \left| \sum_{i=0}^{k} \chi(a+id) \right| : a, d, k \in \mathbb{Z}^+ \text{ and } a+kd \le n \right\},\$$

where the minimum is taken over all 2-colorings $\chi : [1, n] \rightarrow \{1, -1\}$. Then $j(n) \ge O(n^{1/4})$.

Example 7.18. To help get some insight into what the above theorem is saying, consider j(12). For a given coloring of [1, 12], we want the most "unbalanced" arithmetic progression A, in the sense that the difference in the sizes of the sets $A \cap [1, 12]_1$ and $A \cap [1, 12]_{-1}$ is

7.4. Permutations Avoiding Arithmetic Progressions 191

maximized. For example, let χ^* be the coloring such that $[1, 12]_1 = \{3, 4, 5, 6, 8, 9, 10\}$ and $[1, 12]_{-1} = \{1, 2, 7, 11, 12\}$. Then it is easy to check that the arithmetic progression $A = \{3, 4, 5, 6, 7, 8, 9, 10\}$ is the arithmetic progression we seek because the difference between the sizes of $A \cap [1, 12]_1$ and $A \cap [1, 12]_{-1}$ is six, and no other arithmetic progression yields as great a difference (check this). Using the notation of Theorem 7.17, this may be stated (equivalently) as

$$\max_{a,d,k} \left\{ \left| \sum_{i=0}^{k} \chi^{\star}(a+id) \right| : a+kd \le 12 \right\} = 6$$

Denoting this maximum by $M(\chi^*)$, we have $M(\chi^*) = 6$, and this is obtained with a = 3, d = 1, and k = 7 (it is, of course, possible that there is more than one arithmetic progression that gives the maximum). Now, if we do this for each 2-coloring χ of [1,12], we get its associated maximum $M(\chi)$. Then by taking the minimum, j(12), over all 2-colorings χ , we are saying that, for each coloring χ , there will always be some arithmetic progression A in [1,12] having the property that the difference between the number of members of A with color 1 and the number of members of A with color -1 is at least j(12). We leave it to the reader (Exercise 7.7) to determine the value of j(12).

7.4. Permutations Avoiding Arithmetic Progressions

In this section we are concerned with the question of whether permutations (i.e., arrangements) of certain sequences (possibly infinite) contain subsequences of a desired length that form an arithmetic progression. To clarify, we begin with a definition.

Definition 7.19. A sequence (a_1, a_2, \ldots, a_n) or (a_1, a_2, \ldots) has a *k*-term monotone arithmetic progression if there is a set of indices $\{i_1 < i_2 < \cdots < i_k\}$ such that the subsequence $a_{i_1}, a_{i_2}, \ldots, a_{i_k}$ is either an increasing or a decreasing arithmetic progression.

Example 7.20. The sequence (3, 1, 4, 5) contains a 3-term monotone arithmetic progression (namely, 3, 4, 5), but none of four terms. The sequence (4, 2, 3, 1) has no 3-term monotone arithmetic progression.

The sequence (1, 9, 7, 6, 2, 4, 5, 3, 8) has a 4-term monotone arithmetic progression (the subsequence 9, 7, 5, 3 is one), but none of length 5.

Here is one simple question: are there any positive integers n such that *every* permutation of the sequence (1, 2, ..., n) has a 3-term monotone arithmetic progression? This is relatively easy to answer.

Theorem 7.21. Let $n \ge 1$. There is a permutation of (1, 2, ..., n) that does not contain a 3-term monotone arithmetic progression.

Proof. First note that if $1 \leq n_1 < n_2$, and if the theorem holds for n_2 , then it also holds for n_1 (simply take the permutation that works for n_2 , and delete those integers greater than n_1). Hence, it is sufficient to prove the theorem for $n \in \{2^k : k \geq 0\}$. We do this by induction on k.

Clearly, the result holds for $n = 2^0$. Now assume that $k \ge 0$, that $n = 2^k$, and that there is a permutation (c_1, c_2, \ldots, c_n) of $(1, 2, \ldots, n)$ that does not contain a 3-term monotone arithmetic progression. Consider the sequence

 $S = (2c_1 - 1, 2c_2 - 1, \dots, 2c_n - 1, 2c_1, 2c_2, \dots, 2c_n).$

Note that S is a permutation of the sequence $(1, 2, ..., 2n = 2^{k+1})$, and that the first 2^k terms of S are odd and the other 2^k terms of S are even.

In any 3-term monotone arithmetic progression, the first and third terms must be of the same parity. Therefore, if S contains a 3-term monotone arithmetic progression P, then either

$$P \subseteq \{2c_1 - 1, 2c_2 - 1, \dots, 2c_n - 1\} = S_1$$

or

 $P \subseteq \{2c_1, 2c_2, \dots, 2c_n\} = S_2.$

This is impossible, for if $\{a, b, c\} \subseteq [1, n]$ is not a monotone arithmetic progression, then neither is $\{2a - 1, 2b - 1, 2c - 1\}$ nor $\{2a, 2b, 2c\}$. Therefore, S is a permutation of $\{1, 2, \ldots, 2^{k+1}\}$ that contains no 3-term monotone arithmetic progression, and the proof is complete. \Box

Now that we know that for each n there is some permutation of (1, 2, ..., n) containing no 3-term monotone arithmetic progression, a

7.4. Permutations Avoiding Arithmetic Progressions 193

patural question to ask is: how many such permutations are there? Clearly, for $n \ge 3$, not all n! permutations of [1, n] avoid 3-term monotone arithmetic progressions (for example, (1, 2, ..., n)). Let us denote by $\theta(n)$ the number of permutations of (1, 2, ..., n) containing no 3-term monotone arithmetic progressions. No formula for $\theta(n)$ is known. However, both upper and lower bounds are known for $\theta(n)$. The following theorem gives a lower bound.

Theorem 7.22. For all $n \ge 1$, $\theta(n) \ge 2^{n-1}$.

Proof. We use induction on n. First note that $\theta(1) = 1$, $\theta(2) = 2$, and $\theta(3) = 4$, so the inequality holds for $n \leq 3$. Now assume $n \geq 4$, and that it holds for any n' < n. We consider two cases.

Case 1. n = 2m is even. By the reasoning used in the proof of Theorem 7.21, if S_1 and S_2 are any permutations of [1, m] that avoid **3**-term monotone arithmetic progressions, then $((2S_1 - 1), 2S_2)$ and $(2S_2, (2S_1-1))$ are permutations of [1, n] that also avoid 3-term monotone arithmetic progressions. Therefore,

$$\theta(2m) \ge (\theta(m))^2 + (\theta(m))^2 = 2(\theta(m))^2 \ge 2(2^{m-1})^2 = 2^{n-1}.$$

Case 2. n = 2m + 1 is odd. As in Case 1, if S_1 and S_2 are permutations of [1, m] and [1, m+1], respectively, that avoid 3-term monotone arithmetic progressions, then $(2S_1, (2S_2 - 1))$ and $((2S_2 - 1), 2S_1)$ are permutations of [1, n] that do likewise. Therefore,

$$\theta(2m+1) \ge 2\theta(m)\theta(m+1).$$

From the induction hypothesis, if n = 2m, then

$$\theta(n) \ge 2(2^{m-1})^2 = 2^{n-1},$$

and if n = 2m + 1, then

$$\theta(n) \ge 2(2^{m-1})(2^m) = 2^{n-1}.$$

We now turn to upper bounds for $\theta(n)$.

Theorem 7.23. Let $n \ge 1$. If n = 2m - 1, then $\theta(n) \le (m!)^2$. If n = 2m, then $\theta(n) \le (m + 1)(m!)^2$.

Proof. Denote by $\Theta(t)$ the set of permutations of [1, t] that avoid 3-term monotone arithmetic progressions. Note that each member S' of $\Theta(n + 1)$ may be obtained from some member S of $\Theta(n)$ by the insertion of n + 1 somewhere into $S = (a_1, a_2, \ldots, a_n)$. If a_i is such that

(7.8)
$$\left\lceil \frac{n+3}{2} \right\rceil \le a_i \le n,$$

then $\{n + 1, a_i, 2a_i - n - 1\}$ is an arithmetic progression that, from the meaning of $\Theta(n + 1)$, cannot occur as a monotone progression in S'. Therefore, for each a_i that satisfies (7.8), S' cannot have n + 1 to the immediate right of a_i if $2a_i - n - 1$ is to the left of a_i . Likewise, n + 1 cannot be to the immediate left of a_i if $2a_i - n - 1$ is to the right of a_i .

Now, if i < n and if n + 1 is allowed neither to the right of a_i nor to the left of a_{i+1} , S cannot be extended to an element of $\Theta(n + 1)$. Similarly, S cannot be extended if i = 1 and n + 1 is not allowed to be placed to the left of a_i ; and S cannot be extended if i = n and n + 1is not allowed to be placed to the right of a_i . Therefore, each of the $n - \lfloor \frac{n+3}{2} \rfloor + 1$ values of a_i satisfying (7.8) eliminates at least one of the n + 1 positions in S' as a possible location for n + 1. Subtracting this quantity from n + 1 leaves us with at most $\lfloor \frac{n+3}{2} \rfloor$ positions where n + 1 may be placed. Hence

(7.9) $\theta(n+1) \le \left\lfloor \frac{n+3}{2} \right\rfloor \theta(n).$

By a straightforward induction argument, using the fact that $\theta(1) = 1$ and $\theta(2) = 2$, (7.9) implies the theorem. We leave the details to the reader as Exercise 7.9.

We now turn our attention to permutations of \mathbb{Z}^+ , i.e., of the infinite sequence (1, 2, 3, ...). For example, S = (2, 1, 4, 3, 6, 5, 8, 7, ...) and (12, 1, 2, 3, 4, ...) are such permutations. Theorem 7.21 tells us that not all permutations of the interval [1, n] yield 3-term monotone arithmetic progressions. Is this also the case for the permutations of \mathbb{Z}^+ ? The following theorem tells us the answer.

Theorem 7.24. There is no permutation of \mathbb{Z}^+ that avoids 3-term monotone arithmetic progressions.

7.5. Exercises

Proof. Let $S = (a_1, a_2, ...)$ be any permutation of \mathbb{Z}^+ . Let j be the least positive integer such that $a_j > a_1$. Consider $a_k = 2a_j - a_1$. Since $a_j > a_1$, we have $a_k > a_j$. Also, $a_k - a_j = a_j - a_1$, so that $A = \{a_1, a_j, a_k\}$ is an arithmetic progression. By the definition of j, we know that k > j, and therefore A is a monotone arithmetic progression in S.

As for guaranteeing monotone arithmetic progressions of more than three terms in the permutations of \mathbb{Z}^+ , the following theorem is known, the proof of which we omit.

Theorem 7.25. There exist permutations of \mathbb{Z}^+ that avoid 5-term monotone arithmetic progressions

An intriguing question, whose answer is still unknown, is whether there are any permutations of \mathbb{Z}^+ that avoid 4-term monotone arithmetic progressions.

7.5. Exercises

- **7.1** Determine the value of $\Omega_3(3)$.
- **7.2** Prove Theorem 7.6.
- **7.3** Verify, by direct computation, the observation in Example 7.8 that $9 \leq \Omega_2(5) \leq \Gamma_3(5)$.
- 7.4 Complete the proof of Theorem 7.12 by considering the two remaining ways in which $\{2s, s+t, 2t\}$ may be colored.
- **7.5** What is the value of $w^*(3,2)$? of $w^*(3,1)$? How do these examples generalize to arbitrary values of k (the length of the arithmetic progression)?
- **7.6** Determine the value of $w^*(4, 1)$.
- **7.7** Determine the value of j(12) (see Example 7.18).
- **7.8** Find $\theta(4)$ and $\theta(5)$.
- **7.9** Complete the proof of Theorem 7.23 by using induction to show that (7.9) implies the statement of the theorem.

196 7. Other Variations on van der Waerden's Theorem

7.6. Research Problems

- 7.1 Improve the lower bound of Theorem 7.9. References: [53], [63]
- 7.2 Run a computer program to calculate values of $\Gamma_2(k)$. Try to find an upper bound on $\Gamma_2(k)$. References: [53], [63], [195], [209]
- **7.3** Run a computer program to calculate values of $\Gamma_3(k)$. Along the lines of Theorem 7.9, try to find a lower bound for $\Gamma_3(k)$. References: [53], [63], [195], [209]
- **7.4** Is it true that $RR(S, ; r) < \infty$ if and only if, for every prime p, every r-coloring of [1, p 1], every $a \ge 1$, and every $b \ge 0$, $a(S + b) \pmod{p}$ is monochromatic? Reference: [72]
- **7.5** Characterize those pairs (S, r) for which RR(S; r) exists. Reference: [72]
- **7.6** Find an upper bound and/or a lower bound on $w^*(k, 1)$. A computer program to calculate values may give interesting results.

References: [23], [92], [99], [257]

- 7.7 Determine if $\lim_{k\to\infty} [w^*(k,1)]^{1/k} < \infty$. References: [23], [92], [99], [257]
- **7.8** Determine if $\lim_{k\to\infty} [w^*(k,\sqrt{k})]^{1/k} < \infty$. References: [23], [92], [99], [257]
- 7.9 Improve on Theorem 7.17, i.e., find an improved lower bound. References: [84], [92], [99], [228], [257]
- **7.10** Improve the best known upper bound for j(n) (see the References section below). References: [99], [257]
- 7.11 Determine if there exists a permutation of Z⁺ that contains no 4-term monotone arithmetic progressions.
 References: [43], [44], [76], [200], [223], [244]
- **7.12** Does there exist c > 0 such that for all $n \in \mathbb{Z}^+$, $[\theta(n)]^{1/n} \leq c$? Does $\lim_{n\to\infty} [\theta(n)]^{1/n}$ exist? Reference: [**76**]

7.7. References

- **7.13** Investigate the question of avoiding monotone arithmetic progressions by permutations of \mathbb{Z} (rather than \mathbb{Z}^+). Reference: [76]
- 7.14 Determine if it is possible to partition Z⁺ into two sets, each of which can be permuted to avoid 3-term monotone arithmetic progressions. What if we replace Z⁺ with Z? Reference: [76]
- * 7.15 Let $k \ge 1$. Let $\{a_1 < a_2 < \cdots\}$ be a sequence of positive integers such that $a_{i+1} a_i \le k$ for all $i \ge 1$. Must there be a 3-term arithmetic progression a_x, a_y, a_z such that x, y, z is also an arithmetic progression? This is known to be true for $k \le 4$.

7.7. References

§7.1. Theorems 7.4–7.7 are due to Nathanson [195]. Earlier, Rabung [209] showed that the existence of $\Gamma_r(k)$ for all $r, k \ge 1$ is equivalent to van der Waerden's theorem. Theorem 7.9 is due to Alon and Zaks [16].

§7.2. Theorems 7.11–7.13 are from [72], which also includes more on reverse regular sets.

§7.3. Theorem 7.16 is due to Spencer [260]. Roth [228] proved Theorem 7.17. In the other direction, Spencer [257] showed that $j(n) < c\sqrt{n \frac{\ln \ln n}{\ln n}}$ for some constant c. Further work on j(n) may be found in [22], [99], [234], [235], [236], [237]. Valko [269] generalizes the known upper and lower bounds on j(n) to higher dimensions. Some other results related to j(n) are found in [84], [88], [238].

§7.4 The proof of Theorem 7.21 is from [202]. Theorems 7.22–7.25 and their proofs appear in [76]. Sidorenko [250] gives a permutation of \mathbb{Z}^+ in which there are no 3 terms such that both their values and positions form arithmetic progressions. Modular analogs of some of this work are consider in [196].

Additional References: This chapter covered only a few of the many interesting problems related to van der Waerden's theorem that have been considered. A stronger form of van der Waerden's theorem states that for every *r*-coloring of \mathbb{Z}^+ , for each *k* there is a monochromatic arithmetic progression $A = \{a + id : 0 \le i \le k - 1\}$ such that *d* has the same color as *A*. For proofs of this theorem and additional information, see [3], [51], [127], [215]. A related result of Graham, Spencer, and Witsenhausen [129] determines how large a subset of [1, n] can be, that fails to contain both a *k*-term arithmetic progression and its gap *d* (the answer is a function of *n* and *k*).

Another question is this: what is the maximum number s of subsets C_1, C_2, \ldots, C_s of [1, n] such that for all $i \neq j$, $C_i \cap C_j$ is an arithmetic progression? If we include the empty set as an arithmetic progression, then it is known that

$$s \le \left(\begin{array}{c}n\\3\end{array}\right) + \left(\begin{array}{c}n\\2\end{array}\right) + \left(\begin{array}{c}n\\1\end{array}\right) + 1$$

and that this is the best possible upper bound on s [128]. If we do not allow the empty set to be an arithmetic progression, then it is known that there is a constant c such that $s < cn^2$ [252]. For other interesting variations involving arithmetic progressions, see [18], [60], and [205].

Chapter 8

Schur's Theorem

Until now we have devoted most of our study to Ramsey-type theorems dealing with variations of van der Waerden's theorem. There are many other interesting aspects of Ramsey theory on the integers that we may explore, and we do so in the remaining chapters. We begin with a result that came before van der Waerden's theorem: Schur's theorem.

Van der Waerden's theorem proves the existence, in particular, of w(3; r). In other words, any *r*-coloring of \mathbb{Z}^+ must admit a monochromatic 3-term arithmetic progression $\{a, a+d, a+2d\}$ for some $a, d \ge 1$. Letting x = a, y = a + 2d, and z = a + d, this may also be described as a monochromatic solution to x + y = 2z, where $x, y, z \in \mathbb{Z}^+$ and $x \ne y$. Since x + y = 2z is the equation of a plane, it is natural to ask other questions about coloring points in a plane.

Consider the equation of the simple plane z = x + y. Let P be the set of the points in this plane whose coordinates are positive integers. Thus, for example, (1, 1, 2) and (3, 4, 7) are in P (note that we are not insisting that x and y be distinct). Next, using any finite set of colors, assign a color to each positive integer.

Now, for each $(a, b, c) \in P$, perform the following. If the colors of a, b, and c are identical, then color (a, b, c) (in the plane) with that color. Otherwise, mark (a, b, c) (in the plane) with an X. The question is: can all of the points in the plane be marked with an X,
or must there be a colored point? We would like to know whether *every* coloring yields a colored point; in other words, is it possible to finitely color \mathbb{Z}^+ so that no point $(x, y, z) \in P$ is colored?

This question was answered by Issai Schur in 1916, and is one of the first Ramsey-type theorems. However, Erdős and Szekeres' rediscovery of Ramsey's theorem in 1935 is due most of the credit for popularizing the subject.

Schur was not motivated by the idea of coloring points in the plane, but rather by perhaps the most famous and elusive of all mathematical problems, Fermat's Last Theorem (which was not officially a theorem until Wiles proved it in 1995). His result, which states that the answer to the above question is "no, there must be a colored point," has become known as Schur's theorem, but was only used as a lemma for the main theorem in his paper, which is given below as Theorem 8.1. We will prove this theorem after we have acquired a necessary tool.

Theorem 8.1. Let $n \ge 1$. There exists a prime q such that for all primes $p \ge q$ the congruence $x^n + y^n \equiv z^n \pmod{p}$ has a solution in the integers with $xyz \not\equiv 0 \pmod{p}$.

From Theorem 8.1 we can garner some insight into why Fermat's Last Theorem was so difficult to prove, and why no elementary proof (if one exists) has surfaced yet: Fermat's Last Theorem is false if we replace the equation by a congruence. Because of Schur's result, we know that we cannot prove Fermat's Last Theorem just by considering congruences.

8.1. The Basic Theorem

We now state and prove Schur's theorem. We make use of Ramsey's theorem (see Section 1.2). We remind the reader of the following terminology.

Given a coloring χ of a set of positive integers and an equation \mathcal{E} in the variables x_1, x_2, \ldots, x_n , we say that \mathcal{E} has a monochromatic solution under χ if there exist values of x_1, x_2, \ldots, x_n that satisfy \mathcal{E} and that are monochromatic under χ .

Theorem 8.2 (Schur's Theorem). For any $r \ge 1$, there exists a least positive integer s = s(r) such that for any r-coloring of [1, s] there is a monochromatic solution to x + y = z.

Proof. Ramsey's theorem states, in particular, that for any $r \ge 1$ there exists an integer n = R(3; r) such that for any *r*-coloring of K_n (the complete graph on *n* vertices) there is a monochromatic triangle. We will use a specific coloring, described as follows. Number the vertices of K_n by $1, 2, \ldots, n$. Next, arbitrarily partition $\{1, 2, \ldots, n-1\}$ into *r* sets. In other words, randomly place each $x \in \{1, 2, \ldots, n-1\}$ into exactly one of the *r* sets. These sets will correspond to the *r* colors. Color the edge that connects vertices *i* and *j* according to the set of which |j-i| is a member. By Ramsey's theorem, a monochromatic triangle be a < b < c. Hence, b - a, c - b, and c - a are all the same color. To finish the proof, let x = b - a, y = c - b, and z = c - a, and notice that x + y = z.

Definition 8.3. We call the numbers that satisfy Schur's theorem the *Schur numbers* and denote them by s(r). (Some books define the Schur number as the maximal number m = m(r) such that there exists an *r*-coloring of [1, m] that avoids monochromatic solutions to x + y = z. We prefer to use the same type of notation as that of the van der Waerden numbers.)

Definition 8.4. A triple $\{x, y, z\}$ that satisfies x + y = z is called a *Schur triple*.

The only values known for the Schur numbers are for r = 1, 2, 3, 4: s(1) = 2, s(2) = 5, s(3) = 14, and s(4) = 45. In the next example we show that s(2) = 5.

Example 8.5. In Section 1.4 we showed that $s(2) \ge 5$ (color 1 and 4 red; color 2 and 3 blue). We now show that $s(2) \le 5$. Consider any 2-coloring of [1,5]. Without loss of generality we may assume that 1 is colored red. Assume, by way of contradiction, that there is no monochromatic Schur triple. Since 1 + 1 = 2, we must color 2 blue. Since 2 + 2 = 4, we must color 4 red. Since 1 + 4 = 5, we must color 5 blue. All that remains is to color 3. However, if 3 is red, then $\{1, 3, 4\}$

is a red Schur triple and, if 3 is blue, then $\{2,3,5\}$ is a blue Schur triple.

Having Theorem 8.2 under our belt, we are now in a position to prove Theorem 8.1. The proof uses elementary group theory.

Proof of Theorem 8.1. Let p > s(n) be a prime and let

$$\mathbb{Z}_p^* = \{1, 2, \dots, p-1\}$$

be the group (under multiplication) of nonzero residues modulo p. Let $S = \{x^n \pmod{p}: x \in \mathbb{Z}_p^*\}$. Notice that S is a subgroup of \mathbb{Z}_p^* . Hence, we can write \mathbb{Z}_p^* as a union of cosets $\mathbb{Z}_p^* = \bigcup_{i=1}^k a_i S$, where $k = \frac{n}{\gcd(n,p-1)}$.

Next, define a k-coloring of \mathbb{Z}_p^* by assigning the element $t \in \mathbb{Z}_p^*$ the color j if and only if $t \in a_j S$. Since $k \leq n$ and $p-1 \geq s(n)$, by Schur's theorem there exists a monochromatic triple $\{a, b, c\} \subseteq \mathbb{Z}_p^*$ such that a + b = c. That is, for some i, $1 \leq i \leq k$, there exist $a, b, c \in a_i S$ with a + b = c. Hence, there exist $x, y, z \in \mathbb{Z}_p^*$ such that $a_i x^n + a_i y^n \equiv a_i z^n \pmod{p}$. Multiplying through by a_i^{-1} completes the proof.

Although Schur's original proof of Theorem 8.2 did not involve Ramsey numbers (in fact, Ramsey's theorem was not proved until 1928, while Schur proved his result in 1916), it is clear from our proof that the Schur numbers and Ramsey numbers are related. In particular, we have the following corollary. (Recall that $R_r(3)$ denotes $R(3, 3, \ldots, 3)$, where we are using r colors.)

Corollary 8.6. For $r \ge 1$, $s(r) \le R_r(3) - 1$.

Proof. Using the coloring given in the proof of Theorem 8.2, we have a correspondence between edgewise colorings of K_n and colorings of [1, n - 1]. More explicitly, for any *r*-coloring of [1, n - 1] we derive an *r*-coloring of K_n by numbering the vertices of K_n and considering the differences between all pairs of vertices. By Ramsey's theorem, for $n = R_r(3)$ we are guaranteed a monochromatic triangle. By the definition of our coloring, this monochromatic triangle corresponds to a monochromatic Schur triple. Hence, if $n = R_r(3)$, we have $s(r) \le n-1 = R_r(3) - 1$.

Corollary 8.6 gives us an upper bound. However, the upper bound is not very useful unless we have an explicit upper bound for $R_r(3)$. Fortunately, we do.

Lemma 8.7. For $r \ge 1$, $R_r(3) \le 3r!$.

Proof. For r = 1 we have $R_1(3) = 3$, so that the given bound is true. We now assume that $r \ge 2$. For $1 \le i \le r$, denote by $R_r^i(3)$ the Ramsey number $R(3, 3, \ldots, 3, 2, 3, 3, \ldots, 3)$. We begin by showing

r-i

that

(8.1)
$$R_r(3) \le \sum_{i=1}^r R_r^i(3).$$

i-1

To prove (8.1) we will use the same method of proof as that employed to prove Theorem 1.15. Let $m = \sum_{i=1}^{r} R_r^i(3)$ and consider any rcoloring of the edges of K_m . Select one vertex, say v. Let the colors be $1, 2, \ldots, r$, and let $C_i, i = 1, \ldots, r$, denote the set of vertices connected to v by an edge of color i. By the pigeonhole principle there must exist j such that $|C_j| \ge R_r^j(3)$. Hence, the complete graph on C_j must contain either a K_2 of color j or a monochromatic triangle of color $c \in \{1, 2, \ldots, j-1, j+1, j+2 \ldots, r\}$. If it contains a monochromatic triangle, we are done. If it contains a K_2 of color j, then these two vertices, together with v, create a monochromatic triangle of color j, thereby proving (8.1).

Next, we show that for any $i \in \{1, 2, \ldots, r\}$,

8.2)
$$R_r^i(3) = R_{r-1}(3).$$

Let $n = R_r^i(3)$. Clearly, $n \ge R_{r-1}(3)$. It remains to show that $n \le R_{r-1}(3)$. By the definition of n, there must exist an r-coloring of the edges of K_{n-1} that avoids monochromatic triangles of color $c \in \{1, 2, \ldots, i-1, i+1, i+2 \ldots, r\}$ and avoids a K_2 of color i. Since we must avoid a K_2 of color i, no edge may have color i. Hence, we actually have an (r-1)-coloring of the edges of K_{n-1} that avoids

monochromatic triangles. Thus $R_{r-1}(3) \ge n$, and we conclude that (8.2) holds.

Using (8.2), we see that $\sum_{i=1}^{r} R_r^i(3) = rR_{r-1}(3)$. Hence, by (8.1), $R_r(3) \leq rR_{r-1}(3)$ for $r \geq 2$. By repeated application of this inequality and the fact that $R_2(3) = 6$, we get $R_r(3) \leq 3r!$.

From Corollary 8.6 and Lemma 8.7, we have the following result.

Theorem 8.8. For $r \ge 1$, $s(r) \le 3r! - 1$.

Now that we have an upper bound on the Schur numbers, we turn our attention to a lower bound.

Theorem 8.9. For $r \ge 1$, $s(r) \ge \frac{3^r + 1}{2}$.

Proof. Let $n \geq 1$ and assume $\chi : [1, n] \to \{1, 2, \ldots, r\}$ is an *r*-coloring of [1, n] that has no monochromatic Schur triple. Define an (r + 1)-coloring $\hat{\chi} : [1, 3n + 1] \to \{1, 2, \ldots, r + 1\}$ that extends χ as follows. For all $x \in [n + 1, 2n + 1]$, let $\hat{\chi}(x) = r + 1$; and for all $x \in [1, n] \cup [2n+2, 3n+1]$, let $\hat{\chi}(x) = \chi(y)$, where $x \equiv y \pmod{2n+1}$.

We now argue that [1, 3n + 1] contains no monochromatic Schur triple under $\hat{\chi}$. Let $\{x, y, z\}$ with $x \leq y$ be a Schur triple. First, consider the color r + 1. Since 2(n + 1) > 2n + 1, $\{x, y, z\}$ cannot be a monochromatic Schur triple of color r + 1. Now consider any color $j \neq r + 1$. Since $\hat{\chi}$ is identical to χ on [1, n], $\{x, y, z\} \subseteq [1, n]$ cannot be a monochromatic Schur triple of color j. Furthermore, since (2n + 2) + (2n + 2) = 4n + 4 > 3n + 1, it is not possible that $x, y \in [2n + 2, 3n + 1]$. Thus, [2n + 2, 3n + 1] does not contain a Schur triple of color j. Hence, any Schur triple of color j must have $x \in [1, n]$ and $y \in [2n + 2, 3n + 1]$.

However, if $\{x, y, z\}$ is such a Schur triple of color j, by taking $y', z' \in [1, n]$ with $y' \equiv y \pmod{(2n+1)}$ and $z' \equiv z \pmod{(2n+1)}$, we see that $\{x, y', z'\}$ is a Schur triple of color j contained in [1, n], a contradiction.

Thus, we have shown that if $s(r) \ge n+1$, then $s(r+1) \ge 3n+2$. Hence,

(8.3)
$$s(r+1) \ge 3s(r) - 1.$$

The proof is completed by induction on r. Clearly, $s(1) = 2 \ge \frac{3^1+1}{2}$. Now assume that $r \ge 1$ and that $s(r) \ge \frac{3^r+1}{2}$. Then, by (8.3), $s(r+1) \ge 3s(r) - 1 \ge 3\left(\frac{3^r+1}{2}\right) - 1 = \frac{3^{r+1}+1}{2}$.

Since we have the existence of at least one monochromatic Schur triple in any r-coloring of [1, s(r)], we must have an infinite number of Schur triples in any r-coloring of the natural numbers. To see this, note that if [1, n] contains a monochromatic Schur triple, then so does $k[1, n] = \{k, 2k, \ldots, nk\}$, for any positive integer k. This holds by the trivial observation that if x + y = z then kx + ky = kz. Now consider the sets $S_j = s(r)^j [1, s(r)]$ for $j = 0, 1, 2, \ldots$. For each j we have a monochromatic Schur triple that resides in S_j . Consequently, any r-coloring of the positive integers must contain an infinite number of monochromatic Schur triples.

With regards to the infinitude of monochromatic Schur triples, we consider the following problem: find the minimum number of monochromatic Schur triples that a 2-coloring of [1, n] must have. We will answer this question in Theorem 8.15, below. As is often the case with such questions, the answer is given asymptotically, and so we will be using the O(n) notation (see Section 1.5). To solve the above problem, we have need of the following notation and lemmas.

Notation. We denote by $\Delta_{\{a < b < c\}}$ a triangle on vertices a, b, c. We denote by $M_{\chi}(n)$ the number of monochromatic Schur triples under χ , where χ is a given 2-coloring of [1, n].

Lemma 8.10. In every edgewise 2-coloring of the complete graph K_n , there are at least $\frac{n^3}{24} + O(n^2)$ monochromatic triangles.

The proof of Lemma 8.10 is Exercise 1.10.

Lemma 8.11. Over all 2-colorings of [1, n], the minimum number of monochromatic Schur triples is $O(n^2)$, i.e.,

$$\min_{\chi}(M_{\chi}(n)) = O(n^2).$$

Proof. We note first that for any 2-coloring χ , $M_{\chi}(n) \leq cn^2 + O(n)$ for some positive constant c which is independent of χ (we leave this to the reader; see Exercise 8.2). To complete the proof, we use Lemma

8.1. The Basic Theorem

8. Schur's Theorem

8.10 to show that $M_{\chi}(n) \geq \frac{n^2}{48} + O(n)$ for any 2-coloring χ . Examining the proof of Schur's theorem, we see that there is a natural connection between triangles and Schur triples: with each triangle $\Delta_{\{a < b < c\}}$ we can associate the Schur triple $\{b - a, c - b, c - a\}$.

Define $a \oplus b = j$, where $a + b \equiv j \pmod{n}$ and $1 \leq j \leq n$.

Next, we notice that, for j = 1, ..., n, the triangles

 $\Delta_{\{a \oplus j < b \oplus j < c \oplus j\}}$ and $\Delta_{\{a \oplus j < (a+c-b) \oplus j < c \oplus j\}}$

have the same associated Schur triple. Consequently, each Schur triple corresponds to at most 2n triangles (why?). Combining this with Lemma 8.10, we have at least $\frac{n^2}{48} + O(n)$ monochromatic Schur triples, thereby proving the lemma.

Lemma 8.12. Let c > 0 and let n be sufficiently large. Let χ be a 2-coloring of [1, n + 1] and let $\bar{\chi}$ be χ restricted to [1, n]. If $M_{\bar{\chi}}(n) = cn^2 + O(n)$, then $M_{\chi}(n+1) = cn^2 + O(n)$. Hence, we may take n to be even when determining the rate of growth of $M_{\chi}(n)$.

Proof. The only monochromatic Schur triples counted in $M_{\chi}(n+1)$ that are not counted in $M_{\bar{\chi}}(n)$ are of the form $\{x, y, n+1\}$ with x + y = n + 1. There are only $\left\lceil \frac{n}{2} \right\rceil$ possibilities, since $\{x, y\}$ must belong to $\{\{1, n\}, \{2, n-1\}, \dots, \{\left\lfloor \frac{n+1}{2} \right\rfloor, n+1-\left\lfloor \frac{n+1}{2} \right\rfloor\}$. Hence,

$$0\leq M_\chi(n+1)-M_{ar\chi}(n)\leq rac{n}{2}+1,$$

Lemma 8.13. Let $\chi : [1, n] \to \{\text{red, blue}\}$ be described by R, the set of red integers under χ , and B, the set of blue integers under χ . Let N^+ be the set of non-monochromatic pairs $\{a, b\} \subseteq [1, n]$ such that a + b > n. Then

(8.4)
$$2M_{\chi}(n) = \binom{n}{2} - 2|R||B| + |N^+|.$$

so the conclusion of the lemma holds.

Proof. To justify (8.4), we will view the Schur triples as ordered sets, so that we consider (a, b, a+b) and (b, a, a+b) to be distinct (for $a \neq b$). Thus, we will actually count each Schur triple twice. This is the reason why $2M_{\chi}(n)$ is on the left-hand side of (8.4), instead of just $M_{\chi}(n)$.

First, note that $\binom{n}{2}$ counts all ordered Schur triples by choosing y and z in x + y = z. Next, let $a \in R$ and $b \in B$. Then the ordered Schur triples (|b - a|, a, b) and (a, b, a + b) are not monochromatic, and thus we want to remove these from the set of $\binom{n}{2}$ Schur triples. Hence, we subtract 2|R||B| from $\binom{n}{2}$. However, the ordered Schur triples (a, b, a + b) with a + b > n were not counted in $\binom{n}{2}$. Hence, since these were subtracted once in -2|R||B|, we need to add $|N^+|$. This establishes (8.4).

We are now in a position to give the asymptotic minimum number of monoch_zomatic Schur triples over all 2-colorings of [1, n]. We start with Theorem 8.14, which gives an upper bound on this minimum.

Theorem 8.14. Over all 2-colorings of [1, n], the minimum number of monochromatic Schur triples is at most $\frac{n^2}{22} + O(n)$.

Proof. It is sufficient to prove the result for those *n* that are multiples of 11, since if $11 \nmid n$ we may apply Lemma 8.12 at most 10 times to get the desired result. Hence, assume $11 \mid n$ and consider the 2-coloring of [1, n] defined by $R = \left[\frac{4n}{11}, \frac{10n}{11}\right]$ and $B = \left[1, \frac{4n}{11} - 1\right] \cup \left[\frac{10n}{11} + 1, n\right]$, where R is the set of red integers and B is the set of blue integers. To prove this theorem we will show that this coloring admits, asymptotically, only $\frac{n^2}{22} + O(n)$ monochromatic Schur triples. By letting n = 11k, our coloring is R = [4k, 10k] and $B = [1, 4k - 1] \cup [10k + 1, 11k]$.

We start by counting the red Schur triples. Thus, we will count the number of solutions to x + y = z in the interval [4k, 10k]. First, note that $z \ge 8k$. It follows that the red Schur triples are the triples

$$\left\{ \{4k+c, z-4k-c, z\} : 8k \le z \le 10k, 0 \le c \le \frac{z-8k}{2} \right\}$$

(So that we do not count any red Schur triple more than once, we require $4k + c \leq z - 4k - c$, i.e., $0 \leq c \leq \frac{z-8k}{2}$.) Now, by summing over all possible values for z we have

$$\sum_{k=8k}^{10k} \left(\left\lfloor \frac{z-8k}{2} \right\rfloor + 1 \right) = 2 \sum_{z=1}^{k} z + (k+1)$$
$$= k(k+1) + (k+1)$$
$$= k^2 + 2k + 1$$

red Schur triples.

2

8.1. The Basic Theorem

8. Schur's Theorem

Next, we count the blue Schur triples. The same analysis as above shows that within the interval [1, 4k - 1] there are $4k^2 + O(k)$ blue Schur triples. The details for this are left to the reader as Exercise 8.3. Furthermore, there is obviously no blue Schur triple within the interval [10k + 1, 11k]. However, we have not counted all of the blue Schur triples yet; for example, $\{1, 10k + 1, 10k + 2\}$ has not been counted.

In order to count the remaining blue Schur triples $\{x, y, z\}$, let $x \in [1, 4k - 1]$ and let $z \in [10k + 1, 11k]$. Notice that we must have $y \in [10k + 1, 11k]$ in order to have x + y = z. Writing x = 1 + c and y = z - c - 1 with $c \ge 0$, we must have $z - 1 - c \ge 10k + 1$, i.e., $c \le z - 10k - 2$. Hence, we wish to count the number of triples $\{1 + c, z - c - 1, z\}$ with $0 \le c \le z - 10k - 2$ and $z \in [10k + 1, 11k]$. Summing over all possible values of z, we have

$$\sum_{z=10k+2}^{11k} (z-10k-1) = \frac{1}{2} \left((11k)^2 - (10k)^2 \right) - 10k^2 + O(k)$$
$$= \frac{k^2}{2} + O(k)$$

remaining blue Schur triples.

Adding all cases together, we have

$$\frac{11}{2}k^2 + O(k) = \frac{11}{2}\left(\frac{n}{11}\right)^2 + O(n) = \frac{n^2}{22} + O(n)$$

monochromatic Schur triples.

We now show that the upper bound given in Theorem 8.14 is actually the minimum.

Theorem 8.15. Over all 2-colorings of [1, n], the minimum number of monochromatic Schur triples is $\frac{n^2}{22} + O(n)$.

Proof. Denote by R and B the sets of red integers and blue integers, respectively, in any red-blue coloring χ . Let N^+ be the set of non-monochromatic pairs $\{a, b\} \subseteq [1, n]$ such that a + b > n, and let N^- be the set of non-monochromatic pairs $\{a, b\} \subseteq [1, n]$ such that $a + b \leq n$.

From Lemma 8.13, we have $2M_{\chi}(n) = \binom{n}{2} - 2|R||B| + |N^+|$. Now, since |R| + |B| = n, we have

(8.5)
$$|R| = n(\frac{1}{2} + \alpha) \text{ and } |B| = n(\frac{1}{2} - \alpha)$$

for some $\alpha \in \left[-\frac{1}{2}, \frac{1}{2}\right]$. Notice that for α and n fixed, |R| and |B| are determined, so that by minimizing $|N^+|$ we are minimizing $M_{\chi}(n)$. Hence, our first goal is to minimize $|N^+|$ for α and n fixed. To this end, notice that

(8.6)
$$|N^+| + |N^-| = |R||B| = n^2 \left(\frac{1}{4} - \alpha^2\right)$$

Our next step is to find an upper bound for $|N^-| - |N^+|$. Such an upper bound, when coupled with (8.6), will give a lower bound for $|N^+|$, thereby bounding the number of monochromatic Schur triples from below. From Lemma 8.12, we may assume that n is even, so that to obtain an upper bound for $|N^-| - |N^+|$ we may decompose [1, n] into disjoint 2-element subsets $\{x, n + 1 - x\}, x = 1, 2, \ldots, \frac{n}{2}$.

For the pair of sets $X = \{a, n+1-a\}$ and $Y = \{b, n+1-b\}$ with $a \neq b$, we will count the contribution of the elements of these sets to N^+ and to N^- . In the following we will only consider x+y with $x \in X$ and $y \in Y$. Thus, we do not count pairs of the form $\{x, n+1-x\}$. The reason for this is that each such pair only contributes one to N^+ , and since the number of such pairs is $\frac{n}{2} = O(n)$, by Lemma 8.11 we may safely ignore these pairs in the asymptotic calculation.

We consider four cases. In each case, let

$$P = \{(x, y) : x \in X, y \in Y\}.$$

We will determine the number of pairs in P contributing to N^- and the number contributing to N^+ . By doing so, we will find a function f(n) such that $|N^-| - |N^+| \le f(n)$. Such a function, together with (8.6), will provide a lower bound for $|N^+|$.

Case 1. X and Y are monochromatic of different colors. These two sets produce four non-monochromatic pairs in P whose sums are a + b, n + 1 + a - b, n + 1 + b - a, and 2n + 2 - a - b. Clearly two of these sums exceed n and the other two are less than n. Consequently, $|P \cap N^+| = |P \cap N^-|$, i.e., the number of pairs in P contributing to N^+ is equal to the number of pairs in P contributing to N^- .

Case 2. X and Y are non-monochromatic with a and b being different colors. These sets produce two non-monochromatic pairs with sums a + b and 2n + 2 - a - b. Hence, $|P \cap N^+| = |P \cap N^-|$.

Case 3. X and Y are non-monochromatic with a and b being the same color. These sets produce two non-monochromatic pairs with sums n + 1 + b - a and n + 1 + a - b. Thus, $|P \cap N^+| = |P \cap N^-|$.

Case 4. X is monochromatic and Y is non-monochromatic with a and b different colors. These sets produce two non-monochromatic pairs with sums a + b and n + 1 + b - a. Since in our decomposition a + b < n, both of these sums are at most n. In this situation the contribution of these pairs to N^- is 2 greater than their contribution to N^+ , i.e., $|P \cap N^+| + 2 = |P \cap N^-|$.

Since our goal is to find some f(n) such that $|N^-| - |N^+| \le f(n)$, we note that only Case 4 gives us a situation where $|N^-|$ is larger than $|N^+|$ (so that we must have f(n) > 0 for n > 0). We now analyze Case 4 to determine f(n).

Let βn be the number of non-monochromatic sets $\{x, n+1-x\}$. Then exactly one of x, n+1-x is in R. Thus, using (8.5), we see that the number of monochromatic sets $\{x, n+1-x\}$ in R is $\frac{1}{2}\left(n\left(\frac{1}{2}+\alpha\right)\right)-\frac{\beta n}{2}=\left(\frac{1}{2}+\alpha-\beta\right)\frac{n}{2}$. Looking at the details of Case 4, we see that for every non-monochromatic set $\{b, n+1-b\}$, a monochromatic set $\{a, n+1-a\}$ with a and b the same color will have sums that contribute exactly 2 more to N^- than to N^+ . Therefore,

(8.7)
$$|N^{-}| - |N^{+}| = 2\beta n \left(\frac{1}{2} + \alpha - \beta\right) \left(\frac{n}{2}\right) + O(n)$$
$$\leq \frac{\left(\frac{1}{2} + \alpha\right)^{2}}{4} n^{2} + O(n),$$

where the second expression equals the third expression precisely when $\beta = \frac{1}{2} (\frac{1}{2} + \alpha)$. Thus, we have found an upper bound on $|N^{-}| - |N^{+}|$ as sought.

Combining (8.6) and (8.7), we get

(8.8)
$$|N^+| \ge \left(\frac{\frac{1}{4} - \alpha^2}{2} - \frac{\left(\frac{1}{2} + \alpha\right)^2}{8}\right) n^2 + O(n).$$

8.2. A Generalization of Schur's Theorem

Using (8.4) with (8.8), we have

8.9)
$$2M_{\chi}(n) \ge \left(\frac{11\left(2\alpha - \frac{1}{11}\right)^2}{32} + \frac{1}{11}\right)n^2 + O(n).$$

We can easily see that (8.9) is minimized when $\alpha = \frac{1}{22}$, giving $2M_{\chi}(n) \geq \frac{n^2}{11} + O(n)$, i.e., $M_{\chi}(n) \geq \frac{n^2}{22} + O(n)$.

The proof is complete, since the upper bound given in Theorem 8.14 matches this lower bound. $\hfill \Box$

Now that we see that the coloring in Theorem 8.14 attains the minimum, we remark that it can be shown that this coloring is essentially the only coloring that attains the minimum number of mono-chromatic Schur triples.

We have an extension of Theorem 8.14 that gives the following upper bound for the analogous r-colored question $(r \ge 2)$.

Corollary 8.16. Let $r \ge 2$. The minimum number of monochromatic Schur triples over all r-colorings of [1, n] is no greater than $\frac{n^2}{11 \cdot 2^{2r-3}} + O(n)$.

Proof. Extending the coloring defined in the proof of Theorem 8.14, we define, for $n \ge 2^r$, the coloring $\chi : [1, n] \to \{1, 2, \ldots, r\}$ by

$$\chi(i) = \begin{cases} j & \text{if } \frac{n}{2^j} < i \le \frac{n}{2^{j-1}} \text{ for } 1 \le j \le r-2, \\ r-1 & \text{if } 1 \le i \le \frac{4n}{2^{r-2}11} \text{ or } \frac{10n}{2^{r-2}11} < i \le \frac{n}{2^{r-2}}, \\ r & \text{if } \frac{4n}{2^{r-2}11} < i \le \frac{10n}{2^{r-2}11}. \end{cases}$$

The calculation showing that χ admits only $\frac{n^2}{11\cdot 2^{2r-3}} + O(n)$ monochromatic Schur triples is left to the reader as Exercise 8.6.

8.2. A Generalization of Schur's Theorem

We turn our attention to a generalization of Schur's theorem, given below as Theorem 8.17. Another, more powerful, generalization was done by Richard Rado, a student of Schur, and is investigated in Chapter 9.

It will be convenient to use the following notation.

8. Schur's Theorem

Notation. Let $\mathcal{L}(t)$ represent the equation $x_1 + x_2 + \cdots + x_{t-1} = x_t$, where x_1, x_2, \ldots, x_t are variables.

Theorem 8.17. Let $r \ge 1$ and, for $1 \le i \le r$, assume that $k_i \ge 3$. Then there exists a least positive integer $S = S(k_1, k_2, \ldots, k_r)$ such that for every r-coloring of [1, S] there is a solution to $\mathcal{L}(k_j)$ of color j for some $j \in \{1, 2, \ldots, r\}$.

The numbers $S(k_1, k_2, \ldots, k_r)$ are called the generalized Schur numbers. We denote the case $k_1 = k_2 = \cdots = k_r = k$ more simply by $S_r(k)$.

To fully understand Theorem 8.17, consider the following examples.

Example 8.18. Any red-blue-green coloring of [1, S(3, 4, 5)] must contain a solution to $\mathcal{L}(3)$ consisting of only red integers, or a solution to $\mathcal{L}(4)$ consisting of only blue integers, or a solution to $\mathcal{L}(5)$ consisting of only green integers. Theorem 8.17 tells us that S(3, 4, 5) is the least positive integer such that the above condition is met.

Example 8.19. If $k_i = 3$ for $1 \le i \le r$ we have $S(3, 3, \ldots, 3) = s(r)$, the *r*-colored Schur number we have already investigated. Considering $S_r(k) = S(k, k, \ldots, k)$, Theorem 8.17 states that any *r*-coloring of $[1, S_r(k)]$ must contain a monochromatic solution to $\mathcal{L}(k)$.

Example 8.20. We will determine S(4,5). To see that $S(4,5) \ge 14$, consider the 2-coloring of [1, 13] defined by $R = \{1, 2, 12, 13\}$ and B = [3, 11], where R is the set of red integers and B is the set of blue integers. It is easy to check that this coloring admits no red solution to $\mathcal{L}(4)$ and no blue solution to $\mathcal{L}(5)$.

To show that $S(4,5) \leq 14$, let R be the set of red integers and B be the set of blue integers in a given red-blue coloring of [1,14]. We must show that there is either a red solution to $\mathcal{L}(4)$ or a blue solution to $\mathcal{L}(5)$.

We proceed by contradiction, so assume that no red solution to $\mathcal{L}(4)$ and no blue solution to $\mathcal{L}(5)$ exist in the given coloring. First consider the case in which $1 \in R$. Since 1 + 1 + 1 = 3 we must have $3 \in B$. Next, since 3 + 3 + 3 = 12 we must have $12 \in R$. This implies that $10, 14 \in B$, for otherwise 1 + 1 + 10 = 12 or 1 + 1 + 12 = 14

8.2. A Generalization of Schur's Theorem

would be a red solution. Next, since 2+2+3+3=10, we must have $2 \in R$. This implies that we must have $5 \in B$ to avoid the red solution 1+2+2=5, which gives 3+3+3+5=14 as a blue solution, a contradiction. The case when $1 \in B$ is left to the reader in Exercise 8.7.

Just as the proof of Schur's theorem follows easily from Ramsey's theorem, so does the proof of Theorem 8.17.

Proof of Theorem 8.17. Let $n = R(k_1, k_2, ..., k_r)$ be the *r*-colored Ramsey number. Consider the same edgewise coloring as in the proof of Schur's theorem. That is, number the vertices of K_n by 1, 2, ..., n and arbitrarily partition the set $\{1, 2, ..., n-1\}$ into *r* subsets, with each of these subsets corresponding to a different color. Color the edge connecting vertices *i* and *j* according to the subset of which |j - i| is a member.

By Ramsey's theorem, this coloring of K_n must admit a monochromatic K_{kj} subgraph for some color $1 \leq j \leq r$. For ease of notation let $k = k_j$. Let the vertices of this monochromatic subgraph be $\{v_0, v_1, \ldots, v_{k-1}\}$, and define the differences $d_i = v_i - v_0$. By ordering and renaming the vertices we may assume that $d_{i-1} < d_i$ for $2 \leq i \leq k-1$. Since K_k is monochromatic of color j, we have that the edges $\overline{v_{i-1}v_i}$, $i = 1, 2, \ldots, k-1$, and $\overline{v_{k-1}v_0}$ must all have color j. Since $v_i - v_{i-1} = d_i - d_{i-1}$, for $i = 2, 3, \ldots, k-1$, we see that $d_i - d_{i-1}$, $2 \leq i \leq k-1$, d_1 , and d_{k-1} must all have color j. Hence we have the solution to $\mathcal{L}(k)$ of color j given by $d_1 + \sum_{i=2}^{k-1} (d_i - d_{i-1}) = d_{k-1}$ in any r-coloring of $\{1, 2, \ldots, n-1\}$.

Now that we have extablished the existence of generalized Schur numbers, we investigate the values and bounds for some of these numbers. We start with Theorem 8.21, which gives the exact values for all 2-colored generalized Schur numbers. We give part of the proof of Theorem 8.21, and leave the remainder to the reader as Exercise 8.8.

8. Schur's Theorem

Theorem 8.21. Let $k, \ell \geq 3$. Then

$$S(k,\ell) = \left\{ egin{array}{ll} 3\ell-4 & \mbox{if } k=3 \mbox{ and } \ell \mbox{ is odd}, \ 3\ell-5 & \mbox{if } k=3 \mbox{ and } \ell \mbox{ is even}, \ k\ell-\ell-1 & \mbox{if } 4\leq k\leq\ell. \end{array}
ight.$$

Proof. For all colorings below, we denote by R the set of red integers and by B the set of blue integers.

We start by showing that the given expressions serve as lower bounds for their respective cases. To do this, for each case we exhibit a valid 2-coloring, i.e., one that avoids both a red solution to $\mathcal{L}(k)$ and a blue solution to $\mathcal{L}(\ell)$.

Case 1. k = 3 and $\ell \ge 3$ is odd. Consider the coloring of $[1, 3\ell - 5]$ given by

 $\begin{aligned} R &= \{n: 1 \leq n \leq \ell - 2, \ n \text{ odd} \} \cup \{n: 2\ell - 2 \leq n \leq 3\ell - 5, \ n \text{ even} \}; \\ B &= [1, 3\ell - 5] - R. \end{aligned}$

We will first establish that there is no red solution to $\mathcal{L}(3)$. Let $x_1 \leq x_2 < x_3 \leq 3\ell - 5$ be red integers. If $\{x_1, x_2\} \subseteq [1, \ell - 2]$ then $x_1 + x_2$ belongs to $[2, 2\ell - 4]$ and is even. Thus $x_1 + x_2$ is colored blue, so that $\{x_1, x_2, x_3\}$ is not a red solution to $\mathcal{L}(3)$. Hence, we assume that $x_1 \in [1, \ell - 2]$ and $x_2 \in [2\ell - 2, 3\ell - 5]$. Here we have $x_1 + x_2 \in [2\ell - 1, 4\ell - 7]$ and odd. This shows that either $x_1 + x_2$ is colored blue or out of bounds, so that $\{x_1, x_2, x_3\}$ is not a red solution to $\mathcal{L}(3)$. Finally, we assume $\{x_1, x_2\} \subseteq [2\ell - 2, 3\ell - 5]$. This gives $x_1 + x_2 \geq 4\ell - 4 > 3\ell - 5$. In this situation, since the sum is out of bounds, $\{x_1, x_2, x_3\}$ cannot be a red solution to $\mathcal{L}(3)$.

Next, we show that there is no blue solution to $\mathcal{L}(\ell)$. Assume $x_1 \leq x_2 \leq \cdots \leq x_{\ell-1} < x_\ell \leq 3\ell-5$ are ℓ integers all colored blue. If $\{x_1, x_2, \ldots, x_{\ell-1}\} \subseteq [2, \ell-3]$ then $\sum_{i=1}^{\ell-1} x_i \geq 2\ell-2$ and is even. This implies that $\sum_{i=1}^{\ell-1} x_i$ is either colored red or is out of bounds. Hence $\{x_1, x_2, \ldots, x_\ell\}$ is not a blue solution to $\mathcal{L}(\ell)$. If, on the other hand, there exists $j \in \{1, 2, \ldots, \ell-1\}$ such that $x_j \notin [2, \ell-3]$, then $x_1, x_2, \ldots, x_{j-1} \geq 2$ implies that $\sum_{i=1}^{j} x_i \geq 2(j-1) + \ell - 3 = 3j - 5$. However, since $x_\ell \leq 3\ell - 5$, we may assume that $\sum_{i=1}^{\ell-1} x_i = 3l - 5$,

8.2. A Generalization of Schur's Theorem

which is colored red. This shows that $\{x_1, x_2, \ldots, x_\ell\}$ cannot be a blue solution to $\mathcal{L}(\ell)$.

This completes the proof of the lower bound for k = 3 and ℓ odd. The lower bounds for the other cases are left to the reader in Exercise 8.8.

We now move on to the upper bounds. We prove one case and leave the other cases to the reader in Exercise 8.8. We will consider $4 \le k \le \ell$ and $1 \in R$. We prove the equivalent statement:

 $S(k+1, \ell+1) \le k\ell + k - 1$ for $3 \le k \le \ell$.

Assume, for a contradiction, that there exists a 2-coloring of $[1, k\ell + k - 1]$ that avoids both a red solution to $\mathcal{L}(k + 1)$ and a blue solution to $\mathcal{L}(\ell + 1)$.

Since $1 \in R$, it follows that $k \in B$, and hence $k\ell \in R$, which in turn implies that $\ell \in B$. Since $1, k\ell \in R$, we have $k\ell + k - 1 \in B$. We deduce from this that $2k - 1 \in R$ (or else the $(\ell + 1)$ -tuple $\{k, k, \ldots, k, 2k - 1, k\ell + k - 1\}$ would be a blue solution). Since 2k - 1 is red, we must have $2 \in B$ or else the (k+1)-tuple $\{1, 2, 2, \ldots, 2, 2k - 1\}$ would be red. This implies that $3\ell - 2 \in R$ in order to avoid the blue solution given by the $(\ell + 1)$ -tuple $\{2, 2, \ldots, 2, \ell, 3\ell - 2\}$. This in turn shows that we must have $3\ell + k - 3 \in B$ in order not to have the (k + 1)-tuple $\{1, 1, \ldots, 1, 3\ell - 2, 3\ell + k - 3\}$ be red.

We next show that $\ell+1 \in B$. First note that $3 \in R$, for otherwise $\{3, 3, \ldots, 3, k, 3\ell + k - 3\}$ would be blue. Since $1, 3 \in R$, it follows that $k + 2 \in B$. Now assume, for a contradiction, that $\ell + 1 \in R$. Under this assumption, we have $2\ell + k \in B$, so that the (k+1)-tuple $\{1, 1, \ldots, 1, \ell + 1, \ell + 1, 2\ell + k\}$ is not red. This leads to the blue solution to $\mathcal{L}(\ell + 1)$ given by $\{2, 2, \ldots, 2, k + 2, 2\ell + k\}$, the desired contradiction. Thus, $\ell + 1 \in B$.

We have shown that $2, k, \ell + 1$ and $3\ell + k - 3$ are all blue, so that the $(\ell + 1)$ -tuple $\{2, 2, \ldots, 2, k, \ell + 1, 3\ell + k - 3\}$ gives a blue solution to $\mathcal{L}(\ell + 1)$, a contradiction.

We conclude this section with upper and lower bounds for the generalized Schur number $S_r(k)$. We start with an upper bound, the proof of which is very similar to the proof of Corollary 8.6.

8. Schur's Theorem

Theorem 8.22. Let $r \ge 2$. If $k \ge 3$, then $S_r(k) \le R_r(k) - 1$, where $R_r(k)$ is the r-colored Ramsey number.

Proof. The proof of Theorem 8.17 gives a correspondence between the edgewise *r*-colorings of K_n and the *r*-colorings of [1, n-1]. Hence, if $n = R_r(k)$, we have $S_r(k) \le n-1 = R_r(k) - 1$.

As we did in the proof of Lemma 8.7, it is possible to use Ramsey's theorem to obtain an upper bound independent of the Ramsey numbers. However, the formula is rather cumbersome, and we will not present it here.

On the other hand, we do have a nice formula for a lower bound. The lower bound is a generalization of Theorem 8.9. The proof is quite similar to that of Theorem 8.9 and we leave much of it to the reader as Exercise 8.9.

Theorem 8.23. Let $r \ge 2$. If $k \ge 3$, then $S_r(k) \ge \frac{k^{r+1}-2k^r+1}{k-1}$.

Proof. Let $\chi : [1, n] \to \{1, 2, ..., r\}$ be an *r*-coloring of [1, n] with no monochromatic solution to $\mathcal{L}(k)$. Define an (r + 1)-coloring

 $\hat{\chi}: [1, kn + k - 1] \rightarrow \{1, 2, \dots, r + 1\}$

that extends χ as follows: for $x \in [n+1, (k-1)n+k-2]$ let $\hat{\chi}(x) = r+1$; otherwise let $\hat{\chi}(x) = \chi(y)$, where $x \equiv y \pmod{((k-1)n+k-2)}$. We leave it to the reader to deduce that under $\hat{\chi}$, [1, kn + k - 1]contains no monochromatic solution to $\mathcal{L}(k)$. Thus, we have that if $S_r(k) \ge n+1$ then $S_{r+1}(k) \ge kn+k-1$. Hence, $S_{r+1}(k) \ge kS_r(k)-1$. Noting that $S_1(k) = k - 1$, we leave it to the reader to show that $S_r(k) \ge \frac{k^{r+1}-2k^r+1}{k-1}$.

8.3. Refinements of Schur's Theorem

Schur's theorem (Theorem 8.2) tells us that any r-coloring of [1, s(r)] must contain a monochromatic solution to x + y = z. However, x and y need not be distinct, and in fact this was crucial in showing that s(2) = 5 (see Example 8.5). You may have asked yourself: does Schur's theorem hold if we require that x and y be distinct? This is obviously a stricter condition and, as we have seen, this can turn

Ramsey-type statements into false statements. However, in this case, the resulting Ramsey-type statement is true.

Theorem 8.24. For $r \ge 1$, there exists a minimal integer $\hat{s} = \hat{s}(r)$ such that every r-coloring of $[1, \hat{s}]$ admits a monochromatic solution to x + y = z with x and y distinct.

Proof. For r = 1 we have $\hat{s}(1) = 3$. We proceed by induction on r. Let $r \geq 2$ and assume that $\hat{s}(r-1)$ exists. We will show that $\hat{s}(r) \leq w(2\hat{s}(r-1)+1;r)$, where w(k;r) is the van der Waerden function.

For any r-coloring of $[1, w(2\hat{s}(r-1)+1; r)]$ there is a monochromatic arithmetic progression $\{a, a + d, a + 2d, \cdots, a + 2\hat{s}(r-1)d\}$ of color, say, red. If there exists $i, 1 \leq i \leq \hat{s}(r-1)$, such that idis red, then x = id, y = a + id, and z = a + 2id are all red, so that we have x + y = z with x and y distinct. Hence, assume that $d[1, \hat{s}(r-1)] = \{id : 1 \leq i \leq \hat{s}(r-1)\}$ is void of the color red. Then, $d[1, \hat{s}(r-1)]$ is (r-1)-colored. By the inductive assumption and Proposition 2.30 we have a monochromatic solution to x + y = z with x and y distinct. \Box

From previous chapters we have seen that the best known upper bounds on the van der Waerden numbers are very large. Hence, the upper bound for $\hat{s}(r)$ provided by the proof of Theorem 8.24 is not very useful. For example, the proof gives $\hat{s}(2) \leq w(7;2)$. Gowers' bound (Theorem 2.21) gives the smallest known (at least to our knowledge) upper bound for w(7;2):

 $2^{2^{2^{65536}}}$,

a number whose decimal representation cannot be written down because it exceeds the number of atoms in the universe (estimated as 10^{78} atoms).

As it turns out, much better upper bounds for $\hat{s}(r)$ are known. We can fairly easily provide one such bound; this bound is given in Theorem 8.25. The best known upper bound is given in Theorem 8.26, which we state without proof.

Theorem 8.25. For $r \ge 1$, $\hat{s}(r) \le 2^{3r!-1}$.

8.4. Exercises

8. Schur's Theorem

Proof. Let n = 3r!. Let $\{a_i\}_{i=1}^n$ be an increasing sequence of positive integers with no arithmetic progression of length 3.

We first claim that any r-coloring of $[1, a_n]$ must contain a monochromatic solution to x + y = z with $x \neq y$. To see this, we show that there exist $a_i < a_j < a_k$ such that $a_k - a_j, a_k - a_i, a_j - a_i \in [1, a_n]$ all have the same color.

Recall that $R_r(3)$ is an r-color Ramsey number. From Lemma 8.7 we have $R_r(3) \leq n$. Hence, any edgewise r-coloring of K_n must contain a monochromatic triangle. Label the vertices of K_n with the a_i 's. We use the same coloring as defined in the proof of Schur's theorem. Hence, we color the edge between vertices $a_i < a_j$ depending upon $a_j - a_i$. As a result, we have a monochromatic triangle on three vertices, say $a_i < a_j < a_k$.

Now, since $a_k - a_i = (a_k - a_j) + (a_j - a_i)$ and $a_k - a_j \neq a_j - a_i$ (because a_i, a_j, a_k are not in arithmetic progression), we have proven the claim.

To complete the proof, let $a_i = 2^{i-1}$, i = 1, 2, ..., n, and note that $\{2^i\}_{i=0}^{n-1}$ contains no arithmetic progression of length three. \Box

Theorem 8.26. For $r \ge 1$, $\hat{s}(r) \le |r!re| + 1$.

There is still room for improvement on the upper bound of Theorem 8.26. Via a computer search it can be proved that $\hat{s}(2) = 9$; the bound from Theorem 8.26 gives $\hat{s}(2) \leq 11$.

We conclude this chapter with a result similar to Theorem 8.24 for the generalized Schur numbers. Recall that we denote by $\mathcal{L}(t)$ the equation $x_1 + x_2 + \cdots + x_{t-1} = x_t$.

Theorem 8.27. For any $r \geq 1$, there exists a least positive integer $\hat{S} = \hat{S}(k_1, k_2, \ldots, k_r)$ such that for any r-coloring of $[1, \hat{S}]$ there is a monochromatic solution to $\mathcal{L}(k_j)$ of color j for some $j \in \{1, 2, \ldots, r\}$ with $x_1, x_2, \cdots, x_{k_j}$ distinct.

Proof. Let $n = R(k_1, k_2, ..., k_r)$. Let $\{a_i\}_{i=1}^n$ be an increasing sequence of positive integers such that

$$a_m - a_\ell \neq a_j - a_i$$
 for all $1 \le i < j \le \ell < m \le n$.

We first claim that any r-coloring of $[1, a_n]$ must contain a monochromatic solution to $\mathcal{L}(k_j)$ of color j with $x_1, x_2, \ldots, x_{k_j}$ distinct. To establish this claim, we show that for some $j \in \{1, 2, \ldots, r\}$ there exist $a_{i_1} < a_{i_2} < \cdots < a_{i_{k_j}}$ such that $\{a_{i_{t+1}} - a_{i_t} : 1 \le t \le k_j - 1\} \subseteq [1, a_n]$ is monochromatic.

By the definition of n, any edgewise r-coloring of K_n must contain a *j*-colored K_{k_j} for some $1 \le j \le r$. Label the vertices of K_n with the a_i 's. For all pairs of vertices $a_s < a_t$, coloring the edge between a_s and a_t with the color $a_t - a_s$, we have, for some $j \in [1, r]$, a monochromatic K_{k_j} of color j on $k = k_j$ vertices, say $a_{n_1} < a_{n_2} < \cdots < a_{n_k}$.

Now, since $a_{n_k} - a_{n_1} = \sum_{i=1}^{k-1} (a_{n_{i+1}} - a_{n_i})$ and all the differences $a_t - a_s$, $1 \le s < t \le n$, are distinct, we have proven the claim.

To complete the proof, let $a_i = 2^{i-1}$, i = 1, 2, ..., n. To show that the differences $a_t - a_s$, $1 \le s < t \le n$, are distinct, assume (for a contradiction) that there exist $0 \le w < x \le y < z \le n - 1$ such that $2^z - 2^y = 2^x - 2^w$. Then $2^y(2^{z-y} - 1) = 2^w(2^{x-w} - 1)$ and hence

 $2^{y-w}(2^{z-y}-1) = 2^{x-w} - 1,$

a contradiction since the left side is even and the right side is odd. \Box

Remark 8.28. We remind the reader of Definition 1.23. We say that an equation \mathcal{L} is regular if, for all $r \geq 1$, for every *r*-coloring of \mathbb{Z}^+ there is a monochromatic solution to \mathcal{L} . From Theorem 8.17, we see that $x_1 + x_2 + \cdots + x_{k-1} = x_k$ is regular.

8.4. Exercises

- 8.1 Regarding the discussion in the introduction of this chapter, explain why the planes x + y = z and x + y = 2z must, under any given finite coloring, contain an infinite number of colored points with positive integer coordinates.
- 8.2 The maximum number, asymptotically, of monochromatic Schur triples over all r-colorings of [1, n] is known to be $cn^2 + O(n)$, for some c > 0. Find c.
- **8.3** Use Exercise 8.2 to complete Theorem 8.14 by showing that within [1, 4k 1] there are $4k^2 + O(k)$ Schur triples.

- **8.4** Use Lemma 8.11 to show that, under any 2-coloring of [1, n]the number of monochromatic 3-term arithmetic progressions is $O(n^2)$.
- **8.5** Show that the 2-coloring of $[1, 1000m] = [1, n], m \in \mathbb{Z}^+$, given by

152m 0 10m 152m 068m 1106m 0212m 1212m 0106m 168m 052m 110m 052myields only $.053382n^2 + O(n)$ monochromatic 3-term arithmetic progressions. (A computer may be helpful here.) (Note: this is slightly more than $\frac{4}{75}n^2 + O(n)$, which the secondnamed author conjectures to be the minimum number, over all 2-colorings of [1, n], of monochromatic 3-term arithmetic progressions.)

- 8.6 Show that the coloring given in Corollary 8.16 admits only $\frac{n^2}{11\cdot 2^{2r-3}} + O(n)$ monochromatic Schur triples.
- 8.7 Finish Example 8.20 by

a) showing that the given coloring of [1, 13] is indeed a valid coloring, i.e., that there is no red solution to $\mathcal{L}(4)$ and no blue solution to $\mathcal{L}(5)$, and

b) concluding that any 2-coloring of [1, 14] with $1 \in B$ must admit either a red solution to $\mathcal{L}(4)$ or a blue solution to $\mathcal{L}(5)$.

8.8 Finish the proof of Theorem 8.21 as follows:

a) Show that the following colorings provide lower bounds for the stated cases:

> i) For $\ell > 4$ and even, the 2-coloring of $[3\ell - 6]$ given by $R = R_1 \cup R_2$ and $B = [1, 3\ell - 6] - R$, where $R_1 = \{n : 1 < n < \ell - 3, n \text{ odd}\}$ and $R_2 = \{n : 2\ell - 2 \le n \le 3\ell - 6, n \text{ even}\};$ ii) For $4 \le k \le \ell$, the 2-coloring of $[k\ell - \ell - 2]$ given by $R = R_1 \cup R_2$ and $B = [1, k\ell - \ell - 2] - R$, where $R_1 = \{n : (k-1)(\ell-1) < n < k\ell - \ell - 2\}$ and $R_1 = \{n : 1 \le n \le k - 2\}.$

b) Finish the upper bound $4 \le k \le \ell$ for $1 \in B$. (Hint: in the case where $1 \in R$, interchange all occurrences of the set R with the set B, of the word 'red' with the word 'blue', and of the value k with the value ℓ . Check to make sure that the resulting argument is correct.)

c) Deduce the upper bounds for k = 3 and $\ell > 4$ and even. and for k = 3 and $\ell > 3$ and odd, in a fashion similar to the case presented in the proof of Theorem 8.21. (Hint: begin by considering two subcases: $1 \in R$ and $1 \in B$.)

8.9 Finish the proof of Theorem 8.23 by showing that a) $\hat{\chi}$ admits no monochromatic Schur triple, and b) $S_r(k) \ge \frac{k^{r+1} - 2k^r + 1}{k-1}$.

8.5. Research Problems

- **8.1.** Prove or disprove the conjecture that s(5) = 160. References: [106], [110]
- *8.2. For r > 3, find the minimum number, asymptotically, over all r-colorings of [1, n], of monochromatic Schur triples. References: [75], [225], [248]
- ***8.3.** Find the minimum number, asymptotically, over all 2-colorings of [1, n], of monochromatic 3-term arithmetic progressions. (The 2-coloring of [1, n] that gives the smallest upper bound (to date) for this minimum is given in Exercise 8.5.) References: [203], [225], [248]
- ***8.4.** Find the minimum number, asymptotically, over all 2-colorings of [1, n], of monochromatic solutions to x + ay = z, for $a \neq 1$. (The existence of such monochromatic solutions is implied by Rado's theorem, given in the next chapter.) References: [225], [248]
- 8.5. Determine new bounds and/or values for the r-colored generalized Schur numbers, for r > 3. References: [36], [224]
- **8.6.** Determine the exact value of the 2-colored *strict* generalized Schur numbers (see Theorem 8.27), i.e., determine the least positive integer $G(k, \ell)$ such that any 2-coloring of $[1, G(k, \ell)]$ must admit either a red solution to $x_1 + \cdots + x_{k-1} = x_k$ with $x_1 < x_2 < \cdots < x_k$, or a blue solution to $x_1 + \cdots + x_{\ell-1} = x_\ell$ with $x_1 < x_2 < \cdots < x_\ell$. References: [41], [127, p. 77], [224]

8.7. Improve upon the upper bound given for the *strict* Schur numbers $\hat{s}(r)$ (from Theorem 8.25). Reference: [45]

8.6. References

§8.1. Theorem 8.1, Schur's theorem, and Theorem 8.9 can be found in [247]. See [273] for a brief summary of Wiles' proof of Fermat's Last Theorem. Erdős and Szekeres' rediscovery of Ramsey's theorem can be found in [101]. Goodman's result on the number of monochromatic triangles in any 2-coloring of the edges of K_n can be found in [120]. The problem of finding the asymptotic minimum number of Schur triples over all 2-colorings of [1, n] was first posed by Graham, Rödl, and Ruciński in [125]. It was first solved, independently, by Robertson and Zeilberger [225] and Schoen [248]. The proof of Lemma 8.11 can be found in [125]. The proof of Theorem 8.15 is due to Datskovsky [75]. The coloring given in Theorem 8.14 is due to Zeilberger. The fact that the coloring in Theorem 8.14 is essentially the only minimal coloring is proved in [248]. Corollary 8.16 is from [225].

§8.2. The proof of Theorem 8.17 is from [222]. Theorem 8.17 proves the existence, in particular, of $S_r(k)$; however, this is implied by Rado's theorem (Theorem 9.2), which was established before Theorem 8.17. The values for the 2-color generalized Schur numbers $S(k, \ell)$ were determined for $k = \ell$ in [36] and for all k and ℓ in [224].

§8.3. The proof of Theorem 8.24 is adapted from the proof of Theorem 2 in [127, p. 70]. The proof of Theorem 8.26 is found in [45], which improves upon Irving's bound given in [149].

§8.4. Exercise 8.5's coloring was discovered by Pablo Parrilo [203].

Additional References: For a tribute to Schur and an overview of his work, see [190]. Exoo [106] and Fredricksen and Sweet [110] give the current best known lower bounds for the Schur numbers s(5), s(6), and s(7). Bergelson, in [27], uses ergodic theory to prove a density statement which generalizes and strengthens Schur's theorem. A combinatorial proof of Bergelson's result is provided in [109]. Additional density results appear in [31]. In [149], Irving considers m = M(r, k),

the minimum number such that any r-coloring of [1, m] must admit a monochromatic solution to $x_1 + x_2 + \cdots + x_{k-1} = x_k$ with all x_i distinct, and provides an upper bound for m. The case r = 2 can be found in [251]. Schaal [239], and Schaal and Bialostocki [41], consider variations of Schur numbers that are not guaranteed to exist. The 1892 result of Hilbert [142] can be specialized to give Schur's theorem. Work on the numbers associated with Hilbert's theorem can be found in [57].

8.6. References

222

Chapter 9

Rado's Theorem

The previous chapter, on Schur's theorem, included the extension of Schur's theorem to equations of the form $\sum_{i=1}^{k-1} x_i = x_k$, where $k \ge 4$. In other words, we found that for any finite coloring of \mathbb{Z}^+ there is a monochromatic solution to an equation of this form. The extension can be taken further, and was taken further, by one of Schur's Ph.D. students, Richard Rado.

In a series of papers in the 1930's, Rado determined, in particular, exactly which equations $\sum_{i=1}^{k} c_i x_i = 0$ are guaranteed to have monochromatic solutions under any finite coloring of the positive integers. In fact, part of this result is contained in his Ph.D. thesis.

9.1. Rado's Single Equation Theorem

Before stating the goal of this section, we remind the reader that a linear homogeneous equation is any equation of the form

$$\sum_{i=1}^{k} c_i x_i = 0,$$

where each $c_i \in \mathbb{Z}$ is a nonzero constant and each x_i is a variable. Since we will be considering only linear equations, when we use "homogeneous" we will mean "linear homogeneous." We saw in Chapter 8 (see Remark 8.28) that

$$x_1 + x_2 + \dots + x_{k-1} = x_k$$

is regular. In this section, we use this result and classify those homogeneous equations that, under any finite coloring of \mathbb{Z}^+ , have a monochromatic solution $(x_1, x_2, \ldots, x_k \in \mathbb{Z}^+$ all are of the same color).

Considering Remark 8.28, we expand upon Definition 1.23 with the following definition.

Definition 9.1. Let S be a system of linear homogeneous equations. We say that S is *r*-regular if, for $r \ge 1$, for every *r*-coloring of \mathbb{Z}^+ there is a monochromatic solution to S. If S is *r*-regular for all $r \ge 1$, we say that S is regular.

We will find a condition on the constants c_i in the equation

$$(9.1) \qquad \qquad \sum_{i=1}^{k} c_i x_i = 0$$

so that (9.1) is regular. Clearly, some restriction on the constants is needed. For example, if all of the constants are positive, then there is no solution in the positive integers, much less a monochromatic one. So, at least one of the constants must be negative. We also know, from Theorem 8.17, that if $c_i = c$ for $i = 1, 2, \ldots, k-1$ and $c_k = -c$, then we are guaranteed a monochromatic solution (by factoring out the c). This heads us in the right direction.

In this section we will be dealing with systems that consist of only one equation. However, in Section 9.3 we consider systems in general.

Theorem 9.2 (Rado's Single Equation Theorem). Let $k \ge 2$. Let $c_i \in \mathbb{Z} - \{0\}, 1 \le i \le k$, be constants. Then (9.1) is regular if and only if there exists a nonempty $D \subseteq \{c_i : 1 \le i \le k\}$ such that $\sum_{d \in D} d = 0$.

Proof. We first prove the 'if' part. We show that for all $r \ge 1$, if there exists a nonempty subset D as above, then (9.1) is r-regular.

9.1. Rado's Single Equation Theorem

We use induction on r, starting with the case of r = 1. Without loss of generality, let $D = \{c_1, \ldots, c_m\}$ with $c_1 > 0$. If m = k, then we may take $x_i = 1$ for $1 \le i \le k$ as our monochromatic solution. Hence, we assume that m < k. Thus, $c_{m+1} + c_{m+2} \cdots + c_k$ is a nonempty sum. Let $s = c_{m+1} + c_{m+2} + \cdots + c_k$ and note that $s \ne 0$.

Let $x_2 = x_3 = \cdots = x_m$ and $x_{m+1} = x_{m+2} = \cdots = x_k$. Equation (9.1) reduces to

$$c_1x_1 + x_2(c_2 + c_3 + \dots + c_m) + x_{m+1}(c_{m+1} + c_{m+2} + \dots + c_k) = 0.$$

Since $c_1 + c_2 + \cdots + c_m = 0$, we have

(9.2)
$$c_1(x_1 - x_2) + sx_{m+1} = 0$$

Any positive integers x_1 and x_2 such that $x_2 - x_1 = s$, together with $x_{m+1} = c_1$, provide a (monochromatic) solution to (9.2), completing the case when r = 1.

Now let $r \ge 2$ and assume that the result holds for $1 \le t \le r-1$. We show that it holds for r. For each $t \le r-1$, let n(t) be the least positive integer such that for every t-coloring of [1, n(t)] there is a monochromatic solution to (9.1) (n(t) exists by the induction hypothesis).

Assume, without loss of generality, that $c_1 + c_2 + \cdots + c_m = 0$, with *m* maximal and $c_1 > 0$. Again, we may assume that $m \neq k$, so that $s = c_{m+1} + c_{m+2} + \cdots + c_k$ is a nonempty sum. Note that $s \neq 0$.

Let $b = \sum_{i=1}^{k} |c_i|$ and let n = n(r-1). We will show that $n(r) \leq bw(n+1;r)$, where w represents the usual van der Waerden function. We will show that every r-coloring of [1, bw(n+1;r)] admits a monochromatic solution to (9.1).

Let $x_2 = x_3 = \cdots = x_m$ and $x_{m+1} = x_{m+2} = \cdots = x_k$. As in the case when r = 1, (9.1) reduces to (9.2).

Let χ be an r-coloring of [1, bw(n+1; r)]. We shall now find x_1 , x_2 , and x_{m+1} satisfying (9.2), with $\chi(x_1) = \chi(x_2) = \chi(x_{m+1})$. We see that by Theorem 4.9, χ must yield an (n+1)-term monochromatic arithmetic progression with gap a multiple of |s| (note that we have $1 \leq |s| < b$). That is, we have that $\{a, a+d|s|, a+2d|s|, \ldots, a+nd|s|\}$ is monochromatic for some $a, d \geq 1$, with $a + nd|s| \leq bw(n+1;r)$.

If there exists $j \in [1, n]$ such that $\chi(jdc_1) = \chi(a)$, then by letting $|x_2 - x_1| = jd|s|$ and $x_{m+1} = jdc_1$, we have a monochromatic solution to (9.2) and are done. If, on the other hand, for all $j \in [1, n]$ we have $\chi(jdc_1) \neq \chi(a)$, then $\{dc_1, 2dc_1, \ldots, ndc_1\} = dc_1[1, n]$ is (r-1)-colored, and by Proposition 2.30 we are done.

For the 'only if' part of the theorem, we prove the contrapositive. Let c_1, c_2, \ldots, c_k be fixed with no subset summing to zero. We prove the existence of an *r*-coloring of \mathbb{Z}^+ , for some $r \ge 1$, that admits no monochromatic solution to (9.1).

Choose a prime p such that for any $C \subseteq \{c_i : 1 \le i \le k\}$ we have $p \nmid \sum_{c \in C} c$. Since C is a finite set, such a choice is obviously possible.

We now define a 2(p-1)-coloring $\chi : \mathbb{Z}^+ \to \{1, 2, \dots, 2(p-1)\}$. For $i \not\equiv 0 \pmod{p}$, let $\chi(i) = \overline{i}$ where $i \equiv \overline{i} \pmod{p}$ and $1 \leq \overline{i} \leq p-1$. For $i \equiv 0 \pmod{p}$, color i as follows. Let $\mu(i)$ be the maximal integer such that $\frac{i}{p^{\mu(i)}} \in \mathbb{Z}^+$. Thus, we may write $i = \sum_{j \geq \mu(i)} a_j p^j$, where the a_j 's are positive integers. Now let $\chi(i) = a_{\mu(i)} + p - 1$ (note that $1 \leq a_{\mu(i)} < p$).

To prove that χ admits no monochromatic solution to (9.1), assume, for a contradiction, that $Y = \{y_1, y_2, \ldots, y_k\}$ is a monochromatic solution under χ . We consider two cases.

Case 1. $\chi(Y) = u < p$. Then we have $\sum_{i=1}^{k} c_i y_i = 0$, which implies that $u \sum_{i=1}^{k} c_i \equiv 0 \pmod{p}$. Since u and $\sum_{i=1}^{k} c_i$ are both nonzero, p divides $u \sum_{i=1}^{k} c_i$. This is a contradiction, since $p \nmid u$ and $p \nmid \sum_{i=1}^{k} c_i$. **Case 2.** $\chi(Y) = u \ge p$. Write $y_i = \sum_{j \ge \mu(y_i)} a_j p^j$, and define $y'_i = y_i - a p^{\mu(y_i)}$ for $1 \le i \le k$, where $a = \chi(y_i) - p + 1$. Let $\mu^* = \min\{\mu(y_1), \mu(y_2), \dots, \mu(y_k)\}$.

By assumption we have $\sum_{i=1}^{k} c_i y_i = 0$. Hence

$$\sum_{i=1}^{k} c_i y'_i + a p^{\mu^*} \sum_{i=1}^{k} c_i p^{\mu(y_i) - \mu^*} = 0$$

Note that p^{μ^*+1} divides $\sum_{i=1}^k c_i y'_i$. Thus,

$$ap^{\mu^{\star}} \sum_{i=1}^{k} c_i p^{\mu(y_i) - \mu^{\star}} \equiv 0 \pmod{p^{\mu^{\star} + 1}}$$

Therefore, p divides $a \sum_{i=1}^{k} c_i p^{\mu(y_i)-\mu^*}$. Since $1 \le a < p$, this implies that p divides $\sum_{i=1}^{k} c_i p^{\mu(y_i)-\mu^*}$. By the definition of μ^* , at least one of the $\mu(y_i)$'s must equal μ^* . Letting $M = \{t : \mu(t) = \mu^*\}$, we have

$$\sum_{i=1}^{k} c_i p^{\mu(i)-\mu^{\star}} = \sum_{i \in M} c_i + \sum_{i \notin M} c_i p^{\mu(i)-\mu^{\star}},$$

where the first sum on the right-hand side is nonempty.

Since p divides $\left(\sum_{i \notin M} c_i p^{\mu(i)-\mu^*}\right)$ and we require p to divide $\left(\sum_{i \in M} c_i + \sum_{i \notin M} c_i p^{\mu(i)-\mu^*}\right)$, we see that p divides $\left(\sum_{i \in M} c_i\right)$, contradicting our choice of p.

Example 9.3. Theorem 8.17 with $k_1 = k_2 = \cdots = k_r$ follows from Rado's single equation theorem, since $x_1 + x_2 + \cdots + x_{k-1} - x_k = 0$ satisfies the necessary subset requirement $(c_{k-1} + c_k = 0)$.

Theorem 9.2 can be strengthened as seen in the following theorem, which we offer without proof.

Theorem 9.4. Let $r \ge 1$. For every r-coloring of \mathbb{Z}^+ there is a monochromatic solution (b_1, b_2, \ldots, b_k) to (9.1), where the b_i 's are distinct, if and only if (9.1) is regular and there exist distinct integers (not necessarily positive) y_1, y_2, \ldots, y_n such that (y_1, y_2, \ldots, y_n) satisfies (9.1).

Example 9.5. Since x + y = 2z is another way of describing 3-term arithmetic progressions, provided $x \neq y$, we see that Theorem 9.4 proves the existence of a monochromatic 3-term arithmetic progression in any finite coloring of \mathbb{Z}^+ . However, Theorem 9.4 does not imply the same result for arithmetic progressions of length more than 3. For this we need Rado's "full" theorem, which is presented later in this chapter as Theorem 9.27.

Rado's single equation theorem deals only with homogeneous equations. An obvious question is: is there a similar result concerning nonhomogeneous linear equations, i.e., equations of the form $\sum_i c_i x_i = b$, where b is a nonzero integer? As we will see, it is possible in this setting to have a nonempty subset of the coefficients summing to 0 while not being guaranteed a monochromatic solution. Knowing this, any regularity result will most probably be dependent upon

9.1. Rado's Single Equation Theorem

the value of b. In the following proposition, we state what is known about one of the simplest nonhomogeneous cases. Before doing so, we introduce the following notation.

Notation. Let \mathcal{E} be a linear equation. Denote by $r(\mathcal{E}; s)$ the minimal integer, if it exists, such that any *s*-coloring of $[1, r(\mathcal{E}; s)]$ must admit a monochromatic solution to \mathcal{E} . For s = 2 we write simply $r(\mathcal{E})$.

Definition 9.6. The numbers $r(\mathcal{E}; s)$ are called *s*-color Rado numbers for equation \mathcal{E} .

Proposition 9.7. Let $b \in \mathbb{Z} - \{0\}$. Then r(x - y = b; 2) does not exist.

Proof. Consider the 2-coloring of \mathbb{Z}^+ defined by $0^{|b|}1^{|b|}0^{|b|}1^{|b|}\dots$ Clearly, under this coloring, no two positive integers having the same color can have their difference equal to b. Hence, we cannot have a monochromatic solution to x - y = b.

We next present a theorem that generalizes the equation of Proposition 9.7.

Theorem 9.8. Let $b \ge 1$, $k \ge 3$, and let $\mathcal{E}(b)$ represent the equation $x_1 + x_2 + \cdots + x_{k-1} - x_k = -b$. Then $r(\mathcal{E}(b))$ does not exist precisely when k is even and b is odd. Furthermore, we have

$$r(\mathcal{E}(b)) = k^2 + (b-1)(k+1)$$

whenever $r(\mathcal{E}(b); 2)$ exists.

Proof. We start by showing that $r(\mathcal{E}(b))$ does not exist if k is even and b is odd. We do this by giving a 2-coloring of \mathbb{Z}^+ with no monochromatic solution to $\mathcal{E}(b)$. To this end, color all even integers with one color and all odd integers with the other color. If all of the x_i are the same color, then $x_1 + x_2 + \cdots + x_{k-1} - x_k$ is even. Since b is odd, we see that a monochromatic solution cannot exist under this coloring.

Now assume that k is odd or b is even, and let

$$n = k^2 + (b-1)(k+1)$$

We show that n is a lower bound for $r(\mathcal{E}(b))$ by providing a 2coloring $\chi : [1, n-1] \to \{0, 1\}$ that does not admit a monochromatic solution to $\mathcal{E}(b)$. Define χ as follows:

$$\chi(j) = 0$$
 if and only if $j \in [k+b-1, k^2 + (b-2)k]$.

Assume, by way of contradiction, that $Y = \{y_i : 1 \le i \le k\}$ is a monochromatic solution.

If
$$\chi(Y) = 0$$
, then $y_i \ge k + b - 1$ for $1 \le i \le k - 1$. Hence
 $b + \sum_{i=1}^{k-1} y_i \ge (k-1)(k+b-1) = k^2 + (b-2)k + 1$,

a contradiction since $y_k \leq k^2 + (b-2)k$.

Now assume $\chi(Y) = 1$. If $Y - \{y_k\} \subseteq [1, k + b - 2]$, then

$$k+b-1 \le b + \sum_{i=1}^{k-1} y_i \le k^2 + (b-3)k + 2$$

Note that since $k \geq 2$ we have $k^2 + (b-3)k + 2 \leq k^2 + (b-2)k$. Hence, $b + \sum_{i=1}^{k-1} y_i \in [k+b-1, k^2 + (b-2)k]$, a contradiction since this interval is of color 0. Therefore, if $\chi(Y) = 1$ we must have $y_{k-1} \in [k^2 + (b-2)k + 1, n-1]$. Then

$$b + \sum_{i=1}^{k-1} y_i \ge b + (k-2) + k^2 + (b-2)k + 1 = n,$$

and again we have a contradiction, since $y_k \leq n-1$.

We conclude that χ is a valid 2-coloring of [1, n - 1], so that $r(\mathcal{E}(b)) \ge k^2 + (b - 1)(k + 1)$.

For the upper bound, let $m \in \mathbb{Z}^+$ and let $\chi : [1, m] \to \{0, 1\}$ be a 2-coloring that admits no monochromatic solution to $\mathcal{E}(b)$. Define the set

$$A(\chi) = \{x : x \in [1, m-1] \text{ and } \chi(x) \neq \chi(x+1)\}.$$

Note that $|A(\chi)|$ is the number of times the color changes as we proceed from 1 to m.

We finish the proof via a series of claims.

Claim 1. If $|A(\chi) \cap [1, k+b-2]|$ is even, then $m \leq k+b-2$.

Proof of Claim 1. Assume, for a contradiction, that $m \ge k + b - 1$. By the definition of $A(\chi)$ we have $\chi(1) = \chi(k + b - 1)$. Hence,

$$x_i = 1, \ 1 \le i \le k - 1;$$

 $x_k = k + b - 1$

is a monochromatic solution to $\mathcal{E}(b)$, a contradiction.

Claim 2. If $|A(\chi) \cap [1, k+b-2]| = 1$, then $m \le k^2 + (b-1)(k+1) - 1$. *Proof of Claim 2.* Without loss of generality, assume there exists $a_1 \ge 1$ such that $\chi([1, a_1]) = 0$ and $\chi([a_1 + 1, k + b - 1]) = 1$. Note that

$$(9.3) a_1 \le k+b-2$$

Since

$$x_i = k + b - 1, \ 1 \le i \le k - 1;$$

 $x_k = k^2 + (b - 2)k + 1$

is a solution to $\mathcal{E}(b)$, either $|A(\chi)| = 1$ and $m \leq k^2 + (b-2)k$, or $|A(\chi)| \geq 2$. If $|A(\chi)| = 1$ we are done, so assume $|A(\chi)| \geq 2$.

Since $|A(\chi)| \ge 2$, we know there exists $a_2 \ge k + b - 1$ such that $\chi([1, a_1]) = 0$, $\chi([a_1 + 1, a_2]) = 1$, and $\chi(a_2 + 1) = 0$. Now, since $\chi(a_1 + 1) = 1$, we have that $\chi((k - 1)(a_1 + 1) + b) = 0$ or $m < (k - 1)(a_1 + 1) + b$. In either case we have

(9.4)
$$a_2 \le (k-1)(a_1+1) + b - 1.$$

Next, since

$$x_i = 1, \ 1 \le i \le k - 2;$$

 $x_{k-1} = a_2 + 1;$
 $x_k = a_2 + k + b - 1$

is a solution to $\mathcal{E}(b)$, either $m \leq a_2 + k + b - 2$ or $\chi(a_2 + k + b - 1) = 1$.

We show that $\chi(a_2 + k + b - 1) = 1$ contradicts the fact that χ is valid. Since $[1, a_1]$ has color 0, we see that $[k + b - 1, (k - 1)a_1 + b]$ has color 1. Thus, $a_2 \ge (k - 1)a_1 + b$. This implies that

$$a_2 + k + b - 1 \ge (k - 1)(a_1 + 1) + 2b_2$$

9.1. Rado's Single Equation Theorem

Since $a_2 \ge 2$, we have $a_2 \ge \frac{k-1}{k-2}$, i.e., $(k-2)a_2 \ge k-1$. From this we see that $a_2 + k + b - 1 \le (k-1)a_2 + b$. Hence,

$$(k-1)(a_1+1) + b \le a_2 + k + b - 1 \le (k-1)a_2 + b.$$

By the above equation and the fact that the interval $[a_1 + 1, a_2]$ has color 1, there exist $x_1, x_2, \ldots, x_{k-1} \in [a_1 + 1, a_2]$ for which

$$\sum_{i=1}^{k-1} x_i + b = a_2 + k + b - 1,$$

and we have found a monochromatic solution to $\mathcal{E}(b)$, a contradiction. Hence, we cannot have $\chi(a_2 + k + b - 1) = 1$. We conclude that

(9.5) $m \le a_2 + k + b - 2.$

Combining (9.5) with (9.3) and (9.4), we see that

$$m \le k^2 + (b-1)(k+1) - 1,$$

thereby completing the proof of Claim 2.

Claim 3. If k is odd and $|A(\chi) \cap [1, k+b-2]| \ge 3$, then we have $m \le k^2 + (b-2)k - 2$.

Proof of Claim 3. Without loss of generality, assume there exist a_1 and a_2 such that $1 < a_1 < a_2 \leq k + b - 2$, $\chi(1) = 0$, $\chi(a_1) = 1$, $\chi([a_1 + 1, a_2]) = 0$, and $\chi(a_2 + 1) = 1$. Since

$$x_i = a_1 + 1, \ 1 \le i \le \frac{k-1}{2};$$

$$x_i = a_2, \ \frac{k+1}{2} \le i \le k-1;$$

$$x_k = \frac{k-1}{2}(a_2 + a_1 + 1) + b$$

satisfies $\mathcal{E}(b)$, we have either

(9.6)
$$m < \left(\frac{k-1}{2}\right)(a_2 + a_1 + 1) + b$$

or

(9.7)
$$\chi\left(\left(\frac{k-1}{2}\right)(a_2+a_1+1)+b\right) = 1.$$

However, if (9.7) holds, then we have the monochromatic solution:

$$x_i = a_1, \ 1 \le i \le \frac{k-1}{2};$$

 $x_i = a_2 + 1, \ \frac{k+1}{2} \le i \le k-1,$

a contradiction. Hence, (9.6) holds. Since $a_1 + 1 \le a_2 \le k + b - 2$, this gives

$$m \leq (k-1)(k+b-2)+b-1 = k^2 - k + (b-2)(k-1) + (b-2) + 1 \leq k^2 + (b-2)k-2,$$

as required. This completes the proof of Claim 3.

Claim 4. If k and b are both even and $|A(\chi) \cap [1, k+b-2]| = i \ge 3$ with i odd, then $m \le k^2 + (b-2)k - 1$.

Proof of Claim 4. Let

$$A(\chi) \cap [1, k+b-2] = \{a_1 < a_2 < \dots < a_i\}.$$

We assume, without loss of generality, that $\chi(a_j) = 0$ and $\chi(a_j+1) = 1$ for $j = 1, 3, 5, \ldots, i$; and that $\chi(a_j) = 1$ and $\chi(a_j + 1) = 0$ for $j = 2, 4, \ldots, i - 1$. If there exists $t \in [1, i - 1]$ such that $a_{t+1} \neq a_t + 1$, then we have the following two solutions to $\mathcal{E}(b)$:

$$\begin{split} x_i &= a_t, \ 1 \leq i \leq \frac{k}{2}; \\ x_i &= a_{t+1} + 1, \ \frac{k}{2} + 1 \leq i \leq k - 1; \\ x_k &= \frac{k}{2}(a_t) + (\frac{k}{2} - 1)(a_{t+1} + 1) + b - 1 \end{split}$$

and

$$\begin{aligned} x_i &= a_t + 1, \ 1 \le i \le \frac{k}{2}; \\ x_i &= a_{t+1}, \ \frac{k}{2} + 1 \le i \le k - 2; \\ x_{k-1} &= a_{t+1} - 1; \\ x_k &= \frac{k}{2}(a_t) + (\frac{k}{2} - 1)(a_{t+1} + 1) + b - 1. \end{aligned}$$

One of these solutions must be monochromatic unless

$$m \le \frac{k}{2}(a_t) + \left(\frac{k}{2} - 1\right)(a_{t+1} + 1) + b - 2k$$

Since $a_t < a_{t+1} + 1 \le k + b - 1$, this implies that

$$m \le (k-1)(k+b-1) + b - 2 = k^2 + (b-2)k - 1$$

Thus, we may assume that $a_{j+1} = a_j + 1$ for all $j \in [1, i-1]$. Since k + b - 2 is even, either $a_1 \neq 1$ or $a_i \neq k + b - 2$. If $a_1 \neq 1$, then we can assume, without loss of generality, that $\chi(a_1 - 1) = \chi(a_1) = 0$. Hence, the following solutions both satisfy $\mathcal{E}(b)$:

$$x_i = a_1 + 1, \ 1 \le i \le k - 1;$$

 $x_k = (k - 1)(a_1 + 1) + b$

and

$$\begin{aligned} x_1 &= a_1 - 1; \\ x_i &= a_1, \ 2 \leq i \leq \frac{k}{2} - 1; \\ x_i &= a_1 + 2, \ \frac{k}{2} \leq i \leq k - 1; \\ x_k &= (k - 1)(a_1 + 1) + b, \end{aligned}$$

one of which is monochromatic unless $m \leq (k-1)(a_1+1) + b - 1$. Since $a_1 + 1 \leq k + b - 3$ and $k \geq 3$, this implies that

$$m \le k^2 + (b-4)k + 2 \le k^2 + (b-2)k - 4.$$

If $a_i \neq k+b-2$, we may also assume that $\chi(a_i-1) = 0$, $\chi(a_i) = 1$, and $\chi([a_i + 1, k + b - 1]) = 0$. The following solutions both satisfy $\mathcal{E}(b)$:

$$x_{i} = a_{i} - 1, \ 1 \le i \le \frac{k}{2};$$

$$x_{i} = a_{i} + 1, \ \frac{k}{2} + 1 \le i \le k - 2;$$

$$x_{k-1} = a_{i} + 2;$$

$$x_{k} = (k-1)a_{i} + b$$

 and

$$x_i = a_i, \ 1 \le i \le k - 1;$$

 $x_k = (k - 1)a_i + b,$

one of which is monochromatic unless $m \leq (k-1)a_i + b - 1$. Since $a_i \leq k + b - 3$, this gives

$$m \le k^2 + (b-4)k + 2 \le k^2 + (b-2)k - 4.$$

This completes the proof of Claim 4.

To complete the proof of the theorem, it suffices to show that the hypothesis of one of the above claims is true. For the cases when $|A(\chi)| \in \{0, 1, 2, 4, 6, ...\}$, we see that the hypothesis of either Claim 1 or Claim 2 holds. If $|A(\chi)| \ge 3$ and $|A(\chi)|$ and k are both odd, then

the hypothesis of Claim 3 holds. Finally, if $|A(\chi)| \ge 3$ with $|A(\chi)|$ odd and k even, then the hypothesis of Claim 4 holds (according to our earlier assumption, since k is even, b must be even).

We can now conclude that for any valid 2-coloring of [1, m] we have $m \leq k^2 + (b-1)(k+1) - 1$. This inequality, along with the lower bound for $r(\mathcal{E}(b))$ given earlier, yields $r(\mathcal{E}(b)) = k^2 + (b-1)(k+1)$. \Box

A result analogous to, and as complete as, Theorem 9.8, but with b negative, has yet to be found. However, some progress has been made. In particular, for k = 3, we have the following result. We omit the proof, which is is similar to that of Theorem 9.8; see also Exercise 9.6.

Theorem 9.9. For $b \ge 1$, $r(x + y - z = b) = b - \lfloor \frac{b}{5} \rfloor + 1$.

We noted before that any regularity result concerning nonhomogeneous equations will most likely be dependent upon the constant b in

$$\sum_{i=1}^{k} c_i x_i = b.$$

Let $\mathcal{E}(b)$ represent the above equation, and let $\overline{\mathcal{E}}(b)$ denote this same equation with the restrictions $c_1 = c_2 = \cdots = c_{k-1} = 1$ and $c_k = -1$. From Proposition 9.7, we have that $\overline{\mathcal{E}}(b)$ is not regular if k = 2 (for $b \neq 0$). Theorem 9.8 specifies those values (with the restriction that b < 0) of k and b for which $\overline{\mathcal{E}}(b)$ is 2-regular and those values for which $\overline{\mathcal{E}}(b)$ is not 2-regular. Ultimately, we would like to determine those values of k and b (if any) for which $\mathcal{E}(b)$ is regular (i.e., r-regular for all $r \geq 1$).

One reason we see more regularity with Theorem 9.8 than Proposition 9.7 is that the more variables the equation has, the easier it is to find solutions to $\mathcal{E}(b)$ in \mathbb{Z}^+ . At the same time, the more colors we use to color \mathbb{Z}^+ , the harder it is to be guaranteed a monochromatic solution to $\mathcal{E}(b)$. Fortunately, we have the following result, which completely determines which equations are regular.

9.1. Rado's Single Equation Theorem

Theorem 9.10. Let $k \geq 2$ and let b, c_1, c_2, \ldots, c_k be nonzero integers. Let $\mathcal{E}(b)$ represent the equation

(9.8)
$$\sum_{i=1}^{k} c_i x_i = b,$$

and let $s = \sum_{i=1}^{k} c_i$. Then $\mathcal{E}(b)$ is regular if and only if one of the following two conditions holds:

(i) $\frac{b}{s} \in \mathbb{Z}^+$; (ii) $\frac{b}{s}$ is a negative integer and $\mathcal{E}(0)$ is regular.

Proof. Let $r \ge 1$. We first prove the 'if' portion of the theorem. If (i) holds, then $x_i = \frac{b}{s}$, $1 \le i \le k$, is a monochromatic solution under any r-coloring of \mathbb{Z}^+ . If (ii) holds, let χ be any r-coloring of \mathbb{Z}^+ . Now, define γ , an r-coloring of $\{\left|\frac{b}{s}\right|, \left|\frac{b}{s}\right| + 1, \left|\frac{b}{s}\right| + 2, \ldots\}$, by $\gamma(i - \frac{b}{s}) = \chi(i)$. By Proposition 2.30 and Rado's single equation theorem, we have a monochromatic solution to $\mathcal{E}(0)$ under γ . We may write $y_1 - \frac{b}{s}, y_2 - \frac{b}{s}, \ldots, y_k - \frac{b}{s}$ for such a solution. Since

$$\sum_{i=1}^{k} c_i \left(y_i - \frac{b}{s} \right) = 0,$$

we have that

$$\sum_{i=1}^{k} c_i y_i = b.$$

By the definition of γ , we know that y_1, y_2, \ldots, y_k is a monochromatic (under χ) solution to (9.8). This completes the proof of the 'if' portion of the theorem.

For the 'only if' part of the theorem, we will prove its contrapositive. We first assume that $s \nmid b$.

Note that we may rewrite (9.8) as

(9.9)
$$\sum_{i=2}^{k} c_i (x_i - x_1) = b - s x_1.$$

We consider two cases.

Case 1. s = 0. Let p be a prime such that p > b. By (9.9), we have

(9.10)
$$\sum_{i=2}^{k} c_i (x_i - x_1) = b.$$

Define $\chi : \mathbb{Z}^+ \to \{0, 1, \dots, p-1\}$ by $\chi(i) = \overline{i}$, where $i \equiv \overline{i} \pmod{p}$. For any monochromatic solution to (9.10) we have, by the definition of χ , that p divides the left-hand side of (9.10). Hence, p must divide b. However, this contradicts our choice of p. Hence, (9.8) is not p-regular.

Case 2. $s \neq 0$. We assume, without loss of generality, that s > 0. Define $\gamma : \mathbb{Z}^+ \to \{0, 1, \ldots, s - 1\}$ by $\gamma(i) = \overline{i}$, where $i \equiv \overline{i} \pmod{s}$. Assume, for a contradiction, that y_1, y_2, \ldots, y_k is a monochromatic solution to (9.9). By the definition of γ , s divides $\sum_{i=2}^k c_i(y_i - y_1)$. Hence, we must have

$$b - sy_1 \equiv 0 \pmod{s},$$

i.e., $b \equiv 0 \pmod{s}$. Thus, s divides b, a contradiction. Hence, (9.8) is not |s|-regular.

Since we have shown that (9.8) is not *p*-regular in Case 1 and not |s|-regular in Case 2, we have shown that if $s \nmid b$, then (9.8) is not regular.

To finish proving the contrapositive, assume that $\frac{b}{s}$ is a negative integer, but that $\mathcal{E}(0)$ is not regular. Then there exist a finite t and a t-coloring χ of \mathbb{Z}^+ such that, under χ , there is no monochromatic solution to $\mathcal{E}(0)$. Assume, for a contradiction, that (9.8) is regular. Define the t-coloring γ by $\gamma(i) = \chi \left(i - \frac{b}{s}\right)$. Since $\frac{b}{s}$ is a negative integer, γ is well-defined. By assumption, there exists, under γ , a monochromatic solution to (9.8), say y_1, y_2, \ldots, y_k . By the definition of γ , this means that $y_1 - \frac{b}{s}, y_2 - \frac{b}{s}, \ldots, y_k - \frac{b}{s}$ are monochromatic under χ . Since

$$\sum_{i=1}^{k} c_i \left(y_i - \frac{b}{s} \right) = \sum_{i=1}^{k} c_i y_i - \frac{b}{s} \sum_{i=1}^{k} c_i$$
$$= b - b$$
$$= 0,$$

9.2. Some Rado Numbers

.

we have that $y_1 - \frac{b}{s}, y_2 - \frac{b}{s}, \dots, y_k - \frac{b}{s}$ is a monochromatic (under χ) solution to $\mathcal{E}(0)$, a contradiction. This concludes the proof.

9.2. Some Rado Numbers

In this section we present several values and bounds for some 2-color Rado numbers associated with various linear equations in 3 variables. Theorems 9.11, 9.12, and 9.15 deal with equations that, by Rado's single equation theorem and Theorem 9.4, are regular.

Theorem 9.11. Let $a, b \ge 1$ with gcd(a, b) = 1. Then

$$r(ax + by = bz) = \begin{cases} a^2 + 3a + 1 & \text{if } b = 1, \\ b^2 & \text{if } a < b, \\ a^2 + a + 1 & \text{if } 2 \le b < a. \end{cases}$$

Proof. We separate the proof into the three obvious cases.

Case 1. b = 1. Let $n = a^2 + 3a + 1$. Color [a + 1, a(a + 2)] one color and its complement in [1, n - 1] the other color. We leave it to the reader in Exercise 9.7 to verify that this is a valid 2-coloring of [1, n - 1], which implies that $r(ax + y = z) \ge n$.

For the upper bound, assume, for a contradiction, that there exists $\chi : [1, n] \rightarrow \{0, 1\}$ that is valid. We may assume that $\chi(1) = 0$, and hence $\chi(a + 1) = 1$, which in turn implies that $\chi((a + 1)^2) = 0$. Since $(x, y, z) = (a + 2, 1, (a + 1)^2)$ satisfies the equation, we must have $\chi(a+2) = 1$. So that $(a + 2, a + 1, a^2 + 3a + 1)$ is not monochromatic we must have $\chi(a^2 + 3a + 1) = 0$. This gives us the monochromatic solution $(1, (a + 1)^2, a^2 + 3a + 1)$, a contradiction.

Case 2. a < b. Let $n = b^2$. We leave it to the reader in Exercise 9.7 to verify that a valid 2-coloring of [1, n - 1] is given by coloring b[1, b - 1] one color and its complement the other color.

For the upper bound, assume, for a contradiction, that there exists $\chi : [1, n] \to \{0, 1\}$ that is valid.

Assume, without loss of generality, that $\chi(b) = 0$. We now show that $\chi(ib) = 0$ for $1 \le i \le b$ by considering the following solutions

9.2. Some Rado Numbers

9. Rado's Theorem

(which require a < b) for $1 \le i \le b - 1$:

$$s_1(i) = (ib, i(b-a), ib),$$

$$s_2(i) = (b, ib, ib+a),$$

$$s_3(i) = ((i+1)b, i(b-a), ib+a)$$

Since none of these solutions is monochromatic (by assumption), we have, for i = 1, ..., b - 1, in order, the following sequence of implications:

- $s_1(i)$ not monochromatic implies $\chi(i(b-a)) = 1;$
- $s_2(i)$ not monochromatic implies $\chi(ib + a) = 1;$
- $s_3(i)$ not monochromatic implies $\chi((i+1)b) = 0$.

Since $b, 2b, ..., b^2$ are all of color 0 and a < b, we have a monochromatic solution given by $(b^2, b(b-a), b^2)$, a contradiction.

Case 3. $2 \le b < a$. Let $n = a^2 + a + 1$. We leave it to the reader in Exercise 9.7 to verify that a valid 2-coloring of [1, n - 1] is given by coloring b[1, a] one color and its complement the other color.

For the upper bound, assume, for a contradiction, that there exists $\chi : [1, n] \to \{0, 1\}$ that is valid. Clearly, there exists a maximal k such that, for all $i, 1 \leq i \leq k$, we have $\chi(ib) = 0$. We may rule out k > a, since, in this case, $(b^2, b, (a + 1)b)$ would be a monochromatic solution.

We now show that $k \ge a - 1$. Assume, for a contradiction, that $\chi(kb) = 0$ and $\chi((k+1)b) = 1$ with k < a - 1. To avoid the triple ((k+1)b, 1, 1 + (k+1)a) being monochromatic, either $\chi(1) = 0$ or $\chi(1 + (k+1)a) = 0$.

First, consider $\chi(1) = 0$. Let $i \in \{1, 2, \ldots, k\}$. We must have $\chi(1 + ia) = 1$ (so that (ib, 1, 1 + ia) is not monochromatic). Next, so that ((k + 1)b, 1 + ia, 1 + (k + i + 1)a) is not monochromatic, we have $\chi(1 + (k + i + 1)a) = 0$, provided $k + i \leq a$. This implies that $\chi((k + i + 1)b) = 1$, so that ((k + i + 1)b, 1, 1 + (k + i + 1)a) is not monochromatic. Now, provided $k + i \leq a - 1$, we have that $\chi((k + 2)b) = 1$ implies $\chi(1 + (k + i + 2)a) = 0$, so that the solution ((k+2)b, 1+ia, 1+(k+i+2)a) is not monochromatic. Hence, we have $\chi(1+(k+2)a) = \chi(1+(k+3)a) = 0$ (where $1+(k+3)a \leq 1+(a+1)a$)

gince k < a-1). Thus, (b, 1+(k+2)a, 1+(k+3)a) is a monochromatic golution, a contradiction. So, if $\chi(1) = 0$, then $k \ge a-1$.

Next, consider $\chi(1 + (k+1)a) = 0$. We may assume $\chi(1) = 1$. So that the solution (ib, 1 + (k - i + 1)a, 1 + (k + 1)a) is not monochromatic, we have $\chi(1 + ia) = 1$ for $1 \le i \le k$. Now, because ((k+1)b, 1+ia, 1+(k+i+1)a) cannot be monochromatic for i = 0, 1, we have $\chi(1 + (k + 1)a) = \chi(1 + (k + 2)a) = 0$. This gives the monochromatic solution (b, 1 + (k+1)a, 1 + (k+2)a), a contradiction. Hence, if $\chi(1 + (k + 1)a) = 0$, then $k \ge a - 1$.

Since $k \ge a - 1$, from the above two paragraphs we have that $\chi(1 + ia) = 1$ and $\chi(ib) = 0$ for $1 \le i \le a - 1$. Because gcd(a, b) = 1, $\{ib \pmod{a}: 1 \le i \le a - 1\}$ is a complete residue system modulo a. Thus, there exists $x \in [1, a - 1]$ such that $xb \equiv 1 \pmod{a}$. Since 1 < xb < ab, we have $xb \in \{1 + ia : 1 \le i \le a - 1\}$. This is a contradiction, since $\chi(1 + ia) = 1$ for all $1 \le i \le a - 1$, while $\chi(xb) = 0$.

Note that as an immediate consequence of Theorem 9.11, we also know the value of r(ax + by = bz) when $gcd(a, b) \neq 1$. To see this, let gcd(a, b) = g, divide both sides of ax + by = bz by g to obtain $\frac{a}{g}x + \frac{b}{g}y = \frac{b}{g}z$ with $(\frac{a}{g}, \frac{b}{g}) = 1$, and then apply Theorem 9.11. This same analysis may also be applied to the next theorem.

Theorem 9.12. Let $a, b \ge 1$ with gcd(a, b) = 1. Define n(a, b) to be the least positive integer such that every 2-coloring of [1, n(a, b)] admits a monochromatic solution to ax + by = (a + b)z, where x, y, z are distinct. Then $n(a, b) \le 4(a + b) + 1$.

Proof. By means of a computer search, it can be shown that n(1,2) = 13, n(1,3) = 11, n(1,4) = 19, n(1,5) = 25, n(1,6) = 29, and n(2,3) = 21. Furthermore, n(1,1) = w(3;2) = 9. These values show that $n(a,b) \le 4(a+b)+1$ for a+b < 8. Hence, in the remainder of the proof we assume that $a+b \ge 8$. Let \mathcal{E} represent the equation ax + by = (a+b)z.

Assume, for a contradiction, that $\chi : [1, 4(a+b)+1] \rightarrow \{red, blue\}$ is a coloring that admits no monochromatic solution to \mathcal{E} .

Consider

 $S = \{1 + xa + yb : 0 \le x, y \le 4\} \subseteq [1, 4(a + b) + 1].$

We first show that the elements of S are distinct. Assume, for a contradiction, that there exist $i_1, i_2, j_1, j_2 \in [0, 4]$ with $(i_1, j_1) \neq (i_2, j_2)$ such that. Since $(i_1 - i_2)a = (j_2 - j_1)b$ and gcd(a, b) = 1, we have $i_1 - i_2 = kb$ and $j_2 - j_1 = ka$ for some $k \ge 1$. By the restriction on S, $i_1 - i_2$ and $j_2 - j_1$ cannot both equal 4, and thus $i_1 - i_2 + j_2 - j_1 \le 7$, a contradiction since $i_1 - i_2 + j_2 - j_1 \ge a + b \ge 8$.

Rewriting \mathcal{E} as

$$a(x-z) = b(z-y),$$

we see that (x', y', z') is a solution if and only if it has the form (d + kb, d - ka, d) for some $d, k \ge 1$ (provided $d + kb \le 4(a + b) + 1$ and $1 \le d - ka$). Hence, for any $k \in \mathbb{Z}^+$, if

$$1 \le 1 + x'a + y'b - ka < 1 + x'a + y'b + kb \le 4(a+b) + 1,$$

then

$$\{1 + (x' - k)a + y'b, 1 + x'a + y'b, 1 + x'a + (y' + k)b\}$$

cannot be monochromatic (or else we would have a contradiction).

Before going on with the proof, we introduce some notation: we denote by $\lambda_{(s,t)}$ the integer 1 + sa + tb. From the previous paragraph we have the following fact.

Fact 1. For $0 \le j \le i \le k$, $k \ne 0$, $\lambda_{(i-k,j)}$, $\lambda_{(i,j)}$, and $\lambda_{(i,j+k)}$ cannot all have the same color.

We call the three integers $\lambda_{(i-k,j)}$, $\lambda_{(i,j)}$, and $\lambda_{(i,j+k)}$ an isosceles triple since, associating $\lambda_{(s,t)}$ with the point (s, t), the associated points form an isosceles (right) triangle in the plane:



9.2. Some Rado Numbers

Applying Fact 1, we have the following two facts.

Fact 2. $\lambda_{(0,0)}$, $\lambda_{(2,2)}$, and $\lambda_{(4,4)}$ cannot all have the same color. **Fact 3.** For $i \in \{0, 1, 2\}$, $\lambda_{(i,i)}$, $\lambda_{(i+1,i+1)}$, and $\lambda_{(i+2,i+2)}$ cannot all have the same color.

Fact 2 holds because otherwise $\lambda_{(0,0)}$, $\lambda_{(2,2)}$, and $\lambda_{(4,4)}$ are all the same color, say red. By Fact 1, this implies that $\lambda_{(2,0)}$, $\lambda_{(4,0)}$ and $\lambda_{(4,2)}$ are all blue, contradicting Fact 1. To prove Fact 3, assume it is false. Then for some $i \in \{0, 1, 2\}$, the integers $\lambda_{(i,i)}$, $\lambda_{(i+1,i+1)}$, and $\lambda_{(i+2,i+2)}$ all have the same color, say red. By Fact 1, this implies that $\lambda_{(i+1,i)}$, $\lambda_{(i+2,i)}$, and $\lambda_{(i+2,i+1)}$ must all be blue, contradicting Fact 1.

Consider all possible 2-colorings of the set

$$T = \{\lambda_{(x,x)} : 0 \le x \le 4\} \subseteq S.$$

By the pigeonhole principle, one color, say red, must occur at least three times. From Facts 1, 2, and 3 the only possible colorings of

 $(\chi(\lambda_{(0,0)}),\chi(\lambda_{(1,1)}),\chi(\lambda_{(2,2)}),\chi(\lambda_{(3,3)}),\chi(\lambda_{(4,4)}))$

(using r for red and b for blue) belong to

 $\{(r, r, b, r, b), (b, r, r, b, r), (r, b, b, r, r), (r, r, b, r, r)\}.$

We illustrate these four colorings graphically as follows, associating $\lambda_{(s,t)}$ with the point (s,t), and using the standard (x,y)-axes.

b	r	r	r
r_{-}	<i>b</i> _	r_{-}	r_{-}
b	<i>r</i>	<i>b</i>	<i>b</i>
r	r	b	<i>r</i>
r	b	r	r
(1)	(2)	(3)	(4)

We finish the proof graphically by determining, for each of the colorings, the colors of most of the remaining elements of T. The steps consist solely of avoiding monochromatic isosceles triples (see Fact 1). By obtaining the colors of enough of the elements, we will arrive at a contradiction in each case.

Coloring 1.

b	b	b	b	b
<i>r</i> _	r _	r _	r _	r r
b	<i>b</i> _ <i>r</i>	b r	<i>b_r</i>	$b_{-}r$
r	\rightarrow $r_{-}b_{-}$	r_b_	\rightarrow $r_{-}b_{-}$	$r _ b r$
r	r b _ b _	$r b r b_{-}$	$r \ b \ r \ b \ b$	r b r b b

Regardless of how we color the remaining points, we have a monochromatic isosceles triple, a contradiction.

Coloring 2.

r	r	r	r	r
<i>b</i> _	<i>b</i> _	$b \ r$	b r	b r
r	$r_{-}b$	$r_{-}b$	$r_{-}b$	$r_{-}b$
r	$\rightarrow r b _ b \rightarrow$	rbrb	$\overrightarrow{r} b r b \overrightarrow{r} b$	r b r b
b	<i>b r</i> _	b r _	<i>b</i> _ <i>b r</i> _	b x b r r

Whichever color we assign to x, we have a monochromatic isosceles triple, a contradiction.

Coloring 3.

r	r	r	r	r
r _	$r \ b$	rb	$r \ b$	r b
b	<i>b</i>	b	b) <i>b _ x</i>
b	$\rightarrow b r_{} \rightarrow$	b r _ r	$\rightarrow b r_{-} r$	$\rightarrow brrr$
r	$r _ _ b b$	$r r _ b b$	rrbbb	rrbbb

We again obtain a monochromatic isosceles triple for each coloring of x, a contradiction.

Coloring 4.

$$\begin{array}{cccc} r & r \\ r_{-} & r_{-} \\ b_{--} & & b_{--} \\ r_{---} & & r_{--} x \\ r_{---} & & r b_{-} b b \end{array}$$

For either coloring of x we have a monochromatic isosceles triple, a contradiction.

Since we are led to a contradiction using each of the possible colorings, the proof is complete. $\hfill \Box$

Before moving on to the next equation, we mention a couple of examples that are interesting consequences of Theorem 9.12 and its proof.

Example 9.13. In this book we focus on coloring integers; however, we may also color the set of real numbers by using *r*-colorings with domain \mathbb{R} and range $\{1, 2, \ldots, r\}$. We can, of course, define an *r*-coloring on any set, for example the real plane \mathbb{R}^2 . Using this domain, as an immediate consequence of the proof of Theorem 9.12, we have the following, rather formidable sounding, result: any 2-coloring of \mathbb{R}^2 must admit an isosceles right triangle with vertices all of the same color. (In fact, any 2-coloring of $\{(x, y) : x, y \in \mathbb{Z}, 0 \le y \le x \le 4\}$ admits such a triangle, but this may not seem as grand a statement.)

Example 9.14. (A Tic-Tac-Toe Type Triangle) A 2-player game is played on a board B, where $B = \{(x, y) : 0 \le y \le x \le 4\}$. Graphically, B looks like

				-
		-	-	-
	-	-	-	-
-	-		-	

This game is played on the above board, with one player being X and one player being O. The object is to be the first player to create an isosceles right triangle with the same orientation as the board (i.e., the right angle is to the bottom right) with all vertices marked by the player's letter (X or O). As we have seen from the proof of Theorem 9.12, unlike the standard Tic-Tac-Toe game, it is impossible not to have a winner.

The next theorem refines Theorem 9.12 for the cases when a = 1 and $a \leq b$.

Theorem 9.15. Let $b \ge 1$ and define n(b) to be the least positive integer such that for any 2-coloring of [1, n(b)], there is a monochromatic solution to x + by = (1 + b)z with x, y, z distinct. Then

$$n(b) = 4b + 5 if 4 \nmid b$$

and

246

$$n(b) \ge 4b + 3$$
 if $4|b$.

Proof. First note that if $4 \nmid b$, then $n(b) \leq 4b + 5$ is immediate from Theorem 9.12.

To establish the desired lower bounds, we consider three cases. For each case we give a valid 2-coloring of the appropriate interval, and leave it to the reader, in Exercise 9.9, to verify that these colorings are indeed valid.

Case 1. 4|b. Let b = 4k. We want to find a valid 2-coloring of [1, 4(1+b)-2] = [1, 16k+2]. Let $S = [4k+1, 8k+1] \cup [8k+3, 12k+3]$. Define the coloring α on [1, 4b+2] as follows:

$$\alpha(i) = \begin{cases} 0 & \text{if } i \in S, i \text{ even,} \\ 0 & \text{if } i \notin S, i \text{ odd,} \\ 1 & \text{otherwise.} \end{cases}$$

Case 2. b is odd. Let T = [1, 2b + 2]. Define the coloring β on [1, 4(1+b)] by

$$\beta(i) = \begin{cases} 0 & \text{if } i \in T, i \text{ even,} \\ 0 & \text{if } i \notin T, i \text{ odd,} \\ 1 & \text{otherwise.} \end{cases}$$

Case 3. $b \equiv 2 \pmod{4}$. Define γ on [1, 4(1+b)] as

$$\gamma(i) = \begin{cases} 0 & \text{if } i \equiv 1 \pmod{4} \text{ or } i \equiv 2 \pmod{4}, \\ 1 & \text{if } i \equiv 0 \pmod{4} \text{ or } i \equiv 3 \pmod{4}. \end{cases}$$

9.2. Some Rado Numbers

By Theorems 9.2 and 9.4, the Rado numbers given in the last three theorems are guaranteed to exist since the corresponding equations each satisfy the subset condition of Theorem 9.2. We have seen in Theorem 9.8 that some equations that do not satisfy the requirements of Theorem 9.10 (and hence are not regular) are, in fact, r-regular for some $r \ge 2$. In the following three theorems, we consider some homogeneous equations that do not satisfy the subset condition of Theorem 9.2 (which is needed for regularity) that are, nevertheless, 2-regular.

Theorem 9.16. For $a \ge 1$, $r(ax + ay = z) = a(4a^2 + 1)$.

Proof. Let \mathcal{E} represent the equation ax + ay = z. To show that $r(ax + ay = z) \ge a(4a^2 + 1)$, we give a 2-coloring of $[1, a(4a^2 + 1) - 1]$ with no monochromatic solution to \mathcal{E} . Color $[2a, 4a^2 - 1]$ one color and its complement a different color. It is left to the reader in Exercise 9.10 to show that no monochromatic solution exists under this coloring.

We complete the proof by showing that any given 2-coloring of $[1, a(4a^2+1)]$ must admit a monochromatic solution to \mathcal{E} . Assume, for a contradiction, that $\chi : [1, a(4a^2+1)] \rightarrow \{0, 1\}$ admits no monochromatic solution.

We may assume, without loss of generality, that $\chi(1) = 0$. Therefore, $\chi(2a) = 1$. Define $k \ge 2$ to be the maximal integer so that $\chi([1, k - 1]) = 0$. Note that we must have $k \le 2a$, and that $\chi(k) = 1$. This implies that $\chi(2ak) = 0$, since (x, y, z) = (k, k, 2ak) is a solution. Since $\chi(1) = \chi(k - 1) = 0$, we have $\chi(ak) = 1$ (to avoid (1, k - 1, ak)being of color 0). Since $\chi(ak) = 1$, we have $\chi(2a^2k) = 0$. This, in turn, gives $\chi(2ak - (k - 1)) = 1$, so that $(k - 1, 2ak - (k - 1), 2a^2k)$ is not of color 0. To avoid (2ak - (k - 1), k, a(2ak + 1)) being of color 1, we have $\chi(a(2ak + 1)) = 0$. Therefore, (1, 2ak, a(2ak + 1)) is a monochromatic solution, a contradiction.

Theorem 9.17. For $a \ge 1$, $r(ax + ay = 2z) = \frac{a(a^2+1)}{2}$.

Proof. Let \mathcal{E} represent the equation ax + ay = 2z. Let n(a,b) = r(ax + ay = bz) and let n = n(a, 2). From Theorem 9.16, if a is even we have $n = \frac{a(a^2+1)}{2}$, since $n = n(\frac{a}{2}, 1)$. Furthermore, for a = 1 we

have the trivial monochromatic solution x = y = z = 1. Hence, we assume that $a \ge 3$ is odd.

To show that $n \geq \frac{a(a^2+1)}{2}$, we will give a 2-coloring of the interval $I = \begin{bmatrix} 1, \frac{a(a^2+1)}{2} - 1 \end{bmatrix}$ with no monochromatic solution to \mathcal{E} . Color $[a, a^2 - 1]$ red and its complement in I blue. It is left to the reader in Exercise 9.12 to show that under this coloring there is no monochromatic solution to \mathcal{E} .

To finish the proof we show that $n \leq \frac{a(a^2+1)}{2}$. Assume, for a contradiction, that there exists a 2-coloring $\chi : [1, \frac{a(a^2+1)}{2}] \to \{0, 1\}$ that admits no monochromatic solution to \mathcal{E} . Then neither (1, 1, a) nor (a, a, a^2) is monochromatic; therefore 1 and a^2 must have the same color. We may assume that $\chi(1) = \chi(a^2) = 0$ and $\chi(a) = 1$.

Consider the solutions

$$s_1(j) = (1, a^2 - (a-1)j, \frac{a}{2}(a^2 - (a-1)j+1)), \text{ and}$$

$$s_2(j) = (1, a^2 - (a-1)(j+1), \frac{a}{2}(a^2 - (a-1)j+1))$$

for $0 \leq j \leq a - 1$.

Denote $s_i(j)$ by $(1, s_i^{(2)}(j), s_i^{(3)}(j))$ for i = 1, 2 and $0 \le j \le a - 1$. Since none of these solutions $s_i(j)$ may be monochromatic, we have, for $j = 0, 1, \ldots, a-2$, in order, the following sequence of implications:

$$s_1(j) \text{ and } s_2(j) \text{ not monochromatic implies } \chi\left(s_1^{(3)}(j)\right) = 1$$
$$\chi\left(s_1^{(3)}(j)\right) = 1 \text{ implies } \chi\left(s_2^{(2)}(j)\right) = 0,$$
$$\chi\left(s_2^{(2)}(j)\right) = 0 \text{ implies } \chi\left(s_1^{(3)}(j+1)\right) = 1.$$

Since $\chi\left(s_2^{(2)}(a-1)\right) = \chi\left(a^2 - (a-1)^2\right) = 0$, we have that $\left(1, a^2 - (a-1)^2, a^2\right)$ is a monochromatic solution, a contradiction.

The above two theorems are implied by the the following, broader, result, which gives the value of r(ax + ay = bz) for all a and b. We present this without proof.

Theorem 9.18. Assume gcd(a, b) = 1 and let r = r(ax + ay = bz). Then

$$r = \begin{cases} a(4a^2 + 1) & \text{if } b = 1, \\ \frac{a(a^2 + 1)}{2} & \text{if } b = 2, \\ 9 & \text{if } b = 3 \text{ and } a = \\ 10 & \text{if } b = 3 \text{ and } a = \end{cases}$$

1,

2.

For b = 3 and $a \ge 4$,

$$r = \begin{cases} \frac{a(4a^2+2a+3)}{9} & \text{if } a \equiv 1 \pmod{9}, \\ \frac{a(4a^2+a+9)}{9} & \text{if } a \equiv 2 \pmod{9}, \\ \frac{a(4a^2+2a+9)}{9} & \text{if } a \equiv 4 \pmod{9}, \\ \frac{a(4a^2+4a+6)}{9} & \text{if } a \equiv 5 \pmod{9}, \\ \frac{a(4a^2+5a+3)}{9} & \text{if } a \equiv 7 \pmod{9}, \\ \frac{a(4a^2+a+6)}{9} & \text{if } a \equiv 8 \pmod{9}. \end{cases}$$

For $b \geq 4$,

$$r = \begin{cases} r = \binom{b+1}{2} & \text{if } 1 \le a \le \frac{b}{4}, \\ r = \lceil \frac{b}{2} \rceil & \text{if } \frac{b}{4} < a < \frac{b}{2}, \\ r = ab & \text{if } \frac{b}{2} < a < b, \\ r = \lceil \frac{a^2}{b} \rceil a & \text{if } b < a. \end{cases}$$

9.3. Generalizations of the Single Equation Theorem

We present two main results in this section, one without proof, both of which generalize Rado's single equation theorem (Theorem 9.2). The way that the first result generalizes Theorem 9.2 is similar to the way in which the Ramsey numbers $R(k_1, k_2, \ldots, k_r)$ generalize $R_r(k)$. Recall, for example (taking r = 2), that $R_2(k) = R(k, k)$ is the least positive integer such that every 2-coloring of the edges of the complete graph on R(k, k) vertices admits a monochromatic complete graph on k vertices. Meanwhile, the more general Ramsey number $R(k, \ell)$ denotes the least positive integer such that for every red-blue coloring of the edges on $R(k, \ell)$ vertices, there is either a red complete graph on k vertices or a blue complete graph on ℓ vertices. The numbers R(k, k) are called *diagonal* Ramsey numbers, and when $k \neq l$, $R(k, \ell)$ is called an *off-diagonal* Ramsey number

Now consider Rado's single equation theorem. The theorem guarantees a monochromatic solution to certain equations. This has a flavor similar to that of the diagonal Ramsey numbers, and we call the analogous Rado numbers *diagonal* Rado numbers. Now, if we have two different homogeneous linear equations, is it true that under any red-blue coloring of \mathbb{Z}^+ there must be either a red solution to the first equation or a blue solution to the second? What we are considering here are *off-diagonal* Rado numbers (if they exist). The next theorem proves their existence and is easily derived from the proof of Theorem 9.2.

Theorem 9.19. Let $r \geq 1$. For $1 \leq j \leq r$, let $n_j \geq 2$ and let \mathcal{E}_j represent the equation $\sum_{i=1}^{n_j} c_i^{(j)} x_i = 0$. If each \mathcal{E}_j , $1 \leq j \leq r$, is *r*-regular, then there is a least positive integer *n* so that for every *r*-coloring of [1, n], there exists $t \in \{1, 2, \ldots, r\}$ such that \mathcal{E}_t has a solution of color *t*.

If a set S of r homogeneous equations satisfies the conditions of Theorem 9.19, we will say that S is r-regular.

Before delving into the proof of this theorem, we look at an example to help solidify the implication of Theorem 9.19.

Example 9.20. Consider the three equations given below:

$$\begin{aligned} -x_1 + 2x_2 + x_3 &= 0, \\ 3x_1 - x_2 + x_3 + 7x_4 &= 0, \\ 2x_1 + 4x_2 - 3x_3 + x_4 + 6x_5 &= 0 \end{aligned}$$

Each of these equations is regular by Rado's theorem, since each has a nonempty subset of coefficients that sums to zero: (using the notation of Theorem 9.19) $c_1^{(1)} + c_3^{(1)} = 0$, $c_2^{(2)} + c_3^{(2)} = 0$, and $c_1^{(3)} + c_3^{(3)} + c_4^{(3)} = 0$. Applying Theorem 9.19, we are guaranteed that any 3-coloring, say red-blue-green, of \mathbb{Z}^+ must contain one of the following:

a red solution to $-x_1 + 2x_2 + x_3 = 0;$

a blue solution to $3x_1 - x_2 + x_3 + 7x_4 = 0;$

a green solution to $2x_1 + 4x_2 - 3x_3 + x_4 + 6x_5 = 0$.

We now present a sketch of the proof of Theorem 9.19. The proof is very similar to that of Theorem 9.2 so we leave some justification to the reader as Exercise 9.13.

Proof of Theorem 9.19 (sketch). Let $S = \{\mathcal{E}_1, \mathcal{E}_2, \ldots, \mathcal{E}_r\}$. The proof is by induction on r. The case r = 1 holds by Theorem 9.2. Let n = n(S; r - 1) be the least positive integer such that for any (r - 1)-coloring of [1, n] there is a solution to \mathcal{E}_i of color i for some $i \in \{1, 2, \ldots, r - 1\}$.

Assume that for each $j, 1 \leq j \leq r$,

$$c_1^{(j)} + c_2^{(j)} + \dots + c_{m_j}^{(j)} = 0$$

with $m_j < n_j$ maximal, and that $s_j = c_{m_j+1}^{(j)} + c_{m_j+2}^{(j)} \cdots + c_{n_j}^{(j)}$ is a nonempty, nonzero sum (this needs to be justified).

Let $b_j = \sum_{i=1}^{n_j} |c_i^{(j)}|$. Define $b = \prod_{j=1}^r b_j$ and $s = \prod_{j=1}^r |s_j|$. We claim that $n(\mathcal{S}; r) \leq b^r w(n+1; r)$. We show that any *r*-coloring of $[1, b^r w(n+1; r)]$ must admit a solution to \mathcal{E}_i of color *i*, for some $i \in \{1, 2, \ldots, r\}$.

Let $\chi : [1, b^r w(n+1; r)] \to \{0, 1, \ldots, r\}$. Let $x_2 = x_3 = \cdots = x_{m_j}$ and $x_{m_j+1} = x_{m_j+2} = \cdots = x_{n_j}$ for each $j, 1 \leq j \leq r$. Hence, we can write \mathcal{E}_j , for $1 \leq j \leq r$, under these conditions, as

$$c_1^{(j)}(x_1 - x_2) + s_j x_{m_j+1} = 0.$$

By Theorem 4.9, χ admits an (n+1)-term monochromatic arithmetic progression with gap a multiple of s. Let $\{a, a+ds, \ldots, a+nds\}$ be one such arithmetic progression of color, say, t. If there exists $j \in \frac{s}{s_t}[1,n]$ such that $\chi\left(jdc_1^{(t)}\right) = t$, then we are done (this needs to be justified). If, on the other hand, for all $j \in \frac{s}{s_t}[1,n]$ we have $\chi\left(jdc_1^{(t)}\right) \neq t$, then $dc_1^{(t)}\frac{s}{s_t}[1,n]$ is (r-1)-colored (verify that $dc_1^{(t)}\frac{s}{s_t}n \leq b^r w(n+1;r)$). Let $S' = S - \mathcal{E}_t$. Since we have an (r-1)-coloring of $\frac{s}{s_t}[1,n]$ using the colors $1, 2, \ldots, t-1, t+1, t+2, \ldots, r$, by the inductive assumption there exists $c \in [1,r] - \{t\}$ such that \mathcal{E}_c has a solution of color c. This completes the proof.

Example 9.21. Let $S = \{\mathcal{E}_1, \mathcal{E}_2, \dots, \mathcal{E}_r\}$ with $\mathcal{E}_1 = \mathcal{E}_2 = \dots = \mathcal{E}_r$. Then Theorem 9.19 reduces to Theorem 9.2. Just as Rado's single equation theorem was strengthened to show the existence of monochromatic solutions consisting of distinct integers, Theorem 9.19 can be strengthened in the same way. This is given by the next theorem, which we state without proof.

Theorem 9.22. Let $r \geq 1$. Let $S = \{\mathcal{E}_1, \mathcal{E}_2, \ldots, \mathcal{E}_r\}$, where the \mathcal{E}_i 's are defined as in Theorem 9.19. There exist $j, 1 \leq j \leq r$, and distinct $b_i, 1 \leq i \leq n_j$, of color j, that satisfy \mathcal{E}_j , if and only if S is r-regular and for all $j, 1 \leq j \leq r$, there exist distinct $y_i^{(j)} \in \mathbb{Z}, 1 \leq i \leq n_j$, that satisfy \mathcal{E}_j .

As an interesting application of Theorem 9.22, we look at the following example.

Example 9.23. Consider the set of equations $\{x+y=z, x+y=2z\}$. By Theorem 9.22, there exists a minimal integer n such that any redblue coloring of [1, n] must contain either a red solution to x + y = z(of distinct integers) or a blue solution to x + y = 2z (of distinct integers). In other words, there must exist either a red Schur triple or a blue 3-term arithmetic progression. In fact, n = 8. To see that $n \ge 8$, color 1, 4, 7 red, and color 2, 3, 5, 6 blue. The fact that every 2-coloring of [1, 8] admits either a red Schur triple or a blue 3-term arithmetic progression is left to the reader as Exercise 9.4.

We now state, without proof, the full version of Rado's theorem, for which Rado's single equation theorem (Theorem 9.2) is a special case. This is a very powerful theorem, as it describes precisely when we are guaranteed monochromatic solutions to a homogeneous *system* of linear equations. In order to state Rado's full theorem clearly, we make the following definition.

Definition 9.24. Let $C = (\vec{c_1} \ \vec{c_2} \ \dots \ \vec{c_n})$ be a $k \times n$ matrix, where $\vec{c_i} \in \mathbb{Z}^k$ for $1 \le i \le n$. We say that C satisfies the columns condition if we can order the columns $\vec{c_i}$ in such a way that there exist indices $i_0 = 1 < i_1 < i_2 < \dots < i_t = n$ such that the following two conditions hold for $\vec{s_j} = \sum_{i_{j-1}+1}^{i_j} \vec{c_i} \ (2 \le j \le t)$:

1. $\vec{s}_1 = \vec{0}$.

2. \vec{s}_j can be written as a linear combination of $\vec{c}_1, \vec{c}_2, \ldots, \vec{c}_{i_{j-1}}$ for $2 \leq j \leq t$.

9.3. Generalizations of the Single Equation Theorem 253

To help clarify this rather cumbersome definition, we look at a couple of examples.

Example 9.25. Consider the following matrix:

To see that this matrix satisfies the columns condition, rearrange the columns to obtain

$\begin{pmatrix} 1 \end{pmatrix}$	2	-3	0	2	
0	2	-2	-1	1	
$\setminus 4$	-5	1	2	6 /	

Using the notation of Definition 9.24, take t = 2, $i_1 = 3$, and $i_2 = 5$. Then $\vec{s}_1 = \vec{0} \in \mathbb{Z}^3$ and $\vec{s}_2 = (2 \ 0 \ 8)^t = 2\vec{c}_1$ (where t denotes the transpose).

Example 9.26. The following matrix does not satisfy the columns condition:

1	1	3	0	-3	1
1	-2	$^{-1}$	2	7	0
ł	5	3	-5	2	-3
$\left(\right)$	1	-2	3	4	-5 /

By inspection (check!) we see that there does not exist a set of columns that sums to the zero vector.

The key to comprehending Rado's full theorem is having a clear understanding of the columns condition. We now state Rado's full theorem. We do not include its proof, as it is beyond the scope of this book.

Theorem 9.27 (Rado's Full Theorem). Let S be a system of linear homogeneous equations. Write S as $A\vec{x} = \vec{0}$. S is regular if and only if A satisfies the columns condition. Furthermore, S has a monochromatic solution of distinct positive integers if and only if S is regular and there exist distinct (not necessarily monochromatic) integers that satisfy S.

9.4. Exercises

Rado's full theorem opens the door to many new Ramsey-type numbers, one of which is presented in the next two examples.

Example 9.28. For $r \ge 1$, there exists n = n(r) such that under any *r*-coloring of [1, n] we must have both a monochromatic Schur triple and a monochromatic, of the same color, 3-term arithmetic progression.

To show the existence of such a monochromatic structure, consider the following system in the variables x_1, x_2, \ldots, x_6, y :

$$x_1 + x_2 = x_3,$$

 $x_4 - x_5 = y,$
 $x_5 - x_6 = y.$

If x_i , $1 \le i \le 6$, are all the same color, then $x_1 + x_2 = x_3$ is a monochromatic Schur triple and x_6, x_5, x_4 is a monochromatic 3-term arithmetic progression with gap y.

Writing the above system in matrix form (using the notation of Definition 9.24 and Theorem 9.27) with $\vec{x} = (x_1 \ x_2 \ \dots x_6 \ y)^t$, we get $C\vec{x} = \vec{0}$, where

	(1	1	-1	0	0	0	0 \	
C =		0	0	0	1	-1	0	-1	
	l	0	0	0	0	1	-1^{1}	-1 /	

Rearranging the rows of C, we have

	$\begin{pmatrix} 1 \end{pmatrix}$	-1	0	0	0	1	0 \	
$(\vec{c}_1 \ \vec{c}_2 \ \dots \ \vec{c}_7) =$	0	0	1	-1	0	0	-1	
	0 /	0	0	1	-1	0	-1 /	

Since $\sum_{i=1}^{5} \vec{c}_i = \vec{0} \in \mathbb{Z}^3$, $\vec{c}_6 = \vec{c}_1$, and $\vec{c}_7 = \vec{c}_4 + 2\vec{c}_5$, we have satisfied the columns condition. By Rado's full theorem we know that for any finite coloring of \mathbb{Z}^+ , there is a monochromatic solution to the given system of equations. Hence, we have a monochromatic Schur triple and a monochromatic 3-term arithmetic progression (with gap y) of the same color. Note that, in addition, y has this same color.

Example 9.29. Let n be the least positive integer such that any 2-coloring of [1, n] must contain a Schur triple and a 3-term arithmetic

progression, both monochromatic of the same color. We show that $n \leq 16$.

We know that s(2) = 5 and w(3) = 9, where s and w are the usual Schur and van der Waerden functions, respectively. Clearly, in any red-blue coloring of [1, 16] there exist both a monochromatic Schur triple and a monochromatic 3-term arithmetic progression. Assume, for a contradiction, that the colors of every such pair of triples are different.

Let a, a + d, a + 2d be, say, a red 3-term arithmetic progression. Since a + a = 2a, a + (a + d) = 2a + d, and a + (a + 2d) = 2a + 2d, we must have 2a, 2a + d, and 2a + 2d all be blue in order to avoid a red Schur triple. Note that $2a + 2d \le 16$, since w(3) = 9 implies that $a + 2d \le 9$ and $a \le 7$.

Hence, we have that 2a, 2a + d, 2a + 2d is a blue 3-term arithmetic progression, a contradiction since we now have a Schur triple and a 3-term arithmetic progression with the same color.

In this section we have seen two generalizations of Rado's single equation theorem: Theorem 9.19 and Rado's full theorem. Rado's full theorem is clearly more powerful, as it completely classifies regular systems. Its strength is also made apparent by comparing Examples 9.23 and 9.29.

9.4. Exercises

- **9.1** Prove that in the statement of Theorem 9.2 we may take $\mathcal{E} \in \mathbb{Q}[x]$, i.e., the coefficients may be rational.
- 9.2 Which of the following equations are regular?
 - a) x y = 7z;b) $r_1 + 2r_2 + 3r_3 + 4r_4 - 5r_5$

$$x_1 + 2x_2 + 3x_3 + 4x_4 = 5x_5;$$

$$x_1 - 2x_2 + 4x_3 - 8x_4 + 16x_5 - 32x_6 = 0;$$

d)
$$\frac{1}{3}x_1 - \frac{1}{4}x_2 + 2x_3 - \frac{1}{12}x_5 = 0.$$

- **9.3** Give two different equation(s)/system(s) whose *r*-regularity proves the existence of monochromatic 3-term arithmetic progressions in any *r*-coloring of \mathbb{Z}^+ .
- **9.4** Show that the minimal integer n such than any red-blue coloring of [1, n] must admit either a red Schur triple, or a blue

255

3-term arithmetic progression is n = 8. The fact that $n \ge 8$ comes from the coloring given in Example 9.23.

9.5 The following system is regular according to Rado's full theorem (Theorem 9.27). Prove this. Also, what monochromatic object is guaranteed to exist?

9.6 Let $b \neq 0$. Let $\mathcal{E}(b)$ represent the equation x + y - z = b. Define the 2-coloring $\chi : \left[1, b - \left\lceil \frac{b}{5} \right\rceil\right] \rightarrow \{0, 1\}$ by

$$\chi(x) = \begin{cases} 0 & \text{if } x \in \left[\left\lceil \frac{b}{5} \right\rceil + 1, \left\lfloor \frac{b + \left\lceil \frac{b}{5} \right\rceil}{2} \right\rfloor \right] \\ 1 & \text{otherwise.} \end{cases}$$

Show that χ is a valid coloring for $\mathcal{E}(b)$.

- **9.7** Verify the fact that the three colorings used in the proof of Theorem 9.11 are each valid colorings.
- **9.8** In the proof of Theorem 9.12 it is stated that n(1,2) = 13, n(1,3) = 11, n(1,4) = 19, n(1,5) = 25, and n(2,3) = 21. Verify this. (A computer may be helpful, but is not necessary.)
- **9.9** Verify the fact that the three colorings used in the proof of Theorem 9.15 are each valid colorings. Also, what is the standard notation for n(1) in this context?
- **9.10** Do the following problems concerning the proof of Theorem 9.16.

a) Show that the coloring given for the lower bound is valid. b) The statement that (1, 2ak, a(2ak+1)) is a monochromatic solution, which is given at the end of the proof, is not valid unless $k \leq 2a$. Verify this.

c) What is the standard notation for n(1) in this context?

9.11 Prove that for $a \ge 4$, we have $r(x + y = az) = \binom{a+1}{2}$.

- **9.12** Verify the fact that the coloring given in Theorem 9.17 is valid. Also, for the upper bound, why must we insist that $a \ge 3$?
- 9.13 Fill in the details for the proof of Theorem 9.19.
- **9.14** Deduce the existence of the off-diagonal generalized Schur numbers using one of the theorems in this chapter.
- **9.15** Find the exact value of the number n of Example 9.29.

9.5. Research Problems

9.5. Research Problems

9.1 Find an extension of Rado's full theorem which is analogous to Theorem 9.19's extension of Rado's single equation theorem.

Reference: [69]

- *9.2 Theorem 9.10 characterizes those nonhomogeneous equations that are not regular. Further characterize these equations by the greatest number, m, of colors for which they are *m*-regular (i.e., by their degrees of regularity).
- **9.3** The exact value of r(ax + by = (a + b)z) is known when a = 1 and $b \not\equiv 0 \pmod{4}$. Determine r(ax + by = (a + b)z) for all other pairs (a, b). Bounds for these remaining pairs are given by Theorems 9.12 and 9.15. Reference: [69]
- *9.4 Determine the 2-color Rado numbers for aw + bx + cy = cz for all a, b, and c. Start with a = b = 1, c > 1. References: [139], [140], [224]
- **9.5** Complete the result analogous to Theorem 9.8 for b < 0 (the references listed directly below indicate which cases remain open).

References: [159], [239]

- *9.6 Extend Theorem 9.8 to $r \ge 3$ colors. Reference: [239]
- *9.7 Let χ be a given red-blue coloring of [1, n]. Define S_{χ} to be the number of red Schur triples in [1, n] under χ , and V_{χ} to be the number of blue 3-term arithmetic progressions in

258

[1, n] under χ . Determine the asymptotic minimum sum of the number of red Schur triples and the number of blue 3term arithmetic progressions; i.e., find $\min(S_{\chi} + V_{\chi})$, where the minimum is over all red-blue colorings χ of [1, n]. References: [75], [225], [248]

9. Rado's Theorem

*9.8 Let χ be a given red-blue coloring of [1, n]. Define S_{χ} to be the number of monochromatic Schur triples in [1, n] under χ , and V_{χ} to be the number of monochromatic 3-term arithmetic progressions in [1, n] under χ . Determine the asymptotic minimum sum of the number of monochromatic Schur triples and the number of 3-term arithmetic progressions; i.e., find min $(S_{\chi} + V_{\chi})$, where the minimum is over all red-blue colorings χ of [1, n].

References: [75], [225], [248]

9.6. References

§9.1. Bergelson, Deuber, and Hindman [29] extend Rado's theorem to hold over finite fields. The proof of Theorem 9.4 can be found in [127, p. 77]. Theorem 9.8 and its proof appear in [239]. Rado [214] classifies those nonhomogeneous linear systems that are regular. In [70], Theorem 9.9 is given. Extensions of Rado's full theorem are given in [32] and [181]. Generalizations to abelian groups can be found in [80] and [81].

§9.2. Burr and Loo [69] present Theorems 9.11, 9.12, and 9.15 with different proofs for 9.11 and 9.12. Theorem 9.16 and Exercises 9.6 and 9.11 may be found in [70]. Harborth and Maasberg prove Theorem 9.17 in [140] and Theorem 9.18 in [139]. Theorem 9.19 is noted, without proof, in [224].

§9.3. Rado's full theorem's (Theorem 9.27) original proofs are in the series of papers [215], [214], and [213]. Other proofs can be found in [77] and [127].

Additional References: For a survey with an extensive reference list, see [77]. In [108], bounds are given for the number of monochromatic solutions in a regular system. Probability results concerning Rado's full theorem can be found in [226]. Deuber [79] gives a strong

generalization of Rado's theorem. An extension of Deuber's result is in [178].

Systems of linear homogeneous inequalities are dealt with in [242]. Values and bounds for such inequalities are given by Schaal in [241], while in [240] he gives values and bounds for linear nonhomogeneous inequalities. Homogeneous systems whose solutions satisfy a linear inequality are presented in [147]. Nonlinear homogeneous equations are studied in [67] and [180].

Chapter 10

Other Topics

The previous chapters were primarily devoted to three classical theorems of Ramsey theory – van der Waerden's theorem, Schur's theorem, and Rado's theorem – and to many generalizations, extensions, and other modifications of these three theorems. There are quite a few interesting topics and problems that are not in the aforementioned category, but which definitely belong to the area of Ramsey theory on the integers. In this chapter we touch upon a few such topics.

10.1. Folkman's Theorem

Folkman's theorem, also known as the Folkman-Rado-Sanders theorem, involves the existence of certain sets of sums that are monochromatic under any finite coloring of the positive integers. We begin with a definition.

Definition 10.1. Let $T \subseteq \mathbb{Z}^+$ be finite. We define

$$\mathcal{S}(T) = \left\{ \sum_{r \in R} r : R \subseteq T, R \neq \emptyset \right\}$$

and call $\mathcal{S}(T)$ the sumset of T.

Note that the sumset of a set T consists only of sums of distinct elements of T. In particular, if $t \in T$, then $2t \notin S(T)$.

261

Example 10.2. Let $T = \{2, 5, 8\}$. Then $S(T) = \{2, 5, 7, 8, 10, 13, 15\}$. The number 2+2=4, or any sum with the same summand appearing more than once, is not a member of S(T).

We now state Folkman's theorem.

Theorem 10.3 (Folkman's Theorem). For all $k, r \ge 1$, there exists a least positive integer F = F(k;r) such that for every r-coloring of [1, F], there is a k-element subset $T \subseteq [1, F]$ such that S(T) is monochromatic.

Note that the statement of Theorem 10.3 is the "finite" form of Folkman's theorem. This is equivalent (by means of the compactness principle) to the "infinite" form: if $r \ge 1$, then for every *r*-coloring of \mathbb{Z}^+ , there are arbitrarily large sets *T* such that $\mathcal{S}(T)$ is monochromatic.

Example 10.4. Consider F(2; r). If n = F(2; r), then for each *r*-coloring of [1, n] there are integers *a* and *b* such that the set $\{a, b, a+b\}$ is monochromatic. This may look familiar: it is the statement of Schur's theorem, which says that for any $r \ge 1$, there is an integer n = s(r) such that for every *r*-coloring of [1, n], there is a monochromatic solution to x + y = z.

The astute reader may notice that Folkman's theorem is a special case of Rado's full theorem (Theorem 9.27); it follows by elementary, although somewhat untidy, means. However, Folkman's theorem is of sufficient independent interest to warrant our providing a different proof. We will use the following lemma (which is similar to Definition 2.33) to prove Theorem 10.3.

Lemma 10.5. For all $k, r \ge 1$, there is an integer n = n(k; r) so that for any r-coloring of [1, n], there exist $x_1 < x_2 < \cdots < x_k \in [1, n]$ with $\sum_{i=1}^{k} x_i \le n$ such that

$$S_t = \left\{ \sum_{r \in R} x_r : R \subseteq [1, k], \max_{r \in R} r = t \right\}$$

is monochromatic for $t = 1, 2, \ldots, k$.

L. Folkman's Theorem

Proof. We use induction on k, with k = 1 being trivial. Let $k \ge 1$, let r be arbitrary, and assume that n(k;r) exists. We will show that $n(k+1;r) \le 2w(n(k;r)+2;r)$, where w(k;r) is the usual van der Waerden function.

Let m = 2w(n(k; r) + 2; r), and consider an arbitrary *r*-coloring of [1, m]. By van der Waerden's theorem, using Proposition 2.29, there is a monochromatic arithmetic progression

$$A = \{a + jd : 0 \leq j \leq n(k;r) + 1\} \subseteq \left[rac{m}{2},m
ight]$$

Now consider the set

$$D = \{d, 2d, \ldots, n(k; r)d\}.$$

Using Proposition 2.30, along with the inductive assumption, there exist $x_1 < x_2 < \cdots < x_k$ in D such that the associated S_t 's are monochromatic for $t = 1, 2, \ldots, k$. Our goal is to find an x_{k+1} so that S_{k+1} is also monochromatic.

We will show that we may take $x_{k+1} = a + d$. Since $a > \frac{m}{2}$ and $a + n(k;r)d \le m$, we see that $n(k;r)d < \frac{m}{2}$. Hence, $a + d > x_k$, since $x_k \in D$. Because $(a + d) + D \subseteq A$, by taking $x_{k+1} = a + d$ we have $x_{k+1} > x_k$ and

$$S_{k+1} \subseteq (a+d) + D \subseteq A.$$

Hence, S_{k+1} is monochromatic, thereby completing the induction. \Box

We now apply the above lemma to prove Folkman's theorem.

Proof of Theorem 10.3. We show that $F(k;r) \le n((k-1)r+1;r)$, where n(k;r) is defined as in Lemma 10.5.

Let $x_1 < x_2 < \cdots < x_{(k-1)r+1}$ satisfy Lemma 10.5, with associated sets $S_1, S_2, \ldots, S_{(k-1)r+1}$. By the pigeonhole principle, k of these sets must have the same color. Denote these k sets by $S_{i_1}, S_{i_2}, \ldots, S_{i_k}$.

Consider $T = \{i_1, i_2, ..., i_k\}$. Then

$$\mathcal{S}(T) = \left\{ \sum_{r \in R} x_r : R \subseteq T, R \neq \emptyset \right\}$$

is monochromatic, since $\max\{r : r \in R\} \in \{i_1, i_2, \dots, i_k\}$.

263

Now that we have proven the existence of F(k;r) for all $k, r \ge 1$, we refer to F(k;r) as a Folkman number. We also offer, after a definition, an immediate corollary of Folkman's theorem.

Definition 10.6. Let $T \subseteq \mathbb{Z}^+$ be finite. Define

$$\mathcal{P}(T) = \left\{ \prod_{r \in R} r : R \subseteq T, R \neq \emptyset \right\}$$

and call $\mathcal{P}(T)$ the product-set of T.

Example 10.7. For $T = \{2, 5, 8\}, \mathcal{P}(T) = \{2, 5, 8, 10, 16, 40, 80\}.$

Corollary 10.8. For all $k, r \ge 1$, there exists a least positive integer $F^* = F^*(k;r)$ such that for every r-coloring of $[1, F^*]$, there is a k-element subset $T \subseteq [1, F^*]$ with $\mathcal{P}(T)$ monochromatic.

Proof. We will show that $F^*(k;r) \leq 2^{F(k;r)}$. Let $n = 2^{F(k;r)}$ and let χ be an *r*-coloring of [1, n]. Consider $\{2^i : 1 \leq i \leq F(k;r)\}$. Let γ be the *r*-coloring of [1, F(k;r)] defined by $\gamma(i) = \chi(2^i)$.

Let $T = \{t_1 < t_2 < \cdots < t_k\} \subseteq [1, F(k; r)]$ be a k-element set such that $\mathcal{S}(T) \subseteq [1, F(k; r)]$ is monochromatic under γ . By the definition of γ , this implies that $Q = \{2^{t_1} < 2^{t_2} < \cdots < 2^{t_k}\}$ is monochromatic under χ . Since $\mathcal{S}(T)$ is monochromatic and

$$\prod_{r \in R} 2^{t_r} = 2^{\sum_{r \in R} t_r}$$

for any $R \subseteq T$, we see that $\mathcal{P}(Q) \subseteq [1, n]$ is monochromatic. \Box

With regard to bounds on the Folkman numbers, we see from the proofs of Theorem 10.3 and Lemma 10.5 that F(k;r) is bounded from above by 2w(t;r) for some suitably large t, where w is the usual van der Waerden function. As such, Gower's bound for the van der Waerden numbers (Remark 2.22) also provides an upper bound for the Folkman numbers. For a lower bound, we turn to a result of Erdős and Spencer, which we offer without proof.

Theorem 10.9. $F(k; 2) > 2^{ck^2/\log k}$ for some fixed constant c > 0.

A question you may have considered, based on the above results, is: for any $k, r \ge 1$, does every r-coloring of \mathbb{Z}^+ admit a k-element subset $T \subseteq \mathbb{Z}^+$, such that $\mathcal{S}(T) \cup \mathcal{P}(T)$ is monochromatic? The answer to this question is unknown, even for k = 2.

10.2. Doublefree Sets

Taking k = 2 in Folkman's theorem, we have a "strengthened" Schur's theorem, which says that, under any given *r*-coloring of \mathbb{Z}^+ , there exist $x \neq y$ such that $\{x, y, x + y\}$ is monochromatic. We call this a strengthened Schur's theorem since we are requiring $x \neq y$, something that is not required in Schur's theorem. So, can we go the other way and require x = y? In other words, does every *r*-coloring of \mathbb{Z}^+ admit a monochromatic set of the form $\{x, 2x\}$? The answer is no. Consider the 2-coloring χ of \mathbb{Z}^+ defined as follows: for $i = 2^j q$, where *q* is odd, let $\chi(i) = 0$ if *j* is even, and $\chi(i) = 1$ if *j* is odd. After a moment of thought, we see that such a coloring is indeed a well-defined coloring of \mathbb{Z}^+ with no monochromatic set of the form $\{x, 2x\}$.

Definition 10.10. A subset $T \subseteq [1, n]$ is called *doublefree* if T contains no set of the form $\{x, 2x\}$.

Since colorings of [1, n] are just partitions of [1, n] into subsets, we consider the question: what is the size of the largest doublefree subset of [1, n]? The following result answers this question.

Theorem 10.11. The largest size of a doublefree subset of [1, n] is |T|, where

$$T = \{u4^i : u \text{ odd and } u4^i \le n\}$$

Proof. Each positive integer can be expressed uniquely in the form $u2^i$, where u is odd and $i \ge 0$. For each fixed odd positive integer $u \le n$, let i_u denote the largest integer such that $u2^{i_u} \le n$.

Consider $T_u = \{u2^i : 0 \le i \le i_u\}$. Assume T is a double-free subset of [1, n]. Then T cannot contain a pair of elements $u2^{i-1}$ and $u2^i$, with u odd and $i \ge 1$. Therefore, for each fixed odd positive integer u, a maximum-sized subset of T_u that is contained in T is $A_u = \{u2^0, u2^2, u2^4, \ldots, u2^m\}$, where

$$m = \begin{cases} i_u & \text{if } 4|i_u, \\ i_u - 1 & \text{if } 4 \nmid i_u. \end{cases}$$

10.3. Diffsequences

10. Other Topics

Therefore, the size of T is no larger than the size of

 $S = \{u2^{2k} : u \text{ is odd and } u2^{2k} \le n\} = \{u^{4k} : u \text{ is odd and } u4^k \le n\}.$ Since S is doublefree, we are done.

10.3. Diffsequences

In Chapter 4 we investigated the Ramsev properties of certain subfamilies of the family of arithmetic progressions, where the common gap between consecutive integers is restricted to a prescribed set. In particular, recall that if $D \subset \mathbb{Z}^+$, then A_D denotes the family of all arithmetic progressions with gaps belonging to D. We remind the reader also that a set D is called *large* if for all k, r > 1, there is a least positive integer $R(A_D, k; r)$ such that every r-coloring of \mathbb{Z}^+ yields a k-term member of A_D (when this condition is satisfied for a fixed r, D is said to be r-large). In this section we consider a notion related to largeness. Yet, the topic of this section could be considered more like the families we encountered in Chapter 3, where we "loosened" the requirements in the sense that the gaps were allowed some "slack." How can these two contrasting concepts be somehow blended? Actually, it is quite simple: we restrict the gaps to a prescribed set D, but we throw out the requirement that the terms of the sequence form an arithmetic progression.

We begin with some terminology.

Definition 10.12. Let $D \subseteq \mathbb{Z}^+$. A sequence of positive integers $x_1 < x_2 < \cdots < x_k$ is called a k-term *D*-diffsequence if $x_i - x_{i-1} \in D$ for $i = 2, 3, \ldots, k$.

Example 10.13. Let $D = \{1, 5\}$ and let $E = 2\mathbb{Z}^+$. The sequence of integers 3, 8, 9, 14, 19 is a 5-term *D*-diffsequence, and the sequence 1, 2, 7, 8 is a 4-term *D*-diffsequence. A set is an *E*-diffsequence if and only if it consists entirely of even numbers or entirely of odd numbers. Thus, for example, 3, 7, 21, 25, 31, 41 is an *E*-diffsequence.

Definition 10.14. Let $r \ge 1$. A set of positive integers D is called *r*-accessible if for every $k \ge 1$, there exists a least positive integer $\Delta = \Delta(D, k; r)$ such that whenever $[1, \Delta]$ is *r*-colored, there is a

monochromatic k-term D-diffsequence. If D is r-accessible for all positive integers r, we say that D is accessible.

Note that by the compactness principle, the following is an alternate definition for a set D being *r*-accessible: whenever \mathbb{Z}^+ is *r*colored, there are arbitrarily long monochromatic D-diffsequences.

Let us consider some examples.

Example 10.15. Consider $D = \{10n : n \ge 1\}$. We know from Theorem 4.9 (or Example 2.2) that for every finite coloring of the positive integers, there are arbitrarily long monochromatic arithmetic progressions whose gaps belong to D. Certainly, if we remove the requirement of being an arithmetic progression, and instead require merely that the difference between consecutive terms of the progression be a multiple of 10, the corresponding Ramsey property will still hold; i.e., D is accessible.

Example 10.16. Let D be the set of odd positive integers. Coloring \mathbb{Z}^+ as 101010... shows that D is not 2-accessible (in fact, this coloring even avoids monochromatic 2-term D-diffsequences).

We remark that, as made evident in Example 10.15, if a set is r-large, then it must also be r-accessible.

We start our investigation of the Ramsey theory of diffsequences with a very useful lemma. Recall that, for S a set and $c \in \mathbb{R}$, S + cdenotes the set $\{s + c : s \in S\}$.

Lemma 10.17. Let $c \ge 0$ and $r \ge 2$, and let D be a set of positive integers. If every (r - 1)-coloring of D yields arbitrarily long monochromatic (D + c)-diffsequences, then D + c is r-accessible.

Proof. Let $D = \{d_i : i = 1, 2, ...\}$ and assume every (r-1)-coloring of D admits arbitrarily long monochromatic (D + c)-diffsequences. Let χ be an r-coloring of \mathbb{Z}^+ . By induction on k, we show that, under χ , there are k-term monochromatic (D + c)-diffsequences for all k. Since there are obviously 1-term sequences, assume $k \ge 1$ and that under χ there is a k-term monochromatic (D + c)-diffsequence $X = \{x_1, x_2, \ldots, x_k\}$. We may assume X has color red. Consider the set $A = \{x_k + d_i + c : d_i \in D\}$. If some member of A is colored red,

10. Other Topics

then we have a red (k + 1)-term (D + c)-diffsequence. Otherwise we have an (r-1)-coloring of A and therefore, by the hypothesis, A must contain arbitrarily long monochromatic (D + c)-diffsequences.

Remark 10.18. The converse of Lemma 10.17 is false. As one example, let S be the set of odd positive integers, and let $D = S \cup \{2\}$. Let χ be the 2-coloring of D defined by $\chi(x) = 1$ if $x \equiv 1 \pmod{4}$ or x = 2, and $\chi(x) = 0$ if $x \equiv 3 \pmod{4}$. Then χ does not yield arbitrarily long monochromatic D-diffsequences (there are none of length four). On the other hand, Theorem 10.24 below tells us that D is 3-accessible (and, in fact, $\Delta(D, k; 3) \leq 6k^2 - 13k + 6$).

We now present two additional lemmas, the proofs of which are left to the reader as Exercises 10.3 and 10.4.

Lemma 10.19. Let $m \ge 2$ and $i \ge 1$, and assume gcd(i, m) = 1. Let $D = \{x \in \mathbb{Z}^+ : x \equiv i \pmod{m}\}$. Then D is not 2-accessible.

Lemma 10.20. If D is not r-accessible and E is not s-accessible, then $D \cup E$ is not rs-accessible.

We next investigate the accessibility of some specific sets.

Theorem 10.21. Let $D = \{2^i : i = 1, 2, ...\}$. Then

 $8(k-3) + 1 \le \Delta(D,k;2) \le 2^k - 1.$

Proof. We start with the upper bound. Let $\alpha : [1, 2^k - 1] \rightarrow \{0, 1\}$. We show that under α there must be a monochromatic k-term *D*diffsequence. We do this by induction on k. Obviously, it holds for k = 1. Now assume $k \ge 2$, and that $\Delta(D, k - 1; 2) \le 2^{k-1} - 1$. Let $X = \{x_1, x_2, \ldots, x_{k-1}\}$ be a monochromatic *D*-diffsequence, say of color 0, that is contained in $[1, 2^{k-1} - 1]$. Consider the set

 $A = \{x_{k-1} + 2^i : i = 0, 1, \dots, k-1\}.$

Note that $A \subseteq [1, 2^k - 1]$. If there exists $y \in A$ of color 0, then $X \cup \{y\}$ is a monochromatic k-term D-diffsequence. If, on the other hand, no such y exists, then A is a monochromatic k-term D-diffsequence.

For the lower bound, first note that by direct calculation we find that $\Delta(D,3;2) = 7$ and $\Delta(D,4;2) = 11$ (see Table 10.1 at the end of this section). To complete the proof we show by induction on k that, for $k \geq 5$, the 2-coloring χ_k of [1, 8(k-3)] that is represented by $(10010110)^{k-3}$ avoids monochromatic k-term *D*-diffsequences. It is easy to check that this statement is satisfied by k = 5. So now assume $k \geq 5$, that χ_k avoids k-term *D*-diffsequences, and consider χ_{k+1} .

Let $X = \{x_1, x_2, \ldots, x_m\}$ be a maximal length monochromatic *D*-diffsequence under χ_{k+1} . We wish to show that $m \leq k$. Assume, by way of contradiction, that $m \geq k+1$. Then x_{m-1} and x_m both belong to [8(k-3)+1, 8(k-2)], or else the inductive assumption would be contradicted. We have the following two cases.

Case 1. $\chi_{k+1}(X) = 1$. We consider 2 subcases.

10.3. Diffsequences

Subcase i. $x_{m-2} \in [8(k-3)+1, 8(k-2)]$. We have $x_{m-2} = 8k-20$, $x_{m-1} = 8k-18$, and $x_m = 8k-17$. By the structure of χ_k , we see that $x_{m-3} \equiv 4 \pmod{8}$. Hence, there exists, under χ_k , a monochromatic *D*-diffsequence of length m-1, contradicting our assumption about χ_k .

Subcase ii. $x_{m-2} \notin [8(k-3)+1, 8(k-2)]$. Then $x_{m-1} = 8k-18$ and $x_m = 8k-17$. By the structure of χ_k , this implies $x_{m-2} \equiv 6 \pmod{8}$. Then there is an (m-1)-term monochromatic *D*-diffsequence under χ_k , a contradiction.

Case 2. $\chi_{k+1}(X) = 0$. We again consider 2 subcases.

Subcase i. $x_{m-2} \in [8(k-3)+1, 8(k-2)]$. We have $x_{m-2} = 8k-22$, $x_{m-1} = 8k-21$, and $x_m = 8k-19$. Then either $x_{m-3} = 8(k-3)$ or $x_{m-3} \equiv 2 \pmod{8}$. If $x_{m-3} = 8(k-3)$, then $m-3 \leq k-3$, because there can be only one term of a *D*-diffsequence per 10010110-string, a contradiction. If $x_{m-3} \equiv 2 \pmod{8}$, then there is an (m-1)-term *D*-diffsequence of color 0 under χ_k , a contradiction.

Subcase ii. $x_{m-2} \notin [8(k-3)+1, 8(k-2)]$. Then $x_{m-1} = 8k-21$ and $x_m = 8k-19$, and hence $x_{m-2} \equiv 3 \pmod{8}$. This is not possible, since there would then be a monochromatic (m-1)-term *D*-diffsequence under χ_k .

Before continuing, we need one more definition.

Definition 10.22. If $D \subset \mathbb{Z}^+$ is not accessible, the *degree of accessibility* of D is the largest value of r such that D is r-accessible.

10.3. Diffsequences

We denote the degree of accessibility of a set D by doa(D).

As a corollary to Theorem 10.21, we have the following result concerning the degree of accessibility of the set of powers of 2. The proof is left to the reader as Exercise 10.5.

Corollary 10.23. Let $D = \{2^i : i = 1, 2, ...\}$. Then doa(D) = 2.

As mentioned in Remark 10.18, if D consists of the set of odd positive integers along with 2, then $doa(D) \ge 3$. The following theorem tells us more.

Theorem 10.24. Let $D = \{2i + 1 : i \ge 0\} \cup \{2\}$. Then doa(D) = 3. Furthermore,

(10.1)
$$\Delta(D,k;2) = \begin{cases} 3k-4 & \text{if } k \text{ is odd,} \\ 3k-3 & \text{if } k \text{ is even.} \end{cases}$$

Proof. We first show that $doa(D) \geq 3$. Assume, for a contradiction, that $\gamma : \mathbb{Z}^+ \to \{0, 1, 2\}$ is a 3-coloring with no arbitrarily long monochromatic *D*-diffsequence. Let $s_1 < s_2 < \cdots < s_m$ be a monochromatic *D*-diffsequence of maximal length. We may assume this diffsequence has color 2. Then

 $\{s_m + 2\} \cup \{s_m + j : j \in \mathbb{Z}^+, j \text{ odd}\}$

must be void of color 2. We may assume, without loss of generality, that $\gamma(s_m + 2) = 0$. Since *m* is maximal, at most m - 1 elements of $\{s_m + j : j \in \mathbb{Z}^+ \text{ odd}\}$ may have color 0. Let $\gamma(s_m + n) = 0$ with *n* maximal. Then $\{s_m + j : j \in \mathbb{Z}^+ \text{ odd}, j > n\}$ must be of color 1, a contradiction since this is a monochromatic *D*-diffsequence of infinite length.

The fact that doa(D) < 4 follows from Lemma 10.20. To see this, note that by Example 10.16 the set of odd positive integers is not 2-accessible, and that the 2-coloring of \mathbb{Z}^+ represented by 001100110011... shows that {2} is not 2-accessible. Thus, by Lemma 10.20, $doa(D) \leq 3$, and hence doa(D) = 3.

Let f(k) be the function on the right side of (10.1). We next prove that f(k) is an upper bound for $\Delta(D, k; 2)$. First, by direct computation, we find that this is true for k = 2 and k = 3. To prove that f(k) serves as an upper bound if $k \ge 4$, it is sufficient to show that for every $\chi : [1, f(k)] \to \{0, 1\}$, there exist *D*-diffsequences $S = \{s_1, s_2, \ldots, s_{k_1}\}$ and $T = \{t_1, t_2, \ldots, t_{k_2}\}$ such that $\chi(S) = 0$, $\chi(T) = 1$, and $k_1 + k_2 \ge 2k - 1$. We leave it as Exercise 10.6 to show that this condition holds for k = 4 and k = 5. To show it holds for all k, we proceed by induction on k, showing that its truth for k + 2follows from its truth for k.

Assume that $k \geq 4$, and that every 2-coloring of [1, f(k)] admits monochromatic sequences S and T as described above. Let $\chi : [1, f(k+2)] \rightarrow \{0, 1\}$ be an arbitrary 2-coloring. To complete the proof we show that there exist a k'_1 -term D-diffsequence of color 0 and a k'_2 -term D-diffsequence of color 1 such that

(10.2)
$$k_1' + k_2' \ge 2k + 3.$$

We may assume, without loss of generality, that $k_1 \ge k_2$. Let

$$U = \{s_{k_1} + 1, s_{k_1} + 2, \dots, s_{k_1} + 6\}.$$

We consider three cases.

Case 1. There exist at least four elements of U that have color 0. It is easy to see that by appending these four elements to S, we have a monochromatic D-diffsequence, and hence (10.2) holds.

Case 2. Exactly three elements of U have color 0. Then there exist two of these three elements, a and b, such that $S \cup \{a, b\}$ forms a $(k_1 + 2)$ -term D-diffsequence. Likewise there exist two elements, c and d, of T, with color 1 and such that $T \cup \{c, d\}$ forms a $(k_2 + 2)$ -term T-diffsequence. This implies that (10.2) holds for this case.

Case 3. At most two elements of U have color 0. Then we may extend T to a D-diffsequence, monochromatic of color 1, having length $k'_2 \ge k_2 + 4$. Again (10.2) holds.

To complete the proof, we show that $\Delta(D, k; 2) \geq f(k)$ by exhibiting a specific 2-coloring of [1, f(k) - 1] that avoids monochromatic k-term D-diffsequences. Starting with the case in which k is even, let

$$\chi_k : [1, 3k - 4] \to \{0, 1\}$$

be the 2-coloring represented by $1(000111)^{\frac{k-2}{2}}0$. By symmetry it suffices to show that, under χ_k , there is no k-term D-diffsequence with
10.3. Diffsequences

10. Other Topics

color 1. We show this by induction on j, where k = 2j. Obviously the coloring 10 avoids 2-term monochromatic *D*-diffsequences, and the coloring 10001110 avoids 4-term monochromatic *D*-diffsequences, and hence the result holds for j = 1 and j = 2.

Now assume $j \geq 2$, and that χ_k does not yield any k-term monochromatic D-diffsequences with color 1. Note that χ_{k+2} may be represented by $\chi_k 001110$. Let A be a monochromatic D-diffsequence, under χ_k , of color 1, and having maximal length. Obviously, at least one of 3k-7, 3k-6 belongs to A, which implies that 3k-5 also belongs to A. Therefore, at most two members of $\{3k-1, 3k, 3k+1\}$ may be tacked on to A to form a monochromatic D-diffsequence. Thus, under the coloring χ_{k+2} , there is no (k+2)-term D-diffsequence with color 1. This completes the proof for k even.

Next, consider k odd. Let λ_k be the coloring represented by $11(000111)^{\frac{k-3}{2}}00$. The proof is completed in a straightforward manner, similar to the even case, by induction on $\ell = \frac{k-1}{2}$, by showing that the longest *D*-diffsequence with color 1 cannot have length greater than k-1. We leave the details to the reader as Exercise 10.7.

One very intriguing area of research that remains rather open is the relationship between accessibility and largeness. Clearly, no matter how we decide to do the counting, for any given (nontrivial) kand D, there are a lot more k-term D-diffsequences than there are kterm members of A_D (arithmetic progressions with gaps in D). Yet, it is currently unknown whether there exist any accessible sets that are not large. Perhaps settling this question is not a very difficult problem.

There are, however, some fundamental differences between accessibility and largeness that are known. For example, currently there are no known sets that are r-large for some $r \ge 2$, but which are not large. In fact, it has been conjectured that any set that is 2-large is also large (the conjecture remains open). In contrast to this, we have the following theorem concerning r-accessibility.

Theorem 10.25. For each $r \ge 1$, there exists a set $D \subseteq \mathbb{Z}^+$ such that doa(D) = r.

As stated, it may seem a daunting task to prove this theorem, especially given the rather complete lack of such information about r-large sets. However, Theorem 10.25 is an immediate corollary of the following result, which is, itself, rather easy to prove.

Theorem 10.26. For $m \ge 2$, let V_m be the set of all positive integers not divisible by m. Then $doa(V_m) = m - 1$.

Proof. We first show that $doa(V_m) \leq m-1$ by proving the following more general result: if $r \in \mathbb{Z}^+$ and D contains no multiples of r, then D is not r-accessible. To see this, consider $\chi : \mathbb{Z}^+ \to \{0, 1, \ldots, r-1\}$ defined by $\chi(x) = i$ if $x \equiv i \pmod{r}$. It is obvious that χ avoids 2-term monochromatic D-diffsequences. Thus, since (by definition) V_m contains no multiple of m, we conclude that $doa(V_m) \leq m-1$.

Since V_2 is the set of odd positive integers which, as we have already seen is not 2-accessible, we need only consider $m \ge 3$. Let χ be any (m-2)-coloring of V_m . By the pigeonhole principle, some color must yield an infinite number of elements from each of at least two of the congruence classes $1 \pmod{m}$, $2 \pmod{m}$, \dots , $(m-1) \pmod{m}$. Thus, some color admits arbitrarily long V_m -diffsequences. By Lemma 10.17, V_m is (m-1)-accessible, and the proof is complete.

As mentioned above, the question as to whether every accessible set is also a large set is still open. If there do exist any accessible sets that are not large, then the following theorem seems like a good place to start looking.

Theorem 10.27. Let $T \subseteq \mathbb{Z}^+$ be infinite. Then

$$T \ominus T = \{t - s : s < t \text{ and } s, t \in T\}$$

is accessible.

Proof. Let $r \ge 1$, and consider any *r*-coloring of $T \ominus T$. Fix $s \in T$. Let $\{t_1 < t_2 < \cdots\} = \{t \in T : t > s\}$ and define

$$A = \{t_i - s : i = 1, 2, \dots\}.$$

Obviously, there exists a monochromatic infinite $B \subseteq A$. Since B is a $(T \ominus T)$ -diffsequence, by Lemma 10.17, $T \ominus T$ is (r + 1)-accessible. Since r is arbitrary, $T \ominus T$ is accessible.

From the above theorem, we see that a set can be very "sparse" and still be accessible. For example, the set $\{t! - s! : 1 \le s < t\}$ is accessible. It would be desirable to know if it is large.

The last stop on our tour of accessibility concerns the set of primes, which we will denote by P. We know that P is not a large set; in fact, it is not even 2-large, because a necessary condition for a 2-large set is that it contain a multiple of every positive integer (P contains no multiple of 4, for example). An open question is whether any translation of P (i.e., a set c + P where $c \in \mathbb{Z}^+$) is large, or perhaps r-large for some $r \geq 2$. We may ask analogous questions in the context of accessibility; that is, what can we say, if anything, about the degree of accessibility of P + c for $c \geq 0$.

First of all, what about P itself? Is it 2-accessible? As we explained in the previous paragraph, P is not 2-large; but the reason we gave does not apply to 2-accessibility (for example, the set of powers of 2 is 2-accessible, but has no multiple of 3). On the other hand, we do know, by Theorem 10.26, that P is not 4-accessible, since it is, in particular, a subset of V_4 . In fact, we do not know whether P is 2-accessible. By the next theorem, however, we do know for sure that it is not 3-accessible.

Theorem 10.28. Let P be the set of primes. Then $doa(P) \leq 2$.

Proof. It suffices to find a 3-coloring of the positive integers that does not yield arbitrarily long monochromatic *P*-diffsequences. Consider the coloring $\chi : \mathbb{Z}^+ \to \{0, 1, 2\}$ defined by

$$\chi(i) = \begin{cases} 0 & \text{if } 9|i, \\ 1 & \text{if } i \text{ is even and } 9 \nmid i, \\ 2 & \text{if } i \text{ is odd and } 9 \nmid i. \end{cases}$$

It is clear that there is not even a 2-term *P*-diffsequence of color 0. Consider the elements of color 1. For any sequence $a_1 < a_2 < \cdots < a_9$ of color 1, there must be some $i, 2 \le i \le 9$, such that $a_i - a_{i-1}$ exceeds 2 (and is even). Hence there is no 9-term *P*-diffsequence of color 1. Similarly, there is no 9-term *P*-diffsequence of color 0.

If we consider P + c, with $c \ge 2$ even, then it is not hard to show that $doa(P + c) \le 3$. However, we do not know if any such even translations are 2-accessible, let alone 3-accessible.

Based on what we have presented above, we could say that we do not have any positive results about the accessibility of even translations of P (we still do not know if there exists an even translation that is 2-accessible). We do, however, have a nice positive result for odd translations, presented below as Theorem 10.30. Before stating it, we give a lemma that, when used with Lemma 10.17, gives us Theorem 10.30.

The next lemma we present says that for any odd positive integer c, there exist arbitrarily (P + c)-diffsequences consisting of only primes.

Lemma 10.29. Let $c \in \mathbb{Z}^+$ be odd. Then for any $k \ge 2$, there exist $p_1, p_2, \ldots, p_k \in P$ such that $p_i - p_{i-1} \in P + c$ for all i, 2 < i < k.

We do not include the proof of Lemma 10.29, which is beyond the scope of the book. Using Lemma 10.29, the following result can be shown. We leave the proof to the reader as Exercise 10.12.

Theorem 10.30. Let P be the set of primes. If c is an odd positive integer, then P + c is 2-accessible.

There are many interesting questions left unanswered about accessibility. We present several of these in Section 10.8.

We end this section with a table of values of $\Delta(D, k; 2)$ for several choices of D and small k. The symbols T, F, P, and V_n denote $\{2^i : i \geq 0\}$, the set of Fibonacci numbers, the set of primes, and the set of positive integers not divisible by n, respectively.

$D\setminus k$	2	3	4	5	6	7	8	9	10
T	3	7	11	17	25	35	51	?	?
F	3	5	9	11	15	19	21	?	?
P	5	9	13	21	25	33	37	42	49
P+1	7	13	21	27	35	?	?	?	?
P+2	9	17	25	33	?	?	?	?	?
P+3	11	21	31	42	?	?	?	?	?
P+4	13	25	37	?	?	?	?	?	?
P+5	15	29	?	?	?	?	?	?	?
P+6	17	33	?	?	?	?	?	?	?
P+7	19	37	?	?	?	?	?	?	?
V_5	3	5	7	11	13	15	19	?	?
V_6	3	5	7	9	13	15	17	?	?
$ \begin{array}{c} P+7\\ V_5\\ V_6 \end{array} $	19 3 3	37 5 5	? 7 7	? 11 9	? 13 13	? 15 15	? 19 17	???	? ? ?

Table 10.1: Values of $\Delta(D, k)$

10.4. Brown's Lemma

The main result of this short section could have been included in the last section, since it is so closely related to diffsequences. However, the result is of sufficient interest to warrant its own section. We will formalize this result after presenting a definition.

Definition 10.31. A set $S \subseteq \mathbb{Z}^+$ is called *piecewise syndetic* if there exists $d \in \mathbb{Z}^+$ such that S contains arbitrarily long $\{1, 2, \ldots, d\}$ -diffsequences.

Theorem 10.32 (Brown's Lemma). Let $r \ge 1$. Any r-coloring of \mathbb{Z}^+ admits a monochromatic piecewise syndetic set.

Proof. We induct on the number of colors. For r = 1, the result is trivial since \mathbb{Z}^+ is clearly piecewise syndetic. Assume the result holds for $r \ge 1$ colors and let $\chi : \mathbb{Z}^+ \to \{0, 1, \ldots, r\}$ be an arbitrary (r+1)-coloring.

If the color 0 occurs only finitely many times, then there exists n such that $\{n, n + 1, ...\}$ is r-colored. By Proposition 2.30 and the inductive hypothesis, we are done. Hence, we assume that the color 0 occurs infinitely often. Let $R = \{r_1 < r_2 < ...\}$ be the set of integers with color 0.

If R is piecewise syndetic we are done, so we assume that R is not piecewise syndetic. Hence, the differences between consecutive elements of R are not bounded. Let $r_2 - r_1 = k_0$. Then there exists an $i_1 \geq 2$ such that $r_{i_1+1} - r_{i_1} = k_1 > k_0$, so that the interval $[r_{i_1} + 1, r_{i_i+1} - 1]$ is void of color 0. Similarly, we may choose $i_2 \geq i_1$ such that $r_{i_2+1} - r_{i_2} = k_2 > k_1$. Repeating this process, for each $j \geq 2$, denote by I_j the interval $[r_{i_j} + 1, r_{i_j+1} - 1]$, where i_j is chosen so that $r_{i_j+1} - r_{i_j} > r_{i_{j-1}+1} - r_{i_{j-1}}$ and $i_j \geq i_{j-1}$. Hence, the interval I_j is void of color 0 and $|I_j| > |I_{j-1}|$ for $j \geq 2$.

We now define γ , an *r*-coloring of \mathbb{Z}^+ , as follows. Denote by $I_j(k)$ the k^{th} smallest element of I_j . Within $\chi(I_1(1)), \chi(I_2(1)), \chi(I_3(1)), \ldots$ some color (not 0, since the I_j 's are void of the color 0) must occur an infinite number of times. Call this color c_1 and let $\gamma(1) = c_1$. Let \mathcal{T}_1 be the set consisting of those intervals I_j such that $\chi(I_j(1)) = c_1$. Within $\{\chi(I_j(2)) : I_j \in \mathcal{T}_1\}$ there must be some color c_2 that occurs an infinite number of times. Let $\gamma(2) = c_2$ and let \mathcal{T}_2 be the set of those intervals $I_j \in \mathcal{T}_1$ such that $\chi(I_j(2)) = c_2$. We continue in this way to find, for each $i \geq 2$, some color c_i such that

$$\mathcal{T}_i = \{I_j \in \mathcal{T}_{i-1} : \chi(I_j(i)) = c_i\}$$

is infinite, and we define $\gamma(i) = c_i$. The coloring $\gamma : \mathbb{Z}^+ \to \{1, 2, \ldots, r\}$ has the property that for any $n \geq 1$, \mathcal{T}_n is the set of intervals I_j such that $\chi(I_j(i)) = \gamma(i)$ for $i = 1, 2, \ldots, n$.

By the inductive assumption, γ yields a monochromatic piecewise syndetic set, say X. Thus, X contains arbitrarily long $\{1, 2, \ldots, d\}$ diffsequences for some $d \in \mathbb{Z}^+$. Let $\{a_1 < a_2 < \cdots < a_n\} \subseteq X$ be one such diffsequence. Let $I \in \mathcal{T}_{a_n}$ and consider

$$S = \{ I(a_1) < I(a_2) < \dots < I(a_n) \}$$

By the definition of γ , S is monochromatic. By construction, S is piecewise syndetic. Since n can be arbitrarily large, we have found an arbitrary large monochromatic (under χ) piecewise syndetic set. \Box

There is also a "finite form" of Brown's lemma. The fact that this is equivalent to Theorem 10.32 is left to the reader as Exercise 10.13. To state it, we define, for a finite set $A = \{a_1 < a_2 < \cdots < a_n\} \subseteq \mathbb{Z}^+$,

10. Other Topics

the gap size of A to be

$$gs(A) = \max\{a_{j+1} - a_j : 1 \le j \le n - 1\}$$

If |A| = 1, we set gs(A) = 1.

Theorem 10.33. Let $f : \mathbb{Z}^+ \to \mathbb{Z}^+$ be any function, and let $r \ge 1$. Then there exists a least positive integer B(f;r) such that for every r-coloring of [1, B(f, r)], there exists a monochromatic set A with |A| > f(gs(A)).

Proof. Assume, without loss of generality, that f is nondecreasing. For simplicity of notation, denote B(f,r) by B(r). We use induction on r.

Clearly B(1) = f(1)+1, since if the interval [1, f(1)+1] is colored with one color, the interval itself constitutes a monochromatic set Awith gs(A) = 1, so that |A| > f(gs(A)) = f(1).

Now, let $r \ge 2$ and assume that B(r-1) exists. Let

(10.3) m = rf(B(r-1)) + 1.

We will show that $B(r) \leq m$. Assume, for a contradiction, that there is an r-coloring χ of [1, m] such that for every monochromatic set A we have $|A| \leq f(gs(A))$. Let $C_i = \{j : \chi(j) = i\}, 1 \leq i \leq r$. Then $|C_i| \leq f(gs(C_i))$.

Also, $gs(C_i) \leq B(r-1)$ for each $i, 1 \leq i \leq r$; otherwise, for some $a \geq 1$, the set $\{a + 1, a + 2, \ldots, a + B(r-1)\} \subseteq [1, m]$ would have only r-1 colors (it would be void of color i). By Proposition 2.30 and the inductive assumption, this would give a monochromatic set T with |T| > f(gs(T)), contradicting our assumption about χ .

Since f is nondecreasing, $f(gs(C_i)) \leq f(B(r-1))$. Hence,

$$|C_i| \le f(gs(C_i)) \le f(B(r-1))$$

for all $i, 1 \leq i \leq r$. Since $[1,m] = C_1 \cup C_2 \cup \cdots \cup C_r$, we get $m \leq rf(B(r-1))$, contradicting (10.3). Thus, any *r*-coloring of [1,m] satisfies the conditions of the theorem. This proves the existence of B(r), since $B(r) \leq m = rf(B(r-1)) + 1$.

We may rephrase Brown's lemma (Theorem 10.32) as follows: for any r-coloring of \mathbb{Z}^+ , there exists $d \ge 1$ such that for any $n \ge 2$, there exists a monochromatic set $A = \{a_1 < a_2 < \cdots < a_n\}$, where $a_{j+1} - a_j \leq d$ for $1 \leq j \leq n-1$. Comparing this statement to the statements given in Theorem 2.5, we see that Brown's lemma is very reminiscent of van der Waerden's theorem. However, it is known that Brown's lemma neither implies, nor is implied by, van der Waerden's theorem.

10.5. Patterns in Colorings

Most of this book has dealt with the presence of specific monochromatic structures under finite colorings of the integers. In this section, rather than being concerned with finding structures, we will investigate colorings themselves.

For convenience, we will be working with colorings of \mathbb{Z}^+ or [1, n] $(n \in \mathbb{Z}^+)$; however, the integers are used only as placeholders for the colors.

Definition 10.34. Let $n, r \ge 1$ and let χ be an *r*-coloring of [1, n] or \mathbb{Z}^+ . We call χ squarefree if we cannot write $\chi = xzzy$ where each of x, y, z is an *r*-coloring (written as a string of colors) and z is nonempty. We call *s* cubefree if we cannot write s = xzzzy where x, y, z are *r*-colorings and *z* is nonempty.

Note that Definition 10.34 does not say that x or y must be nonempty.

Example 10.35. There are no 2-colorings of [1, 4] that are squarefree. To see this, assume $\chi : [1, 4] \rightarrow \{0, 1\}$ is a squarefree coloring. Without loss of generality, we may assume that $\chi(1) = 0$. Since χ is squarefree, $\chi(2)$ must be 1. Likewise, $\chi(3) = 0$, and $\chi(4) = 1$. Now we have the coloring zz, where z = 01, a contradiction.

Since it is quite easy to show that there is no squarefree 2-coloring of an interval of length more than three (by Example 10.35), the next result might seem somewhat remarkable.

Theorem 10.36. Let $r \geq 3$. There exists $\gamma : \mathbb{Z}^+ \to \{0, 1, \dots, r-1\}$ such that γ is squarefree.

Concerning cubefree colorings, we have the following result.

10. Other Topics

Theorem 10.37. Let $r \geq 2$. There exists $\gamma : \mathbb{Z}^+ \to \{0, 1, \dots, r-1\}$ such that γ is cubefree.

We next provide the colorings that are used to establish Theorems 10.36 and 10.37. The proofs that these colorings yield the desired conclusions are left to the reader as Exercise 10.14.

Definition 10.38. Let c be a string of colors 0 and 1. Let f act on c from left to right by

$$f \cdot 0 = 01$$
 and $f \cdot 1 = 10$.

The *Thue-Morse sequence*, also known as the Prouhet-Thue-Morse sequence, is

$$\lim_{n \to \infty} f^n \cdot 0 = 01101001100101101\dots$$

To understand the Thue-Morse sequence, consider the first few iterations of f acting on 0. We have

 $0 \xrightarrow{f} 01 \xrightarrow{f} 0110 \xrightarrow{f} 01101001 \xrightarrow{f} 0110100110010110 = f^4 \cdot 0.$

We see that the action of f appears to be such that $f^{n-1} \cdot 0$ is a leftmost substring of $f^n \cdot 0$. This is indeed true, but will not be proven here.

Using the Thue-Morse sequence, we can derive a related sequence.

Definition 10.39. Let $\{t_1, t_2, t_3, ...\}$ be the Thue-Morse sequence. Let $s = \{s_1, s_2, s_3, ...\}$ be the sequence defined by $s_i = t_{i-1}t_i$ (concatenation, not multiplication) so that

 $S = \{01, 11, 10, 01, 10, 00, 01, 11, 10, \dots\}.$

Define M to be the sequence obtained from S by replacing every 01 by 0, every 10 by 1, and every 00 and every 11 by 2 to get

$$M = \{0, 2, 1, 0, 1, 2, 0, 2, 1, \dots\}.$$

We call M the Morse-Hedlund sequence.

We leave it to the reader as Exercise 10.14 to prove that the Thue-Morse sequence is an example of a cubefree 2-coloring of \mathbb{Z}^+ , and that the Morse-Hedlund sequence is an example of a squarefree 3-coloring of \mathbb{Z}^+ .

Some natural questions we may now ask are:

1. For $n \ge 1$, how many squarefree 3-colorings of length n are there?

2. For $n \ge 1$, how many cubefree 2-colorings of length n are there?

In investigating these questions we will use the following notation.

Notation. For $n \in \mathbb{Z}^+$, let sq(n) denote the number of squarefree 3-colorings of length n, and let c(n) denote the number of cubefree 2-colorings of length n.

For small n, the exact values of sq(n) and c(n) are known. We give the first 10 values of each:

n	sq(n)	c(n)
1	1	1
2	3	2
3	6	4
4	12	6
5	18	10
6	30	16
7	42	24
8	60	36
9	78	56
10	108	80

Table 10.2: Number of squarefree and cubefree colorings

Although no exact formula for sq(n) or c(n) is known, the following result, which we present without proof, gives upper and lower bounds for these functions.

Theorem 10.40. Let $n \ge 1$. There exist constants a, b > 0 such that

$$(1.118)^n < 110^{\overline{42}} \le sq(n) < a(1.3021)^n,$$

$$2 \cdot (1.080)^n < 2^{\frac{n}{9}+1} \le c(n) < b(1.4576)^n.$$

10.6. Zero-sums

This section seems a fitting way to conclude this book, insomuch as it involves results obtained from the pigeonhole principal, the most basic principle of Ramsey theory. We begin with a definition.

Definition 10.41. Let $k, r \ge 1$ and let $\chi : \mathbb{Z}^+ \to \{0, 1, \ldots, r-1\}$ be an *r*-coloring. For $x_1, x_2, \ldots, x_k \in \mathbb{Z}^+$ we say that (x_1, x_2, \ldots, x_k) is a zero-sum sequence if $\sum_{i=1}^k \chi(x_i) \equiv 0 \pmod{r}$.

10. Other Topics

It is convenient to use the language "S is zero-sum" when we want to say that S is a zero-sum sequence.

Before stating perhaps the most well-known zero-sum theorem, we start with two nice applications of the pigeonhole principle.

Theorem 10.42. Let $r \ge 1$. Let $\chi : [1, r] \rightarrow \{0, 1, \dots, r-1\}$ be an *r*-coloring. There exist $i, j \in [1, r]$ with $i + j \le r$ such that

$$(i, i+1, i+2, \ldots, i+j)$$

is zero-sum.

Proof. Assume, for a contradiction, that no such *i* and *j* exist. For k = 1, 2, ..., r define

$$s_k = \sum_{i=1}^k \chi(i) \,(\mathrm{mod}\,r)$$

By assumption, $s_k \neq 0$ for k = 1, 2, ..., r. By the pigeonhole principle, there exist $x, y, 1 \leq x < y \leq r$, such that $s_x = s_y$. Thus,

$$\sum_{i=1}^{x} \chi(i) \equiv \sum_{i=1}^{y} \chi(i) \, (\text{mod} \, r)$$

This yields

$$\sum_{i=x+1}^{y} \chi(i) \equiv 0 \,(\text{mod}\,r),$$

and we are done by taking i = x + 1 and j = y - x - 1.

Lemma 10.43. Let $r \geq 2$, and let $\chi : \mathbb{Z}^+ \to [0, r-1]$ be an *r*-coloring. Let $A \subseteq \mathbb{Z}^+$ be such that $|\{\chi(a) : a \in A\}| = k > 1 + \frac{r}{2}$ and write $\{\chi(a) : a \in A\} = \{c_1, c_2, \ldots, c_k\}$. Then, for any $c \in [0, r-1]$, there exist $i, j, 1 \leq i < j \leq k$, such that $c \equiv (c_i + c_j) \pmod{r}$.

Proof. Let χ and A be as in the statement, and let $c \in [0, r-1]$ be given. Let

 $B = \{ (c - c_1) \, (\operatorname{mod} r), (c - c_2) \, (\operatorname{mod} r), \dots, (c - c_k) \, (\operatorname{mod} r) \}$

(so that |B| = k). Since $|A \cap B| = |A| + |B| - |A \cup B|$ and |A| + |B| > 2 + r, while $|A \cup B| \le r$, we see that A and B must have at least 3 elements in common. Say $x, y, z \in A \cap B$. Thus,

 $x \equiv (c - x') \pmod{r}, \quad y \equiv (c - y') \pmod{r}, \quad z \equiv (c - z') \pmod{r}$

for some $x', y', z' \in A$. If we have $x \neq x', y \neq y'$, or $z \neq z'$, then we are done because we have two distinct elements of A summing (modulo r) to c. Hence, assume x = x', y = y', and z = z'. Since $2x \equiv 2y \equiv 2z \equiv c \pmod{r}$ we have

$$2(x-y) \equiv 0 \pmod{r}, \quad 2(x-z) \equiv 0 \pmod{r}, \quad 2(y-z) \equiv 0 \pmod{r},$$

i.e., there are three solutions to $2t \equiv 0 \pmod{r}$. Since we may have at most two distinct solutions to $2t \equiv 0 \pmod{r}$ (check!), one of the following must hold:

(i)
$$x - z \equiv (y - z) \pmod{r};$$

(ii) $x - y \equiv (x - z) \pmod{r};$
(iii) $x - y \equiv (y - z) \pmod{r}.$

If (i) holds, then x = y, contradicting the fact that x and y are distinct. If (ii) holds, then y = z, again a contradiction. If (iii) holds, we have $x + z \equiv 2y \equiv c \pmod{r}$, which implies that x = z since $2x = x + x \equiv c \pmod{r}$, contradicting the fact that x and z are distinct.

We now state and prove the main theorem of this section.

Theorem 10.44 (Erdős-Ginzburg-Ziv Theorem). Let $r \ge 2$. Let S be a set of 2r-1 elements. For any r-coloring $\chi : S \to \{0, 1, \ldots, r-1\}$, there exist distinct $t_1, t_2, \ldots, t_r \in S$ such that (t_1, t_2, \ldots, t_r) is zero-sum.

Proof. Let χ be a given *r*-coloring. We use induction on the number of (not necessarily distinct) prime factors of *r*. We start with the base case: *r* is prime. Let $0 \le c_1 \le c_2 \le \cdots \le c_{2r-1}$ be the colors used by χ (all of the colors are not necessarily used). If $c_i = c_{i+r-1}$ for some $i \le r-1$, then we are done since $\sum_{j=i}^{i+r-1} c_j \equiv 0 \pmod{r}$. Hence, assume $C_i = \{c_i, c_{i+r-1}\}, i = 1, 2, \ldots, r-1$, are 2-element sets.

10. Other Topics

Define, for
$$1 \le k \le r - 1$$
,
 $X_k = \left\{ \sum_{i=1}^k x_i \, (\text{mod } r) : x_i \in C_i, i = 1, 2, \dots, k \right\}.$

We will show that $|X_k| \ge k+1$ by means of induction on k. For k = 1 the result is trivial. Let k < r-1 and assume $|X_k| \ge k+1$. We will show that $|X_{k+1}| \ge k+2$. Clearly, the set

$$Y = \{ (s_i + c_{k+1}) \pmod{r} : s_i \in X_k, i = 1, 2, \dots, |X_k| \}$$

consists of $|X_k|$ distinct elements. We may assume that $|X_k| = k+1$, or else $|Y| \ge k+2$ and we are done since $Y \subseteq X_{k+1}$.

Hence, it remains to show that there exists $s_j \in X_k$ such that $(s_j + c_{r+k}) \pmod{r} \notin Y$. Assume, for a contradiction, that no such s_j exists, i.e., for each s_j , $j = 1, 2, \ldots, k+1$, there exists $s'_j \in X_k$ such that

(10.4)
$$s'_{j} + c_{k+1} \equiv (s_{j} + c_{r+k}) \pmod{r}.$$

Let $d = c_{k+1} - c_{r+k}$. Then $d \neq 0$, and hence we can rewrite (10.4) as

(10.5)
$$s'_{j} \equiv (s_{j} - d) \pmod{r}.$$

Since $|X_k| = k + 1$, there exist $t_1, t_2, \ldots, t_{r-k-1}$, distinct residues modulo r, that are not members of X_k . Note that k < r-1 gives us $r-k-1 \ge 1$. Since r is prime, there exists $s \in X_k$ such that $s-d \equiv t_i \pmod{r}$ for some $i, 1 \le i \le r-k-1$, contradicting the existence of an $s'_j \in X_k$ satisfying (10.5). This completes the proof that $|X_k| \ge k+1$ for $1 \le k \le r-1$.

Let $X = X_{r-1}$. Hence, |X| = r (since there are only r residues modulo r). Consider c_{2r-1} , which is not a member of any C_i . Since X contains all residues modulo r, there exist $x_i \in C_i$, $1 \le i \le r-1$, such that

$$\sum_{i=1}^{r-1} x_i \equiv -c_{2r-1} \, (\operatorname{mod} r).$$

Hence,

$$x_1 + x_2 + \dots + x_{r-1} + c_{2r-1} \equiv 0 \pmod{r},$$

and we have r elements that satisfy the conclusion of the theorem. This completes the base case.

Now, let r = pm with p a prime and $m \neq 1$. Clearly, the number of prime factors of m is less than the number of prime factors of r. Taking all colors modulo p, there exists a p-element subset $T_1 \subseteq S$ such that $\sum_{t \in T_1} \chi(t) \equiv 0 \pmod{p}$. Consider $S_1 = S - T_1$, so that $|S_1| = 2(m-1)p - 1$. Again, taking all colors modulo p, there exists a p-element subset $T_2 \subseteq S_1$, such that $\sum_{t \in T_2} \chi(t) \equiv 0 \pmod{p}$. Let $S_2 = S_1 - T_2$. Continuing to view all colors modulo p, for each i, $2 \leq i \leq 2m - 1$, there exists a p-element subset $T_i \subseteq S_{i-1}$ where $S_{i-1} = S_{i-2} - T_{i-1}$ (taking $S_0 = S$). Hence, there exist 2m - 1pairwise disjoint subsets of S, say, $T_1, T_2, \ldots, T_{2m-1}$, with each T_i satisfying $\sum_{t \in T_i} \chi(t) \equiv 0 \pmod{p}$.

For i = 1, 2, ..., 2m - 1, let $k_i p = \sum_{t \in T_i} \chi(t)$ and consider the set $S' = \{k_i : i = 1, 2, ..., 2m - 1\}$. By the induction hypothesis, S' contains $k_{i_1}, k_{i_2}, ..., k_{i_m}$ such that $\sum_{j=1}^m k_{i_j} \equiv 0 \pmod{m}$. Hence, $T = T_{i_1} \cup T_{i_2} \cup \cdots \cup T_{i_m}$ is a subset of S with pm elements such that $\sum_{t \in T} \chi(t) \equiv 0 \pmod{pm}$, which completes the proof. \Box

As can be seen from the above proof, the fact that the set S has 2r - 1 elements is crucial. To see that we cannot r-color a set with only 2r - 2 elements and expect the same result to hold, consider the following example.

Example 10.45. For all $r \ge 2$, there exists $\chi : [1, 2r - 2] \rightarrow [0, r - 1]$ such that for any distinct $t_1, t_2, \ldots, t_r \in [1, 2r - 2], (t_1, t_2, \ldots, t_r)$ is not zero-sum. To show this, let χ to be the coloring defined by $\chi(i) = 0$ for $i = 1, 2, \ldots, r - 1$, and $\chi(i) = 1$ for $i = r, r + 1, \ldots, 2r - 2$ (χ uses only 2 of the possible r colors). Since $\sum_{i=1}^{r} \chi(x_i) \le r - 1$ and is positive, for any distinct $t_1, t_2, \ldots, t_r \in [1, 2r - 2]$, we see that (t_1, t_2, \ldots, t_r) is not zero-sum.

In Example 10.45, we used only 2 colors. In fact, it is known that for any r-coloring $\chi : S \to \{0, 1, \ldots, r-1\}$, where S is a set of 2r-2 elements, if there do not exist $t_1, t_2, \ldots, t_r \in S$ such that (t_1, t_2, \ldots, t_r) is zero-sum, then, in fact, it must be the case that $\chi : S \to \{x, y\}$, where r-1 elements of S have color x, and the remaining r-1 elements of S have color y. Hence, if χ is an r-coloring that uses at least three colors, then we may use |S| = 2r - 2 in Theorem 10.44.

Based on the previous example, we refine Theorem 10.44 by means of the following definition and subsequent results.

Definition 10.46. For $1 \le k \le r$, define g = g(r, k) to be the least integer such that for all S with |S| = g and all $\chi : S \to \{0, 1, \ldots, r-1\}$, whenever $|\chi(S)| = k$ (i.e., the range of χ has size k), there exist $t_1, t_2, \ldots, t_r \in S$ such that (t_1, t_2, \ldots, t_r) is zero-sum.

Note that Theorem 10.44 implies that g(r, k) is well-defined, since $g(r, k) \leq 2r - 1$.

From the discussion after Example 10.45, we see that the following result is true.

Proposition 10.47. Let $2 < k \leq r$. Then g(r, 2) = 2r - 1 and $g(r, k) \leq 2r - 2$.

Investigating g(r, k) further, we have the following two theorems.

Theorem 10.48. *For* $r \ge 1$ *,*

$$g(r,r) = \begin{cases} r & \text{if } r \text{ is odd,} \\ r+1 & \text{if } r \text{ is even} \end{cases}$$

Proof. By definition, $g(r,r) \ge r$ for all r. For an upper bound, let |S| = r and let χ be a coloring of S such that $\chi(S) = \{0, 1, \dots, r-1\}$.

First consider r odd. By definition, all colors are used by χ , so that

(10.6)
$$\sum_{s \in S} \chi(s) = \sum_{i=0}^{r-1} i = \frac{r(r-1)}{2}.$$

Since r is odd we see that $\frac{r-1}{2} \in \mathbb{Z}^+$, and hence the elements of S are zero-sum.

Now let r be even, so that $\frac{r}{2} \in \mathbb{Z}^+$. By (10.6), since $r \ge 2$, we have gcd(r, r-1) = 1, so that $\frac{r}{2} \not\equiv 0 \pmod{r}$. Hence, $g(r, r) \ge r+1$.

Now let $T = \{t_1, t_2, \ldots, t_{r+1}\}$ and $\gamma: T \to \{0, 1, \ldots, r-1\}$, with $|\gamma(S)| = r$. We may assume that $\gamma(t_{r+1})$ is the sole duplicate color.

Hence, for r even,

$$\sum_{i=1}^{r} \gamma(t_i) = \sum_{i=0}^{r-1} i \equiv s \pmod{r}$$

for some $s \neq 0$. Let $d = s + \gamma(t_{r+1})$. Then there exists $j \in \{1, 2, ..., r\}$ such that $\gamma(t_j) = d$. Thus,

$$\sum_{i=1\atop i\neq j}^{r} \gamma(t_i) \equiv (s-d) \, (\text{mod} \, r)$$

which gives us

$$\gamma(t_{r+1}) + \sum_{\substack{i=1\\i\neq j}}^r \gamma(t_i) \equiv 0 \,(\text{mod}\,r),$$

and hence $(t_1, t_2, \ldots, t_{j-1}, t_{j+1}, \ldots, t_{r+1})$ is zero-sum. Since γ is arbitrary, this implies that $g(r, r) \leq r+1$. Hence, we have g(r, r) = r+1 for r odd, thereby completing the proof.

Theorem 10.49. If $r \ge 5$ and $1 + \frac{r}{2} < k \le r - 1$, then g(r, k) = r + 2.

Proof. To show that $g(r,k) \leq r+2$, let $\chi : [1, r+2] \rightarrow \{0, 1, \ldots, r-1\}$ be a coloring that uses exactly k colors. Let $s = \sum_{i=1}^{r+2} \chi(i) \pmod{r}$. By Lemma 10.43, there exist $x, y \in [1, r+2]$, with x < y, such that $s \equiv (x+y) \pmod{r}$. Hence,

$$(1, 2, \ldots, x - 1, x + 1, x + 2, \ldots, y - 1, y + 1, y + 2, \ldots, r + 2)$$

is zero-sum. This proves that $g(r, k) \leq r + 2$.

To prove that $g(r,k) \ge r+2$ we present a valid 2-coloring of [1, r+1]. We leave to the reader, as Exercise 10.15, the existence of $x \in \{2, 3, \ldots, k\}$ such that

$$2x \equiv \left(\binom{k}{2} + 1\right) \; (\bmod r).$$

Let $\chi : [1, r + 1] \rightarrow \{0, 1, \dots, x - 1, x + 1, \dots, k\}$ be the k-coloring represented by

$$\underbrace{0\dots 0}_{r-k+1} 1 1 2 3 \dots (x-1) (x+1) (x+2) \dots k.$$

Let

288

$$s = 1 + \sum_{\substack{i=1\\i \neq x}}^{k} i$$

so that

$$s = \binom{k}{2} - x + 1 \equiv x \,(\text{mod}\,r).$$

For any r elements in [1, r + 1], the sum of their colors, under χ , is $(s-t) \pmod{r}$ for some $t \in \{0, 1, \ldots, x-1, x+1, x+2, \ldots, k\}$. By our choice of x, we have $s-t \equiv (x-t) \not\equiv 0 \pmod{r}$. Hence, there is no collection of r elements which is zero-sum under χ . This proves that $g(r,k) \geq r+2$, and completes the proof.

10.7. Exercises

- **10.1** Find F(3; 2).
- **10.2** Let \widehat{F} be the least positive integer such that for any 2-coloring of $[1, \widehat{F}]$ there exist $x, y \in [1, \widehat{F}]$, not necessarily distinct, such that $\{x, y, x + y, xy\}$ is monochromatic. Show that $\widehat{F} = 39$.
- **10.3** Prove Lemma 10.19.
- **10.4** Prove Lemma 10.20. (Hint: see Theorem 4.32.)
- 10.5 Prove Corollary 10.23 by showing that D is not 3-accessible.
- **10.6** Using the notation of the proof of Theorem 10.24, show that for k = 4 and k = 5, for every 2-coloring of [1, f(k)], there exist S and T as described in the proof.
- 10.7 Complete the proof of Theorem 10.24.
- **10.8** Prove that a = 2 is the only value of $a, a \ge 2$, for which $\{a^i : i = 0, 1, 2, ...\}$ is 2-accessible.
- **10.9** Let $F = \{F_0, F_1, ...\}$ be the set of Fibonacci numbers initialized by $F_0 = F_1 = 1$. Show that $\Delta(F, k; 2) \leq F_{k+3} 2$.
- **10.10** Let $k \ge 2$. Show that $\Delta(S_3, k; 2) = 4k 5$ and

$$\Delta(S_4, k; 2) = \begin{cases} 3k - 4 \text{ if } k \text{ is odd,} \\ 3k - 3 \text{ if } k \text{ is even.} \end{cases}$$

- m nor n. Do parts (a) and (b) to show that $\Delta(V_{3,4}, k; 2) = 7k 12$ for $k \ge 3$.
 - a) Show that $\Delta(V_{3,4}, k; 2) \ge 7k 12$ for $k \ge 3$ by proving:
 - i) for k even, the coloring $1(10011000110011)^{\frac{k-2}{2}}$ avoids monochromatic k-term $V_{3,4}$ -diffsequences, and
 - ii) for k odd, the coloring $1(10011000110011)^{\frac{k-3}{2}}(1001100)$ avoids monochromatic k-term $V_{3,4}$ -diffsequences.

b) Show that $\Delta(V_{3,4}, k; 2) \leq 7k - 12$ for $k \geq 3$. (Hint: show that every 2-coloring of [1, 7k - 12] has monochromatic $V_{3,4}$ -diffsequences $X = \{x_1, x_2, \ldots, x_m\}$ and $Y = \{y_1, x_2, \ldots, y_n\}$, of different colors, such that $m + n \geq 2k - 1$. Then consider $x_m - y_n \equiv i \pmod{12}$ for $i = 0, 1, \ldots, 11$.)

10.12 Prove Theorem 10.30.

10.8. Research Problems

- 10.13 Prove that Theorems 10.32 and 10.33 are equivalent.
- **10.14** Prove that the Thue-Morse sequence is cubefree and that the Morse-Hedlund sequence is squarefree.
- 10.15 Finish the proof of Theorem 10.49 by proving the existence of x. Also, where is $r \ge 5$ needed?

10.8. Research Problems

- *10.1 Prove or disprove: For all $r \ge 1$, any *r*-coloring of \mathbb{Z}^+ must admit a monochromatic set of the form $\{x, y, x + y, xy\}$. References: [212], [232]
- **10.2** Define f(n) to be the least positive integer such that for every 2-coloring of [1, f(n)] there is an *n*-element set with no pair of elements differing by exactly 2. For example, the subset $\{1, 2, 5\}$ of [1, 5] shows that $f(3) \leq 5$. On the other hand, it is impossible to find a 3-element subset of [1, 4] for which no pair differs by exactly 2. Hence $f(3) \geq 5$. Do a study of f(n). Try the same problem when the difference 2 is replaced by some other constant. Try more than two colors. References: [156], [157], [158]
- *10.3 Determine if P, the set of primes, is 2-accessible. More generally, does there exist an even number e such that P + e

is 2-accessible? If so, what is the order of magnitude of $\Delta(P+e,k;2)$? Reference: [174]

- *10.4 Let c be an odd positive integer. What is doa(P + c)? In particular, is P + c 3-accessible? Run a computer program to calculate various values of $\Delta(P + c, k; 2)$ (some are given in Table 10.1) and $\Delta(P + c, k; 3)$. Reference: [174]
- **10.5** Let $S = \{2\} \cup \{2i + 1 : i \ge 1\}$. Study the function $\Delta(S, k; 2)$. Reference: [174]
- **10.6** Determine the true order of magnitude of $\Delta(F, k; 2)$, where F is the set of Fibonacci numbers. Reference: [174]
- 10.7 Let $T = \{2^i : i \ge 0\}$. Determine the true order of magnitude of $\Delta(T, k; 2)$ (the present authors have conjectured that $\Delta(T, k; 2) = 8k 23$, the lower bound given by Theorem 10.21).

References: [174]

10.8 Extend the results on $\Delta(T, k; 2)$ (see Research Problem 10.7) and $\Delta(F, k; 2)$ (see Research Problem 10.6) to more than 2 colors.

Reference: [174]

- 10.9 Does there exist an accessible set that is not large (see Definition 4.28)?References: [62], [174]
- **10.10** Find a formula for $f(V_m, k; 2)$ for $m \ge 5$. Reference: [174]
- **10.11** Find a formula for $f(V_{m,n}, k; 2)$ for $3 \le m < n$. (See Exercise 10.11 for the definition of $V_{m,n}$.) Reference: [174]
- 10.12 Improve the bounds found in Theorem 10.40. References: [50], [83], [201], [262]
- **10.13** Find g(r,k) for $5 \le k \le 1 + \frac{n}{2}$ (see Definition 10.46). References: **[11]**, **[37]**, **[40]**, **[48]**, **[49]**, **[117]**

10.9. References

§10.1. The original proof of Folkman's theorem is unpublished, but reproduced in [127]. The proof presented in this section is essentially this proof. It was also proved by Rado [212] and, independently, by Sanders [232]. The proof of Theorem 10.9 can be found in [97]. A proof based only on Ramsey's theorem (and not van der Waerden's theorem) can be found in [197]. A powerful generalization of Folkman's theorem is due to Hindman [143] (it is known as Hindman's theorem). A simpler proof of Hindman's theorem was given by Baumgartner [19]. Related work is found in [144] and [145]; also see [146] for an excellent survey, and [253].

§10.2. Extensions of doublefree sets can be found in [98].

§10.3. All results presented concerning diffsequences are from [174], which contains further work on the subject. The conjecture that the set of large sets is equal to the set of accessible sets was posed by Tom Brown.

§10.4. Brown's lemma is due to Tom Brown and is taken from [**52**]. We have denoted the numbers in Theorem 10.33 by B(f;r) in honor of his contributions to the field of Ramsey theory on the integers. A generalization of Brown's lemma is investigated in [**54**].

§10.5. The Thue-Morse sequence was defined originally by Prouhet [**208**], and rediscovered by Thue (see [**268**]) with additional work by Morse [**193**]. For a good expository article about Thue, see [**35**]. The Morse-Hedlund sequence is from [**194**]. Regarding Theorem 10.40, the lower bound for sq(n) is from [**262**], the upper bound for sq(n) is from [**201**], the lower bound for c(n) is from [**50**], and the upper bound for c(n) is from [**83**]. Good general references for this section are [**13**], [**182**], and [**183**].

§10.6. There are at least five different proofs of the Erdős-Ginzburg-Ziv theorem. The one given here follows the original [**91**], which can be found in [**14**], along with four other proofs of the Erdős-Ginzburg-Ziv theorem. Some generalizations of the Erdős-Ginzburg-Ziv theorem are discussed in [**38**]. The definition of g(n, k) is given in [**40**]. Theorem 10.42 can be found in [**11**]. The proof of Lemma 10.43 is from [**117**], but was first proved in [**48**]. In [**49**], Brakemeier also

considers g(n,k) for *n* prime. Proposition 10.47 is from [**37**]. The cases k = 3, 4 of g(n, k) are studied in [**40**]. Theorem 10.49 was first proved in [**49**] and then independently in [**117**].

Additional References: There are many other Ramsey-type problems that have not been discussed in this book, but which are certainly of interest. Abbott, Liu, and Riddell [10] considered a function much like the Erdős and Turán function (see Section 2.5), but where one tries to avoid arithmetic progressions in a set of n real numbers, rather than in [1, n]. Let m(n) represent the largest integer m such that there exists a k-term sequence in [1, m] with the property that no member of the sequence is equal to the mean of the other members. Bounds on m(n) are given in [1], [2], [47], [96], and [100].

Notation

Notation	Description	Page
٢٠٦	Ceiling function	10
ſſ	Floor function	10
\oplus	Modular addition	107
[a, b]	$\{a,a+1,\ldots,b\}$	9
A - B	$\{x\in A: x\not\in B\}$	9
A_D	Family of arithmetic progressions with gaps	104
	in D	
AP	Family of arithmetic progressions	14
$AP_{a(m)}$	Family of arithmetic progressions with gaps	163
	congruent to $a \pmod{m}$	
$AP^*_{a(m)}$	$AP_{a(m)} \cup A_{\{m\}}$	168
$AP_{(m)}$	Set of arithmetic progressions \pmod{m}	164
AUG_b	Family of augmented progressions with tail b	152
B(f;r)	Brown number	278
cul_j	Culprit of color j	32
$\Delta_{\{a,b,c\}}$	Triangle on vertices a, b, c	205
$\Delta(D,k;r)$	Diffsequence Ramsey-type number	266
doa	Degree of accessibility	270
dor	Degree of regularity	138
dor_k	Degree of regularity for $T_{k-1}(a)$	145
DW(k)	Descending wave 2-color Ramsey-type number	70
F(k;r)	Folkman number	262
$\Gamma_m(k)$	Least s guaranteeing k -term arithmetic	73
	progressions in all s-term $[1, m]$ -gap	
	sequences	
$GQ_{\delta}(k)$	2-color Ramsey-type function for generalized	67
	quasi-progressions	

294

Notation

Notation	Description	Page
g(r,k)	Least integer such that for all S with $ S = g$	286
	and all $\chi: S \to \{0, \ldots, r-1\}$, whenever	
	$ \chi(S) = k$ there exist $t_1, t_2, \ldots, t_r \in S$ with	
	(t_1, t_2, \ldots, t_r) zero-sum	1
gs	Gap size	278
(k, n, d)-progression	k-term quasi-progression with diameter n	57
TT(x)	and low-difference d	1.477
$H(s_1,\ldots,s_k)$	2-color Ramsey-type function for nomothetic	147
λ (μ μ μ)	copies of $\{1, 1+s_1, \ldots, 1+s_1+\cdots+s_k\}$	110
$\lambda(c,\kappa;r)$	Special r-coloring of $[1, cr(k-1)]$ that	112
$\mathcal{C}(\mathbf{A})$	Equation multiplication multiplicati	010
$\mathcal{L}(l)$	Equation $x_1 + \cdots + x_{t-1} = x_t$ min $(\mathcal{L} \cdot \Gamma - (V, \mathcal{L}))$ is a human mathematic	212
$\mu(\kappa)$	satisfying Property B and $ E = k$	- 30
	for all $E \in \mathcal{E}$ }	
$M_{\chi}(n)$	Number of monochromatic Schur triples	205
	in $[1, n]$ under χ	
u(k)	Erdős and Turán function	41
$\Omega_m(k)$	Least n so that every $\{x_1, \ldots, x_n\}$ with	182
	$x_i \in [(i-1)m, im-1]$ contains a k-term	
	arithmetic progression	
P_n	Family of sequences generated by iteration	81
	of a polynomial of degree n	
$P_{n,k}$	Family of k -term members of P_n	81
$Q_n(k)$	2-color Ramsey-type function for	57
	quasi-progressions	
$R(AP^*_{a(m)},k,l;r)$	Generalization of $R(AP^*_{a(m)}, k; r)$	168
$r(\mathcal{E};s)$	Rado number for equation ${\cal E}$	230
$R(\mathcal{F},k;r)$	Ramsey-type number for family ${\cal F}$	14
$R(k_1,\ldots,k_r)$	r-color (classical) Ramsey number	8
$R_r(k)$	$R(\underbrace{k,\ldots,k})$	8
BB(S,r)	r Beverse r-regular number	185
$S(k_1, k_2)$	Generalized Schur number	212
$\hat{S}(k_1,\ldots,k_r)$	Strict generalized Schur number	218
S_{n}	Family of sequences generated by iteration	82
$\sim n$	of a polynomial of degree at most n	
S. h	Family of k-term members of S_{m}	82
$SP_m(k)$	Ramsey-type number for semi-progressions	72
s(r)	Schur number	201
$\hat{s}(r)$	Strict Schur number	217
$T_{a,b}$	Set of (a, b) -triples	136
T(a, b; r)	Ramsey-type number for (a, b) -triples	136
		1

Notation

Notation	Description	Page
$T(a_1,\ldots,a_{k-1})$	Ramsey-type function for generalization	145
	of (a, b) -triples	110
$\Theta(n)$	Set of permutations of $[1, n]$ with no 3-term	194
	arithmetic progression	101
heta(n)	$ \Theta(n) $	103
V_m	$\{x \in \mathbb{Z}^+ : m \nmid x\}$	273
$V_{m,n}$	$\{x \in \mathbb{Z}^+ : m \nmid x, n \nmid x\}$	213
w(k)	w(k;2)	203
w(k;r)	van der Waerden number	11
$w(k_1,\ldots,k_r;r)$	Mixed van der Waerden number	22
w'(c,k;r)	Ramsey-type number for arithmetic progressions	111
	with gaps at least c	111
w'(f(x),k;r)	Ramsey-type number for f-a p 's	116
$w^*(k, j)$	Bamsey-type number for arithmetic and mark	110
(, , , ,	with color discrepancy at least i	187
Z ⁺	Positive integers	
-	I OSITIVE INTEREIS	9

- H. Abbott, On a conjecture of Erdős and Straus on non-averageing sets of integers, in <u>Proceedings of the Fifth British Combinatorial</u> <u>Conference</u>, 1-4, Utilitas Mathematica Publishing, Inc., Winnipeg, Canada, 1976.
- [2] H. Abbot, On the Erdős-Straus non-averaging set problem, Acta Mathematica Academiae Scientiarum Hungaricae 47 (1986), 117-119.
- [3] H. Abbott and D. Hanson, Lower bounds for certain types of van der Waerden numbers, J. Combinatorial Theory Series A 12 (1972), 143-146.
- [4] H. Abbott and D. Hare, Families of 4-sets without property B, Ars Combinatoria 60 (2001), 239-245.
- [5] H. Abbott and A. Liu, On partitioning integers into progression free sets, J. Combinatorial Theory Series A 13 (1972), 432-436.
- [6] H. Abbott and A. Liu, On property B(4) of families of sets, Ars Combinatoria 26 (1988), 59-68.
- [7] H. Abbott and A. Liu, On property B(s), Ars Combinatoria 7 (1979), 255-260.
- [8] H. Abbott and A. Liu, On property B(s). II, Discrete Math. 37 (1981), 135-141.
- [9] H. Abbott and A. Liu, Property B(s) and projective planes, Ars Combinatoria 20 (1985), 217-220.
- [10] H. Abbott, A. Liu, and J. Riddell, On sets of integers not containing arithmetic progressions of prescribed length, J. Australian Math. Society 18 (1974), 188-193.

- [11] M. Aigner and G. Ziegler, <u>Proofs from The Book</u>, Springer-Verlag, Berlin, 1st ed., 1998.
- [12] M. Ajtai, J. Komlós, and E. Szemerédi, A note on Ramsey numbers, J. Combinatorial Theory Series A 29 (1980), 354-360.
- [13] J.-P. Allouche and J. Shallit, <u>Automatic Sequences: Theory</u>, <u>Applications, Generalizations</u>, Cambridge University Press, Cambridge, U.K., 2003.
- [14] N. Alon and M. Dubiner, Zero-sum sets of prescribed size, in Combinatorics, Paul Erdős is Eighty, Vol. 1, 33-50, Bolyai Society Math. Studies, Budapest, 1993.
- [15] N. Alon and J. Spencer, Ascending waves, J. Combinatorial Theory Series A 52 (1989), 275-287.
- [16] N. Alon and A. Zaks, Progressions in sequences of nearly consecutive integers, J. Combinatorial Theory Series A 84 (1998), 99-109.
- [17] P. Anderson, A generalization of Baudet's conjecture (van der Waerden's theorem), American Math. Monthly 83 (1976), 359-361.
- [18] J. Baumgartner, Partitioning vector spaces, J. Combinatorial Theory Series A 18 (1975), 231-233.
- [19] J. Baumgartner, A short proof of Hindman's theorem, J. Combinatorial Theory Series A 17 (1974), 384-386.
- [20] J. Beck, On 3-chromatic hypergraphs, Discrete Math. 24 (1978), 127-137.
- [21] J. Beck, A remark concerning arithmetic progressions, J. Combinatorial Theory Series A 29 (1980), 376-379.
- [22] J. Beck, Roth's estimate of the discrepancy of integer sequences is nearly sharp, *Combinatorica* 1 (1981), 319-325.
- [23] J. Beck and V. Sós, Discrepancy theory, in <u>Handbook of</u> <u>Combinatorics</u>, Vol. 2, 1405–1446, Elsevier, Amsterdam, 1995.
- [24] M. Beeler, A new van der Waerden number, Discrete Applied Math. 6 (1983), 207.
- [25] M. Beeler and P. O'Neil, Some new van der Waerden numbers, Discrete Math. 28 (1979), 135-146.
- [26] F. Behrend, On sets of integers which contain no three terms in arithmetical progression, *Proceedings National Academy Science U. S. A.* 32 (1946), 331-332.
- [27] V. Bergelson, A density statement generalizing Schur's theorem, J. Combinatorial Theory Series A 43 (1986), 338-343.
- [28] V. Bergelson, Ergodic Ramsey theory an update, in Ergodic Ramsey Theory of \mathbb{Z}^d -actions, 1-61, Cambridge, U.K., 1996.

- [29] V. Bergelson, W. A. Deuber, and N. Hindman, Rado's theorem for finite fields, sets, graphs and numbers, *Colloquia Mathematica Societatis* János Bolyai 60 (1992), 77-88.
- [30] V. Bergelson, H. Furstenburg, N. Hindman, and Y. Katznelson, An algebraic proof of van der Waerden's theorem, L'Enseignement Mathématique 35 (1989), 209-215.
- [31] V. Bergelson and N. Hindman, Desnity versions of two generalizations of Schur's theorem, J. Combinatorial Theory Series A 48 (1988), 32-38.
- [32] V. Bergelson and N. Hindman, Partition regular structures contained in large sets are abundant, J. Combinatorial Theory Series A 93 (2001), 18-36.
- [33] V. Bergelson and A. Leibman, Polynomial extensions of van der Waerden's and Szemerédi's theorems, J. American Mathematical Society 9 (1996), 725-753.
- [34] E. Berlekamp, A construction for partitions which avoid long arithmetic progressions, *Canadian Math. Bulletin* **11** (1968), 409-414.
- [35] J. Berstel, Axel Thue's work on repetitions in words, Séries Formelles et Combinatoire Algébrique 11 (1992), LaCIM, Université de Québec à Montréal, 65-80.
- [36] A. Beutelspacher and W. Brestovansky, Generalized Schur numbers, Lecture Notes in Math. 969 (1982), 30-38.
- [37] A. Bialostocki and P. Dierker, On the Erdős-Ginzburg-Ziv theorem and the Ramsey numbers for stars and matchings, *Discrete Math.* 110 (1992), 1-8.
- [38] A. Bialostocki, Paul Erdős, and H. Lefmann, Monochromatic and zerosum sets of nondecreasing diameter, *Discrete Math.* 137 (1995), 19-34.
- [39] A. Bialostocki, H. Lefmann, and T. Meerdink, On the degree of regularity of some equations, *Discrete Math.* **150** (1996), 49-60.
- [40] A. Bialostocki and M. Lotspeich, Some developments of the Erdős-Ginzburg-Ziv theorem I, Colloquia Mathematica Societatis János Bolyai 60 (1992), 97-117.
- [41] A. Bialostocki and D. Schaal, On a variation of Schur numbers, Graphs and Combinatorics 16 (2000), 139-147.
- [42] T. Bisztriczky and G. Tóth, A generalization of the Erdős-Szekeres convex n-gon theorem, Journal für die Reine und Angewandte Mathematik 395 (1989), 167-170.
- [43] M. Bóna, Permutations avoiding certain patterns: the case of length 4 and some generalizations, *Discrete Math.* 175 (1997), 55-67.
- [44] M. Bóna, Permutations with one of two 132-subsequences, Discrete Math. 181 (1998), 267-274.

- [45] P. Bornsztein, On an extension of a theorem of Schur, Acta Arithmetica 101 (2002), 395-399.
- [46] E. Boros, $PG(2, p^s)$, p > 2, has property B(p + 2), Ars Combinatoria **25** (1988), 111-113.
- [47] A. Bosznay, On the lower estimate of non-averaging sets, Acta Mathematica Academiae Scientiarum Hungaricae 53 (1989), 155-157.
- [48] W. Brakemeier, Ein Beitrag zur additiven Zahlentheorie, *Ph.D. thesis*, Braunschweig, 1973.
- [49] W. Brakemeier, Eine Anzahlformel von Zahlen modulo n, Monatshefte für Mathematik 85 (1978), 277-282.
- [50] F.-J. Brandenburg, Uniformly growing kth power-free homomorphisms, Theoretical Computer Science 23 (1983), 69-82.
- [51] A. Brauer, Über Sequenzen von Potenzresten, Sitzungsber. Preuss. Akad. Wiss. Phys. Math. Kl. (1928), 9-16.
- [52] T. C. Brown, An interesting combinatorial method in the theory of locally finite semigroups, *Pacific J. Math.* 36 (1971), 285-289.
- [53] T. C. Brown, On van der Waerden's theorem and a theorem of Paris and Harrington, J. Combinatorial Theory Series A **30** (1981), 108-111.
- [54] T. C. Brown, Monochromatic forests of finite subsets of N, Integers: Electronic J. Combinatorial Number Theory 0 (2000), A4.
- [55] T. C. Brown, Some new van der Waerden numbers (preliminary report), Notices American Math. Society 21 (1974), A-432.
- [56] T. C. Brown, Variations on van der Waerden's and Ramsey's theorems, American Math. Monthly 82 (1975), 993-995.
- [57] T. C. Brown, Paul Erdős, F. Chung, and R. Graham, Quantitative forms of a theorem of Hilbert, J. Combinatorial Theory Series A 38 (1985), 210-216.
- [58] T. C. Brown, Paul Erdős, and A. R. Freedman, Quasi-progressions and descending waves, J. Combinatorial Theory Series A 53 (1990), 81-95.
- [59] T. C. Brown and A. R. Freedman, Arithmetic progressions in lacunary sets, Rocky Mountain J. Math. 17 (1987), 587-596.
- [60] T. C. Brown and A. R. Freedman, Small sets which meet all the k(n)-term arithmetic progressions in the interval [1, n], J. Combinatorial Theory Series A 51 (1989), 244-249.
- [61] T. C. Brown, A. Freedman, and P. J.-S. Shiue, Progressions of squares, Australasian J. Combinatorics 27 (2003), 187-192.
- [62] T.C. Brown, R.L. Graham, and B. Landman, On the set of common differences in van der Waerden's theorem on arithmetic progressions, *Canadian Mathematical Bulletin* 42 (1999), 25-36.

- [63] T. C. Brown and D. Hare, Arithmetic progressions in sequences with bounded gaps, J. Combinatorial Theory Series A 77 (1997), 222-227.
- [64] T.C. Brown and B. Landman, Monochromatic arithmetic progressions with large differences, Bulletin Australian Math. Society 60 (1999), 21-35.
- [65] T.C. Brown and B. Landman, The Ramsey property for collections of sequences not containing all arithmetic progressions, *Graphs and Combinatorics* 12 (1996), 149-161.
- [66] T. C. Brown, B. Landman and M. Mishna, Monochromatic homothetic copies of {1, 1+s, 1+s+t}, Canadian Math. Bulletin 40 (1997),149-157.
- [67] T. C. Brown and V. Rödl, Monochromatic solutions to equations with unit fractions, Bulletin Australian Math. Society 43 (1991), 387-392.
- [68] T. C. Brown and J.-S. Shiue, On the history of van der Waerden's theorem on arithmetic progressions, *Tamkang J. Math.* 32 (2001), 335-241.
- [69] S. Burr and S. Loo, On Rado numbers I, unpublished.
- [70] S. Burr and S. Loo, On Rado numbers II, unpublished.
- [71] V. Chvátal, Some unknown van der Waerden numbers, in Combinatorial Structures and their Applications, 31-33, Gordon and Breach, New York, 1970.
- [72] K. Compton, A van der Waerden variant, Electronic J. Combinatorics 6 (1999), R22.
- [73] E. Croot, On a coloring conjecture about unit fractions, Annals of Math. (2) 157 (2003), 545-556.
- [74] C. Culver and B. Landman, Some new mixed van der Waerden numbers, preprint.
- [75] B. Datskovsky, On the number of monochromatic Schur triples, Advances in Applied Math. 31 (2003), 193-198.
- [76] J. Davis, R. Entringer, R. Graham, and G. Simmons, On permutations containing no long arithmetic progressions, Acta Arithmetica 34 (1977), 81-90.
- [77] W. Deuber, Developments based on Rado's dissertation "Studien zur Kombinatorik," London Math. Society Lecture Note Series 141 (1989), 52-74.
- [78] W. Deuber, On van der Waerden's theorem on arithmetic progressions,
 J. Combinatorial Theory Series A 32 (1982), 115-118.
- [79] W. Deuber, Partitionen und lineare Gleichungssysteme, Mathematische Zeitschrift 133 (1973), 109-123.
- [80] W. Deuber, Partition theorems for abelian groups, J. Combinatorial Theory Series A 19 (1975), 95-108.

- [81] W. Deuber and H. Lefmann, Partition regular systems of homogeneous linear equations over abelian groups: the canonical case, Annals New York Acadamy Science 555 (1989), 167-170.
- [82] P. Ding and A. R. Freedman, Semi-progressions, J. Combinatorial Theory Series A 76 (1996), 99-107.
- [83] A. Edlin, The number of binary cube-free words of length up to 47 and their numerical analysis, J. Difference Equations and Applications 5 (1999), 353-354.
- [84] Paul Erdős, Extremal problems in number theory II, Matematikai Lapok 17 (1966), 135-155.
- [85] Paul Erdős, On a combinatorial problem. Nordisk Matematisk Tidskrift 11 (1963) 5-10, 40.
- [86] Paul Erdős, On a combinatorial problem. II, Acta Mathematica Academiae Scientiarum Hungaricae 15 (1964), 445-447.
- [87] Paul Erdős, On a combinatorial problem. III, Canadian Math. Bulletin 12 (1969), 413-416.
- [88] Paul Erdős, Problems and results on combinatorial number theory, in Survey of Combinatorial Number Theory, J.N. Srivastava et al., ed., 117-138, North Holland, Amsterdam, 1973.
- [89] Paul Erdős, Some new problems and results in graph theory and other branches of combinatorial mathematics, *Lecture Notes in Math.* 885 (1981), 9-17.
- [90] Paul Erdős, Some remarks on the theory of graphs, Bulletin American Math. Society 53 (1947), 292-294.
- [91] Paul Erdős, A. Ginzburg, and A. Ziv, Theorem in additive number theory, Bulletin Research Council of Isreal 10 (1961-2), Section F, 41-43.
- [92] Paul Erdős and R. L. Graham, <u>Old and New Problems and Results</u> in Combinatorial Number Theory, *Monographies de L'Enseignement Mathematique* 28 (1980), University of Geneva.
- [93] Paul Erdős and A. Hajnal, On a property of families of sets, Acta Mathematica Academiae Scientiarum Hungaricae 12 (1961), 87-123.
- [94] Paul Erdős and L. Lovasz, Problems and results on 3-chromatic hypergraphs and some related questions, *Colloquia Mathematica Societatis* János Bolyai 10 (1975), 609-627.
- [95] Paul Erdős and R. Rado, Combinatorial theorems on classifications of subsets of a given set, *Proceedings London Math. Society* 3 (1952), 417-439.
- [96] Paul Erdős and A. Sarkozy, On a problem of Straus, in Disorder in Physical Systems, 55–66, Oxford University Press, New York, 1990.

- [97] Paul Erdős and J. Spencer, Monochromatic sumsets, J. Combinatorial Theory Series A 50 (1989), 162-163.
- [98] Paul Erdős and J. Spencer, A problem in covering progressions, *Studia* Scientiarum Mathematicarum Hungarica **30** (1995), 149-154.
- [99] Paul Erdős and J. Spencer, <u>Probabilistic Methods in Combinatorics</u>, Academic Press, 1974.
- [100] Paul Erdős and E. Straus, Nonaveraging sets, II, Colloquia Mathematica Societatis János Bolyai 4 (1970), 405-411.
- [101] Paul Erdős and G. Szekeres, A combinatorial problem in geometry, Composito Math. 2 (1935), 464-470.
- [102] Paul Erdős and P. Turán, On some sequences of integers, J. London Math. Society 11 (1936), 261-264.
- [103] Paul Erdős, Z. Tuza, and P. Valtr, Ramsey-remainder, European J. Combinatorics 17 (1996), 519-532.
- [104] Péter Erdős, Some generalizations of property B and the splitting property, Annals Combinatorics **3** (1999), 53-59.
- [105] F. Everts, Colorings of sets, Ph.D. thesis, University of Colorado, 1977.
- [106] G. Exoo, A lower bound for Schur numbers and multicolor Ramsey numbers of K_3 , *Electronic J. Combinatorics* 1 (1994), R8.
- [107] G. Exoo On constructing hypergraphs without Property B, Ars Combinatoria 30 (1990), 3-12.
- [108] P. Frankl, R. Graham, and V. Rödl, Quantitative theorems for regular systems, J. Combinatorial Theory Series A 47 (1988), 246-261.
- [109] P. Frankl, R. Graham, and V. Rödl, Quantitative Versionen von kombinatorischen Partitionssätzen (German), Jahresbericht der Deutschen Mathematiker-Vereinigung 92 (1990), 130-144.
- [110] H. Fredricksen and M. Sweet, Symmetric sum-free partitions and lower bounds for Schur numbers, *Electronic J. Combinatorics* 7 (2000), R32.
- [111] H. Furstenberg, <u>Dynamical Methods in Ramsey Theory</u>, American Math. Society, 2003.
- [112] H. Furstenberg, Ergodic behavior of diagonal measures and a theorem of Szemerédi on arithmetic progressions, J. Analyse Math. 31 (1977), 204-256.
- [113] H. Furstenberg, Poincaré recurrence and number theory, Bulletin American Math. Society 5 (1981), 211-234.
- [114] H. Furstenberg, A polynomial Szemerédi theorem, in Combinatorics, Paul Erdős is Eighty, Vol. 2, 253-269, Colloquia Mathematica Societatis János Bolyai, Budapest, 1996.

- [115] H. Furstenberg, <u>Recurrence in Ergodic Theory and Combinatorial</u> Number Theory, Princeton University Press, Princeton, NJ, 1981.
- [116] H. Furstenburg and B. Weiss, Topological dynamics and combinatorial number theory, J. Analyse Math. 34 (1978), 61-85.
- [117] L. Gallardo, G. Grekos, and J. Pihko, On a variant of the Erdős-Ginzburg-Ziv problem, Acta Arithmetica 89 (1999), 331-336.
- [118] J. Gerver, J. Propp, and J. Simpson, Greedily partitioning the natural numbers into sets free of arithmetic progressions, *Proceedings Amereri*can Math. Society **102** (1988), 765-772.
- [119] J. Gerver and L. Ramsey, Sets of integers with nonlong arithmetic progressions generated by the greedy algorithm, *Math. Computation* 33 (1979), 1353-1359.
- [120] A. W. Goodman, On sets of acquaintances and strangers at a party, American Math. Monthly 66 (1959), 778-783.
- [121] W. T. Gowers, A new proof of Szemerédi's theorem, Geometric and Functional Analysis 11 (2001), 465-588.
- [122] R. Graham, <u>Rudiments of Ramsey theory</u>, American Math. Society, 1981.
- [123] R. Graham, M. Grötschel, and L. Lovász editors, <u>Handbook</u> of <u>Combinatorics</u>, Vol. 1, The MIT Press, North-Holland, 1995.
- [124] R. Graham, M. Grötschel, and L. Lovász editors, <u>Handbook</u> of <u>Combinatorics</u>, Vol. 2, The MIT Press, North-Holland, 1995.
- [125] R. Graham, V. Rödl, and A. Ruciński, On Schur properties of random subsets of integers, J. Number Theory 61 (1996), 388-408.
- [126] R. Graham and B. Rothschild, A short proof of van der Waerden's theorem on arithmetic progression, *Proceedings American Math. Soci*ety 42 (1974), 385-386.
- [127] R. Graham, B. Rothschild, and J. Spencer, <u>Ramsey Theory</u>, Wiley-Interscience, 2nd ed., 1990.
- [128] R. Graham, M. Simonovits, and V. Sós, A note on the intersection properties of subsets of integers, J. Combinatorial Theory Series A 28 (1980), 106-110.
- [129] R. Graham, J. Spencer, and H. Witsenhausen, On extremal density theorems for linear forms, in <u>Number Theory and Algebra</u>, 103-109, ed. by H. Zassenhaus, Academic Press, New York, 1977.
- [130] J. Graver and J. Yackel, Some graph theoretic results associated with Ramsey's theorem, J. Combinatorial Theory 4 (1968), 125-175.
- [131] R. N. Greenwell and B. Landman, On the existence of a reasonable upper bound for the van der Waerden numbers, J. Combinatorial Theory Series A 50 (1989), 82-86.

- [132] R. Greenwood and A. Gleason, Combinatorial relations and chromatic graphs, Canadian J. Math. 7 (1955), 1-7.
- [133] C. Grinstead and S. Roberts, On the Ramsey numbers R(3,8) and R(3,9), J. Combinatorial Theory Series B **33** (1982), 27-51.
- [134] D. Gunderson and V. Rödl, Extremal problems for affine cubes of integers, *Combinatorics, Probability and Computing* 7 (1998), 65-79.
- [135] R. Guy, <u>Unsolved problems in number theory</u>, Springer-Verlag, New York, 1994.
- [136] A. Hales and R. Jewett Regularity and positional games, Transactions American Math. Society 106 (1963), 222-229.
- [137] D. Hanson, Lower bounds for certain types of van der Waerden numbers, J. Combinatorial Theory Series A 12 (1972), 143-146.
- [138] F. Harary, A tribute to F. P. Ramsey, J. Graph Theory 7 (1983), 1-7.
- [139] H. Harborth and S. Maasberg, All two-color Rado numbers for a(x + y) = bz, Discrete Math. **197/198** (1999), 397-407.
- [140] H. Harborth and S. Maasberg, Rado numbers for a(x + y) = bz, J. Combinatorial Theory Series A 80 (1997), 356-363.
- [141] Y. R. Haung and J. S. Yang, New upper bounds for van der Waerden numbers W(3, n) (Chinese), Chinese Annals of Math. Series A 21 (2000), 631-634.
- [142] D. Hilbert, Uber die Irreducibilität ganzer rationaler Functionen mit ganzzahligen Coefficienten, Journal für die Reine und Angewandte Mathematik 110 (1892), 104-129.
- [143] N. Hindman, Finite sums from sequences within cells of a partition of N, J. Combinatorial Theory Series A 17 (1974), 1-11.
- [144] N. Hindman, Partitions and sums and products of integers, Transactions American Math. Society 247 (1979), 227-245.
- [145] N. Hindman, Partitions and sums and products two counterexamples, J. Combinatorial Theory Series A 29 (1980), 113-120.
- [146] N. Hindman, Ultrafilters and combinatorial number theory, Lecture Notes in Math 751 (1979), Springer-Verlag, Berlin, 119-184.
- [147] N. Hindman and I. Leader, Partition regular inequalities, European J. Combinatorics 19 (1998), 573-578.
- [148] P. Hoffman, <u>The Man Who Loved Only Numbers</u>, Hyperion, New York, 1998.
- [149] R. Irving, An extension of Schur's theorem on sum-free partitions, Acta Arithmetica 25 (1973-74), 55-64.
- [150] D. Johnson, On property B_r , J. Combinatorial Theory Series B 20 (1976), 64-66.

- [151] V. Jungic, Elementary, Topological, and Experimental Approaches to the Family of Large Sets, *Ph.D. thesis*, Simon Fraser University, Burnaby, BC, Canada, 1999.
- [152] J. Justin, Théorème de van der Waerden, lemme de Brown, et demigroupes répétitifs (unpublished preprint), in: Journees sur la theorie algebrique des demi-groups (Faculty de Sciences de Lyon, 1971).
- [153] J. Kalbfleisch, Chromatic Graphs and Ramsey's Theorem, *Ph.D. the*sis, University of Waterloo, Waterloo, Ontario, Canada, 1966.
- [154] G. Kéry, On a theorem of Ramsey, Matematikai Lapok 15 (1964), 204-224.
- [155] J. Kim, The Ramsey number R(3,t) has order of magnitude $t^2/\log t$, Random Structures and Algorithms 7 (1995), 173-207.
- [156] J. Konvalina, On the number of combinations without unit seperation, J. Combinatorial Theory Series A 31 (1981), 101-107.
- [157] J. Konvalina and Y.-H. Liu, Subsets without q-separation and binomial products of Fibonacci numbers, J. Combinatorial Theory Series A 57 (1991), 306-310.j
- [158] J. Konvalina and Y.-H. Liu, Subsets without unit separation and Fibonacci numbers, *Fibonacci Quarterly* 29 (1991), 141-144.
- [159] W. Kosek and D. Schaal, Rado numbers for the equation $\sum_{i=1}^{m-1} x_i + c = x_m$ for negative values of c, Advances Applied Math. 27 (2001), 805-815.
- [160] B. Landman, Avoiding arithmetic progressions (mod m) and arithmetic progressions, Utilitas Math. 52 (1997), 173-182.
- [161] B. Landman, Generalized van der Waerden numbers, Graphs and Combinatorics 2 (1986), 351-356.
- [162] B. Landman, Generalized van der Waerden Numbers, Ph.D thesis, Virginia Tech., 1983.
- [163] B. Landman Monochromatic sequences whose gaps belong to $\{d, 2d, \dots, md\}$, Bulletin Australian Math. Society 58 (1998), 93-101.
- [164] B. Landman, On avoiding arithmetic progressions whose common differences belong to a given small set, J. Combinatorial Math. and Combinatorial Computing 30 (1999), 221-229.
- [165] B. Landman, On some generalizations of the van der Waerden number w(3), Discrete Math. **207** (1999), 137-147.
- [166] B. Landman, Ramsey functions associated with second order recurrences, J. Combinatorial Math. and Combinatorial Computing 15 (1994), 119-127.
- [167] B. Landman, Ramsey functions for quasi-progressions, Graphs and Combinatorics 14 (1998), 131-142.

- [168] B. Landman, Ramsey functions related to the van der Waerden numbers, Discrete Math. 102 (1992), 265–278.
- [169] B. Landman, An upper bound for van der Waerden-like numbers using k colors, Graphs and Combinatorics 9 (1993), 177-184.
- [170] B. Landman and R. N. Greenwell, Accurate bounds for new van der Waerden type numbers, J. Combinatorial Math. and Combinatorial Computing 8 (1990), 103-106.
- [171] B. Landman and R. N. Greenwell, Some new bounds and values for van der Waerden-like numbers, *Graphs and Combinatorics* 6 (1990), 287-291.
- [172] B. Landman and R. N. Greenwell, Values and bounds for Ramsey numbers associated with polynomial iteration, *Discrete Math.* 68 (1988), 77-83.
- [173] B. Landman and A. Long, Ramsey functions for sequences with adjacent differences in a specified congruence class, *Congressus Numerantium* **103** (1994), 3-20.
- [174] B. Landman and A. Robertson, Avoiding monochromatic sequences with special gaps, preprint.
- [175] B. Landman and A. Robertson, On generalized van der Waerden triples, Discrete Math. 256 (2002), 279-290.
- [176] B. Landman and B. Wysocka, Collections of sequences having the Ramsey property only for few colours, *Bulletin Australian Math. Soci*ety 55 (1997), 19-28.
- [177] J. Layman and B. Landman, Note on the local growth of iterated polynomials, Aequationes Math. 27 (1984), 150-156.
- [178] H. Lefmann, A canonical version for partition regular systems of linear equations, J. Combinatorial Theory Series A 41 (1986), 95-104.
- [179] H. Lefmann, Note on monoton waves, J. Combinatorial Theory Series A 50 (1989), 316-318.
- [180] H. Lefmann, On partition regular systems of equations, J. Combinatorial Theory Series A 58 (1991), 35-53.
- [181] H. Lefmann and H. Si-Kaddour, On a maximality property of partition regular systems of equations, *Canadian Math. Bulletin* 36 (1993), 96-102.
- [182] M. Lothaire, Algebraic Combinatorics on Words, Cambridge University Press, Cambridge, U.K., 2002
- [183] M. Lothaire, <u>Combinatorics on Words</u>, Addison-Wesley Publishing Co., Reading, MA, 1983.
- [184] R. McCutcheon, Elemental Methods in Ergodic Ramsey Theory, Lecture Notes in Mathematics 1722, Springer-Verlag, Berlin, 1999.

- [185] R. McCutcheon, An infinitary van der Waerden theorem, J. Combinatorial Theory Series A 86 (1999), 214-231.
- [186] B. McKay and Z. K. Min, The value of the Ramsey number R(3,8),
 J. Graph Theory 19 (1995), 309-322.
- [187] B. McKay and S. Radziszowski, R(4,5) = 25, J. Graph Theory 19 (1995), 309-322.
- [188] D. H. Mellor, The eponymous F. P. Ramsey, J. Graph Theory 7 (1983), 9-13.
- [189] G. Mills, A quintessential proof of van der Waerden's theorem on arithmetic progressions, *Discrete Math.* **34** (1975), 292-299.
- [190] L. Mirsky, The combinatorics of arbitrary partitions, Bulletin of the Institute of Mathematics and its Applications 11 (1975), 6-9.
- [191] L. Moser, Notes on number theory. II. On a theorem of van der Waerden, *Canadian Math. Bulletin* **3** (1960), 23-25.
- [192] L. Moser, On non-averaging sets of integers, Canadian J. Math. 5 (1953), 245-252.
- [193] M. Morse, Recurrent geodesics on a surface of negative curvature, Transactions American Math. Society **22** (1921), 84-100.
- [194] M. Morse and G. Hedlund, Unending chess, symbolic dynamics, and a problem in semigroups, *Duke Math. J.* **11** (1944), 1-7.
- [195] M. Nathanson, Arithmetic progressions contained in sequences with bounded gaps, *Canadian Math. Bulletin* **23** (1980), 491-493.
- [196] M. Nathanson, Permutations, peridicity and chaos, J. Combinatorial Theory Series A 22 (1977), 61-68.
- [197] J. Nešetřil and V. Rödl, Another proof of the Folkman-Rado-Sanders theorem, J. Combinatorial Theory Series A 34 (1983), 108-109.
- [198] J. Nešetřil and V. Rödl, Partitions of vertices, Commentationes Mathematicae Universitatis Carolinae 17 (1976), 85-95.
- [199] J. Nešetřil and V. Rödl, editors, <u>Mathematics of Ramsey Theory</u>, Springer-Verlag, Berlin, 1990.
- [200] J. Noonan, The number of permutations containing exactly one increasing subsequence of length three, *Discrete Math.* 152 (1996), 307-313.
- [201] J. Noonan and D. Zeilberger, The Goulden-Jackson cluster method: extensions, applications, and implementations, J. Difference Equations and Applications 5 (1999), 355-377.
- [202] T. Odda, Solution to Problem E2440, American Math. Monthly 82 (1975), 74.
- [203] P. Parrilo, personal communication (2000).

- [205] C. Pomerance and A. Sárközy, Combinatorial number theory, in <u>Handbook of Combinatorics</u>, Vol. 1, 967-1018, Elsevier, Amsterdam, 1995.
- [206] H.-J. Prömel and B. Rothschild, A canonical restricted version of van der Waerden's theorem, *Combinatorica* 7 (1987), 115-119.
- [207] H.-J. Prömel and B. Voigt, <u>Aspects of Ramsey Theory</u>, Springer-Verlag, 1995.
- [208] M. Prouhet, Mémoire sur quelques relations entre les puissances des nombres, Comptes Rendus des Séances de l'Académie des Sciences, Paris 33 (1851), 225.
- [209] J. Rabung, On applications of van der Waerden's theorem, Math. Magazine 48 (1975), 142-148.
- [210] J. Rabung, Some progression-free partitions constructed using Folkman's method, Canadian Math. Bulletin 22 (1979), 87-91.
- [211] R. Rado, Note on combinatorial analysis, Proceedings of the London Math. Society 48 (1943), 122-160.
- [212] R. Rado, Some partition theorems, in Combinatorial Theory and its Applications, Vol. 3, 929-936, North-Holland, Amsterdam, 1970.
- [213] R. Rado, Some recent results in combinatorial analysis, Congrès International des Mathematiciens, Oslo, 1936.
- [214] R. Rado, Studien zur Kombinatorik, Mathematische Zeitschrift 36 (1933),424-480.
- [215] R. Rado, Verallgemeinerung eines Satzes von van der Waerden mit Anwendungen auf ein Problem der Zahlentheorie, Sonderausg. Sitzungsber. Preuss. Akad. Wiss. Phys.-Math. Klasse 17 (1933), 1-10.
- [216] S. Radziszowski, Small Ramsey numbers, *Electronic J. of Combina*torics Surveys (2002), DS1.
- [217] F. Ramsey, <u>The Foundations of Mathematics and Other Logical</u> <u>Essays</u>, ed. by R. Braithwaite, *International Library of Psychology*, <u>Philosophy and Scientific Method</u>, Littlefield, Adams & Co., Paterson, NJ, 1960.
- [218] F. Ramsey, On a problem of formal logic, Proceedings London Math. Society 30 (1930), 264-286.
- [219] R. Rankin, Representations of a number as the sum of a large number of squares, Proceedings Royal Society Edinburgh Section A 65 (1960/1961), 318-331.

- [220] P. Ribenboim, <u>The New Book of Prime Number Records</u>, Springer-Verlag, New York, 1996.
- [221] J. Riddell, A lattice point problem related to sets containing no *l*-term arithmetic progression, *Canadian Math. Bulletin* **14** (1971), 535-538.
- [222] A. Robertson, Difference Ramsey numbers and Issai numbers, Advances Applied Math. 25 (2000), 153-162.
- [223] A. Robertson, Permutations restricted by two distinct patterns of length three, Advances Applied Math. 27 (2001), 548-561.
- [224] A. Robertson and D. Schaal, Off-diagonal generalized Schur numbers, Advances Applied Math. 26 (2001), 252-257.
- [225] A. Robertson and D. Zeilberger, A 2-coloring of [1, n] can have $\frac{n^2}{22} + O(n)$ monochromatic Schur triples, but not less!, *Electronic J. Combinatorics* 5 (1998), R19.
- [226] V. Rödl and A. Ruciński, Rado partition theorem for random subsets of integers, *Proceedings London Math. Society* 74 (1997), 481-502.
- [227] K. Roth, On certain sets of integers, J. London Math. Society 28 (1953), 104-109.
- [228] K. Roth, Remark concerning integer sequences, Acta Arithmetica 9 (1964), 257-260.
- [229] K. Roth, Sur quelques ensembles d'entiers, Comptes Rendus des Séances de l'Académie des Sciences, Paris 234 (1952), 388-390.
- [230] R. Salem and D. Spencer, On sets of integers which contain no three terms in arithmetical progression, *Proceedings National Academy Sci*ence U. S. A. 28 (1942), 561-563.
- [231] R. Salem and D. Spencer, On sets which do not contain a given number of terms in arithmetical progression, *Nieuw Archief voor Wiskunde* 23 (1950), 133-143.
- [232] J. Sanders, A Generalization of Schur's Theorem, Ph.D. thesis, Yale University, 1969.
- [233] A. Sarkozy, On difference sets of sequences of integers. III, Acta Mathematica Academiae Scientiarum Hungaricae 31 (1978), 355-386.
- [234] A. Sarkozy, Some remarks concerning irregularities of distribution of sequences of integers in arithmetic progressions I, Colloquia Mathematica Societatis János Bolyai 13 (1976), 287-303.
- [235] A. Sarkozy, Some remarks concerning irregularities of distribution of sequences of integers in arithmetic progressions II, Studia Scientiarum Mathematicarum Hungarica 11 (1976), 79-103.
- [236] A. Sarkozy, Some remarks concerning irregularities of distribution of sequences of integers in arithmetic progressions III, *Periodica Mathematica Hungarica* 9 (1978),127-144.

- Bibliography
- [237] A. Sarkozy, Some remarks concerning irregularities of distribution of sequences of integers in arithmetic progressions IV, Acta Mathematica Academiae Scientiarum Hungaricae **30** (1977), 155-162.
- [238] A. Sarkozy and C. Stewart, On irregularities of distribution in shifts and dilations of integer sequences I, Mathematische Annalen 276 (1987), 353-364.
- [239] D. Schaal, On generalized Schur numbers, Congressium Numerantium 98 (1993), 178-187.
- [240] D. Schaal, Rado numbers for some inequalities and an arbitrary number of colors, *Congressus Numerantium* 121 (1996), 147-153.
- [241] D. Schaal, Upper and lower bounds for some Rado numbers for inequalities, Utilitas Math. 53 (1998), 63-69.
- [242] M. Schäffler, Partition regular systems of linear inequalities, Documenta Mathematica 3 (1998), 149-187.
- [243] B. Schechter, <u>My Brain is Open</u>, Simon and Schuster, New York, 1998.
- [244] F. Schmidt and R. Simion, Restricted permutations, European J. Combinatorics 6 (1985), 383-406.
- [245] W. Schmidt, Ein kombinatorisches Problem von P. Erdős und A. Hajnal, Acta Mathematica Academiae Scientiarum Hungaricae 15 (1964), 373-374.
- [246] W. Schmidt, Two combinatorial theorems on arithmetic progressions, Duke Math. J. 29 (1962), 129-140.
- [247] I. Schur, Über die Kongruenz $x^m + y^m = z^m \pmod{p}$, Jahresbericht der Deutschen Mathematiker-Vereinigung **25** (1916), 114-117.
- [248] T. Schoen, The number of monochromatic Schur triples, European J. Combinatorics 20 (1999), 855-866.
- [249] S. Shelah, Primitive recursive bounds for van der Waerden numbers, J. American Math. Society 1 (1988), 683-697.
- [250] A. Sidorenko, An infinite permutation without arithmetic progressions, Discrete Math. 69 (1988), 211.
- [251] W. Sierpiński, <u>Elementary Theory of Numbers</u>, Państwowe Wydawnictwo Naukowe, Warsaw, 1964.
- [252] M. Simonovits and V. Sós, Intersection theorems on structures, Annals Discrete Math. 6 (1980), 301-313.
- [253] G. Smith, Partitions and (m and n) sums of products, J. Combinatorial Theory Series A 72 (1995), 77-94.
- [254] H. Y. Song, H. Taylor, and S. W. Golomb, Progressions in every twocoloration of Z_n, J. Combinatorial Theory Series A 61 (1992), 211-221.

- [255] J. Spencer, Asymptotic lower bounds for Ramsey functions, Discrete Math. 20 (1977), 69-76.
- [256] J. Spencer, The Erdős existence argument, Algorithms and Combinatorics 13 (1997), 337-344. Springer, Berlin, 1997.
- [257] J. Spencer, A remark on coloring integers, Canadian Math. Bulletin 15 (1972), 43-44.
- [258] J. Spencer, Ramsey theory and Ramsey theoreticians, J. Graph Theory 7 (1983), 15-23.
- [259] J. Spencer, Restricted Ramsey configurations, J. Combinatorial Theory Series A 19 (1975), 278-286.
- [260] J. Spencer, Solution to Problem P. 185, Canadian Math. Bulletin 16 (1973), 464.
- [261] R. Stevens and R. Shantaram, Computer-generated van der Waerden partitions, *Math. Computation* **32** (1978), 635-636.
- [262] X. Sun, New lower bound on the number of ternary square-free words, preprint.
- [263] Z. Szabó, An application of Lovász's local lemma a new lower bound for the van der Waerden number, *Random Structures and Algorithms* 1 (1990), 343-360.
- [264] E. Szemerédi, On sets of integers containing no four elements in arithmetic progression, Acta Mathematica Academiae Scientiarum Hungaricae 20 (1969), 89-104.
- [265] E. Szemerédi, On sets of integers containing no k elements in arithmetic progression, Acta Arithmetica **27** (1975), 199-245.
- [266] A. Taylor, A note on van der Waerden's theorem, J. Combinatorial Theory Series A 33 (1982), 215-219.
- [267] J. Thouvenot, La démonstration de Furstenberg du théorème de Szemerédi sur les progressions arithmétiques, *Lecture Notes in Math.* 710 (1979), Springer-Verlag, Berlin, 1979.
- [268] A. Thue, <u>Selected Mathematical Papers</u>, T. Nagell, A. Selberg, S. Selberg, and K. Thalberg editors, Universitetsforlaget, Oslo, 1977.
- [269] B. Valko, Discrepancy of arithmetic progressions in higher dimensions, J. Number Theory 92 (2002), 117-130.
- [270] B. L. van der Waerden, Beweis einer baudetschen Vermutung, Nieuw Archief voor Wiskunde 15 (1927), 212-216.
- [271] B. L. van der Waerden, How the proof of Baudet's conjecture was found, in <u>Studies in Pure Mathematics (Presented to Richard Rado)</u>, 251-260, Academic Press, London, 1971.

- [272] M. Walters, Combinatorial proofs of the polynomial van der Waerden theorem and the polynomial Hales-Jewett theorem, J. London Math. Society 61 (2000), 1-12.
- [273] A. Wiles, Modular forms, elliptic curves, and Fermat's last theorem, in: Proceedings of the International Congress of Mathematicians, Volume 1 (Zurich, 1994), 243-245, Birkhauser, Basel, 1995.
- [274] J. Winn, Asymptotic bounds for classical Ramsey numbers, Polygonal Publishing House, Washington, NJ, 1988.
- [275] J. Wróblewski, A nonaveraging set of integers with a large sum of reciprocals, Math. Computation 43 (1984), 261-262.

Index

(a, b)-triple: 135
Abbott, H.: 292
Accessible set: 266
Ackermann, W.: 39
Ackermann function: 39, 40, 53
Alon, N.: 185, 197
a (mod m)-progression: 163
Arithmetic progression: 11
Monotone: 191
Arithmetic progression (mod m): 163
Ascending wave: 100
Asymptotic notation: 10
Big-O notation: 10
Little-o notation: 10
Augmented progression: 152

Baudet's conjecture: 54 Baumgartner, J.: 291 Bergelson, V.: 134, 222, 258, 291 Berlekamp, E.: 35 Bialostocki, A.: 223 Big-O notation: 10 Brakemeier, W.: 291 Brown's lemma: 276, 278, 291 Brown, T.: 100, 134, 291 Burr, S.: 258

Cantor's diagonal argument: 25 Ceiling function: 10 Coloring edge-coloring: 7 *r*-coloring: 9 Columns condition: 252 Compactness principle: 25 Complete graph: 7 Composition function: 38, 117 Cubefree coloring: 279 Culprit: 31

Datskovsky, B.: 222 Degree of accessibility: 269 Degree of regularity: 138, 139, 147, 176 Derived coloring: 44 Descending wave: 70, 100, 176 Deuber, W.: 258 Difference coloring: 19 Diffsequence: 267, 276 Doublefree set: 265

Edge: 6 Edge-coloring: 7 Equation Homogeneous: 226 Nonhomogeneous: 229 Regular: 13, 219, 226 Erdős, P.: 18, 19, 20, 35, 40, 41, 43, 51, 52, 53, 54, 100, 134, 200, 222, 264 Erdős-Ginzburg-Ziv theorem: 283, 291 Erdős and Turán function: 41, 53, 100 Ergodic theory: 53, 128, 134, 222 Everts, F.: 53 Exoo, G.: 222

Fermat's last theorem: 200, 222 Fibonacci numbers: 275, 288, 290 Fields medal: 38 Floor function: 10 Folkman number: 264 Folkman's theorem: 261, 262, 291 Folkman-Rado-Sanders theorem: 261

Fredricksen, H.: 222 Freedman, A.: 100 Furstenburg, H.: 53 Gap set: 104 Finite: 105 Infinite: 111 Gap size: 278 Generalized quasi-progression: 66 Generalized Schur number: 212 Strict: 221 Geometric progression: 125 Gleason, A.: 20 Goodman, A.: 222 Gowers, T.: 38, 40, 41, 50, 53, 176, 264 GQ_{δ} -progression: 66 Graham, R.: 40, 41, 134, 198, 222 Graph: 6 Complete graph: 7 Subgraph: 6 Graver, J.: 20 Greatest integer function: 10 Greenwood, R.: 20 Grinstead, C.: 20 Hainal, A.: 53

Hales-Jewett theorem: 162 Harary, F.: 1, 20 Harborth, H.: 258 Hilbert, D.: 11, 20, 39, 223 Hindman, N.: 258, 291 Hindman's theorem: 291 Homogeneous: 226 Homothetic copy: 147 Hyperedge: 35 Hypergraph: 35

Irving, R.: 222 Isosceles triple: 242 Iterated function: 81

Kalbfleisch, J.: 20 Kéry, G.: 20

Landman, B.: 180 Large set: 124, 133, 266, 268, 272, 273, 274, 291 r-large set: 124, 266, 274 Lattice point: 17, 53 Least integer function: 10 Lefmann, H.: 100 Leibman, A.: 133, 134 Linear homogeneous equation: 226 Linear nonhomogeneous equation: 228 Little-o notation: 10 Liu, A.: 292 Long, A.: 180 Loo, S.: 258 Lovasz's local lemma: 53 Low-difference: 57 Maasberg, S.: 258 McKay, B.: 20 Min, Z.: 20 Mixed van der Waerden number: 33, 34, 49, 51, 53 Monochromatic: 7, 9 Monotone arithmetic progression: 191 Monotonic sequence: 4 Morse-Hedlund sequence: 258, 280, 291

Nathanson, M.: 197 Nonhomogeneous: 229

Off-diagonal Rado number: 250 Off-diagonal Ramsey number: 250

Parrilo, P.: 222 Party problem: 6 Permutation: 191 Piecewise syndetic: 276 Pigeonhole principle: 3, 17 p_n -sequence: 81 p_n -function: 81 Polynomial Generating: 81 Iterated: 81 Pomerance, C.: 54 Probabilistic method: 18, 50, 54 Product-set: 264 Property B: 36, 53 Prouhet-Thue-Morse sequence: 280

Quasi-progression: 56

Rabung, J.: 100, 197 Rado, R.: 13, 53, 211, 225, 258, 277 Rado numbers: 13, 230, 239, 257 Off-diagonal Rado numbers: 250 Rado's full theorem: 253, 258, 262 Rado's selection principle: 25 See compactness principle: 25 Rado's single equation theorem: 13, 225. 226 Ramsey, F.: 1, 8, 20 Ramsey number: 8, 15, 20, 202, 203, 218, 249, 250 Off-diagonal: 250 Ramsey's theorem: 2, 7, 19, 200, 202, 222 Ramsey-type: 11, 14 Rankin, R.: 53 *r*-coloring: 9 Recurrence relation: 91 Refined triple: 45 Regular: 13, 15, 138, 173, 219, 226 Regular system of equations: 226

Index

r-regular: 13, 15, 138, 173, 226, 250 Reverse regular: 185 Riddell, J.: 54 292 Roberts, S.: 20 Robertson, A.: 222 Rödl. V.: 222 Roth, K.: 43, 53, 197 Ruciński, A.: 222 Sanders, J.: 291 Schaal, D.: 223, 259 Schmidt, W.: 35 Schmidt's lemma: 37, 53 Schoen, T.: 222 Schur, I.: 11, 12, 199, 200, 202, 222 Schur number: 12, 19, 201, 214, 222 Generalized: 212, 222 Strict: 221 Strict generalized: 221 Schur's theorem: 12, 19, 20, 200, 201 Schur triple: 201, 222 Scope: 72 Semi-progression: 72 Shelah, S.: 40 Shiue, P.: 100 Sidorenko, A.: 197 Sós, V.: 19 Spencer, J.: 20, 54, 197, 198, 262 Squarefree coloring: 279 Stirling's formula: 18 Strict generalized Schur numbers: 221 Strict Schur numbers: 221 Subgraph: 6 Sumset: 261 Superset: 56 Sweet, M.: 222 Syndetic: 276 Szekeres, G.: 20, 100, 200, 222 Szemerédi, E.: 43, 53 Szemerédi's theorem: 43 Tail: 152 Thue-Morse sequence: 280, 289, 291 Tic-tac-toe game: 245

Valid coloring: 15, 29, 99, 117 Valko, B.: 197 Vandermonde determinant: 83 Van der Waerden, B. L.: 11 Van der Waerden number: 11, 12, 21, 34, 38, 52, 69 Algorithms to compute: 30, 32 Asymptotic bounds: 35, 37, 38 General bounds: 35, 38, 41 Mixed van der Waerden number: 19, 33, 34, 49, 52, 53

Tower function: 38, 39, 40, 53

Turán, P.: 41, 43, 53, 99

Van der Waerden's theorem: 11, 12, 20, 23, 43, 53 Equivalent statements: 27 Finite version: 25 Infinite version: 25 Vertices: 6

Walters, M.: 134 Wave Ascending: 100 Descending: 70, 93, 100 Wiles, A.: 200, 222 Witsenhausen, H.: 198 Wow function: 39, 40, 53 Wróblewski, J.: 53

Yackel, J.: 20

Zaks, A.: 185, 197 Zeilberger, D.: 222 Zero-sum sequence: 282

Titles in This Series

- 24 Bruce M. Landman and Aaron Robertson, Ramsey theory on the integers, 2004
- 23 S. K. Lando, Lectures on generating functions, 2003
- 22 Andreas Arvanitoyeorgos, An introduction to Lie groups and the geometry of homogeneous spaces, 2003
- W. J. Kaczor and M. T. Nowak, Problems in mathematical analysis III: Integration, 2003
- 20 Klaus Hulek, Elementary algebraic geometry, 2003
- 19 A. Shen and N. K. Vereshchagin, Computable functions, 2003
- 18 V. V. Yaschenko, Editor, Cryptography: An introduction, 2002
- 17 A. Shen and N. K. Vereshchagin, Basic set theory, 2002
- 16 Wolfgang Kühnel, Differential geometry: curves surfaces manifolds, 2002
- 15 Gerd Fischer, Plane algebraic curves, 2001
- 14 V. A. Vassiliev, Introduction to topology, 2001
- 13 Frederick J. Almgren, Jr., Plateau's problem: An invitation to varifold geometry, 2001
- 12 W. J. Kaczor and M. T. Nowak, Problems in mathematical analysis II: Continuity and differentiation, 2001
- 11 Michael Mesterton-Gibbons, An introduction to game-theoretic modelling, 2000
- John Oprea, The mathematics of soap films: Explorations with Maple[®], 2000
- 9 David E. Blair, Inversion theory and conformal mapping, 2000
- 8 Edward B. Burger, Exploring the number jungle: A journey into diophantine analysis, 2000
- 7 Judy L. Walker, Codes and curves, 2000
- 6 Gérald Tenenbaum and Michel Mendès France, The prime numbers and their distribution, 2000
- 5 Alexander Mehlmann, The game's afoot! Game theory in myth and paradox, 2000
- 4 W. J. Kaczor and M. T. Nowak, Problems in mathematical analysis I: Real numbers, sequences and series, 2000
- 3 Roger Knobel, An introduction to the mathematical theory of waves, 2000
- 2 Gregory F. Lawler and Lester N. Coyle, Lectures on contemporary probability, 1999
- 1 Charles Radin, Miles of tiles, 1999

Errata for <u>Ramsey Theory on the Integers</u> by Bruce M. Landman and Aaron Robertson Date: June 23, 2004

FRONT MATTER

p. xi, Table 4.1: change "w" to "w'"

CHAPTER 1

- p. 12, line -11: change "We" to "The"
- p. 12, line -4: insert "monochromatic" before "integers"
- p. 15, line -5: change "F" to " \mathcal{F} "
- p. 16, line 2: change "exists" to "exist"
- p. 17, line 17: change "M(k;1)" to "M(k+1;1)"
- p. 20, line -3: change "contain" to "contains"

CHAPTER 2

- p. 24, line -2: change "exist" to "exists"
- p. 29, line -15: change "w(3,4)" to "w(3;4)"
- p. 34, Table 2.1: change "w(5,3,3;3) = 77" to "w(5,3,3;3) = 80"
- p. 37, line 3: change " $k \ge 1$ " to " $k \ge 2$ "
- p. 42, line -2: change " $\frac{m}{r}$ " to " $\frac{m-1}{r}$ "
- p. 48, lines 5–8: replace the definitions of the x'_i 's by: $x'_0 = d; x'_i = x_{i-1}$ for $1 \le i \le t+1$.

p. 49, exercise 1: Should read: "Show that within [1, n] there are $\frac{n^2}{2(k-1)}(1 + o(1))$ k-term arithmetic progressions."

p. 50, exercise 6: change "w(k, 2; r)" to "w(k, 2; 2)"

p. 50, exercise 8: in (b) change "
$$\sum_{i=1}^{\frac{n^2}{4}+O(n)} 2^{1-k} = \frac{n^2+O(n)}{2^{k+1}}$$
" to "
$$\sum_{i=1}^{\frac{n^2}{2(k-1)}(1+o(1))} 2^{1-k} = \frac{n^2}{(k-1)2^k}(1+o(1)).$$
"

p. 50, exercise 8: in (c) change " $\frac{n^2 + O(n)}{2^{k+1}} < 1$ " to " $\frac{n^2}{(k-1)2^k}(1+o(1)) < 1$ "

- p. 50, exercise 8: in (d) change " $n = 2^{k/2}$ " to " $n = \sqrt{k-1} \cdot 2^{\frac{k-1}{2}}$ "
- p. 52, line -3: change "w(5;2)" to "w(5,5;2)"

CHAPTER 3

p. 61, line -8: change the first occurrence of "t" to "i"

p. 69, Table 3.2: replace the functions in the first column by the functions x - 1, x - 2, x - 3, x - 4 and x - 5 (in that order)

p. 71, line 2: change " $x_0 = 0$ and $x_1 = 1$ " to " $x_0 = n + 1$ and $x_1 = n$ "

- p. 71, line 4: change "min" to "max" and change " $y x_i$ " to " $x_i y$ "
- p. 71, line 5: change " x_1, x_2, \ldots, x_k " to " $x_k, x_{k-1}, \ldots, x_1$ "

p. 71, line 6: change "and that $x_k \leq n$ " to "that is contained in [1, n]" (Note: the proof of Theorem 3.21 given in the text is for *ascending* waves.)

- p. 84, line 8: change "(1-a)d" to "(1-a) + d"
- p. 88, line 10: change " $AP \cup P_{k-2}$ " to " $R(AP \cup P_{k-2}, k)$ "

CHAPTER 4

p. 105, line -12: change "k" to "n + 1"

- p. 109, line 9: change "j" to " ℓ "
- p. 111, Theorem 4.9: change " $m\mathbb{Z}^+$ " to " $A_{m\mathbb{Z}^+}$ "
- p. 111, Corollary 4.10: change " $\mathbb{Z}^+ F$ " to " $A_{\mathbb{Z}^+-F}$ "

CHAPTER 5

- p. 136, line 5: change "k" to "3"
- p. 137, line 9: change "w" to "w'"
- p. 138, proof: there are 3 occurrences of w that should be w'
- p. 151, line 3: change " \leq " to "<"

CHAPTER 7

p. 192, line -7 and -9: delete "= S_1 " and "= S_2 "

CHAPTER 8

p. 210, line -9: change "the same color" to "of different colors"

CHAPTER 9

p. 242, line 5: insert " $1 + i_1a + j_1b = 1 + i_2a + j_2b$ " after "such that"

p. 251, line 17: change " $\{0, 1, ..., r\}$ " to " $\{1, 2, ..., r\}$ "

p. 252, line 6: change "There" to "For every r-coloring of \mathbb{Z}^+ , there"

CHAPTER 10

- p. 261, last line: change "if $t \in T$, then $2t \notin T$ " to " $t \in T$ does not imply that $2t \in T$ "
- p. 262, line 2: insert "obtained only" after "sum"
- p. 263, lines 3, 5, and 8: change "n(k;r)" to "kn(k;r)"
- p. 263, line 8: change " $[\frac{m}{2}, m]$ " to " $(\frac{m}{2}, m]$,"
- p. 263, line of change $\lfloor 2, \dots \rfloor$ by (a + d) + D to " $(a + d) + \sum_{r \in R \subseteq [1,k]} x_r$ "

p. 268, Lemma 10.20: before "then" insert "and if either $D + E = \{d + e : d \in D, e \in E\} \subseteq D$ or $D + E \subseteq E$," p. 270, replace the first paragraph of the proof of Theorem 10.24 with the following:

We first show that $doa(D) \geq 3$. Assume, for a contradiction, that $\gamma : \mathbb{Z}^+ \to \{0, 1, 2\}$ is a 3-coloring without arbitrarily long monochromatic D-diffsequences. Let $s_1 < s_2 < \cdots < s_m$ be a monochromatic D-diffsequence of maximal length. We may assume this diffequence has color 2. Then $S = \{s_m + j : j \text{ odd}\}$ is void of color 2. Let $S = \bigcup_{i \ge 0} S_i$, where $S_i = \{s_m + 2i(m+1) + j : j \in \{1, 3, 5, \dots, 2m+1\}\}$ are sets of m+1 elements. Define, for $i \ge 0$, $T_i = \{s_m + 2i(m+1) + j : j \in \{2, 4, 6, ..., 2m+2\}\}$, which are also sets of m+1 elements. Each T_i must contain an element of either color 0 or color 1, for otherwise T_i would be an (m + 1)-term D-diffsequence of color 2, contradicting the choice of m. Furthermore, each S_i must contain elements of both color 0 and color 1 (since it is void of color 2), for otherwise S_i would be an (m + 1)-term monochromatic D-diffsequence. Since some color, say color 0, must occur an infinite number of times in the T_i 's, there exist $x_{i_1} \in T_{i_1}, x_{i_2} \in T_{i_2}, x_{i_3} \in T_{i_3}, \ldots$, where $i_{j+1} > i_j + 1$, all of color 0. For each $j \ge 1$, let $y_{i_j} \in S_{i_j+1}$ be of color 0. Then $x_{i_1}, y_{i_1}, x_{i_2}, y_{i_2}, \ldots$ is an infinitely long *D*-diffsequence of color 0, contradicting the existence of *m*.

- p. 275, line 15: insert "long" after "arbitrarily"
- p. 285, line 7: change "2(m-1)p 1" to "(2m-1)p 1"
- p. 288, exercise 9: change each occurrence of " F_0 " to " F_1 " and each occurrence of " F_1 " to " F_2 "
- p. 288, exercise 10: change " S_3 " to " V_3 " and " S_4 " to " V_4 "
- p. 290, exercises 10 and 11: change "f" to " Δ " in each problem